University of Strathclyde

The impact of sub-optimal international portfolio allocations on cost of capital, stock market development and investor protection standards

Frank Obenpong Kwabi

Department of Accounting and Finance

Submitted for the degree of Doctor of Philosophy

The impact of sub-optimal international portfolio allocations on cost of capital, stock market development and investor protection standards

By

Frank Obenpong Kwabi



Strathclyde Business School

Department of Accounting and Finance

A thesis submitted in fulfilment of the requirement for the degree of

Doctor of Philosophy

February 2015

This thesis is the result of the author's original research. It has been composed by the author and has not been previously submitted for examination which has led to the award of a degree.

'The copyright of this thesis belongs to the author under the terms of the United Kingdom Copyright Acts as qualified by University of Strathclyde Regulation 3.50. Due acknowledgement must always be made of the use of any material contained in, or derived from, this thesis.'

Signed:

Date: February 2015

Abstract

Both developed and emerging markets have liberalised their stock markets by removing investment restrictions on equity capital inflows and outflows. The aim is to attract foreign investors and also to allow domestic investors to diversify internationally. However, existing literature shows that local investors overweight the domestic market (home bias), whilst foreign investors under or overweight foreign markets (foreign bias). Current studies have mainly investigated factors that determine home and foreign bias. The study uses comprehensive macro and micro level data to examine the implications of home and foreign bias on three research questions.

The first empirical study investigates the impact of home and foreign bias on cost of capital. We mainly use five measures to proxy for cost of capital. We find compelling evidence supporting the hypothesis, those countries that exhibit higher home bias, experience higher cost of capital. Similarly, consistent with theory, we find that countries that have higher foreign bias enjoy lower cost of capital.

In the second empirical study, we examine the impact of home and foreign bias on stock market development. Economic reasoning suggests, that countries that have home bias should have lower level of stock market development, while the countries where foreign equity portfolio investors invest more, should be associated with higher development. Our findings, based on rigorous analysis, confirm that prevalence of higher degree of home bias impedes stock market development. Likewise, higher foreign bias in equity portfolio allocations has significant positive implications for the development level of domestic stock market.

Finally, in our third empirical research, we examine whether varying degrees of home and foreign bias have any impact on country level investor protection standards. We report two findings. First, we find strong evidence that supports the hypothesis that home bias leads to weak investor protection. Second, consistent with theory, countries that experience higher foreign bias, tend to have better investor protection.

The findings suggest that provision of encouraging optimal international portfolio allocations to increase risk sharing, could be a crucial policy measure for governments. Policy makers in emerging countries can improve macroeconomic fundamentals and good governance to attract and retain foreign investors.

Acknowledgement

I am most grateful to God for the good health I have had throughout the period of undertaking the PhD. I would like express my deepest appreciation to my principal supervisor, Dr Chandra Thapa for his guidance and motivation necessary to complete this research. I would like to express my gratitude to my second supervisor, Professor Krishna Paudyal and all members of staff at the Accounting and Finance Department of the Strathclyde Business School for their support. My appreciation also goes to my fellow PhD students for their warm friendship, generous support and comments.

Last but not the least; I would like to express my deepest appreciation to my family, for their unending support and tolerance. I am very grateful to my little son Aaron for his patience when I have to excuse myself from family playing time to do some research work.

Tables of Contents

Chapter	1:	Introduction and summary of findings	1
1.1	Intro	oduction	1
1.2	Over	rview of the Study	2
1.3	Rese	earch Importance and Motivation	5
1.4	Rese	earch Objectives and Hypotheses	6
1.5	Rese	earch Gaps and Contributions	7
1.6	Rese	earch Findings	. 12
1.7	The	Structure of the Thesis	. 13
Chapter	2:	Background and literature review	. 15
2.1	Intro	oduction	. 15
2.2	Back	ground	. 15
2.2	.1	Portfolio theory	. 15
2.2	.2	The case for international portfolio diversification	. 16
2.2	.3	Stock market liberalization and integration	. 22
2.2	.4	Home and foreign bias phenomenon	. 27
2.3	Liter	rature review for Chapter 6: Sub-optimal portfolio allocation and cost of capital	. 29
2.3	.1	Importance of cost of capital	. 29
2.3	.2	Financial globalization and cost of capital	.31
2.3	.3	Gaps in the cost of capital literature	. 37
2.4 devel	Liter opmer	rature review for chapter 7: International portfolio investment and stock market nt	. 38
2.4	.1	Importance of stock market development	. 38
2.4	.2	Foreign portfolio investment and stock market development	. 39
2.4	.3	Adverse effect of foreign investment flow on stock market development	.43
2.4	.4	Gaps in the stock market development literature	.46
2.5 standa	Liter ards	rature review for chapter 8: International portfolio investment and investor protection	on .48
2.5	.1	Importance of investor protection	.48
2.5	.2	Investor protection and foreign investors	. 52
2.5	.3	Domestic institutional investors and investor protection	.54
2.5	.4	Foreign institutional investors' activism and investor protection	. 55
2.5	.5	Institutional investors' activism	.60
2.5	.6	Gaps in the investor protection standards literature	.63

Chapter 3:		Research questions and hypothesis development	66
3.	1 Res	earch questions	66
3.	2 W	nat is the impact of sub-optimal international portfolio allocation on cost of capital?	66
	3.2.1	Financial liberalisation and cost of capital	66
	3.2.2	Theoretical framework	69
	3.2.3	Contributions	75
	3.2.4	Summary of findings	78
3.	3 Wh	at is the role of international portfolio investment on stock market development?	79
	3.3.1	Equity portfolio investment and stock market development	79
	3.3.2	Contributions	82
	3.3.3	Summary of findings	83
3.	4 Wh	at is the effect of international portfolio investment on investor protection?	84
	3.4.1	Sub-optimal portfolio allocation and investor protection	84
	3.4.2	Contributions	89
	3.4.3	Summary of findings	91
Chap	oter 4:	Data	92
4.	1 Ho	me and foreign bias measures	92
	4.1.1	Equity home bias	93
	4.1.2	Equity foreign bias	94
4.	2 Sub	p-optimal international portfolio allocations and cost of capital	95
	4.2.1	Historical realised returns of the market	95
	4.2.2	Sovereign credit-risk rating measures of cost of capital	96
	4.2.3	Country equity risk premium	97
	4.2.4	Dividend Yield	97
	4.2.5	Control variables	98
4.	3 Inte	ernational portfolio investment and stock market development1	.00
	4.3.1	Market capitalisation as a percentage of GDP1	.00
	4.3.2	Stock value traded as a percentage of GDP1	.01
	4.3.3	Turnover ratio1	.01
	4.3.4	Transaction cost1	.02
	4.3.5	Control variables1	.03
4.	4 Inte	ernational portfolio investment and investor protection1	.08
	4.4.1	Government effectiveness (quality of institutions)1	.09
	4.4.2	Control of corruption1	.10

	4.4.3	3	Regulatory quality	110
	4.4.4	4	Rule of law	111
	4.4.5	5	Control variables	
4.	.5	Sun	nmary of data	116
Cha	pter !	5:	Methodology	121
5.	.1	Sum	imary analysis	121
5.	.2	Corr	elation analysis	
5.	.3	Роо	led OLS	
5.	.4	Auto	ocorrelation and heteroskedasticity standard error correction	124
	5.4.2	1	Autocorrelation	125
	5.4.2	2	Heteroskedascity	
	5.4.3	3	Multicollinearity	129
	5.4.4	4	Newey-West autocorrelation and heteroskedasticity standard error	129
5.	.5	Spee	cifications issues	130
	5.5.2	1	Omitted variable bias	130
	5.5.2	2	Endogeneity (reverse causality)	132
	5.5.3	3	Lagged values	133
	5.5.4	4	Heckman selection model	134
Cha	pter (6 Firs	t empirical study: Sub-optimal portfolio allocation and cost of capital	137
6.	.1	Emp	pirical analysis	137
6.	.2	Sum	ımary analysis	138
6.	.3	Corr	elation analysis	140
6.	.4	Ecor	nometric issues	140
	6.4.2	1	Autocorrelation	140
	6.4.2	2	Heteroskedasticity	140
	6.4.3	3	Multicollinearity	141
	6.4.4	4	Omitted variable bias	141
6.	.5	Reg	ression results	141
	6.5.2	1	Cost of capital and equity home bias	142
	6.5.2	2	Cost of capital and equity foreign bias	142
	6.5.3	3	Cost of capital and fund level foreign bias measure	143
	6.5.4	4	Control variables	143
6.	.6	Rob	ustness test	144
	6.6.2	1	Additional measure of cost of capital: Tobin's Q	145

6.6.2		Reverse causality: Lagged pre-determined variables	146
6.6.	.3	Heckman selection bias	147
6.7	Cha	pter summary	148
Chapter	7 Sec	ond empirical study: International portfolio investment and stock market	
develop	ment		164
7.1	Sum	nmary analysis	164
7.2	Corr	relation coefficients	166
7.3	Ecoi	nometric issues	167
7.3.	.1	Autocorrelation	167
7.3.	.2	Heteroskedasticity	167
7.3.	.3	Multicollinearity	168
7.3.	.4	Omitted variable bias	168
7.4	Reg	ression results	168
7.4.	.1	Stock market development and equity home bias	169
7.4.	.2	Stock market development and equity foreign bias	169
7.4.	.3	Stock market development and fund level foreign bias measures	170
7.4.	.4	Control variables	170
7.5	Rob	ustness tests	171
7.5.	.1	Reverse causality: Lagged pre-determined variables	171
7.5.	.2	Heckman selection bias	173
7.6	Cha	pter summary	174
Chapter	8 Thi	rd empirical study: International portfolio investment and investor protection	. 190
8.1	Sum	imary statistics	191
8.2	Corr	relation analysis	192
8.3	Ecoi	nometric issues	192
8.3.	.1	Autocorrelation	193
8.3.	.2	Heteroskedasticity	193
8.3.	.3	Multicollinearity	193
8.3.	.4	Omitted variable bias	194
8.4	Reg	ression results	194
8.4.	.1	Investor protection and equity home bias	194
8.4.	.2	Investor protection and equity foreign bias	195
8.4.	.3	Investor protection and fund level foreign bias measures	196
8.4.	.4	Control variables	196

8.5	Robustness tests	196
8.5.	1 Reverse causality: Lagged pre-determined variables	197
8.5.	2 Heckman selection bias	
8.5.	3 Alternative investor protection measures	
8.6	Chapter summary	
Chapter	9 Summary, conclusions and limitations	218
9.1	Introduction	218
9.2	Restatement of the research questions	218
9.3	Hypothesis, data and methodology	219
9.4	Research findings	220
9.5	Key contributions	222
9.6	Potential limitations of the research and avenues for future research	223
9.7	Policy implications of the research	224
9.8	Concluding remarks	224
Appendix		226
References		233

List of Tables

Table 6-1: Summary statistics of depende and key independent variables	150
Table 6-2: Summary statistics of control control variables	151
Table 6-3: Pearson's pairwise correlation coefficient between the dependent and independe	ent
variables	153
Table 6-4: The relation between equity home bias and four cost of capital proxies	154
Table 6-5: The relation between equity foreign bias and four cost of capital proxies	155
Table 6- 6: The relation between equity foreign bias using global fund and four different cost	of
capital proxies	156
Table 6-7: Relation between foreign bias, home bias, country bias and Tobin's Q	157
Table 6-8: Using one-year lag CPIS_HB	158
Table 6-9: Using one-year lag CPIS_FB	159
Table 6-10: Using one-year lag GF_FB	160
Table 6-11: CPIS_HB regression based on the Heckman two stage treatment model	161
Table 6-12: CPIS_FB regression based on the Heckman two stage treatment model	162
Table 6-13: GF_FB regression based on the Heckman two stage treatment model	163
Table 7-1: Summary statistics of the dependent variables	176
Table 7-2: Summary statistics of control control variables	178
Table 3: Pearson's pairwise correlation coefficients between the dependent and independer	nt
variables	180
Table 7-4: The relation between equity home bias and four measures of stock market develo	pment
Table 7-5: The relation between equity foreign bias and four measures of stock market devel	181 onment
	182
Table 7-6: The relation between equity country bias and four measures of stock market	
development	183
Table 7-7: Using one-year lag equity home bias	184
Table 7-8: Using one-year lag equity foreign bias	185
Table 7-9: Using one-year lag equity country hias	186
Table 7-10: CPIS HB regression based on the Heckman two stage treatment model	187
Table 7-11: CPIS FB regression based on the Heckman two stage treatment model	188
Table 7-12: GF_FB regression based on the Heckman two stage treatment model	189
Table 8-1: Summary statistics of dependent variables	204
Table 8-2: Summary statistics of control control	206
Table 8-3: Pearson's pairwise correlation coefficients between the dependent and independent	ont
variables	
Table 8-4: The relation between equity home bias and four investor protection proxies	
Table 8-5: The relation between equity foreign bias and four investor protection proxies	
Table 8-6: The relation between equity foreign bias using global fund and four different investigation	stor
protection proxies	210
Table 8-7: Using one-year lag CPIS_HB	211
Table 8-8: Using one-year lag CPIS FB	212
Table 8-9: Using one-year lag GF_FB	213
Table 8-10: CPIS_HB regression based on the Heckman two stage treatment model	214
Table 8-11: CPIS FB regression based on the Heckman two stage treatment model	215

Table 8-12: GF_FB regression based on the Heckman two stage treatment model	216
Table 8-13: Using alternative investor protection measures	217

Chapter 1: Introduction and summary of findings

1.1 Introduction

Several studies provide theoretical and empirical evidence of how equity investors could benefit from international portfolio diversification (see Sonlink, 1974; Bekaert and Urias, 1996; Stulz, 1999a; and De Roon et al. 2001). Following on the mean-variance approach of Markowitz (1952) and Solnik (1974), the international capital asset pricing model (ICAPM) suggests that equity portfolio investors should hold the implied world market portfolio as the optimal benchmark (see Adler and Dumas, 1983; and Lewis, 1999). However, in real world practice, a number of studies show that investors in both developed and emerging economies ignore the theoretical prescription of ICAPM, by sub-optimally investing across international markets. Such sub-optimal international allocations generate what the literature terms as the varying degrees of home and foreign biases in their international portfolio allocations. Home bias relates to an overinvestment in the domestic stock market, relative to the theoretical optimal investment based on the countries' world market capitalisation weight. Similarly, foreign bias refers to the phenomenon whereby foreign investors either overweight or underweight their investment across foreign countries. In this study we refer to the phenomenon of home and foreign bias as sub-optimal portfolio allocations.¹

It is conceivable that the sub-optimal international portfolio allocations, which lead to the home and foreign biases, should reduce the potential benefits that the countries can obtain from financial liberalisation or optimal international portfolio diversification. So far, the review of existing literature reveals that most of the studies have focussed on explaining the causes of home and foreign biases.² However, exceptionally scant studies examine the implications of such sub-optimal investments. In this study we empirically examine the impact of sub-optimal portfolio allocations on cost of capital, stock market development and investor protection. This chapter provides the introduction and structure of the study.

¹ For further details on the differences between home and foreign bias see Dahlquist et al. (2003) and Chan et al (2005).

² Such explanations include differences in stock market development, corporate governance, accounting standards, hedging motives, departure from purchasing power parity, political risk, liquidity risk, information asymmetry and capital control (see Alder and Dumas, 1983 Solnik, 1974, and Cooper and Kaplanis, 1986, 1994 and Stulz, 2005).

We introduce the current study in this chapter which is structured as follows: Section 1.2 offers an overview of the study; Section 1.3 provides the importance and motivation of the study; Section 1.4 explains the objectives, the research design and the methods of the research; Section 1.5 offers the analysis of the key research findings; Section 1.6 provides the contributions of the research; and Section 1.7 shows the structure of the thesis.

1.2 Overview of the Study

The central aim of the research is to examine the impact of sub-optimal equity portfolio allocation (home and foreign bias) on three research questions. Current studies show, that in spite of financial liberalisation and the suggestions by ICAPM, both domestic and foreign investors' deviate from holding optimal portfolio allocation. Therefore, the study specifically seeks to examine whether, and how, the phenomenon of home and foreign bias have any varying impact on cost of capital, stock market development and investor protection.

The first empirical study examines the impact of home and foreign bias on cost of capital. The conjecture is that when a country attracts sufficient foreign investment, it will subsequently lead to more sharing of risk between domestic and foreign investors, increase good governance and stock market liquidity, which, in turn, reduces cost of capital. Conversely, countries that experience higher home bias tend to have segmented markets, and therefore experience less risk sharing and higher cost of capital. We use five alternative cost of capital measures which have been widely used in the existing literature: historical realised returns of the market (HRR_m) , sovereign credit risk rating (rCred), country equity risk premium (CERP), dividend yield (DY), and Tobin's Q to proxy for cost of capital. This is mainly to ensure that home and foreign bias measures are not sensitive to a particular cost of capital proxy. We use data from the Coordinated Portfolio Investment Survey (CPIS) of the International Monetary Fund (IMF) to construct home bias (CPIS_HB) and foreign bias (CPIS_FB). The study therefore examines the impact of home and foreign bias on cost of capital. Additionally, we construct global fund country bias (GF_FB) using a unique and novel global fund data to investigate the impact of country bias on cost of capital. Stulz (1999a) and Henry (2000a) provide a theoretical argument of why financial globalisation reduces cost of capital. Previous studies have largely investigated the effects of financial liberalisation on cost of capital, using event studies and cross-listings on the US stock exchange.

The second empirical study examines the impact of home and foreign bias on stock market development. Errunza (2001) theoretically argues that when a country attracts sufficient foreign investment, stock market development improves because foreign investors will demand better investor protection. He notes that the active participation of foreign equity portfolio investors (FEPIs) infuses confidence among the local investors to trade nationally and internationally, driving up the level of competition in domestic markets. As such, with the increasing presence of FEPIs, the trading activities of domestic investors also grow, which further helps the market to become more active/liquid and price efficient, leading to efficient allocation of resources in the economy. Using a sample of 46 countries, Vagias and Dijk (2011) provide strong evidence, confirming the conjecture that foreign portfolio investors enhance market liquidity.

Kar (2001) shows that India liberalised its equity market in 1992, the increased activities of foreign investors began to play a crucial role in the institutionalisation of the market. He notes that post 1992, there were significant reforms observed within a span of six years, such as automation of trading system which increased transparency and efficiency, introduction of electronic book entry transfer system helped remove the inefficiencies and risks associated with a paper-based system, and the establishment of regulatory framework by securities exchange board of India to protect investors etc.

We use four different stock market development measures which have vastly been employed in the development and finance literature; market capitalisation as a percentage of GDP (*MGDP*), stock value traded as a percentage of GDP (*TRGDP*), turnover ratio (*TURN*), and transaction cost (*TRCOST*). This is to ensure that our sub-optimal portfolio allocation measures (home and foreign bias), are robust to all the stock market development proxies. Stulz (1999a) theoretically suggests that home bias increases cost of capital. Therefore, home bias will subsequently impede on stock market development, as home bias will lower the valuations of firms because there is an inverse relationship between cost of capital and valuation of firms.

Existing studies mainly examine the importance of stock market development in relation to economic growth (see King and Levine, 1993; Rajan and Zingales, 1998; and Carlin and Mayer, 2003). If home bias persists in a country, the domestic country stock market will be segmented. Theory suggests that, the segmentation of the domestic market, as a result of

home bias, will reduce stock prices and value trading, increase transaction cost and lower the importance of the stock market in the economy. However, when foreign investors allocate their investment towards the implied world market capitalisation weight, then it implies that the country is far more favoured by foreign investors. This will then lead to high participation in the stock market by investors. Subsequently, it will create an efficient trading technology, greater liquidity and improved transparency. This suggests that higher foreign bias will lead to better stock market development.

The third and final empirical study investigates the impact of home and foreign bias on investor protection. Existing studies have extensively examined the importance of investor protection on several variables, particularly in the areas of stock market development, economic growth and development. For instance, Eleswarapu and Venkataraman (2006) demonstrate that companies providing better investor protection enjoy high turnover and liquidity. La Porta et al. (1997) show that better investor protection standards relate to high number of listed companies. Stulz (2005) shows that countries providing better investor protection to minority investors, experience economic growth.

Even though investor protection is important for stock market development and economic growth, limited studies have investigated factors that influence or determine investor protection. We conjecture that when a country attracts sufficient foreign investment, the foreign investors will demand better protection and prevent companies from engaging in corrupt practices and expropriation of resources. Beck et al. (2000a) argue that countries will be compelled to improve the level of investor protection if they are to succeed in attracting foreign investors into the country, as they will be competing with other countries for foreign investors. When the domestic equity market is primarily dominated by local investors as a result of home bias, the domestic market becomes segmented, thereby reducing the level of investor protection provided to minority investors. Additionally, government will pursue poor policies that reduce return on investment.

To provide robustness to our analysis, we examine the impact of home and foreign bias on investor protection using four proxies of investor protection standards; government effectiveness (*Gov_Eff*), control of corruption (*Con_Cor*), regulatory quality (*Reg_Qual*), and rule of law (*Rule_Law*) which we obtain from World Governance Indicators (WGI). We also

employ alternative measures of investor protection from two governance sources to check the robustness of our results.

1.3 Research Importance and Motivation

Reviews of the existing literature provide reasons for home and foreign bias. However, there is little research that has examined the implications of domestic and foreign investors, deviating their equity investment allocation from the implied weight suggested by the ICAPM. We are motivated to embark on this empirical research, based on the following three factors:

First, existing literature shows the importance of cost of capital (see section 2.3.1). For instance, financial markets use cost of capital as the discount rate they apply to cash flow to calculate equity prices. A decrease in cost of capital makes projects more profitable as a reduction in cost of capital improves the net present values (NPV) of projects, which increases investment of corporations and thereby increases shareholders' wealth. Stulz (1999a) provides theoretical explanations for why financial liberalisation leads to greater flow of foreign equity investment, which tends to reduce cost of capital in the host country through enhanced risk sharing. However, subsequent study by Henry (2000a) shows that financial liberalisation has little effect on cost of capital. We, show why international portfolio investment has not had the expected impact on cost of capital. Therefore, we are motivated to examine empirically, the varying impact of sub-optimal portfolio allocation (home and foreign biases) on cost of capital.

Second, reviews of the international finance literature show the important role stock markets play with respect to economic growth and development (see section 2.4.1). Rousseau and Sylla (2001) illustrate that a developed stock market leads to efficient allocation of resources, which, in turn, promotes economic growth. Levine and Zervos (1995a) demonstrate that an increase in stock market liquidity relates to economic growth. Errunza (2001) suggests that when a country liberalises the stock market, the country will attract foreign investors and will lead to stock market development. A review of the existing literature shows that even with the apparent importance of stock market development, existing studies have not examined the impact of home and foreign bias on stock market development.

Third, a substantial body of research mainly focus on the role and importance of investor protection standards (see section 2.5.1). For example, Stulz (2005) demonstrates that better investor protection, that protects minority investors, leads to economic growth. La Porta et al. (1997) argue that countries that provide better investor protection, experience high liquidity. Friedman et al. (2003) note that better investor protection, reduce expropriation risk, and provide sustained firm level performance during financial crises. There are limited studies that examine factors that influence country level investor protection. Prevailing studies predominantly examine the impact of investor protection on several variables (see Wei, 2000; Islam and Montenegro, 2002; IMF, 2005). To the best of our knowledge, no studies have examined the effect of home and foreign bias on investor protection standards. We are therefore motivated to examine the impact of home and foreign bias on country level investor protection standards.

1.4 Research Objectives and Hypotheses

The main objective of this research is to examine the impact of home and foreign bias on three research areas; cost of capital, stock market development, and investor protection standards. Our three empirical studies employ home and foreign bias data of 44 countries for a 10 year period, 2001 to 2010. We develop nine hypotheses and test those using Newey-West autocorrelation and heteroskedasticity correction standard errors. We use pooled ordinary least squares (OLS) with panel data modelling, due to the wide cross sectional variations and sequential differences in foreign portfolio investment. Employing panel data provides accurate inference of the model parameters. We use lagged-values, and the Heckman selection model to provide robustness to our result.

The first research hypothesis is that home bias measures increase cost of capital, whilst foreign and country bias reduces cost of capital (see Chapter 6). When domestic investors overinvest in their home market, risk sharing between home and foreign investors reduces, and therefore, increases cost of capital. However, countries that attract sufficient foreign investment, experience an increase in risk sharing amongst investors, which, in turn, reduces the cost of capital.

The second research objective, investigates the impact of home and foreign bias on stock market development (see Chapter 7). We hypothesise that the prevalence of home bias will

inhibit stock market development, whilst foreign bias will improve stock market development. When a country attracts sufficient foreign investors into the domestic market, their participation and demand for good governance will boost the confidence of domestic investors. Consequently, it will cause an increase in stock market liquidity, bid up prices; increase the efficiency of trading technologies, and the number of dealers and brokers, which will improve stock market development.

The third research objective is to examine the impact of home and foreign bias on investor protection (Chapter 8). We hypothesise that when home investors over-allocate their investment in deviation from the implied optimal weight suggested by ICAPM, it reduces the investor protection standards provided to minority investors. Alternatively, when a country is favoured by foreign investors who reside in countries where good governance practices exist, the foreign investors will demand and export good governance to the host countries. This will lead to an improvement in a country's level investor protection standards, which will protect minority investors from expropriation by company insiders and majority shareholders.

1.5 Research Gaps and Contributions

We present the contributions of our research in this chapter. The existence of home and foreign bias, suggests that domestic and foreign investors, allocate a greater percentage of their equity to their domestic market, in deviation from the global weight suggested by ICAPM. Earlier studies mainly provide explanations to the home and foreign bias phenomena. A review of the existing literature suggests, that there is limited research on the implications of sub-optimal portfolio allocation (home and foreign bias), due to unavailability of cross country equity investment data. We, therefore, take advantage of the high value bilateral foreign equity portfolio holdings survey data, provided by the Coordinated Portfolio Investment Survey (CPIS) of the International Monetary Fund (IMF). Additionally, we employ a novel fund level allocation data, provided by the Emerging Portfolio Fund Research (EPFR).³

In our first empirical study, we make several unique and important contributions to the home and foreign bias literature. First, current studies predominantly examine the causes of home

³ See appendix 1 for the Emerging Portfolio Fund Research data.

and foreign bias (see French and Poterba, 1991; Chan et al, 2005; Gelos and Wei, 2005; Bekaert and Wang, 2010), with only Lau et al. (2010) investigating the implications of these biases. In this study, we investigate whether varying degrees of home and foreign biases have implications for cost of capital. Furthermore, the relevant literature is either based on event studies, investigating how the cost of capital changes in the post financial liberalisation period, or analyses the influence of depository receipts on the cost of capital.⁴

Theory suggests that the reduction of cost of capital post-liberalisation is driven by the increased risk sharing between domestic and foreign investors; enhanced stock market liquidity and an improvement of good governance (see Doidge et al. 2004; Foerster and Karolyi, 2000). Similarly, a number of studies investigate the impact of floating ADRs/GDRs on cost of capital.⁵ We use direct measures of sub-optimal international portfolio investments (i.e. home and foreign bias) to study their effects on cost of capital. Furthermore, rather than solely using home bias measure, which focuses on domestic investors (see Lau et al. 2010), we use two additional relative measures of international portfolio investment, known as foreign bias.⁶ However, as Bekaert et al. (2009) note, ADR companies exhibit representation problems and hence, do not provide full exposure to foreign stock markets. Clearly, ADRs ignore the investments of local investors in foreign markets and foreign portfolio investors investing directly in non-cross-listed domestic firms. Countries receiving relatively higher investments, i.e. international investors exhibiting greater foreign bias, should be more integrated with world capital markets and should demonstrate a relatively lower cost of capital. The two foreign bias measures are based on aggregate macro and individual global fund level micro data.

Second, the aggregated measures of foreign and home bias measures used in existing studies do not consider the individual objectives or focus of funds. Although on a theoretical basis, each fund should be globally diversified, however in practice, a fund's objective could be a

⁴ For example, Kim and Singal (2000) and Chari and Henry (2004a) show that the post financial liberalization cost of capital of the liberalized economy significantly decreases. Stulz (1999) and Henry (2000) also find similar results, suggesting that the reduction in cost of capital is driven by increased risk-sharing and improved corporate governance. De Jong and de Roon (2005) document that, the increased time-varying integration (i.e. the process of gradual financial liberalization of the domestic equity market) is associated with a reduction in the cost of capital.

⁵ For example, Foerster and Karolyi, (1999), Errunza and Miller (2000) and Karolyi (2004) demonstrate that firms issuing American Depositary Receipts (ADRs) experience a fall in the cost of capital, driven by the increased global risk sharing effect. For more recent evidence see Edison and Warnock (2008) and Hail and Leuz (2009).

⁶ See section 3.1.1 and 3.1.2 for details on the construction of home and foreign biases.

single country, single region or global diversification, targeted to suit the preferences of different investor groups. For example, if the focus of the fund is diversification only within the European Union (EU), it will have no allocations across countries outside the EU. Additionally, if the focus of a fund is on a single country or region, it will significantly affect the aggregate measure of home and foreign bias. To address this issue of potential bias in the measure of sub-optimal allocations of international investors, we construct a foreign bias measure that uses unique micro firm level global funds' allocation data across 44 countries. These global funds that we use, carry the sole objective of global diversification across all investable countries. The use of global funds is the most restrictive and robust measure of foreign bias, significantly reducing what we refer to as the fund-focus bias in the construction of foreign bias measures used in the existing literature.

In addition to using the two conventional proxy of cost of capital, i.e. dividend yield and historical risk premium, we also use sovereign bond rating, based on implied cost of capital, and expected country equity risk measure (Jewel and Livingston, 1998; Lau et al. 2010; Damodaran, 2012). We use *Tobin's Q* to examine the impact of home and foreign bias on firm performance (firm valuation), as there is an inverse relation between firm valuation and cost of capital.

Consistent with theory, our comprehensive empirical analysis and extensive robustness tests provide strong evidence that a higher degree of home bias is associated with a higher cost of capital. Correspondingly, a higher degree of foreign bias, i.e. foreigners tilting their country specific portfolio weight towards the global optimum, reduces the host country's cost of capital.

In our second empirical study, we explore the impact of home and foreign bias on stock market development. Prior studies on stock market development have investigated the importance of stock market development. Very few studies have examined factors that influence or determine stock market development. For instance, Claessens et al. (2006) show macroeconomic fundamentals and institutional environment affect stock market development. Billmeier and Massa (2009), find that institutions and remittances impact positively on stock market development. In this regard, this study advances the existing literature on stock market development and makes significant contributions towards understanding the impact of home bias on stock market development.

The study makes four important contributions. First, it adds to the scant but growing literature on the implications of home bias in international equity portfolio investments. In spite of several studies explaining the potential reasons of home and foreign biases (see Chan et al, 2005), a comprehensive study, delving into the possible implications of such biases is highly limited, and to our knowledge, none on the consequences of stock market development. The only study that is remotely related to ours is that of Lau et al. (2010), who demonstrate the implications of sub-optimal portfolio investments on cost of capital. Our study, rather than solely using home bias measure, which focuses on domestic investors (see Lau et al. 2010), we use two alternative measures of international portfolio investment, known as foreign bias. Countries receiving relatively higher investments, i.e. international investors exhibiting greater foreign bias, should be more integrated with world capital markets and should experience a relatively lower cost of capital. The two foreign bias measures are based on aggregate macro and individual global fund level micro data.

Further, the aggregated measures of foreign and home biases used in existing studies do not consider the individual objectives or focus of funds. Although on a theoretical basis, each fund should be globally diversified, in practice, a fund's objective could be a single country, single region or global diversification, targeted to suit the preferences of different investor groups. For example, if the focus of the fund is diversification only within the European Union (EU), it will have no allocations across countries outside the EU. Additionally, if the focus of a fund is on a single country or region, it will significantly affect the aggregate measure of home and foreign bias. To address this issue of potential bias in the measure of sub-optimal allocations of international investors, we construct a foreign bias measure that uses unique micro firm level global funds' allocation data across 44 countries. The global funds that we use have the sole objective of global diversification across all investable countries. The use of global funds is the most restrictive and robust measure of foreign bias, significantly reducing what we refer to as the fund-focus bias in the construction of foreign bias measures used in the existing literature.

Second, we examine the impact of foreign bias on stock market development. Previous studies focus largely on factors determining foreign bias. For instance, Chan et al. (2005) investigate factors that influence foreign bias using mutual fund equity allocation. More recently, Chan et al. (2009) investigate the implications of foreign bias on firm value. Whilst, Beugelsdijk and Frijns (2010) examine how culture and cultural distances between stock

markets explains foreign bias. As far as existing literature is concerned, our study is the first to examine the impact of foreign bias on stock market development.

Third, another distinct contribution of our study, relative to existing studies, is that our study uses a unique dataset of micro global fund allocation data across 44 countries, to examine the impact of foreign bias using global fund foreign bias on stock market development.

Additionally, we also add to the growing literature that connects international integration and financial deepening of the local markets, post market liberalisations (Hargis and Ramanlal, 1997; Levine and Zervos, 1998, Errunza, 2001, Chan et al. 2009 and Bekaert et al. 2011). However, this literature is based on the assumption that liberalisation leads to increased participation of the foreign investors in the local market, which as Bekaert and Havery (2003) note, may not be the case, due to the presence of various direct, indirect and emerging market specific risk barriers. Our measure of home and foreign biases are direct measures of the variations in the degree of participation of foreign and domestic investors in international diversifications relative to the normative suggestions of ICAPM.

Our study has important policy implications, because existing studies show that a large and more efficient stock market leads to economic growth (see Levine and Zervos, 1996, Bencivenga et al. 1996; Rousseau and Wachtel, 1998; Levine et al, 2000; Beck et al, 2000b; Rousseau and Sylla, 2001). However, alternative studies, for instance (see Devereux and Smith, 1994; Stiglitz, 2000; Eichengreen, 2002), show that financial liberalisation has not had the expected positive impact on economic growth. We thereby provide evidence that cross-country pervasiveness of home bias inhibits stock market development. Additionally, we show that when foreign investors increase their investment towards the global portfolio weight as implied by ICAM, stock market development improves.

In our third empirical study, we provide novel contributions to the investor protection literature. Existing literature primarily examines the importance of investor protection by explaining economic growth and development. In spite of the importance of investor protection, to the best of our knowledge, very few studies have addressed the factors that determine investor protection. For instance, Wei (2000) shows that corruption negatively relates to an open economy, as resources will be committed to improve institutional quality. Islam and Montenegro (2002) demonstrate that trade openness is robustly associated with

institutional quality. The IMF (2005) shows that trade openness has a positive impact on both institutional transitions and the quality of economic institutions. Busse and Groning (2007) analyse the importance of trade liberalisation towards good governance. They find evidence that suggests trade openness has a positive influence on good governance. We extend the existing studies by examining the impact of home bias on investor protection.

There is no single published study that has addressed the effect of foreign bias on investor protection standards. To the best of our knowledge, our study is the first to examine the impact of foreign bias on investor protection. We employ four proxies of investor protection which aim to capture investors' rights, how these rights are enforced at courts and government policies that enhance return on investment. The proxies also capture government policies that increase return on investment. The proxies are government effectiveness (Gov_Eff), control of corruption (Con_Cor), regulatory quality (Reg_Qual), and rule of law ($Rule_Law$). We obtain data from World Governance Indicators, which has widely been used in the governance literature (see Knack et al. 2003; Andres, 2006, and Liu and San, 2006).

The study also applies global fund foreign bias to examine its effects on investor protection. We construct the variable, using funds which have the sole objective of global diversification. Finally, we also address the concern of endogeneity, using alternative estimation statistical techniques that include lagged predetermined values of home and foreign biases, and the Heckman selection model.

1.6 Research Findings

We present the research findings in consistence with the research objectives. The first empirical study tests three hypotheses, by examining the impact of sub-optimal portfolio allocation (home and foreign bias) on cost of capital. The study uses five measures to proxy for cost of capital. Consistent with the existing theory, the result in Chapter 6, provides strong and robust evidence that high degrees of home bias exhibited by domestic investors, increase cost of capital, due to lack of risk sharing. Nonetheless, the study finds that foreign bias leads to a reduction in cost of capital via increased international risk sharing and higher liquidity. As there is an inverse relation between cost of capital and firm valuation, we perform a robustness test using Tobin's Q. The result shows that home bias reduces firm valuation through an increase in cost of capital. We also find that foreign bias increases firm valuation, as a result of reduction in cost of capital.

The second empirical research tests three hypotheses, by examining the impact of suboptimal portfolio allocation on stock market development. We employ four proxies of stock market development. The research findings in Chapter 7 provide compelling evidence that the prevalence of home bias in a country inhibits stock market development. The reason for this is that the domestic stock market will be dominated by large institutional investors who will buy-and-hold, and will adversely affect the liquidity of the stock market. Consistent with the theory, we find that foreign bias has a statistically and significant positive impact on stock market development at the 1% level. The participation of foreign investors enhances liquidity; increases risk sharing, and improves good governance. We use lagged values, and the Heckman selection to address endogeneity problems, and the results suggest our main analysis is robust.

In our third empirical study, we test three hypotheses by investigating the impact of suboptimal portfolio allocation on investor protection. We use four different proxies of investor protection. In line with the theory, the research findings in Chapter 8 demonstrate that home bias leads to weak investor protection. This is explained by the market being segmented and dominated by domestic institutional investors who tend to be corporate insiders. This is consistent with the suggestions provided by Errunza (2001) that home bias leads to weak investor protection. We also find that foreign bias improves country level investor protection standards. This is characterised by foreign institutional investors from well governed countries, exporting good governance to the host countries. We perform a sensitivity test which confirms that our main result is robust to endogeneity, and not sensitive to different data sources of investor protection measures. The estimated coefficients remain statistically significant and maintain the same sign in all the regressions we run.

1.7 The Structure of the Thesis

The thesis is organized in nine chapters. In Chapter 1, we introduce the thesis, research importance and motivation, research objectives, main research findings and the structure of the thesis. Chapter 2 offers a background and literature review of the study. Chapter 3 presents the research questions and hypotheses' development. Chapter 4 reports the data sources used in the research. Chapter 5 discusses the research methods. Chapter 6 presents the analysis and the findings of the first research question, the sub-optimal portfolio allocation impact on cost of capital. Chapter 7 presents the research findings of the second

research question i.e. the impact of home, foreign, and country bias on stock market development. Chapter 8 reports the research findings for the third research question i.e. the impact of home and foreign bias on investor protection. Chapter 9 summarises and concludes the research. It also provides the limitations, policy implications and recommendations for future studies.

Chapter 2: Background and literature review

2.1 Introduction

This chapter will review and discuss related existing literature. The chapter is organised as follows: Section 2.2 offers a general background to the study; Section 2.3 provides a literature review of the flow of foreign portfolio investment in relation to cost of capital; Section 2.4 reviews flow of foreign portfolio investment in relation to stock market development; Section 2.5 reviews and discusses flow of foreign portfolio investment in relation to stock market in relation to investor protection; Section 2.6 reports gaps in the existing literature; Section 2.7 summarizes and concludes the chapter.

2.2 Background

In this section, we present the general background of the thesis. We mainly report factors that influence foreign portfolio investment. For instance, we provide a literature review on portfolio theory, the case for international portfolio diversification, stock market liberalisation and integration, stock market segmentation and, finally, the phenomenon of home and foreign bias.

2.2.1 Portfolio theory

Markowitz (1952) developed modern portfolio theory, based on a standard mean-variance framework where he demonstrates how an optimum portfolio can be achieved, based on expected return, variance and the covariance of securities' return. In his publication in 1952 "Portfolio Selection," Markowitz applied mathematical methods to explain that the risk of a portfolio reduces by adding securities that are less than perfectly correlated in a portfolio. In other words, he shows that the risk of a portfolio reduces when two assets that are not perfectly correlated are added to a portfolio. Due to the possibility of portfolio risk reduction through diversification, he argues that portfolio risk measured by variance, will not only be influenced by the individual variances of the return on the assets, but also by the degree of the covariance of all the assets. The risk of an asset does not significantly depend on the risk of each individual asset, but on the general contribution of each asset to the portfolio risk. The "law of large numbers" cannot be entirely applied to the diversification of risk in a portfolio choice because, in practice, the returns on individual assets could be correlated. This indicates

that portfolio risk, generally, cannot be eliminated, irrespective of the different types of securities added to a portfolio.

According to ICAPM, a country's risk premium depends on the covariance of that country's market portfolio return with the world market portfolio return. If the covariance with world stock markets is high, then the market portfolio of the country is risky from the perspective of the global markets. The capital asset pricing model (CAPM), suggests that the expected return on a security, is equal to the risk-free rate plus a risk premium. In an efficient market, efforts to reduce the risk level of the portfolio by simply adding less-risky investment also reduces the expected return of the portfolio. International portfolio diversification does not reduce the expected return of the portfolio; however, diversification lowers the risk of the portfolio. Solnik and McLeavey (2004) argue that international portfolio diversification improves the sharp ratio, due to international reduction of unsystematic risk. The next section explains why investors should construct an internationally diversified portfolio.

2.2.2 The case for international portfolio diversification

The fundamental argument for international portfolio diversification is that it assists foreign investors to reduce unsystematic risk without suffering any decline in returns.⁷ The expansion of investment opportunities through international portfolio diversification enhances the risk adjusted performance of a portfolio. The potential gains from international portfolio investment provide the opportunity for international investors to partake in the economic growth of other countries. Different countries have different opportunities for real economic growth as countries have different social, economic and political development. Foreign equity investors will benefit if there is higher return and lower correlation between the domestic and foreign securities. Investment in countries with higher economic growth, particularly in emerging markets, will assist international investors to obtain higher returns on investment.

⁷ With regard to international diversification, systematic risk relates to risks that affect all countries, such as the 2008 global financial credit crisis. Unsystematic risk relates to a particular country, for instance, governance issues.

Studies suggest that foreign investors can therefore, participate and enjoy high economic growth opportunities in emerging countries.⁸ Bartram and Dufey (2001) argue that, there is high volatility in emerging markets; however, due to the low correlation with developed market returns, emerging markets significantly contribute to the reduction of unsystematic risk of a portfolio. When international investors construct portfolios that include developed and emerging markets, the risk associated with the portfolio reduces without any reduction of the portfolio returns. Increasing the number of securities in a portfolio reduces a portfolio's unsystematic risk, but the systematic risk remains unchanged as the securities return will be impacted by sets of factors that are common to all the markets. The variance of a portfolio will significantly be influenced by the degree of international diversification. Odier et al. (1995) examine the risk-return characteristics of emerging markets relative to developed markets. They find evidence of substantial higher returns in emerging markets; however, the returns are associated with higher levels of market volatility. They show a correlation of 0.31 between returns of emerging markets and the world index of developed markets, implying that emerging markets might provide better benefits of diversification than a portfolio solely invested in developed markets.

International portfolio diversification shows that, other things being equal, when there is a low correlation between international stocks markets, it presents an opportunity to international investors to reduce the volatility or the unsystematic risk of the portfolio. This implies that, a low correlation between international stock markets provide, to an active investor, the opportunity to gain higher returns. An experienced and skilful investor can potentially adjust the international asset allocation of global portfolio towards the stock market with higher returns. Solnik and McLeavey (2004) note that international stock markets do not move up and down together as there are variations in macroeconomic conditions in different countries. The case for international portfolio diversification has two constituents. First, the potential benefits of risk reduction by keeping international diversified securities. Second, the potentials of an added foreign exchange risk diversification. As investors add more securities to the portfolio, the risk of the portfolio reduces substantially at the initial stage, then asymptotically towards the level of systematic risk of the market. A

⁸ Several Eastern European countries, Latin America, Asia, and Middle East are classified as emerging countries due to their low income and high economic growth rate.

well-diversified domestic portfolio will have a beta tilting towards 1.0. Companies within the same country tend to be affected by the same macroeconomic conditions.

Apparently, there is a high correlation between firms within the same country. As a result, stock prices of firms within the domestic market usually move up and down together because they encounter similar domestic macroeconomic factors such as monetary announcements, interest rate movement, budget deficits and the national GDP growth, which leads to a positive correlation amongst firms within the same country. A foreign stock market provides opportunities for domestic investors to spread the risk in their portfolio as the domestic country and the foreign countries face different macroeconomic conditions, the stock market return of the two countries will not be highly positively correlated, therefore domestic investors can diversify away the local market risk.

When a portfolio is well diversified, the variance of the portfolio return relative to the variance of the market's return reduces to the level of systematic risk. This implies that international portfolio market risk is lower than that of a domestic portfolio. This typically happens when the returns on domestic stocks are not closely correlated with the returns of foreign stocks. The potential benefits from international diversification should induce investors to participate in foreign securities. The inclusion of foreign securities reduces portfolio risk for a given level of return. There are several theoretical models and empirical studies that provide evidence of benefits from international portfolio diversification.

Driessen and Laeven (2007) find the benefits of international portfolio diversification are larger in developing countries, relative to developed countries. They argue that emerging countries are associated with high investment risk; however, investors from emerging countries can engage in risk sharing by including securities from developed countries. Studies suggest that investors from developing countries benefit most from international diversification, relative to investors from developed countries. A previous study by Huberman and Kandel (1987), Bekaert and Urias (1996) show the potential substantial cost investors from developing countries can incur if they fail to diversify internationally.

Contributing to the work of Markowitz (1952), Evans and Archer (1968) demonstrate empirically that portfolio diversification reduces portfolio risk. Grubel (1968) extends the portfolio theory developed by Markowitz, by showing the effects of a portfolio that includes

long-term assets and claims dominated in foreign currency. He finds that international portfolio diversification offers gains to investors through risk reduction and benefits from growth in other countries. He demonstrates empirically by employing data of 11 major stock markets on ex post-realised rates of return from investment. He further shows that international diversification not only depends on interest rate differential but also on the growth rate in total assets holdings in two countries. Studies in the US with regard to benefits of international portfolio diversification by Huberman and Kandel (1987), Bekaert and Urias (1996), De Roon et al. (2001), provide further support that US investors can obtain substantial gains from international portfolio diversification through increased risk sharing. Other studies suggest that the large size of the global market capitalisation and integration provide justification for international portfolio diversification.

The global stock market has experienced substantial growth since 1970. The New York Stock Exchange, which in 1970 accounted for 60 per cent of the world market capitalisation, was the most significant market in the world (Solnik and McLeavey, 2004). Towards the end of 2001, the US stock market capitalisation was around \$25 trillion which stood at 50 per cent of the world market capitalisation. Solnik (1974) employs weekly data for the period 1966 to 1971 of a large sample of individual stocks from seven European countries and the US. He shows the significant gains that can be achieved by investors through portfolio diversification in foreign and domestic stock markets. By holding a well-diversified equity portfolio, investors combine assets from countries to reduce portfolio risk. When more securities from several countries are added to the portfolio, the less the portfolio is affected by the poor performance of one particular country. The entire portfolio risk will be influenced by not only the number of securities included in the portfolio, but in addition, by the riskiness of each individual security and the extent to which they correlate with each other. For instance, a portfolio of 20 securities from the same country, experience less diversification benefits than a portfolio that includes securities selected from 20 countries. Solnik (1974) suggests that American portfolio managers can improve diversification benefit by selecting stocks listed on the New York and American stock exchanges instead of holding equities from the 30 Dow Jones Industries. Furthermore, he demonstrates that an internationally diversified portfolio is associated with less risk than a domestic dominated portfolio because, as diversification increases, portfolio risk declines.

Bartram and Dufey (2001), show the potential benefits that investors can derive from international portfolio diversification. They demonstrate that international portfolio investors can experience considerable gain through (i) participation in the growth of foreign markets; (ii) hedging of investors' consumption; and, (iii) diversification effects. Fletcher and Marshall (2005) employ the Bayesian approach of Wang to examine the benefits of international portfolio diversification for UK investors, between January 1988 and 2000. They find substantial increase in Sharpe and certainty equivalent return performance via portfolio diversification. Dimson et al. (2002), show that international portfolio investment offers domestic investors the opportunity to reduce portfolio risk. Examining the importance of international portfolio diversification, Hentschel and Long (2004) find considerable benefits to investors in developed equity markets. Li et al. (2003) suggest that even when short sales constraints are imposed, investors still benefit significantly from international portfolio diversification. Shawky et al. (1997) offer a comprehensive review of issues relating to international portfolio diversification. They argue that international portfolio diversification provides reasonable means in reducing portfolio risk, without reducing equity returns.

Lessard (1976) employs multivariate analysis on a group of Latin-American countries. He finds that international diversification consistently reduces portfolio risk. Eun and Resnick (1994) analyse the gains from international portfolio diversification from a Japanese and US perspective. They employed monthly return data for national bond and stock market indices for the period January 1978 to December 1989. They conclude that the potential gains from international diversification are much greater for US investors than for Japanese investors. Additionally, Japanese investors gain more from international diversification through increased international risk sharing, while for US investors, the gains accrue, not so much in terms of lowering risk, but in terms of higher returns. Mayshar (1979) maintains that when investors hold securities concentrated in a relatively few undiversified portfolios, the securities own variance or standard deviation will substantially affect its required equilibrium returns. Bekaert and Harvey (1995) provide evidence on the variations of the degree of the integration of emerging equity markets over time. They show the potential diversification benefits that can be achieved from investing in emerging equity markets. In spite of the theoretical models and empirical evidence of the benefits of portfolio diversification, studies suggest that the gains from international portfolio diversification to a large extent depend on the correlation between the domestic and foreign markets.

Levy and Sarnat (1970) argue that even though Markowitz (1952) provides a positive reason for diversification of risky assets, the extent to which portfolio diversification reduces portfolio risk, depends on the correlations across stock markets. If there is a positive high correlation amongst stock markets, no amount of diversification can reduce the portfolio risk. Portfolio diversification's theoretical and practical importance can be experienced in the sense that, if a country's stock market returns are highly but not perfectly correlated, then there is an opportunity to reduce risk through diversification. When there is a relatively less degree of positive correlation between economies, it implies that there is the opportunity for portfolio risk reduction through international diversification. Markowitz (1952; 1959) suggests that as long as correlations of returns between markets are not perfect, it offers a necessary but not sufficient condition for portfolio diversification.

Studies by Grubel (1968) and Lessard (1973) find that the correlation between international equity markets are low and, therefore, offer the opportunity to diversify portfolios to reduce risk. In a related study on international portfolio diversification, Errunza et al. (1999) argue that investors can gain from international portfolio diversification through investment in multinational companies, American depository receipts, and country funds that trade in the USA. Harvey (1995) demonstrates that the correlations across emerging markets are low, and thereby, provides opportunities for domestic investors to reduce portfolio risk beyond what they could achieve if they had kept only domestic equity. However, in recent times, international stock market linkages have experienced increased correlations between stock markets and therefore reduced the benefits of international diversifications.

A study by Goetzmann et al. (2002) show the seemingly increasing correlations amongst stock markets towards the end of the 20th century, which have resulted in the reduction of diversification benefits. Rajan and Friedman (1997) examine the impact of country risk on portfolio choice, in the context of the Markowitz portfolio selection model. They conclude that international portfolio diversification opportunities exist and that market segmentation plays a substantial role in the generation of those benefits. Nevertheless, as global markets become less segmented over time, international markets will become more highly correlated and the benefits of international diversification may reduce. Heston and Rouwenhorst (1994) re-examine industry versus country diversification by investigating the behaviour of individual shares from Europe. They conclude that country-specific factors are more important than industry factors when explaining diversification gains.

2.2.3 Stock market liberalization and integration

The previous section demonstrates the benefits that investors can derive from international portfolio diversification. This section provides a literature review on stock market liberalisation and integration. Stock market liberalisation is whereby a country removes investment restrictions and relaxes capital controls on equity investment to attract foreign portfolio investments to promote economic growth. Essentially, the national government allows foreign investors to hold domestic companies' equities without restrictions, and domestic investors are simultaneously allowed to purchase securities from foreign countries.

The last four decades have witnessed both developed and emerging countries liberalising their financial markets under the expectation of faster economic growth. There are several forms of financial liberalisation which include capital account liberalisation, banking sector liberalisation, and stock market liberalisation. Theories of stock market liberalisation suggest that the removal of cross-border investment restrictions will lead to global stock market integration.⁹

Existing studies, for instance, Bekaert and Harvey, (2003) suggest that stock market liberalisation will lead to stock market integration. In finance, stock markets are regarded as integrated if financial assets of similar risk have identical expected return irrespective of the country of domicile. The expected return of an integrated market is determined by the covariance of the domestic country index return with the world market index return and possibly with currency deposit rates. In an integrated stock market, common rewards are associated with risk exposure. More significantly, the rewards for risk are not important in an integrated market, as systematic risk is common to all integrated countries. The quest for a deeper understanding of stock market liberalisation and integration has become a major research and policy issue. Generally, evidence suggests that the outcome of stock market liberalisation and integration depends on factors such as the pace of implementation, the sequence of the liberalisation, the degree of macroeconomic stability before, during and after the liberalisation period, and the institutional structures of the liberalising economy.

Studies provide several factors that motivate countries to liberalise and integrate the domestic stock market with the global stock market. Existing studies suggest that political and

⁹ Integration means financial stock markets are globalized or interlinked with each other and are affected by world common factors.

economic factors influence a government's decision to liberalise the stock market. Contributing to factors that influence national governments to liberalise the stock markets, Kaya et al. (2012) show that political and economic factors influence government decision to liberalise the stock market. Among the economic factors, the level of financial development, structure of the economy, the quality of investor protection and the level of the government's involvement in the economy play significant roles towards the likelihood of stock market liberalisation. Political factors take account of the influence of foreign governments and agencies through the allocation of foreign financial aid. The economic factors include the level of financial development, the structure of the economy, the quality of investor protection, and the level of the government's involvement in the economy.

It is widely recognised that stock market liberalisation is an integral part of financial sector development. Theory and empirical evidence suggest that countries can derive substantial economic benefits by allowing foreign investors to access the domestic stock market. Errunza, (2001) postulates that stock market liberalisation via flow of foreign investors will lead to a decline in cost of capital, improvement in stock market development, and better country level investor protection standards.

Financial liberalisation and integration helps international investors to diversify their portfolio by bidding up the prices of local securities and avoiding sectors that are inefficient. Generally, prior to financial liberalisation, the domestic price of risk (variance) exceeds the global price of risk (the covariance). Therefore, we expect equity risk premium to decline when a completely segmented country liberalises its stock market. Holding expected future cash flows constant, the fall in the equity risk premium will lead to a permanent decline in the aggregate cost of capital, increase stock price valuation and subsequently stimulate investment and GDP growth. Studies by Harvey (1995), and Bekaert and Harvey (1995, 1998), demonstrate that stock market integration have a positive influence on stock prices.

Financial theory predicts that when a country liberalises the stock market, foreign investors will participate in the liberalised country's stock market. The positive impact of liberalisation works through a reduction of cost of capital, real interest rates and unsystematic risk as the relevant benchmark for pricing risk changes from the local market index to a world market index. Consequently, companies in the liberalising countries will experience increase in stock prices and the rate of capital accumulation will rise. Existing studies suggest that stock

market liberalisation can play an important role in increasing economic growth. In principle, stock market liberalisation offers a channel for companies to raise capital at a lower cost than from the banks. Due to the fact that, although, interest on loans borrowed from the bank can be deducted on the company's tax return, the debt involves borrowing money to be repaid, plus interest.

Kim and Singal (2000) argue that, the opening of a country's stock market to foreign investors is associated with an increase in stock price indexes in the liberalising country. Henry (2000b) shows that stock market liberalisation is related to an increase in a country's overall level of private investment. Chari and Henry (2004a, b) find that an increase in a firm's investment correlates positively with expectations of increase in its future earnings after financial liberalisation. Gupta and Yuan (2004) find a positive relation between liberalisations and growth using alternative methodologies. Schmukler and Vesperoni (2003) provide evidence of a decrease in a firm's long-term debt following liberalisation. In a sample of 1,141 firms from 28 countries, Mitton (2006) studied the impact of financial liberalisation on five standard measures of operating performance; growth, investment, profitability, efficiency, and leverage. Though the study avoided methodological problems of earlier studies, he provides further evidence that opening stock market to foreign investors is associated with a broad range of improvements in operating performance for the liberalising firms. Galindo et al. (2001), show that stock market liberalisation improves capital allocation efficiency of firms in 12 developing countries. Levine and Zervos (1998) find that stock markets become more liquid following liberalisation in a study of 16 countries. Stulz (1999a) argues that financial globalisation allows financial resources to flow to areas where productivity is efficient, high, and offers countries the opportunity to enjoy the benefits of their comparative advantages.¹⁰

Studies suggest that financial liberalisation may improve the supply and allocation efficiency of funds for investment. For instance, it is often thought that financial liberalisation improves the allocative efficiency of savings. However, Laeven (2003) argues that the consequence of financial liberalisation on the supply of funds for investment is theoretically ambiguous. In a

¹⁰ It is reasonable to expect foreign investors to invest in countries where there are weak regulation poor investor protection standards as a result of benefits of international portfolio diversification and high expected returns especially in emerging countries. The foreign investors will seek to invest in these countries with poor investor protection standards with the hope of higher returns and also influencing investor protection standards.
segmented market, McKinnon (1973) and Shaw (1973) argue that an interest rate ceiling, which is another common feature of repressed financial systems, distorts the allocation of credit and may lead to under-investment in projects that are risky but have a higher expected rate of return. Obstfeld (1994) shows that stock market liberalisation, increases the pool of funds allocated towards risky investment projects due to improved risk sharing.

Laeven (2003) estimates a dynamic investment model using annual panel data for 394 listed firms in 13 developing countries for the period 1988 to 1998. He finds that financial liberalisation affects small and large firms differently. Small firms in developing countries gain from financial liberalisation, but in the case of large firms, the allocation efficiency benefits from financial liberalisation is rather offset by the adverse effects of losing access to preferential credit. Galindo et al. (2002) show a positive correlation between financial liberalisation and improvements in allocative efficiency of investment in some countries using firm-level data from 12 developing countries. Laeven (2003) argues that, successful financial liberalisation requires better political treatment of large and well-connected firms. In recent times, Abiad et al. (2008) find robust evidence that capital account liberalisation and integration improves capital allocation efficiency across countries. Although, countries have removed the implicit and explicit investment constraints global equity investors' face towards their investment, nonetheless, due to the phenomenon of home and foreign bias, it is less likely that financial liberalisation will result in full integration of the global stock market. Therefore, there is a certain degree of segmentation in some countries.

The financial market is segmented when there is a cross-border regulatory restriction on capital outflows and inflows. The restrictions could include discriminatory taxes or separate legal status given to foreign investors in terms of ownership restrictions. Furthermore, prevalence of home bias by domestic investors can also segment a liberalised market. Existing studies show that investors exhibit varying degrees of home and foreign bias for several reasons.¹¹ In a completely segmented market, domestic investors mainly hold securities from their home country without holding a world diversified market portfolio. As a result, resident investors will have to bear all the risk associated with the investment. This subsequently makes the equity risk premium embedded in a segmented market aggregate valuation, proportional to the variance of the country's aggregate return. Consequently,

¹¹ Cross-country variations in the quality of corporate governance mechanisms, accounting standards, legal institutions, stock market regulations, investor protection, information asymmetry, direct and transaction cost.

securities are priced off at the local market return. In a fully segmented market, the country's expected return is determined solely by the variance of return in that market multiplied by the price of variance. Because risk is shared locally, the price of variance will largely depend on the weighted relative risk aversions of investors in the domestic country.

When a financial market is segmented, domestic investors are restricted from purchasing shares in foreign countries, which includes capital mobility and foreign exchange controls. Similarly, foreign investors encounter investment restrictions in the form of barriers on foreign equity investment.¹² Other forms of restrictions include imposing limits on the percentage of a firm's shares that foreign investors can purchase. Domowitz et al. (1998) show that stock market segmentation reduces the gains that investors can achieve from risk sharing via international diversification. In a segmented market, domestic investors are mainly prevented from holding an international diversified portfolio, which increases cost of capital. National governments sometimes place investment restrictions to ensure that it attracts foreign investment, whilst the control of companies remains with domestic citizens. This is to avoid conceding ownership control to foreign investors.

In a segmented market, companies issue restricted shares that can only be purchased by domestic investors and unrestricted shares where foreign investors can buy them. Studies show that unrestricted shares, trade at a premium price relative to restricted shares. For instance, Hietala (1989) on Finland, Lam et al. (1990) on Singapore, Bailey and Jagtiani (1994) on Thailand, Stulz and Wasserfallen (1995) on Switzerland, and Domowitz et al. (1998) on Mexico. They show that there is an economically significant stock price premium for unrestricted shares, relative to restricted shares.

Studies suggest that market segmentation leads to underdevelopment of stock markets as a result of stock market illiquidity, weak regulatory mechanisms and inadequate disclosures by companies. Errunza and Miller (2000) argue that market segmentation leads to higher cost of capital as the funding options available to companies are internally generated. Particularly, in the case of emerging markets, it is only government agencies and large companies that have access to the stock market. Market segmentation has a negative effect on the size and liquidity of a stock market. Segmentation makes stock markets become very small and

¹² Eun and Janakiramanan (1986) provide ownership restrictions in 16 countries.

inactive as a result of only local investors holding shares in the domestic companies. Errunza and Losq (1985) suggest that restricting foreign investors from buying securities in a segmented country will cause securities to trade at a premium, to the extent that they are spanned by investors' home market securities. Errunza and Losq (1989) argue that controlling capital flows, makes it impossible for investors to construct a world market portfolio to reduce portfolio risk. Therefore, the removal of investment barriers through financial liberalisation will lead to an increase in securities' aggregate market value.

2.2.4 Home and foreign bias phenomenon

In spite of financial liberalisation, which has led to the removal of direct and indirect legal investment restrictions, existing studies show that investors still deviate from holding internationally diversified portfolio, due to the home and foreign bias phenomenon (see Erruza and Losq, 1985; Cooper and Kaplanis, 1994; Bekaert and Harvey, 1995; Errunza, 2001; Warnock, 2002; Bekaert, 1995; Bekaert and Harvey, 2003; Dahlquist et al. 2003; Chan et al, 2005; Gelos and Wei, 2005; Hunter, 2006).

Home bias relates to higher inward investment, i.e., over weighting of the home market relative to the suggestion of the ICAPM benchmark. Similarly, foreign bias refers to the tendency of over or under weighting of foreign markets against the suggestion of ICAPM benchmark allocations. A number of studies (see Stapleton and Subrahmanyam, 1977; Errunza and Losq, 1985; Janakiramanan, 1986; Stulz, 1999b; Bekaert and Harvey, 2003) conjecture that the prevalence of home bias in portfolio allocations will segment financial markets from the global markets, and therefore, will reduce international risk sharing benefits. This implies that a higher degree of home bias should be associated with a higher cost of capital and vice versa.

Existing studies provide theoretical and empirical explanations for home bias, which include information asymmetry, differential accounting standards, poor investor protection standards, political risk, currency risk and economic development. Errunza (2001) notes that domestic investors are better informed about their local markets than foreign investors. Therefore, such informational asymmetry leads to home bias, which, in turn, leads to higher cost of equity capital. Chan et al. (2005) argue that emerging markets are characterised with poor information disclosure, therefore, investors from developed countries, experience

informational disadvantage and tend to underweight foreign investment. Other explanations for home bias include: Warnock's (2002) high transaction cost, Dahlquist et al. (2003) differences in corporate governance, Errunza and Losq's (1985) barriers to international capital flows, Baxter and Jermann's (1997) hedging motives, and Cooper and Kaplanis' (1994) departure from purchasing power parity. Other reasons also include the level of protection given to minority investors (see Bekaert and Harvey, 1995; Errunza, 2001; Hunter, 2006), Gelos and Wei (2005) country level transparency, Bekaert (1995) separate legal status given to domestic and foreign investors. Home bias segments stock markets and thus has implications on cost of equity capital, stock market development, and investor protection.

Few studies have examined the implications of home bias (see Lewis, 1999; Karolyi and Stulz, 2003; Chan et al. 2009; Lau et al. 2010). Stulz (1999a) demonstrates, theoretically, that home bias segments markets and thereby reduces firm value through an increase in cost of equity capital. Other studies provide empirical evidence to support the argument that home bias lowers risk sharing (see Errunza and Losq, 1985; Janakiramanan, 1986; Stapleton and Subrahmanyam, 1977). If foreign investors underweight their investment in the domestic market, risk sharing between the foreign and local investors reduces and therefore lowers cost of capital benefit of financial liberalization (Carrieri et al. 2006).

We provide the background to the study in this section. In the subsequent section, we offer a literature review that relates to the empirical studies in chapters 6, 7, and 8. In Chapter 2.3, we provide a literature review for the first empirical study in chapter 6. Similarly, in Chapter 2.4, we offer a literature review for the second empirical study in chapter 7. Finally, in Chapter 2.5, we provide a literature review for the third empirical study in Chapter 8.

2.3 Literature review for Chapter 6: Sub-optimal portfolio allocation and cost of capital

This section provides a literature review for Chapter 6. We are motivated to undertake the empirical studies in Chapter 6 as a result of the importance of cost of capital. Additionally, current studies mainly use event studies to examine the effect of financial liberalisation on cost of capital. A review of the existing literature demonstrates that there are limited studies that directly examine the effects of sub-optimal portfolio allocation on cost of capital.

2.3.1 Importance of cost of capital

Cost of capital is fundamental to a variety of corporate decisions, and is generally conceived to be the expected return investors demand for holding securities in a company. It is used to determine the hurdle rate for investment projects. Additionally, the market applies it as the discount rate to companies' expected cash flows to derive the stock price. We define cost of capital to be the minimum expected rate of return required by equity investors for providing capital to the firm. Existing studies have extensively used average historical market return, log country credit risk rating, country equity risk premium, dividend yield, and implied cost of capital to proxy for cost of capital. When CAPM is used to compute expected return on equity, it comprises of the risk-free rate of interest and risk premium. Cost of equity capital is sometimes referred to as the expected cost of equity capital because it is a forward-looking concept. For publicly traded companies, stock price is observable, but the market's expectations of future cash flows are not. As a result, neither component is directly observable from realised prices or returns.

Streams of research provide factors that affect cost of capital. For instance, Hail and Leuz (2009) examine cost of capital effects of changes in growth expectations around US crosslistings. They find that cross-listing on US exchange reduces cost of capital, due to the improvement of investor protection standards for outside investors. Bhattacharya and Daouka (2002) analyse the impact of insider trading laws on cost of capital and conclude that, introduction of insider trading laws do not change cost of capital but, after first prosecution, leads to a decrease in cost of capital. Hail and Leuz (2006) provide compelling evidence that a country's legal institutions and securities regulations' effectiveness relate to cross-country variations in cost of capital. The optimal investment policies of corporations are directly affected by cost of capital. A good understanding of the precise effect of home and foreign bias on cost of capital is important for managers, governments and equity investors for a number of reasons. First, managers require accurate estimates of cost of equity capital for a company's capital budgeting decisions and project evaluation. When firms consider whether to undertake an investment project, the present value of the project is given by the expected cash flows they expect to receive from the project, discounted at their required rate of return. A decrease in cost of capital makes all projects whose return has a positive covariance with the return on the market portfolio more advantageous, especially if the project has a positive covariance with the return on the market portfolio. Studies suggest that firms from segmented markets that can access an international capital market will experience a lower cost of capital via enhanced risk sharing. A reduction in cost of capital will cause net present value (NPV) to increase and will stimulate corporate investment to maximise shareholders' wealth within the economy. Alternatively, when cost of capital rises, firms will find fewer projects yielding returns higher enough to warrant new investment.

Second, governments and policy makers are concerned with the level of cost of capital in their country. Countries will immensely benefit from a fall in cost of capital as there will be an increase in physical investment following a large flow of foreign portfolio investment as projects which previously had negative NPV prior to the financial liberalisation, will turn into positive NPV. The effects of positive NPVs will generate jobs, result in capital flight reversal and higher growth rates of outputs, and have a broader positive impact on economic welfare than a financial windfall to domestic equity investors. Companies' investment in new assets and capacity is determined by whether they anticipate to generate higher return on those investments than the cost attached to the investment capital. If equity risk premium increases, cost of equity capital will subsequently increase, leading to less investment in the overall economy and lower economic growth.

Third, cost of capital is important to equity portfolio investors. It is useful in determining the fair price of securities as the discounted value of all future dividend payments over the holding period of the securities. There is an inverse relationship between cost of equity capital and firm valuation or share price. If the required return demanded by investors is high, investors will offer to pay low price per share. A fall in cost of capital will cause an increase in share price and maximise shareholders' wealth.

2.3.2 Financial globalization and cost of capital

Both developed and emerging countries liberalised their stock market in the 1970s and 1980s respectively by allowing unrestricted inflow and outflow of foreign equity investment. In equity market liberalisation, countries remove the implicit and explicit investment barriers investors' face towards investment, and therefore, foreign investors can buy or sell local securities without restrictions. Similarly, local investors can also buy or sell foreign securities. Studies suggest that when a country liberalises its stock market, equity risk premium of the country will reduce, due to the improved risk sharing between domestic and foreign investors. Furthermore, foreign investors will demand better corporate governance.

The international capital asset pricing model suggests that investors should hold an optimally world market capitalisation weight to benefit fully from diversification (see Solnik, 1974; Adler and Dumas, 1983; and Lewis, 1999). However, in spite of financial liberalisation, the global financial market has some fragments of segmentation, as investors in both developed and emerging markets exhibit varying degrees of home and foreign bias. Prevailing studies show that domestic investors generally deviate from holding optimal global portfolio weight as a result of the home bias phenomenon, which segments a market from the rest of the world. Recent studies by Chan et al. (2005) show the persistence and prevalence of home bias across 48 developed and emerging countries.¹³

Bekaert and Harvey (2000) find that financial liberalisation reduces cost of capital, even though the reduction is smaller than the theoretical predictions, which may be attributable to home bias. For other studies that have examined the implication of home bias (see Lewis, 1999; Karolyi and Stulz, 2003; Chan et al. 2009; Lau et al. 2010). Stulz (1999b) demonstrates, theoretically, that home bias can reduce firm value through an increase in cost of equity capital. Previous studies mainly investigate factors that determine home and foreign bias. For instance, French and Poterba (1991) empirically examine and provide explanations for home bias, which includes domestic investors' optimistic high expected return on domestic equity market than foreign markets, capital controls which involve explicit limits on cross-border investment, transaction cost and institutional factors. However, this study examines the implications of home and foreign bias on cost of capital.

¹³ See chapter 2.7 for theoretical and empirical reasons for home bias.

Segmented markets are associated with stock market illiquidity. This, therefore, increases transaction cost and limits the shares that firms can issue, as corporations can raise capital from few investors. The equity premium embedded in the aggregate valuation will be proportional to the variance of the country's aggregate cash flow when the country is segmented as a result of home bias. Several studies provide empirical evidence to demonstrate that the impact of globalisation on cost of capital is small due to home bias of domestic investors which reduces risk sharing. Errunza and Losq (1985), Eun and Janakiramanan (1986), and Stulz (1981b) show that cost of capital does not reduce much, owing to the indirect investment restriction explained by the home bias phenomenon which segments stock markets. Stulz (1999a) suggests that the impact of globalisation on cost of capital depends on the degree of home bias. As home bias falls, cost of capital correspondingly falls. The success of globalisation depends on how broad the investor base is in the domestic market.

Intuitively, when a country removes cross-border capital investment restrictions, foreign equity investors will take the opportunity to diversify and share investment risk globally, as prescribed by ICAPM to reduce portfolio unsystematic risk. The increase in global risk sharing reduces cost of capital, partly due to investors' expectations of a lower risk premium. This will, in turn, lower the discount rate use in selecting potentially acceptable investment opportunities, therein improving a firm's chances of accepting a higher number of profitable projects. A lower risk premium demanded by investors will translate into a lower cost capital for the firms. Consequently, investors will be expected to react positively because of higher expected future dividend pay-outs and growth in the market value.

Financial theory suggests that financial liberalisation will lead to financial market integration. Because when a country's government allows foreign investors to buy shares in the domestic stock market, foreign equity investors will take the opportunity to diversify investment risk globally. Adler and Dumas (1983) postulate that in an integrated stock market, the expected return of the country will depend on the covariance of the country's stock index return with the world market portfolio return, and probably with currency deposit rates. The increase in global risk sharing reduces cost of equity capital, partly due to investors' expectations of a lower risk premium and increased liquidity. This will lower the discount rate to improve a firm's chances of accepting a higher number of profitable projects. The theory of the international capital asset pricing model predicts that, sufficient flow of foreign investment to

a country decreases cost of capital due to improved risk sharing between domestic and foreign investors, especially if the correlations between a country's stock market and the world stock market is not positively correlated (see Stapleton and Subrahmanyan, 1977; Errunza and Losq, 1985; Eun and Janakiramana, 1986; and Stulz, 1999a, b). In a partially integrated market, the risk premium of the country depends on the weighted average of the covariance of the country's index return with world portfolio return and the country's variance with the degree of the country's market integration as the weight (Bekaert and Harvey 1995; de Jong and de Roon, 2005).

Stulz (1999b) shows that when a country opens its stock market to foreign investors, cost of capital declines, owing to the increase in risk sharing and good governance. Existing studies demonstrate the effect of home bias on security prices. For example, Chan et al. (2009) examine the effects of home bias on firm valuation via Tobin's Q at both country and firm level. They find that greater risk sharing between domestic and foreign investors, reduces the cost of equity capital and thereby increases firm valuation. Their study confirms earlier theoretical research by Stulz (1999a) who suggests that when a country opens its capital market to foreign investors, cost of equity capital declines through enhanced risk sharing, capital market efficiency, better legal system, shareholders' activism, improved corporate governance and adequate disclosures. Henry (2000a) uses data from 12 emerging markets and finds that equity valuation increases by two per cent in the eight month period before liberalisation. He concludes that stock market liberalisation decreases a country's cost of capital. In a related study, Bekaert and Harvey (2000) show, empirically, that cost of capital falls between 5 and 75 basis points after capital market liberalisation. They find insignificant evidence of changes in realised returns but show a substantial reduction in dividend yields from the pre- to the post-liberalisation period.

Fischer (2003), Rogoff (1999), and Summers (2000) argue that emerging countries should experience a decline in cost of capital, enjoy increased physical investment and GDP growth when they receive a large flow of foreign portfolio investment. Consistent with this, is evidence provided by Tesar and Werner (1998), Bekaert and Harvey (1998), and Errunza and Miller (1998), who document that the local price of risk (variance) exceeds the global price of risk (covariance). Therefore, equity risk premium should expectedly fall when a country's stock market becomes integrated with the rest of the world. Holding expected cash flow constant, the decline in the equity risk premium will lead to a permanent fall in the aggregate

cost of capital and subsequent revaluation of the aggregate equity price index. Kim and Singal (2000) find evidence that emerging markets' stock returns are abnormally high in the months leading to stock market liberalisation.

Other studies suggest that when firms from segmented markets gain access to the international market through ADR, the firms become integrated with the rest of the world and should therefore experience a fall in cost of capital. A study by Foerster and Karolyi (1999) demonstrate a rise in stock prices of companies from segmented markets that cross-list, via a fall in cost of capital. They find significant average excess returns of 19 per cent during the year before listing, and an additional 1.20 per cent during the listing week, but incur a substantial average decline of 14 per cent the year following the listing. The stock market beta, relative to its home market index on the average, declines dramatically from 1.03 to 0.74 per cent. However, the global beta, relative to the world market index, does not experience any significant change. Miller (1999) examines return around the announcement of the ADR programme and the return around the day when the actual listing take place. He finds a positive return both around the announcement date and around the listing date.

Errunza and Miller (2000) use a firm level data set of 126 firms from 32 countries to study the impact of American Depository Receipts (ADR) on cost of capital. They find firms in the sample experience substantial decline in realised abnormal returns. They attribute the reduction of 42.2 per cent to the introduction of ADR after controlling for confounding effects. In contrast, Bekaert and Harvey (1998) find mixed results of the effect of financial liberalisation on the cost of capital using market level data and after controlling for concurrent economic reforms. Henry (2003) argues that, capital stock growth rate increases by 1.1 per cent following liberalisation, from an average of 5.4 per cent per year in the preliberalization period, to an average of 6.5 per cent in the post liberalization period. He shows that developing countries' stock market liberalisation reduces cost of capital and subsequently leads to an investment boom. He concludes that in the face of compelling empirical evidence, arguments that financial liberalisation is not bringing about any real benefit, does not hold any truth.

Miller (1999) investigates the stock price impact of international dual listing. He analysed a sample of 181 firms from 35 countries that participated in a depository receipt programme over the period 1985 to 1995 using cross-sectional regression. Miller finds abnormal returns

during and around the liberalisation announcement date, and concludes that firms gain from listing shares outside their domestic market. In a more recent work by Edison and Warnock (2008), they use monthly data of equity flows from US to emerging markets between 1989 and 1999 and find that cross-listing of emerging market equity on the US exchange, results in a sharp short horizon inflows, whereas the reduction of capital control, results in inflows over longer horizons, mainly in Asia. In a recent study, Hail and Leuz (2009) use a sample of 40,000 firms' yearly observations from 45 countries over the period from 1990 to 2005, to empirically examine whether cross-listing in the US exchange reduces firms' cost of capital. They find convincing evidence that cross-listing on the US exchange, leads to a significant decrease in firms' cost of capital after controlling for traditional proxies for risk, analyst forecast bias and firm-fixed effects.

Bekaert and Harvey (1998) examine the impact of financial liberalisation on cost of capital using event studies. They find that an increase in foreign equity flow, leads to a lower cost of capital, large market size relative to GDP, and economic growth. Similarly, De Jong and de Roon (2005) investigate time-varying market integration and expected returns in 30 emerging markets. They find annual expected return, on average, decreases by 11 basis points. Lau et al. (2010) examine the impact of home bias on cost of equity capital. They find international differences in cost of equity capital are significantly associated with varying degrees of home bias from a sample of 38 countries. Their study shows that countries will experience substantial decline in cost of capital if home bias decreases.

Henry (2000a) finds that cost of capital falls when a country attracts sufficient portfolio investment as a result of financial liberalisation. He shows that countries, on average, enjoy 3.3 per cent abnormal returns per month in real dollar terms in the eight month period leading to stock market liberalisation, due to an increased risk sharing between domestic and foreign investors. The standard international asset pricing model (IAPM) theoretically predicts that countries will experience a lower cost of capital when it liberalises its stock market as a result of increased risk sharing between domestic and foreign investors.

When a country's stock market becomes integrated with the rest of the world through attraction of sufficient foreign investors, the country's equity premium will be proportional to the covariance of the country's aggregate cash flows with the world portfolio. Bekaert and Harvey (1995) suggest that emerging markets that are integrated with the rest of the world,

will experience market equilibrium valuation which incorporates an equity premium that lies in the region between the autarky and fully integrated premium. Prior studies (see Stulz, 1999a, b; Tesar and Werner, 1998; Bekaert and Harvey, 2000; Errunza and Miller, 1998) show that in a segmented market (home bias) the domestic price of risk (variance) is higher than the world price of risk (covariance). Bekaert and Harvey (2000) find that stock market liberalisation reduces aggregate dividend yield owing to the change in cost of capital, but is not attributable to the change in profits of companies or increase in cash flow.

The overwhelming expectation of stock market liberalisation is that countries will attract sufficient foreign investors, which will create competition amongst foreign and domestic investors. Foreign investors will come with financial and technological skills, which can help firms to effectively manage risk. Obstfeld (1994) shows that globalisation allows economies to concentrate more and embark on riskier projects as countries can undertake cross-border investment to diversify risk. Subsequently, aggregate investment increases as diversification reduces cost of capital.

Harvey (1991) provides support to the suggestion of international portfolio diversification using a sample of several countries. However, he finds an inconsistent return of the Japanese portfolio with ICAPM over this sample period because of the extremely large return of Japanese stocks in the 1990s. Chan et al. (1992) provide evidence in support of international portfolio diversification, by showing that the risk premium of the US market depends on the extent of correlation between US portfolios and international stock markets. DeSantis and Gerrard (1997) also provide support to ICAPM of international portfolio diversification. The key intuition of ICAPM is that, the risk premium of countries that are integrated with the global stock market depends on their covariance with the global stock market portfolio. Therefore, for financial globalisation to effectively reduce cost of capital, the shareholder base should be truly global without any home bias.

Other studies show that cost of capital declines, due to better governance after stock market liberalisation. Kim and Singal (2000) argue that foreign investors will demand greater transparency and better disclosure rules, which will enhance capital allocation efficiency. Foreign investors will also require accountability of management and better investor protection against controlling investors. This will subsequently reduce the risk associated with the investment and will thereby reduce cost of capital. Kim and Singal (2000) find that stock returns rise immediately after stock market liberalisation but subsequently fall. Henry (1998) demonstrates that countries that liberalise their stock market, experience higher revaluation of the domestic stocks as a result of the lower cost of capital.

2.3.3 Gaps in the cost of capital literature

Reviews of the existing literature show the prevalence of home and foreign bias in international portfolio allocation. However, there is limited empirical evidence of the impact on cost capital. Previous studies directly examine financial liberalisation and cost of capital. Liberalisation could coincide directly with macroeconomic reforms, and the impact on cost of capital may not necessarily be as a result of the financial liberalisation. Further, existing studies provide explanations for home and foreign bias without investigating their implications. The first empirical chapter examines the implications of sub-optimal portfolio allocation on cost of capital via home and foreign bias. We show that domestic and foreign investors play varying roles in influencing the cross-country cost of capital

The theory notes that relative to the ICAPM, as the magnitude of domestic bias in the equity portfolio allocations decreases, it facilitates the benefits of global risk sharing between foreign and domestic investors. Similarly, following the same ICAPM prescription, the increase in foreign bias towards a particular host country by foreign investors should also positively influence global risk sharing.

The increased international diversification of home investors (i.e. decreasing home bias) and foreign investors (i.e. increasing foreign bias) should integrate the host capital market into the world capital market. The benefits of such growing market integration should ultimately manifest itself in the reduction of cost of capital. This suggests that the cross-sectional and temporal variations in home and foreign bias should, in part, explain the differences in the cross-country cost of capital. To be specific, the empirical result in Chapter 6 shows that home bias is positively associated with a higher cost of capital. Similarly, we demonstrate that foreign bias has a negative impact on cost of capital.

2.4 Literature review for chapter 7: International portfolio investment and stock market development

2.4.1 Importance of stock market development

Existing studies mainly examine the importance of stock market development in relation to economic growth and development. The reasonable expectation of the phenomenon is that developed stock markets experience high liquidity and serve as a channel for investors to diversify risk. Countries that have well developed stock markets are associated with better flow of information regarding companies' activities and good governance. Furthermore, the existence of a well-developed stock market increases the pool of equity funding available for riskier investment projects, which is fundamental and a prerequisite for economic growth. For instance, several projects with high returns require long-run commitments of capital; however, investors are reluctant to relinquish control of their savings for long periods. Without a liquid stock market that assists savers to purchase and sell shares whenever they require access to their savings, less investment may occur in high return projects. Generally, developed stock markets contribute immensely towards capital accumulation and technological innovation. Studies show that countries with well-developed stock markets, experience less information asymmetry and financial resources are allocated to the most productive sectors of the economy (see Rousseau and Wachtel, 1998; Levine et al. 2000; Beck et al. 2000b; Rousseau and Sylla, 2001).

A growing body of literature shows that stock markets provide services that boost economic growth. For instance, Bencivenga et al. (1996), and Levine (1991) argue that stock market liquidity is important for economic growth. Levine and Zervos (1996) use cross-country sample data of 41 countries over the period 1976 to 1993 to empirically examine the relation between stock market development and long-run economic growth. They find that stock market development has a positive impact on economic growth. Cameron (1967) and McKinnon (1973) provide a conceptual description and empirical evidence of how financial systems affect economic growth.

Other studies document the impact of stock market development on economic growth (see Levine, 1997; Rajan and Zingales, 1998; Calderon and Liu, 2002). King and Levine (1993) show that the level of stock market development predicts economic growth, capital accumulation and productivity. Carlin and Mayer (2003) find a significant relationship

between financial systems and economic growth. Garrestsen et al. (2004) show a causal association between economic growth and stock market development. They find that a 0.4 per cent increase in stock market, leads to a one per cent rise in economic growth. Beck et al. (2006) find higher capital market development relates to economic growth. Bose (2005) provides a theoretical explanation to the positive relation between stock market development and economic growth. Claessens et al. (2006) use panel data to demonstrate that stock market development and internationalisation relate positively to stock market liberalisation and economic development.

Diamond and Verrecchia (1982), and Jensen and Murphy (1990) argue that an efficient stock market helps mitigate the principal-agent problem by influencing corporate control. Laffont and Tirole (1988), and Scharfstein (1988) show that a well-functioning stock market makes corporate takeovers possible to mitigate the principal-agent problem and promote efficient allocation of resource to spur economic growth. Stock markets promote the acquisition of information about firms (see Grossman and Stiglitz, 1980; Kyle, 1984; Holmstrom and Tirole, 1993). In larger and more liquid stock markets it is easier for an investor to obtain information about firms, which stimulates investors' interest to research and monitor firms.¹⁴

2.4.2 Foreign portfolio investment and stock market development

The removal of cross-border equity investment restrictions in several economies have led to an upsurge in cross-country equity investment. Studies suggest that flow of foreign equity investment via foreign institutional investors improves stock market development. This is explained by the fact that financial globalisation through cross-border portfolio allocation, enhances risk sharing, reduces cost of capital and subsequently improves the valuation of firms. Furthermore, foreign investors will require greater transparency and better disclosure practices which are important for the efficient allocation of financial resources. Foreign investors will also demand accountability of management and better investor protection to minimize risk of expropriation by managers and corporate insiders.

¹⁴ For literature of the impact of stock market development on economic growth and development (see Schumpeter, 1911; Gelb, 1989; Ghani, 1992; King and Levine, 1993; Atje and Jovanovic, 1993; Levine and Zervos, 1996; Levine and Zervos, 1998; Rousseau and Wachtel, 1998; Demirgüc-Kunt and Maksimovic, 1998; and Beck et al. 2000b).

Previous studies suggest that foreign institutional investors help improve the monitoring of management. They employ direct and indirect measures to influence management to adopt policies that improve shareholders' wealth, which includes threat of selling shares, and active use of ownership rights through voting. The participation of foreign institutional investors has the potential to reduce agency problems of managerial discretion, caused by the separation of ownership and control. Existing studies demonstrate that foreign institutional investors serve as a credible mechanism for transmitting information to domestic investors. Davis and Kim (2007) and Gillan and Starks (2003) show that foreign institutional investors are known for taking a more active role in promoting changes in corporate governance practices as they are independent of management. Ferreira et al. (2010) conclude that international portfolio investors serve as a mechanism to enhance international financial integration. Intuitively, as the demand for capital increases, countries will be motivated to improve investor protection and corporate governance standards to attract foreign equity investment. Poor corporate governance standards deter investors' participation of the stock market. Chan et al. (2005) show foreign investors hold less investment in countries that have poor investor protection, weak disclosure requirements, and poor securities' regulations.

The participation of foreign investors in the domestic market will increase stock market liquidity because foreign investors from well governed countries will demand better investor protection, which will reduce risk and improve stock market development. Studies show that foreign institutional investors have a positive impact on the Chinese stock market development. They demonstrate that through informational frictions, the participation of foreign institutional investors in the domestic stock market, influences state-owned enterprises' liquidity. The finance literature provides two main channels through which foreign investors could influence stock market development. First, through liquidity or the altering of the levels of trading activities on the stock market; and, second, by changing the informational environment on the market. Stoll (2000) refers to the two mechanisms as real and informational friction effect, respectively. He suggests that the participation of foreign investors can influence the real frictional element of market liquidity by altering the level of trading activity of the stock market. Khanna and Palepu, (2000) argue that foreign investors can monitor corporate management better than domestic investors and produce more timely and accurate forecasts than local analysts.

Innately, the presence of foreign investors will elicit more trade and enhance trading activities which reduces real friction cost by spreading fixed real cost over more trades. Concerning the informational impact on liquidity, it is arguable that foreign investors are better informed traders.¹⁵ Stulz (1999a, b) suggests that the liquidity of the stock market increases when international institutional investors increase their participation in the domestic stock market, because foreign institutional investors will demand better information disclosure and engage in high trading activities.

Existing studies have examined the impact of financial liberalisation on stock market liquidity. For instance, Levine and Zervos (1998) and Bekaert et al. (2002) show that financial liberalisation improves stock market liquidity through increased participation of the domestic stock market by foreign investors. A recent study by Bekaert et al. (2007) shows a positive impact of trade openness on emerging market liquidity. Henry (2000b) demonstrates how participation in the stock market by foreign investors improves liquidity, and private investment booms. Apparently, countries with high market liquidity tend to have average transaction costs which rely on the number of investors participating in trading activities.

Theory predicts that greater participation of informed institutional investors will lead to more information efficiency and thereby generate higher liquidity.¹⁶ Kyle (1985) uses a theoretical model to examine the impact of informed traders and their information advantage on liquidity and price efficiency. A number of studies provide evidence of a positive relationship between the participation of foreign institutional investors and stock market development. Mendelson and Tunca (2004) show that stock price captures extra information about security value and trading risk. Studies demonstrate that information efficiency is enhanced by competition amongst informed investors (Subrahmanyam, 1991; Holden and Subrahmanyam, 1992).¹⁷

Other studies find that the participation of foreign equity investors enhances a more liquid stock trading environment. Tesar and Werner (1995) examine foreign portfolio investment in five OECD countries and find a higher stock turnover rate in foreign equity holdings in local

¹⁵ Other studies suggest that domestic investors are better informed than foreign investors (see Choe et al. 2000; Hau, 2001).

¹⁶ The process whereby information is disseminated through increased trading is referred to as information efficiency.

¹⁷ For the impact of multiple informed traders' strategic trading on liquidity (see, Foster and Viswanathan, 1996; Back et al. 2002).

equity holdings. A more recent study by Vagias and van Dijk (2011) investigates foreign equity investment in 46 countries. They provide compelling evidence that a foreign equity portfolio flow leads to higher local stock market liquidity. Foreign institutional investors can enhance stock market liquidity via its beneficial association with positive corporate governance and information environment, particularly in emerging markets. Foreign institutional investors can exert their management expertise to restrain corporate insiders from diverting resources for private use, as well as deterring them from insider information trading. Wei et al. (2005) argue that foreign institutional investors serve as an effective corporate governance mechanism to improve firm value in China. Gul et al. (2010) note that the presence of foreign investors may prompt the management to disclose price-sensitive information in a timely manner. Consequently, it reduces insider trading and improves information environment uncertainty faced by liquidity traders. Therefore, it encourages trading activities and enhances stock market liquidity. Bae et al. (2006), show that participation of foreign institutional investors attracts the coverage of security analysts.

Chou et al. (2012) find that foreign institutional investors are better informed and price stabilising traders, whilst local institutional and individual investors lack these attributes. In essence, foreign institutional investors are net buyers when the market is liquid. However, individual investors are simultaneously net sellers. Their results imply that foreign institutional investors create trading requirements which benefit liquidity. Foreign investors are perceived to have a positive influence on trading activities as a result of having access to sophisticated skills and a pool of investment professionals. Especially, in emerging markets, equity investors closely monitor the trading activities of foreign institutional investors. Several studies, for instance, Grinblatt and Keloharju (2000) and Seasholes (2000) argue that foreign institutional investors are better informed than local investors due to their possession of a vast knowledge of global financial markets. They show that foreign investors have better access to expertise and talent and, therefore, are smarter than domestic investors. Grinblatt and Keloharju (2000) find that foreign investors buy more stocks that perform better in the next 120 trading days than the local individual investors. Seasholes (2000) shows that foreign investors buy (sell) prior to the announcement of good (bad) earnings in Taiwan. Froot et al. (2001), and Froot and Ramadorai (2008) empirically demonstrate that foreign investors are well-informed because their trading activities lead to price momentum. They show that foreign investors' equity demands predict not only prices in foreign markets, but also prices of closed-end country funds, suggesting that foreigners have access to more information than

domestic investors. However, Choe et al. (2005) find no evidence that foreign investors are better-informed in Korea. Similarly, Griffin et al. (2002) find that foreign investors are not better informed because they do not time the market at the daily frequency.

2.4.3 Adverse effect of foreign investment flow on stock market development

Other studies, in contrast, show that there is a negative impact on liquidity when foreign institutional investors vastly increase their investment in the domestic country. A theoretical framework developed by Bolton and Von Thadden (1998) and Kahn and Winton (1998) shows that foreign institutional investors come at a liquidity cost. They suggest that foreign investors become passive investors who engage in buy-and-hold activities. They argue that foreign investors are good at gathering private information which, in the long run, results in information asymmetry and inactive trading. Additionally, by virtue of being large shareholders of the firms, they become corporate insiders who tend to interfere with the company business decisions. These mechanisms are inclined to reduce stock market development via illiquidity and inactive trading. Bolton and Von Thadden (1998) model the cost and benefits of institutional investors' monitoring of managers. They show that the presence of block ownership reduces the number of shares tradable to the public and results in inactive trading of companies' shares. This eventually segments the stock market and reduces the benefits of stock market liberalisation. They argue that a decline in foreign block ownership leads to an immediate reduction in free-float shares, tradable among ordinary investors and such decline is not attributable to holding changes of domestic large shareholders.

Several studies suggest that foreign institutional investors are experienced, better trained and well informed of the activities of companies.¹⁸ They exploit their superior information advantage which, ultimately, dampens stock market liquidity. Studies show that it accordingly increases the degree of information asymmetry (see Dennis and Weston, 2001; Agarwal, 2007; Brockman and Yan, 2009). Kaln and Winton (1998) provide a model, whereby large shareholders have better access to management, which helps them to trade at the expense of uninformed liquidity traders. A study by Rubin (2007) shows that the concentration of institutional ownership negatively relates to stock market liquidity as block holders may have access to private information regarding the value of the firm.

¹⁸ See Ali et al. 2004; Ke and Petroni, 2004; and Bushee and Goodman, 2007.

Foreign institutional investors are less likely to have a positive effect on stock market liquidity, owing to being perceived to be well informed traders (Grinblatt and Keloharju, 2004 and Seasholes, 2004). Kyle (1985) argues that the trading activities of foreign institutions lead to high transaction costs to compensate for losses suffered by liquidity traders. Glosten and Milgrom (1985) show that informed trading is highly likely to widen spreads.

Rhee and Wang (2009) find that financial liberalisation has a positive impact on stock market liquidity as a result of macroeconomic, institutional and regulatory reforms, which coincides with financial liberalisation but not as a result of direct participation of foreign investors. They argue that the presence of foreign investors impacts negatively on the Indonesian stock market development. De la Torre et al. (2007) show that foreign investors' direct participation in the domestic market, has a negative association with emerging market liquidity. Greater institutional ownership leads to larger spreads and smaller quoted depth (see Heflin and Shaw, 2000; Sarin et al. 2000; and Dennis and Weston, 2001). Agarwal (2007) demonstrates that liquidity increases with institutional holding and begins to reduce upon reaching 40%.

Ng et al. (2011) use a sample of 40 countries to examine the liquidity effects of foreign institutional investors. They find a negative impact of foreign institutional investors on stock market liquidity. They show that the number of institutional investors reduces the number of free-float shares available to the public which reduces stock turnover. Their results support the conjecture that inactive trading of foreign institutional investors reduces trading friction in the stock market. Ng et al. (2011) argue that foreign institutional investors are informed investors who reduce the incentives of the participation of liquidity traders.

Several studies argue that foreign institutional investors create wider bid-ask spreads and higher adverse selection components of spreads (Heflin and Shaw, 2000). In an international context, the adverse liquidity effect may be exacerbated by foreign institutional investors, as studies arguably suggest that foreign investors are sophisticated investors who possess superior information about the domestic market (Seasholes, 2004; Froot and Ramadorai, 2008). Brockman and Yan (2009) demonstrate that higher institutional ownership, leads to lower stock turnover. However, other studies, for instance, Heflin and Shaw (2000), argue that the negative liquidity effect could be attributed to the information asymmetry between

institutional shareholders and other investors. Alternative studies argue that it is debatable to suggest that foreign investors' destabilise the domestic stock market and reduce stock market liquidity (see Tesar and Werner, 1995; Choe et al. 1999; Boyer et al. 2006; and Vagias and Dijk, 2011).

In terms of information differential, Chou et al. (2012) find that foreign institutional investors are better informed price stabilising traders, relative to local investors. They note that foreign investors are perceived to have a positive influence on trading activities, as a result of having access to sophisticated skills and a pool of investment professionals. This is particularly true in emerging markets, where all equity investors closely monitor the trading activities of foreign institutional investors. Several other studies, for instance, Grinblatt and Keloharju (2000) and Seasholes (2000), also argue that foreign institutional investors are better informed than local investors, as the former possess better knowledge of global financial markets. Empirically, Froot and Ramadorai (2008) demonstrate that foreign investors are well-informed because their trading activities lead to price momentum. Similarly, Dvorak (2005) also reports that foreign institutions have better information, due to their expertise and experience.

Bekaert and Harvey (2003) and Lau et al., (2010) note that higher domestic portfolio investments, relative to the theory keeps the domestic cost of capital higher, as risks are shared only among domestic investors and the market is globally segmented. Higher cost of capital deters significant real investments and retards the growth of the local stock markets with fewer issuances, concentrated industrial base and inefficient pricing.

Based on the above discussions, economic conjectures suggest that equity market liberalisation and subsequent integration have a positive effect on the development of stock markets, through increased participation of foreign investors in domestic markets and of domestic investors in foreign markets. However, Bekaert and Harvey (2003) and Errunza et al., (2013) note that despite the removal of explicit barriers leading to formal liberalisation of the markets, the observed gains of liberalisation are not as expected. The inadequate benefits have been attributed to the presence of implicit or indirect barriers to international investments, both for developed and, relatively more so, for emerging markets. Such barriers could be weak investor protection, poor accounting standards, higher trading costs, and market liquidity risks etc. These implicit barriers do not allow investors to optimally diversify

their portfolio in line with the theoretical suggestions of the International Capital Asset Pricing Model (ICAPM) (see Bekaert and Wang, 2010), leading to home and foreign biases.¹⁹ Studies confirm that the prevalence of home and foreign biases in equity portfolio investment, segregates stock markets from the world capital markets, and thereby prevents international risk sharing (see, for example, Lau et al., 2010). This implies that the sub-optimal international portfolio diversification, adversely affects the cost of equity capital and other benefits, including the financial deepening of the market. Thus, if the degree of financial integration is influenced by the sub-optimal cross-country portfolio allocations, then the degree of home and foreign biases should influence stock market development. Accordingly, we hypothesize that home bias (domestic portfolio investors allocating disproportionately more investments in the domestic market, i.e. lower presence of foreign investors) should inhibit the development of local stock markets. Alternatively, foreign biases (disproportionately greater presence of foreign investors) should accelerate the development of the host stock markets.

2.4.4 Gaps in the stock market development literature

This study reconciles with existing findings, that a decline in cost of capital, following stock market liberalisation is much smaller than theory predicts (Bekaert and Harvey, 2000). The flow of foreign portfolio investment is expected to improve stock market development through enhanced institutional quality, as foreign institutional investors from developed countries are likely to have well-research skills, strategies and realistic stock valuation skills.

Theoretical conjectures suggest that growing international portfolio investments should drive the development of the host stock markets. Empirical evidence suggests foreign investors may demand better market reforms to induce confidence in the efficient pricing and execution of the market. They may provide the much needed liquidity and international risk sharing benefits, which should further help in the development of domestic stock markets. However, the development effect is based on the assumption that foreign and domestic investors take advantage of the international diversification opportunities by optimally

¹⁹ The presence of home bias indicates that local investors hold a significantly higher percentage of their securities in the domestic market relative to the theoretical prescription of the ICAPM benchmark. Correspondingly, foreign bias relates to the tendency of foreign investors to over or under weight foreign markets compared to the ICAPM benchmark. For further details on the differences between home and foreign bias, see Dahlquist et al., (2003) and Chan et al., (2005).

investing in different international markets. Despite such conjecture, the literature suggests that portfolio investors do not optimally diversify their portfolio internationally, consequently resulting in home and foreign biases. The prevalence of such biases should inhibit the integration of the host stock market and influence stock market development. Drawing on such theoretical conjecture, this study addresses the issue of whether the observed cross-country differences in stock market development are explained by the variations in equity portfolio investors' home and foreign biases.

Therefore, in this study, we examine whether the cross-sectional and temporal variations observed in home and foreign biases in international equity investments can explain the heterogeneous development of stock markets across the globe.²⁰ Empirically, we test such conjecture, using standard measures of home and foreign biases²¹ in equity investments and alternative proxies of country level stock market development.²² Two important conclusions emerge.

In this study (see Chapter 7), we construct the home and foreign bias measures using two sets of comprehensive data from 44 countries, covering a sample period of 2001-2010. We examine the implications of sub-optimal portfolio allocations on stock market development, using the classical measures of country level stock market development. The extensive univariate, correlation, and regression analyses suggest that higher home bias significantly deters stock market development as a result of sub-optimally lower international diversification by domestic investors. Similarly, positive foreign bias, i.e. the tendency to favour a particular foreign (host) country in their country allocations, has a positive influence on the development of stock markets for the host country.

Studies on stock market development are dominated in the area of the importance of stock market development towards economic growth and development. From the theoretical

²⁰ Theoretical justification for this is derived in Appendix A by blending the two strands of literature. The first strand of arguments relates to the implications of globalization on cost of capital and the second shows how the variations in cross-country liquidity risks and stock market development affect expected returns. In its simplest form the Appendix demonstrates how the sub-optimal international diversification of equity portfolio investors affects liquidity and market development.

²¹ They are Home bias (*CPIS_HB*), foreign bias (*CPIS_FB*), and global fund foreign bias (*GF_FB*).

²² Four proxy measures we use are (i) market capitalization as a percentage of GDP, (ii) value of stocks traded as a percentage of GDP, (iii) stock turnover ratio (value of stocks traded as a percentage of market capitalization), and (iv) transaction costs.

perspective, countries that experience high degree of home bias will suffer weak stock market development. The explanation is that the domestic stock market will be dominated by large institutional investors who will buy-and-hold and will, therefore, adversely affect international risk sharing, stock market liquidity and development. However, foreign bias, whereby foreign investors hold a well-diversified equity investment towards a country ICAPM implied weight improves risk sharing and liquidity and, subsequently, enhances stock market development.

The results of our findings in Chapter 7 imply that countries, particularly emerging markets, contemplating the development of the depth and breadth of their local stock markets, should initiate effective policy measures which encourage the international diversification of their domestic investors and attract foreign portfolio investors.

2.5 Literature review for chapter 8: International portfolio investment and investor protection standards

This section provides a literature review that connects two strands of literature. The first focuses on the importance of investor protection. The second strand of the literature focuses on the role domestic and foreign institutional investors' play to influence country level governance across the world. We are motivated to investigate this study as a result of the importance of investor protection standards and the limited studies that examine factors that determine country level investor protection standards.

2.5.1 Importance of investor protection

The issue of investor protection relates to the agency problem as explained by Jensen and Meckling (1976). They show that managers expropriate resources of the company to benefit themselves, instead of returning profit to outside investors. Several studies show the extent managers use resources of the company at the expense of outside investors. For example, managers can make poor investment decisions that destroy shareholders' value through unreasonable acquisitions and empire building. In order to restrain corporate insiders from expropriating company resources, outsider investors will need to rely on the country's judicial system to enforce law, via an efficient and independent judicial system. Contract design should offer outsider investors the right to discipline company insiders or replace

managers (La Porta et al. 1998). Countries that have better investor protection will enact and enforce securities laws that restrain manipulations by corporate insiders.

There are a myriad of researches that examine the important role investor protection plays in an economy. Current studies suggest that improvement in country level investor protection such as control of corruption; institutional quality and judicial efficiency promote economic growth and stock market development. Stulz (2005) argues that strong institutional qualities that provide better investor protection to minority investors and restrain corporate insiders from expropriating firms' resources for their own benefit, leads to economic growth. Better governance assists firms to cheaply raise finance, which is immensely important for firms with huge growth opportunities that cannot internally raise capital but could access the stock market. Empirical evidence supports the theory that quality corporate governance has a positive impact on growth opportunities (see Durnev and Kim, 2005; Klapper and Love, 2004). Haidar (2009) show that country level investor protection explains cross-country variations in GDP growth and by this means, countries with better investor protection

Shleifer and Vishny (1997) show that country level investor protection affects stock market development and the capabilities of firms' quest in raising external financial resources. La Porta et al. (1997) argue that countries that exhibit good governance, experience a large and liquid stock market. Engelen and Van Essen (2010) demonstrate that better legal protection for equity investors, boosts investors' confidence concerning their ability to obtain higher return on their investment. Spamamman (2010) shows that minority investors experience better protection when directors of companies seek to work in their interest. Djankov et al. (2008), Djankov et al. (2007) provide empirical evidence that adequate investor protection, positively relates to increased availability of credit. Existing studies by Laeven and Majnoni (2005) show that the efficiency of the judicial system of a country improves loan recovery. Stronger legal protection for creditors is more likely to improve firm performance during financial crisis, as credit protection provisions for creditors offer assurance to creditors against default. Levine (1999) shows a positive association between investor protection and several measures of capital market development.

Studies in the US show that better firm level governance positively relates to firm value (see Gompers et al. 2003; Bebchuk and Cohen, 2005; and Bebchuck et al. 2009). Strong and well-

enforced outsider rights reduce the acquisition of private control benefits. Countries that have an unbiased judicial system that protects minority investors, reduce corporate insider's ability to obtain private control (see Claessens et al. 2000; Nenova, 2000; Dyck and Zingales, 2002). Other studies show that better investor protection increases companies' insiders cost of diverting resources (see Shleifer and Vishny, 1997; La Porta et al. 2000; Shleifer and Wolfenzon, 2002) and therefore increase firm value (Durnev and Kim, 2005; Dahya et al. 2008; Aggarwal et al. 2007).

Several studies show how better investor protection improves firm performance. For instance, Mitton (2002) demonstrates that better corporate governance is important for firm value. Friedman et al. (2003) suggest that adequate country level governance reduces expropriation risk and provides a sustained firm performance during financial crisis. Leuz et al. (2003) argue that poor financial reporting is associated with countries with weak legal protection and, accordingly, inhibits contribution to the efficient and competitive functioning of financial markets. Doidge et al. (2004) show that the level of investor protection of a country, relates to a country's financial and economic development; their findings suggest that when economic and financial development is poor, the incentives to improve firm-level, governance are low due to the substantial cost involving outside financial resources to adopt better governance mechanisms. They show that good governance is important due to its inverse relation with the cost of capital. Better governance reduces the cost of capital because investors will expect the firm to be well-governed when they provide funds.

Doidge et al. (2004) argue that country level governance characteristics play a significant role to influence governance systems in poorly developed countries. In a poorly developed country that has weak investor protection; it is overwhelmingly difficult for firms to engage in good governance. Therefore, the rights of minority investors will largely be determined by the country characteristics. Country level governance is very important, as it is expensive for firms in poorly investor protected countries to adopt good governance mechanisms when accessing capital markets for the first time. The benefits for firms in implementing good governance may not justify the high cost associated with the implementation. So, country level governance is tremendously important, as is firm-level governance. Reduction in the expropriation of minority investors increases the participation in equity investment. Castro et al. (2004) demonstrate theoretically, the relationship between investor protection and economic growth. They show that an improvement in investor protection enhances risk sharing, which indicates larger demand for capital. The demand effect provides support for the positive association between investor protection and economic growth. Weak investor protection indicates higher interest rates attributable to a shift of demand schedule. A higher interest rate translates into lower income for entrepreneurs.

Mitton (2004) uses Credit Lyonnais Securities Asia (CLSA) data of 365 firms from 19 emerging countries to examine the impact of firm level corporate governance on dividend pay-outs. He reports a positive relation between better governance and high dividend pay-out. He suggests that firm-level investor protections are not a substitute but rather complementary. Country-level investor protection plays a substantial role in preventing expropriation. Leuz et al. (2010) find that foreign investors allocate less investment in firms that are domiciled in countries with weak outsider protection. Therefore, weak country governance inhibits countries from attracting sufficient foreign investment. Dyck and Zingales (2004) suggest that poor investor protection standards create opportunities for expropriation of minority investors by managers and majority shareholders. To a large extent, the value of private benefits of control that insiders may enjoy or consume, relates to the level of investor protection.

La Porta et al. (1997, 1998, 1999a and 2002) show that countries with better legal protection of investors, experience better developed stock markets and higher firm valuations than countries with poor investor protection. Daouk et al. (2006) demonstrate that improvement of capital market governance index, relates to reduction of implied cost of equity capital. Hail and Leuz (2006) show that firms experience a decline of cost of capital when they operate from countries with extensive disclosure requirements, stronger security regulations, and stricter enforcement mechanisms. La Porta et al. (2000) argue that legal protection is very important to outside investors. When outsiders provide finance to a company, they face risk of expropriation and are exposed to the likelihood that they might not realise their expected investment return, due to expropriation by corporate insiders.

Management's expropriations of minority investors occur through several channels, including stealing of profit. In other circumstances, managers sell assets and outputs to companies under their control, below market prices. La Porta et al. (1999b) use a sample data of firms from 27 developed countries and show that countries with better investor protection have higher equity valuation via Tobin's Q, than countries that provide weak investor protection.

They also demonstrate that higher insider cash flow ownership, weakly relates to higher corporate valuations which are more pronounced in countries that provide poor investor protection.

Johnson et al. (2000a) demonstrate that countries that provide inferior investor protection, experience substantial market decline during a financial crisis. They show that during the 1997-1998 Asia crises, the level of investor protection was a key predictor of currency depreciation in 25 countries. Giannetti and Koskinen (2004) show that stocks have lower expected return when investor protection is weak in a country and reduces firms' access to the stock market to raise finance. Recent and growing studies, for instance (see Gompers et al. 2003; Core et al. 2006; Cremers and Nair, 2005; and Yermack, 2006), argue that poor investor protection standards, relates to a lower stock return. McKinsey and Company (2003) report show that foreign and domestic investors put greater emphasis on investor protection when making investment decisions.

Investors domiciled in strong corporate governance countries prefer to invest in their own countries, which reduce the benefits that can be derived from diversification. Weak investor protection reduces the participation of foreign investors in countries with poor investor protection. Guiso et al. (2001, and 2003) show that variations in cross-country investors' participation in the stock market largely depends on whether the countries have good protection for minority investors. Shleifer and Wolfenzon (2002) demonstrate that firms experience higher valuations and dispersed ownership in countries with better protection. La Porta et al. (2002) find better investor protections have a positive association with Tobin's Q.

2.5.2 Investor protection and foreign investors

Investor protection relates to agency problems. Bris and Cabolis (2004) refer to shareholder protection, as protection provided by corresponding Corporate Law or Commercial Code to shareholders of a company. In principle, the law applicable to companies is the law of the country of the nationality of the firm. The relevant investor protection standards against expropriation, is not determined by the law of the country of nationality of the shareholders, but is accordingly determined by the country where the firms operate, or the country where firms' assets are located (Horn, 2001).

Expropriation of investors by the state includes actions taken by the state that reduces return on corporate investment. Governments are required to institute policies that can restrain corporate insiders from expropriating outside investors. However, in most countries, the state is unable to protect investors' rights. Such countries experience high prevalence of ineffective government, weak control of corruption, poor regulatory quality, and an ineffective rule of law which inhibits enforcement of investors' rights. Stulz (2005) provides a model whereby the state expropriates minority investors by taking actions that reduce shareholders' wealth, ranging from outright confiscation, to regulations that favour the constituency of government officials in positions of authority. Stulz (2005) argues that agency problems are higher when institutions that are expected to protect minority investors are weak within a country.

Johnson et al. (2000a) show that controlling shareholders expropriate for their own benefits when a country's institutions are weak. It is conceivable that laws and regulations that protect minority investors will relate positively to the cost of extracting private benefits of control. The laws amongst others, includes laws that make it difficult for controlling shareholders to expropriate resources of the company, providing outside investors opportunities to influence and contribute to the policies of the company, laws that allow outside investors to seek for damages, as well as laws that demand sufficient disclosure from the firm. The participation of foreign investors in the domestic market can help improve the investor protection standards given to minority investors.

Studies suggest foreign institutional investors within a country can influence the level of investor protection in a country. For instance, Shleifer and Vishny (1986) show the role that active foreign institutional investors play in reducing agency problems. They demonstrate that active shareholders can monitor the activities of controlling shareholders to reduce agency problems. Giannetti and Simonov (2006) argue that when foreign investors, from countries with better institutions become controlling shareholders, they can play an important role to reduce expropriation of outside investors.

When corporate managers reduce expropriation, firms will be able to raise finance from foreign investors. Furthermore, foreign investors will expect consumption of private benefits in countries with poor investor protection, and will therefore, seek to buy equity from countries that provide better investor protection. Eventually, countries with weak investor protection will be compelled to improve the level of investor protection standards provided to minority investors.

2.5.3 Domestic institutional investors and investor protection

Countries that experience home bias will have the domestic market dominated by large corporate insiders who will develop close ties and connive with the state to expropriate minority investors. For instance, Stulz (2005) demonstrates that high prevalence of home bias implies that the stock market is controlled by concentrated institutional investors who seek to expropriate minority investors. Furthermore, domestic institutional investors will feel obliged to be loyal, and therefore, compromise with management and the state. A report by BusinessWeek (2006) demonstrates that Fidelity investors played an active role in governance issues in Europe but were relatively passive on governance issues in the US. Studies suggest that in a segmented market, governments will most likely take actions that will reduce return on investment but benefit corporate insiders through corrupt practices. However, when foreign investors participate in the domestic market, they will exert pressure on the state to improve governance. The state can enhance governance by reducing bureaucracy and corruption.

Domestic institutional investors will seek to promote and protect potential business relationships with firms and the state, which will compromise their monitoring role. Likewise, it will reduce the pressure they can exert on the state to improve investor protection. Large domestic institutional investors who are interested in building relationships with companies and the state, will contribute little to good governance. When domestic institutional investors are not interested in monitoring management, they tend to compromise their ability to influence protection provided to minority investors. Since minority domestic investors hold relatively few shares in the company, they will have less incentive to monitor as they can easily liquidate their investment when the company performs poorly. A recent study by Giannetti and Laeven, (2009) provides evidence that in Sweden, corporate ownership by local pension funds affiliated with controlling shareholders does not improve firm valuation, but increases the control premium.

Furthermore, when a stock market is segmented as a result of home bias, Douma et al. (2006), show that domestic institutional investors do not perform their monitoring role

properly, partly due to the fact that in emerging countries there are more owner managers who hold a large percentage of shares and, as a result, they adversely influence the legal system to prevent moral hazard on the part of management. Firth et al. (2008) show that, in several countries, the government is a majority shareholder who exerts political pressure on local investors and state institutions. Therefore, political pressure can impact negatively on the monitoring role of domestic institutional investors.

2.5.4 Foreign institutional investors' activism and investor protection

Financial globalisation through foreign institutional investors has contributed to the increased participation of institutional investors holdings globally. This is mainly to diversify portfolio risk and enjoy an increasingly faster growth in developing economies. Subsequently, in recent years, there has been an increased proportion of foreign investors' participation in the stock markets of developing countries, particularly as investment in emerging markets offers better returns to international institutional investors, such as pensions and mutual funds. This trend has also led to the question of whether foreign investors can influence country level investor protection.

A considerable body of research has focused on the role of institutional investors as corporate monitors. Grossman and Hart (1980) argue that only large institutional investors can achieve sufficient benefits to have incentive to monitor. Indeed, Shleifer and Vishny (1986) suggest that large institutional investors may have a greater incentive to monitor corporate managers and insiders.

The monitoring of management and the state to improve investor protection could be expensive, particularly, when the market is segmented. Nonetheless, when the market is integrated, foreign institutional investors can increase the monitoring of managers. They will exert pressure on companies and the state to improve the governance systems of the country as they can seek investment elsewhere. Stulz (1999b) suggests that foreign institutional investors are more likely to perform arms-length monitoring that ends up increasing firm value to benefit minority shareholders.

Globalisation through foreign investors, increases the monitoring of management and reduces information asymmetry and agency cost. Errunza (2001) conjectures, that foreign investors

from better governed countries will export good governance and demand better investor protection, which can improve the host country's governance mechanisms. Additionally, the participation of foreign investors will create competition for financial resources between countries. It is conceivable that competition for financial resources will compel countries to reduce rent seeking activities and improve investor protection standards (Rajan and Zingales, 2000).

Previous research, by Gillan and Starks (2003) argue that foreign institutional investors have become increasingly willing to use their ownership rights and activism, to exert pressure on managers and the state to act in their best interest. Apparently, foreign investors bring along highly competent analysts and researchers, who have varying degrees of experience in a vastly differing economic and political environment. Therefore, countries seeking to attract foreign portfolio investors will be compelled to improve governance and provide better investor protection, trading and delivery systems, to boost the confidence and participation of domestic investors. Attracting foreign portfolio investors will significantly influence developing countries governance, especially in the absence of large domestic investors.

Globalisation will subject countries to competition for foreign investment and will accordingly coerce countries to invest in building strong institutions. Doidge et al. (2007) argue that financial globalisation helps firms to learn better investor protection standards from countries with good governance. Flow of foreign capital and globalisation influence the level of a country's resource endowment through international technology transfers and productivity. Existing studies suggest that globalisation can improve institutional quality and investor protection. Better institutions relate to greater transparency, less corruption, and better protection of property rights, which promote market fundamentals, leading to stock market development and economic growth. Good governance contributes to positively influencing individuals and corporate investment decisions. Errunza (2001) assesses the benefits of foreign portfolio investment to the host country. He argues that the involvement of foreign investors in the domestic stock market will enhance investor protection standards. He posits that foreign investors will require timely and better access to information concerning protection of minority rights and quality of stock market trading rules and regulations.

Investors from countries that have good governance will demand and influence the governance system prevailing in the host country. This will force the domestic country to

improve on institutional quality to attract foreign investors. Rent seeking and corrupt practices which reduce shareholders' wealth will be harder to engage when several countries compete for investment. As globalisation increases countries' investment base, better institutional quality will be demanded by foreign investors to manage risk. High risk and greater opportunities work together to break the effectiveness of existing networks and rules, which creates demand for better institutions.

Intuitively, foreign investors from countries that provide better investor protection will export good governance to the host countries that offer weak investor protection to minority investors. Kang and Kim (2010) show instances where foreign investors are involved in governance activities in the host country in areas such as hostile takeover threats, proxy contests, expressions of opposition to amending anti-takeover provisions, efforts to seek representatives on the targets board, threat of top executive takeover or involvement in the selection of a new top executive and demands for asset downsizing. Doidge et al. (2004) note that financial globalisation coerces firms to improve investor protection. Further studies argue that an open economy relates positively to good governance. For example, Smarzynsk and Wei (2000), and Bonaglia et al. (2001) find that financial globalisation is positively associated with a better investor protection granted by a country. Their findings are explained by foreign investors exporting and demanding good governance and investor protection in the host country. Internationalisation via ADR can influence corporate managers to improve investor protection by aligning the interests of insiders with that of minority investors. Coffee (1999) and Stulz (1999b) provide discussions on how ADR programmes allow foreign firms to adopt US quality institutions. Particularly, when foreign firms are listed on the US stock market, they become subjected to the US securities laws and regulations. Furthermore, they are monitored by the Securities and Exchange Commission (SEC).

Foreign investors can substantially influence the governance system in the host country through the following channels. First, foreign institutional investors could offer policy-makers, information on better regulatory issues prevailing in other countries and their home country. Second, foreign institutional investors can indirectly compel a government to reform by threatening to exit the country for a more favourable investor protection country, which could have serious repercussions on the economy through share price volatility. Aggarwal et al. (2011) examine the role of foreign institutional investors in promoting better governance practices across 23 countries for the period 2003 to 2008. They find institutional investors

contribute to the promotion of good governance across countries. Foreign institutional investors could potentially influence the host country to adopt good governance practices, by threatening management of the use of voting rights (voice), or threat of selling their shares ("voting with their feet"). Foreign institutional investors usually have the resources to monitor, and the ability to influence managers and policy-makers to improve on good governance. Nesbitt (1994), Smith (1996) and Guercio and Hawkins (1999) provide evidence that institutional investors can force managers to focus more on corporate finance.

Further studies show how financial globalisation through foreign institutional investors reduces risk of expropriation. Stulz (2005) notes that financial globalisation lowers the ability to expropriate because domestic investors will move their investment to foreign markets and foreign investors will avoid investing in countries that have poor governance. Stulz (2005) suggests that financial globalization via foreign institutional investors helps domestic investors learn better governance mechanisms from countries with strong investor protection. In the absence of globalisation, and for that matter, the absence of foreign investors, the financial market will be segmented. Financial globalisation reduces the ability of state rulers redirecting resources toward some firms and away from others, as such practice makes foreign investors avoid domestic firms. Financial globalisation makes it difficult for the state to expropriate investors, because they will exit the host country when they feel threatened.

Studies show that foreign institutional investors actively participate in governance issues in foreign countries. For instance, Gillan and Starks (2003) provide a theoretical argument of the important role foreign institutional investors' play to influence good governance across the world. Ferreira and Matos (2008), using a data set of equity portfolio holdings of 27 countries, examine the role of institutional investors around the world. They find that foreign investors with little business ties with firms, engage in the monitoring of firms globally. There have been instances whereby, due to the monitoring activities of foreign institutional investors, a UK based hedge fund, The Children's Investment Fund (TCIF) in 2005, forced the management of Deutsche Borse to stop a takeover of a firm on the London Exchange, which led to the resignation of both the chief executives and the chain of the supervisory board (Economist, 2008). Becht et al. (2008) show, empirically, foreign investors activism in Central Europe.

Foreign portfolio investors can employ the services of investment bankers to play a key role of certification. Investment banks may risk their reputations in recommending good companies and countries to investors. More often than not, investment banks stand in a better position to evaluate companies' and countries' prospects to foreign institutional investors. Foreign investors will participate actively in monitoring the board, as a board that lacks credibility is less likely to take actions that will improve shareholder value. Active foreign investors will seek to replace a weak board that does not discipline managers. Foreign investors can influence the country to adopt good governance that improves investor protection. At the firm level, companies may seek to improve investor protection provided to minority investors.

Huang and Zhu (2012) employ data of qualified foreign institutional investors in China and find evidence consistent with Stulz (1999b), that foreign institutional investors are less prone to political pressure and more likely to perform arms-length monitoring than the domestic mutual funds. Huang and Zhu (2012) show that flow of large foreign institutional investors help promote the market principle in corporate governance in emerging markets, which reduces the "twin agency" problem from the state ruler's discretion.

Foreign investors can instil the concept of shareholder value and free market culture amongst local investors. La Porta et al. (2000) show that financial markets are more developed in countries where there is better protection of investors' rights. Gillan and Starks (2003), suggest that foreign institutional investors are often believed to play a significant role in prompting changes in corporate governance practices than domestic money managers. Wei (2000) shows that corruption negatively relates to an open economy as resources will be committed to improving institutional quality. Islam and Montenegro (2002) demonstrate that trade openness robustly, and positively, relates to institutional quality. The IMF (2005) shows that trade openness has a positive impact on both institutional transitions and the quality of economic institutions. Rajan and Zingales (2003) analyse the importance of trade liberalisation towards good governance.

2.5.5 Institutional investors' activism

Shleifer and Vishny (1986) show how institutional investors monitor managers and seek ways to improve firm value. According to Watts and Zimmerman (1986), corporate insiders have the tendency to pursue their own interests. However, Monks and Minow (1995) argue, that international investors can ensure that management focus mainly in pursuing the maximisation of shareholders' wealth. Studies by Li et al. (2006) examine the monitoring of managers by institutional investors. They show that a strong governance environment, acts to strengthen the monitoring ability such, that more institutional investors who hold a long-term investment perspective, have a tendency of monitoring management more actively than institutional investors with short-term perspectives (Chen et al. 2007).

Management will be less able to pursue their own interests at the expense of shareholders when there is effective monitoring by active shareholders. In equity finance, the firm receives cash from outside investors without a contractual agreement to give anything back. For equity financing to be possible, investors must expect to receive sufficient cash flows flows, to provide them with an expected return, comparable to what they would expect to earn on other investments of the same risk. However, (i) managers have access to information that investors do not have about the firm's investment, and (ii) managers have incentives, both to issue equity when they feel their stock is overvalued and take projects that do not necessarily increase shareholders value. Monitoring of activities of managers is costly and does not provide a complete solution for corporate governance (Stulz, 1999b). The extent to which management finds it in their own interests to maximise shareholders' wealth, depends critically on the firm's corporate governance system. It is possible for managers to be monitored efficiently, particularly when it also has strong incentives to increase value.

Henry (2003) contends that financial globalisation may affect factor productivity of a country by promoting better corporate governance and signalling higher quality of state governance.²³ In other words, financial globalisation, i.e. encouraging international portfolio diversification, which integrates the local with world capital markets, may have a lasting effect on the improvement of investor protection standard. However, the significant departure from the theoretical benchmarking in international equity portfolio diversification, referred to as home

²³ In modern finance, it is axiomatic that investor protection regulations and practices are pivotal for the welfare of corporates and countries (La Porta et, al, 1997, 1998, 2000, 2002).
and foreign bias, is a ubiquitous phenomenon, and is well documented in the literature (see Cooper et al., 2014 for review). In this study we investigate whether the puzzle of suboptimal diversifications (i.e. home and foreign bias) has any implication on the quality of investor protection standards.

Literature advances sound economic arguments, as we explain below, that encouraging international equity portfolio diversification, particularly allowing inward foreign equity portfolio investments, should improve the quality of investor protection standards (see Smarzynsk and Wei, 2000 and Bonaglia et al., 2001). Consistent with this economic reasoning, our empirical results reveal that countries where investors span their equity portfolio in line with the theoretically optimal international benchmark, are associated with higher quality of investor protection standards. Such positive implications may arise through indirect or 'collateral' benefits of greater financial integration. As elaborated in the following paragraphs, in a financially open economy, foreign equity portfolio investors play a catalytic role in improving the quality of both corporate and state governance.

With regard to influence on corporate governance, studies document several channels through which international portfolio diversification may influence the governance of the firms in host countries. For example, Kang and Kim (2010) note that foreign investors, particularly institutional investors, may play an influential role in domestic governance practices by employing various corporate governance mechanisms. Such disciplinary methods may take the form of hostile takeover threats, proxy contests, expressing opposition to or attempting to amend anti-takeover provisions²⁴, initiating efforts to seek representatives on the target boards, threatening replacement of top executives and demanding asset downsizing. Likewise, Boubakri et al. (2005) note that foreign ownership could lead to improvements in post-privatisation performance of newly privatised firms, because foreign investors normally demand high information disclosure standards, inject funds into newly privatised firms, and, for the sake of reputation, maintain stern control of manager's actions. Stulz (2005) offers strong theoretical conjecture, which argues that the growing presence of foreign investors lowers the ability of insiders to expropriate minority investors. Similarly, the framework of Kho et al. (2009) implies that foreign investors, particularly those from countries with better

²⁴ For example, in 2005 *The Children's Investment Fund (TCI)*, a UK based hedge fund, which had a major share of the German Deutsche Borse, forced the management to stop a takeover of London Exchange which led to the resignation of both the chief executives (Economist, 2008).

investor protection institutions, becomes a valuable inside monitor because the laws of their own countries restrict their ability to consume private benefits from other insiders. On empirical front, using data on China's split-share structure reform, Huang and Zhu (2014) confirm that involving foreign institutional investors in corporate governance practices, can significantly lower the possibility of expropriation by controlling shareholders in emerging markets.

With reference to the standard of state governance, studies report that competition for international financial resources should compel policymakers to reform their state and corporate governance regulatory and observance environment²⁵ (see Errunza, 2001 and Rajan and Zingales, 2003). Addressing the implications of state governance, Stulz (2005) argues that financial globalisation makes it difficult for the state itself to expropriate investors, as they risk losing the much needed foreign investments if they do not heed to the demands of foreign investors.²⁶ Similar sentiments are echoed by Rajan and Zingales (2003) who conjecture, that competition for financial resources becomes stronger when foreign investors become involved in the domestic economy. As a result, the growing interest of foreign investors drives reform²⁷ in the domestic investor protection regulations (see Beck et al., 2000 and Rajan and Zingales, 2000). For example, responding to foreign investors' pressure, domestic regulatory bodies signal their intention to improve the quality of governance through the adoption of international accounting standards. Errunza (2001) also posits, that with their increasing interest, foreign investors demand the formulation and observance of regulations, which compels corporates to disseminate timely and relevant information to the investor fraternity. Using data of emerging markets, Huang and Zhu (2012) show that the flows of foreign institutional investors helps promote the market-based principle of corporate governance, thus reducing the "twin agency" problem from the state ruler's discretion.

²⁵ For example, in 2007 Japan implemented the *Financial Instruments and Exchange Law*, which amended or abolished many laws that regulated foreign securities firm and was intentionally based on UK's Financial Services Authority's framework (Report by Herbert Smith, 2008 on Contemporary issues facing financial services institutions in Asia, <u>http://documents.lexology.com/cd07ed3a-b7d3-4b63-ab50-bcffa0e01dc1.pdf</u>)

²⁶ TCI initiated legal action against the Indian government under the provisions of bilateral investment treaties between India and UK over the under-pricing of coal by Coal India Limited, in which TCI holds 1% stake (see: http://www.business-standard.com/article/economy-policy/tci-starts-legal-action-against-indian-govt-underuk-cyprus-treaties-112032900095_1.html).

²⁷ Demand from foreign investors may also lead to withdrawal/deferment of reforms. For example, in March 2012 India announced imposition of a controversial general anti-avoidance rules (GAAR) on transactions made by foreign investors, without much clarity, to be effective from 1 April 2012. Foreign portfolio investors demanded immediate reversal of the reform. After intense pressure from the foreign institutional investors India deferred the introduction of GAAR until April 2013 and after further negotiations it was postponed until 2016 (Source: Financial Times, 7 May 2012 and 3 September 2012)

The above discussion convincingly underlies the importance of international portfolio diversification for firm and state governance. In terms of theoretical guidance on what should be the optimal international portfolio diversification, the International Capital Asset Pricing Model (ICAPM) suggests that equity portfolio investors should hold the world market as a benchmark of their global portfolio (see Stulz, 1981 and Chan et al, 2005). However, despite the theoretical recommendations, studies note that both domestic and foreign investors deviate from holding the implied world market portfolio, which tends to segment international financial markets, particularly emerging markets (see Lau et al, 2010). Such deviation from theoretically suggested optimal allocations, results into what the literature refers to as home and foreign bias in international portfolio diversification. Home bias refers to the phenomenon in which domestic investors tend to over allocate their home market, relative to the theoretical conjecture, thus leaving a significantly lower share of the country's wealth to be held by foreign investors. Similarly, foreign bias indicates that foreign investors tend to either over or underweight foreign markets, relative to implied benchmark (see Chan et al., 2005 and Bekaert and Wang. 2010). How do home and foreign biases affect the governance quality?

Janakiramanan (1986) and Stulz (1999) conclude that the prevalence of home and foreign bias in equity portfolio investments explains the degree of international integration/segmentation (i.e. degree of financial globalisation) of the domestic equity markets with the world capital markets. Higher degrees of home bias reflect lower degrees of financial globalization and vice versa. Alternatively, higher degrees of foreign bias imply higher degree of financial globalization and vice versa (see Lau et al, 2010 for the theoretical presentation). As a result, greater home bias indicates a more internationally closed market and thus lower presence of foreign investors as domestic investors allocate sub-optimally higher investments in their own market. Alternatively, in a relatively open and financial integrated market, higher foreign bias (i.e. higher level of foreign investments relative to implied theory) signifies greater presence of foreign investors. This should incentivise foreign investors to demand higher quality of IPS (from state and firms).

2.5.6 Gaps in the investor protection standards literature

There is a large body of research on investor protection that is chiefly dominated by the importance of investor protection. Our study adds to the two strands of literature. First, and

as noted above, to the best of our knowledge, this is the first study which examines the effect of sub-optimal international portfolio diversifications, i.e. the implications of home and foreign biases, on the quality of investor protection standards. Few related studies which investigate the determinants of investor protection are focused on the role of economic openness. For example, Islam and Montenegro (2002) demonstrate that trade openness is positively associated with institutional quality, but do not investigate the effect of financial openness. Similarly, Busse and Groning (2009) also demonstrate the importance of trade liberalisation on good governance practices, but again do not account for financial openness.

In this study, we hypothesise that the greater degree of home (foreign) bias should be related to lower (higher) quality of investor protection standards. A vast body of literature is devoted to explain the causes of home and foreign biases²⁸. However, studies investigating the implications of such biases are very limited. To the best of our knowledge, there is no study that examines the implications of home and foreign biases on the quality of investor protection standards. Using a sample of 44 countries over a period of 2001-2010, our study reports the following two important findings.

First, results suggest that the sub-optimal international portfolio diversifications of domestic and foreign investors have significant influence on the quality of investor protection standards. Specifically, markets where investors observe a higher degree of home bias (lower presence of foreign investors) are associated with poor quality of corporate and state investor protection standards. Similarly, relative to more closed markets (lower foreign bias), countries that allow/attract greater foreign portfolio investments (greater foreign bias) demonstrate a higher level of investor protection standards. Our results hold, after carefully accounting for several other possible determinants of investor protection standards, and for the potential reverse causality arising from the possibility that improvement in investor protection may cause a higher presence of foreign investors, i.e. lower home bias and higher foreign bias. The outcome of our study supports the view that financial globalisation and the consequent demand of foreign investors, has significant positive effect on the improvement of state and corporate investor protection standards (Stulz, 1999) and Doidge et al., 2007).

²⁸ See Chan et al, 2005 and Bekaert et al., 2010

Second, the results also confirm that mostly the developed markets exhibit a lower level of home bias, compared to their emerging markets counterpart. We also find that most developed countries report stronger positive foreign bias, i.e. these countries are preferred by international investors, compared to the emerging markets. These findings are consistent with existing studies (see Bekaert and Wang, 2010 and Chan et al., 2005). We further add to this literature, by providing new evidence on the observance of biases in the cross-country allocations by the sophisticated global fund managers whose sole objective is to create a broad based optimal global diversification. We uncover that the manifestation of investment biases, is not only reserved to the aggregate and macro data which may include country, regional or global funds, but such biases are also observed by the most sophisticated global fund managers.

Our study may be remotely related to Lau et al, (2010) who also demonstrate the implications of home and foreign biases. However, their focus is on the level of cost of capital, whereas our study examines the influence of home and foreign biases on the quality of investor protection standards.

Second, the results of our study also add to the growing debate, which states that the impact of international diversification and consequent risk sharing benefits should not only be limited to cost of capital and growth responses (see Henry, 2003 and Kose et al., 2010). Rather, the beneficial results should be examined through the influence of financial globalisation on factor productivity, such as improvement of micro and macro institutional quality, including corporate and state governance.

Our result suggests that when the domestic market is dominated by large institutional investors, they tend to be corporate insiders who will develop close ties with the government to expropriate minority investors. Furthermore, the government will pursue policies that will reduce return on investment. However, when countries attract large foreign investors, the foreign investors from well governed countries will export good governance to the host country, which tends to improve investor protection.

Chapter 3: Research questions and hypothesis development

3.1 Research questions

As noted earlier, there have been an extensive number of studies investigating the causes of sub-optimal international portfolio allocations, i.e. home and foreign biases. However, studies examining the implications of such biases are limited. As such, in this study, we address the following three research questions:

RQ1. What is the impact of sub-optimal portfolio allocations on the cost of capital?

RQ2. What is the role of international portfolio investment on stock market development?

RQ3. What is the effect of international portfolio investment on investor protection?

In the following sections we develop our hypotheses related to the above research questions.

3.2 What is the impact of sub-optimal international portfolio allocation on cost of capital?

3.2.1 Financial liberalisation and cost of capital

In this study, we examine the impact of sub-optimal portfolio allocation on cost of capital across 44 countries. The intriguing question from the perspective of governments and policy makers is whether there is any impact on cost of capital if a country attracts sufficient foreign investment. Specifically, we investigate the impact of home and foreign bias on cost of equity capital.

Cost of capital is an important input in the capital budgeting evaluation of investment projects. Higher cost of capital, leads to lower net present value, undermining the acceptance possibilities of prospective projects. Such non-feasibility of investment projects, particularly owing to higher cost of capital, harms the prospects of making positive effects in the real economy, leading to slower growth and lost employment opportunities. Theory suggests that financial liberalisation should integrate domestic stock markets with global capital markets which, in turn, should reduce countries' cost of capital (see Bekaert and Harvey, 2003) triggered by international risk sharing. The perceived reduction in cost of capital as a result of financial liberalisation should further trigger economic growth, driven by increased real investments. Bekaert and Harvey (2000), Bekaert et al. (2002) and Henry (2000b)

demonstrate that real investment increases in the aftermath of equity market liberalisation. For instance, Bekaert et al. (2001) show that economic growth increases by 0.7% to 1.4% annually post liberalisation. Consequently, from a national policy makers' point of view, appreciating the determinants of cost of capital cannot be understated.

Studies suggest that foreign and domestic equity investors should seek to diversify globally to reduce the country-specific investment risks of their portfolio (Solnik, 1974).²⁹ The theoretical implications of financial liberalisation and the subsequent flow (outward and inward) of foreign equity investments are well documented. For instance, Bekaert and Harvey (2003) posit a simple model, based on the extension of static integration/segmentation equilibrium models,³⁰ by examining the effects of financial liberalisation on cost of capital. The model predicts that in a financially segmented market, the volatility of the local market is much greater than its covariance with world markets. Keeping the variance and covariance constant, the theory predicts an increase in stock price (decrease in cost of capital) as the market moves from being segmented from the world capital market, into being a more integrated state, driven by increased cross-country portfolio investment.

Following on from the above discussion, Errunza (2001) further notes that when markets liberalise, the enhanced quality and quantity of information demanded by international investors should diminish the prevalent informational asymmetries and, hence, lower the cost of capital. For example, Merton (1987), focusing on market segmentation transpiring from incomplete information, demonstrates that the cost of capital, decreases with the size of the investor base, due to more efficient risk sharing. Financial liberalisation should not only affect the investor base but also alter investor composition (domestic versus international) as foreigners actively trade in the local market. The impact of changes in the investor base is pivotal, as the local securities will be priced more favourably by foreign investors. In fact, the very essence of the cost of capital/segmentation hypotheses is the migration from local pricing and shareholder base, to global pricing and more diversified international shareholder base. As such, Errunza (2001) suggests that removing inward capital-flow restrictions and the consequent inflows of foreign equity investments should reduce the cost of capital of domestic equities.

²⁹ Also see Stapleton and Subrahmanyan, 1977 and Stulz, (1999a)

³⁰ See Errunza and Losq, 1985; Eun and Janakiramana, 1986; Errunza et al. 1999 and Martin and Rey, (2000)

Using an equilibrium framework, Errunza and Losq (1985) show that the opening of equity markets to foreign investors should lower the cost of capital. Although there is a strong theoretical case of resultant fall in the cost of capital by allowing foreign investors into local markets, Bekaert and Harvey (2003) note that the removal of direct and indirect legal restrictions, may deter foreign investors from investing optimally in home (host) markets.

Henry (2003) finds that when emerging countries open their stock markets to foreign investors, aggregate dividend yield reduces by 240 basis points. Furthermore, the growth rate of capital stock increases by an average of 1:1 percentage points per year. Several studies suggest that liberalising the stock market will lead to a decline in cost of capital, investment booms, and the growth rate of output per worker increases. Financial liberalisation can, however, help investors diversify globally to share risk. This is contrary to an increasingly popular view that capital account liberalisation brings no real benefits and seems untenable.

Despite the positive effect, Bekaert and Harvey (2000), focusing on emerging markets; find that financial liberalisation has a disappointingly small effect on cost of capital. One potential reason is the sub-optimal allocations of investments by portfolio investors across the foreign markets (see Errunza, 2001). The International Capital Asset Pricing Model (ICAPM) suggests that international investors should hold the implied world market portfolio as the optimal benchmark (see Chan et al. 2005). However, equity portfolio investors in both the developed and emerging markets, exhibit varying degrees of home and foreign bias in their international portfolio allocations, which tends to segment global financial markets.³¹

Investors' deviation from holding optimally portfolio allocation influences the degree of market integration/segmentation. Such sub-optimal allocations by foreign investors, imply that a varying degree of cross-country home and foreign biases should explain the differences in host countries' cost of capital, i.e. higher negative bias (more segmented market) should be correlated with lower cost of capital and vice versa.

³¹ For recent evidence on causes of home and foreign/country bias, see Chan et al., (2005) and Bekaert and Wang (2010)

3.2.2 Theoretical framework

Following the theoretical framework of Lewis (1999) we present the standard model, showing the pricing relationship against the sub-optimal holdings of domestic investors in country l.³²

$$E(r_l) = \gamma \frac{w_l - w_l^*}{1 - w_l^*} Var(r_l) + \gamma \frac{1 - w_l}{1 - w_l^*} Cov(r_l, r_w)$$
(3-1)

Where:

 $E(r_l)$ = is country *l*'s risk premium

 γ = relative risk aversion parameter and we assume identical level of relative risk aversion for all investors in the country *l*.

 w_l = proportion of domestic investors portfolio allocated to domestic equities of country *l*.

 w_l^* = country *l*'s market share in the world market portfolio.

 $Var(r_l)$ = variance of the local market return of country *l*.

 $Cov(r_l, r_w)$ = is the covariance of return between asset *l* and world market.

If we assume that domestic investors in country l only invest in local securities. In this case equation (3-1) reduces to:

$$E(r_l) = \gamma Var(r_l) \tag{3-2}$$

When local investors invest mainly in domestic equities, the expected return of their portfolio is proportional to the variance of the domestic market return. This conjecture in the pricing modelling is similar to that of an asset pricing modelling in a completely segmented market. Equation (3-2) shows the impact of home bias on cost of capital. In the absence of international diversification, the price of the domestic market portfolio is determined only by its own return variance. In a situation where local investors do not exhibit home bias and keep domestic equity in relation to the country's share in the world-market portfolio, i.e. $w_l = w_l^*$, equation (3-1) is then expressed as:

$$E(r_l) = \gamma Cov(r_l, r_w) \tag{3-3}$$

 $^{^{32}}$ As it is standard model, we do not report the derivation of the entire model (for full derivation see Lau et al. (2010).

If domestic investors do not exhibit home bias, it signifies that they are diversifying their portfolio internationally and thus make the local stock market fully integrated with the rest of the world. In reality, equity investors show varying degrees of home bias.³³ The percentage of their domestic equity holdings falls within the interval of $(w_l^*, 1)$ and thus equation (3-1) could be written as:

$$E(r_{l}) = \gamma \frac{w_{l} - w_{l}^{*}}{1 - w_{l}^{*}} \{ Var(r_{l}) - Cov(r_{l}, r_{w}) \} + \gamma Cov(r_{l}, r_{w})$$
(3-4)

Equation (3-4) shows the association between the degrees of home bias on the cost of capital. The greater extent that local investors hold their domestic country's equities (w_l) , the higher the degree of home bias they exhibit.³⁴ The term $(w_l - w_l^*)/(1 - w_l^*)$ in equation (3-4) could be interpreted as the degree of market integration within the framework of Bekaert and Harvey (1995). A country with a lower level of integration or higher segmentation will have a larger value of w_l . The high proportion of home bias or market segmentation will therefore result in higher cost of capital, if the following condition prevails:

$$Var(r_l) > Cov(r_l, r_w) \tag{3-5}$$

It is well documented in the literature that the $Cov(r_l, r_w)$ is smaller than $Var(r_l)$ (see Stulz, 1999).

In theory, foreign investors should also benchmark the optimal allocation, i.e. w_l^* for country l. However, studies (e.g. see Chan et al. 2005) show that in case of bilateral foreign portfolio investments, foreign investors relatively under (over) allocate the benchmark weight suggested by ICAPM exhibiting what is termed as foreign bias. Suppose if the foreign allocation from country k into country l is w_{kl} then foreign bias for country l exhibited by investors in country k is defined as $\log(w_{kl}/w_l^*)$. This ratio could be negative (underallocation) or positive (over-allocation). Clearly, on average, a higher degree of foreign bias form all foreign bias for country l implies relatively lower home bias. This suggests that a higher (lower) degree of foreign bias should be associated with a higher (lower) degree of

³³ See Chan et al, (2005) for evidence.

 $^{^{34}}$ Bekaert and Harvey (1995) develop a regime-switching model to show the effects of the differing proportions of market segmentation which is equal to equation (8).

market integration, which in turn, should be related to lower (higher) cost of capital for investors in country *l*. Thus, this measure of foreign bias, constructed in the literature, should be inversely related to cost of capital.

The flow of resources into a liberalising country will reduce cost of capital, increase investment and raise output (Fischer, 1998; Summers, 2000). Theories suggest that risk-free rate and equity premium which are the two components of cost of capital will reduce when segmented countries liberalise their stock market. Henry (2000a) suggests that investment does increase post-reforms, but the effects of liberalisation on investment remains significant, after controlling for reforms. Bekaert et al. (2001) find that the increase in growth due to liberalisation is slightly larger than one percentage point after controlling for a number of variables. Generally, more financial liberalisation is thought to reduce the imperfections of financial markets, resulting in a reduction in the cost of capital and an increase in the level of investment (Henry, 2000a and Bekaert and Harvey, 2000).

In the context of this paper, we explain financial liberalisation to be when the government of a country decides to allow foreigners to purchase shares in that country's stock market. In other words, government allows the inflow and outflow of foreign equity investment without restrictions, and simultaneously allows domestic equity investors to buy and dispose of foreign stocks. Financial liberalisation can take several forms such as: openness of the bond market, financial sector, particularly, the banking sector and foreign exchange reforms. The process of financial liberalisation is very complex, with no generally agreed economic model that significantly explains the dynamics of the process. This means that even though there are general equilibrium models of economies in integrated states and those economies that are segmented, there is a lack of a model that identifies the mechanisms which transform a country from a segmented to an integrated economy. The ICAPM predicts that stock market liberalisation may virtually reduce the liberalising country's cost of equity capital, corresponding with allowing for risk sharing between domestic and foreign investors (see Stapleton and Subrahmanyan (1977), Errunza and Losq (1985), Eun and Janakiramanan (1986), and Stulz (1999a, b)).

Bekaert and Harvey (2003) prescribe a simple model to attain some intuition of financial globalization, where the model traces the impact of market integration on stock prices from the perspective of an emerging market. The model largely extends the standard static

integration/segmentation model (Martin and Rey, 2000). Within the context of a simple quadratic utility specification, Bekaert and Harvey (2003) examine the three-period problem for the global market and emerging market. They made the assumption that there is a single stock left of each asset. Dividends are paid out in period three, whilst there remain two trading periods. The government in an emerging economy may or may not integrate the market with the global financial market in period two. There is a price taking equity investors in each market who invest in period three.

The theory of ICAPM predicts that sufficient flow of foreign investment to a country, decreases cost of equity capital, due to improved risk sharing between domestic and foreign investors, especially when there is a negative correlation between the country's stock market and the world stock market (see Errunza and Losq, 1985; Eun and Janakiramana, 1986; Stulz, 1999a). The expected return will depend on the global price of risk and global covariance of risk. The implication is that, holding expected future cash flow constant, there should be a significant increase in a country's equity price index. Moreover, there should be an observation of an increase in physical investment following the flow of foreign portfolio investment, as projects which previously had negative net present values (NPV) prior to the liberalisation will turn into positive NPV. The effects of the positive NPVs will lead to growth of GDP, creation of jobs and capital flight reversal.

When financial markets are not segmented Adler and Dumas (1983) postulate that the expected return of the country will depend on the covariance of the country's stock index return with the world market portfolio return and probably with currency deposit rates. They formalize the notion that with the removal of investment restrictions and integration of the country's stock market with the rest of the world, local investors have the option to buy foreign securities, and foreign investors could also purchase domestic securities. This creates the possibility of global risk sharing between domestic and foreign investors and thus reduces expected return. On the other hand, when markets are segmented, the expected return of the country depends on the variance of the country's stock market index return. When financial markets are not integrated globally, securities are mainly owned by domestic investors. In a partially integrated market, the risk premium of the country depends on the weighted average of the covariance of the country's market integration as the weight (Bekaert and Harvey 1995; de Jong and de Roon, 2005).

Studies show that financial liberalisation significantly reduces cost of capital. Stulz (1999b) suggests that when a country opens its stock market to foreign investors, cost of capital declines, due to increased risk-sharing and improved corporate governance. Even though there are opportunities of global risk sharing between domestic and foreign investors, investors face varying degrees of investment constraints. In spite of financial liberalisation, the global financial market has some fragments of segmentation, as equity portfolio investors in both the developed and emerging markets, exhibit varying degrees of home and foreign bias. Existing studies demonstrate the effect of home bias on security prices. Chan et al. (2009) provide evidence using mutual fund data, to show that home bias affects firm value.

Earlier research by Stulz (1999b) suggests that, when a country opens its stock market to foreigners, its cost of equity capital is expected to decline through enhanced risk-sharing, corporate governance, capital market efficiency, better legal systems, shareholders' activism and adequate disclosures. Henry (2000a) using data from 12 emerging markets, finds that equity valuation rises by two per cent in the eight month period before liberalisation and concludes that stock market liberalisation decreases countries' cost of equity capital. Another stream of research by Bekaert and Harvey (2000) shows that cost of capital usually declines after capital market liberalisation with varying effects between five and 75 basis points. They find insignificant evidence of change in realized returns but find substantial reduction in dividend yields between the pre and post-liberalisation period. Recent empirical work by Hail and Leuz (2009) investigates whether cross-listing in US exchange, reduces a firm's cost of capital after controlling for traditional proxies, for risk, analyst forecast bias and firm-fixed effects.

Fischer (2003); Rogoff (1999), and Summers (2000) argue that, emerging countries should experience a decline in cost of capital, and enjoy increased physical investment and output (GDP) when they receive a large flow of foreign portfolio investment from developed countries. Consistent with this, is empirical evidence produced by Tesar and Werner (1998), Bekaert and Harvey (1998), and Errunza and Miller (1998) who document that the local price of risk (variance) exceeds the global price of risk (covariance), so it is expected that equity risk premium falls when a segmented country liberalises its stock market. Holding expected cash flow constant, the decline in the equity risk premium will lead to a permanent fall in the aggregate cost of capital and attendant revaluation of the aggregate equity price index. Kim

and Singals (2000) find evidence that emerging markets' stock returns are abnormally high in the months leading to liberalisation. A study by Foerster and Karolyi (1999) demonstrates that stock prices of companies from segmented markets that cross-list, are expected to rise as a result of a subsequent fall in expected returns, as the additional built-in-risk premium, compensating for these barriers dissipates. They find significant average excess returns of 19 per cent during the year before listing and an additional 1.20 per cent during the listing week, but a substantial average decline of 14 per cent is incurred during the year following the listing. The stock market beta, relative to its home market index declines dramatically from 1.03 to 0.74 on average but its global beta, relative to the world market index does not experience any significant change.

Errunza and Miller (2000) use firm level data of a sample of 126 firms from 32 countries to study the impact of financial liberalisation, thus the introduction of American Depository Receipts (ADR) on cost of capital. They generally conclude that market liberalization reduces cost of capital. All the firms in their sample, experienced substantial decline in realised abnormal returns. They attribute the reduction of 42.2% to the introduction of ADR after controlling for impact of market and confounding effects. They suggest that it was extremely difficult for previous research to find the relationship between stock market liberalisation and cost of capital. Accordingly, they argue that segmented emerging markets, prior to financial integration, do have a high, steady, state of return due to the high risk premium. However, during liberalisation, the discount rate declines and equity valuation increases. In contrast, Bekaert and Harvey (1998) find mixed results on the impact of financial liberalisation on the cost of capital using market level data, after controlling for concurrent economic reforms. Henry (2003) further argues that capital stock growth rate increases by 1.1 per cent following liberalisation, from an average of 5.4 per cent per year in the pre-liberalisation period, to an average of 6.5 per cent in the post-liberalization period. He concludes that liberalisation of the stock market by developing countries, reduces cost of capital and leads to an investment boom; additionally, the growth rate of output per worker increases. He concludes that the arguments of financial liberalisation do not bring about any real benefit and do not hold any truth in the face of empirical evidence.

Miller (1999) investigates the stock price impact of international dual listing, by analysing a sample of 181 firms from 35 countries that participated in a depository receipt programme during the period 1985 to 1995, using cross-sectional regression. He finds abnormal returns

during and around the announcement date, and concludes that firms gain from listing shares outside their domestic market. Firms listing on major US exchanges such as the NYSE, NASDAQ instead of OTC, experience greater abnormal returns. Bekaert and Harvey (1998) examine the impact of stock market liberalisation on cost of capital by focusing on changes in returns and dividend yields pre, post, and during the liberalisation period to account for the re/evaluation effect, using cross-section time-series for a sample of emerging markets. They find that increases in equity flows are associated with lower cost of capital, higher correlation with world market returns, lower asset concentration, lower inflation, large market size relative to GDP, greater trade, and slightly higher per capital economic growth.

More recent work by Edison and Warnock (2008) use monthly portfolio equity data flows from the US to emerging markets between 1989 and 1999 and find that cross-listing of emerging market-equity on the US exchange, results in sharp, short-horizon inflows, whereas the reduction of capital controls, results in inflows over longer horizons mainly in Asia. Emerging markets do not enjoy wide foreign buying, nor does foreign investment in the cross-listed firm increase. They find that firms obtain funding during the time of cross-listing and that ends the story. De Jong and de Roon (2005) examine time-varying market integration and expected returns in 30 emerging markets. They find that the average annual decrease in segmentation of 0.055 reduces cost of capital by 11 basis points. Hail and Leuz (2009) analyse cost effects and changes in growth expectations around US cross-listings, using large panel data of more than 40,000 firms' yearly observations from 45 countries during the period 1990 to 2005. They find strong evidence that cross-listings on US exchange are associated with a significant decrease in firms' cost of capital after controlling for traditional proxies for risk; analyst forecast bias, and firm-fixed effects. More recently, Lau et al. (2010) investigated the impact of home bias on cost of equity capital and find that international differences in cost of equity capital are significantly associated to varying degrees of home bias for 38 countries in their sample. Their study implies that countries could experience substantial decline in cost of capital if home bias decreases, and also if risk sharing between domestic and foreign investors increases.

3.2.3 Contributions

Employing five proxies of cost of capital measures and extensive robustness tests, our study provides a comprehensive and rigorous investigation to show how home and foreign bias

explains variations in cross-country cost of capital. Within the context of existing literature, our study makes the following contributions. First, except for Lau et al. (2010), most of the existing studies examine the effect of financial globalisation on cost of capital, either based on event study, investigating how the cost of capital changes post financial liberalisation period, or demonstrate the influence of depository receipts on the cost of capital. For instance, Kim and Singal (2000) and Chari and Henry (2004) show that the post financial liberalisation cost of capital significantly decreases. Stulz (1999) and Henry (2000a) also find similar results, suggesting that the reduction in cost of capital is driven by increased risk sharing and improved corporate governance.³⁵ De Jong and de Roon (2005) document that the increased time varying integration, i.e., the process of gradual financial liberalisation of the domestic equity market is associated with reduction in the cost of capital. However, the assumptions underlying these studies are that domestic investors are actively and optimally involved in international diversifications investing abroad, and foreign investors optimally invest in foreign countries. However, an extensive number of studies, as discussed above, demonstrate that home investors shun the optimal benefits of international markets and overallocate in their home markets, leading to home bias. Similarly, foreign investors do not allocate optimally in foreign markets, resulting in foreign bias.

Furthermore, only a scant number of studies investigate the impact of floating ADRs/GDRs on cost of capital. For example, Errunza and Miller (2000) demonstrate that firms issuing American Depositary Receipts (ADRs) experience a fall in cost of capital, driven by the increased global risk sharing effect (see Foerster and Karolyi, 1999; Doidge et al. 2004 for further evidence).³⁶ However, as Bekaert et al. (2009) note, ADR companies exhibit representation problems and, hence, do not provide full exposure to foreign stock markets. Clearly, ADRs ignore the investments of local investors in foreign markets and foreign portfolio investors investing directly in non-cross-listed domestic firms. Unlike other studies which use liberalisation events or ADRs/GDRs, we directly employ sub-optimal measures of international portfolio investments (i.e. home bias) to study its effects on the cost of capital. Financial liberalisation could coincide with macroeconomic reforms which may make it difficult to assess the actual impact of liberalisation on cost of capital

 ³⁵ Also see Errunza and Miller (2000), Chari and Henry (2004a) and Doidge et al. (2004).
 ³⁶ For more recent evidence see Edison and Warnock (2008) and Hail and Leuz (2009).

Second, instead of only using home bias measures, which focus on domestic over-investment by domestic investors (see Lau et al. 2010), we employ an additional relative measure of international portfolio investment, known as foreign bias. Clearly, following the above discussion on the theoretical arguments of global risk sharing and market integration, countries receiving relatively higher investments, i.e., international investors exhibiting greater foreign bias, should be more integrated with world capital markets, and hence, should demonstrate a relatively lower cost of capital.

Furthermore, we employ micro level data to construct other foreign bias measures (global fund foreign bias) to examine its impact on cost of capital. The former is the standard macro data from the International Monetary Fund's Coordinated Portfolio Investment Survey (CPIS) data. The aggregate measures of foreign and home bias measures used in the existing studies, ignore the individual objectives or focus of funds. Even though, on a theoretical basis, each fund should be globally diversified. On practical grounds, a fund's objective may be single country, single region or global diversifications, to suit the needs of different tastes of portfolio investors. For example, if the focus of the fund is diversification only within the EU, it will have no allocation across countries outside the EU. Additionally, if the focus of a fund is on a single country or region, it will significantly affect the aggregate measure of home and foreign bias. To overcome this issue of potential bias in the measure of sub-optimal allocation of international investors, the second foreign bias measure we construct, uses unique micro firm level global funds' allocation data across 44 countries.

The global funds that we employ carry the sole objective of global diversification across all investable countries. The use of global funds is the most restrictive and robust measure of foreign bias we could compute, significantly mitigating what we call the fund-focus bias in the measurement of foreign bias measures used in the existing literature.

Finally, we further contribute to Lau et al. (2010) by providing robust analysis using *Tobin's* Q as an alternative measure of the cost of capital. We also address the concern of endogeneity using alternative estimation methods, such as lagged predetermined values of home and foreign biases, and the Heckman selection model.

We robustly test our hypotheses using Newey-West autocorrelation and heteroskedasticitycorrected standard errors. Consistent with theory, our comprehensive empirical analysis and extensive robustness tests provide strong evidence that a higher degree of home bias is associated with a higher cost of capital. Correspondingly, a higher degree of foreign bias, i.e., foreigners tilting their country specific portfolio weight towards the global optimum, reduces the host country's cost of capital.

We use four proxies of cost of capital; historical realised return of the market (HRRm), sovereign credit ratings (rCred), country equity risk premium (CERP), and dividend yield (DY) to test the following hypothesis.

- **H**₁ Countries that exhibit higher home bias are associated with higher cost of capital.
- **H**₂ Countries with higher foreign bias relate to lower cost of capital.
- **H**₃ Pervasiveness with higher global fund foreign bias allocation is associated with lower cost of capital.

3.2.4 Summary of findings

The findings of the study demonstrate that after controlling for a multitude of conventional risk variables, which existing studies show to affect cost of capital, our study provides robust support to the hypothesis, that when domestic investors exhibit home bias and over-invest in their home market, cost of capital increases. However, when foreign investors allocate investment towards the implied global weight as suggested by ICAPM, cost of capital reduces as a result of increased international risk sharing. This suggests that the cross-sectional and temporal variations in home and foreign bias, largely explain the differences in cross country cost of capital. To be specific, higher home bias (higher foreign bias) should be associated with higher (lower) cost of capital. In this study, we provide comprehensive robust and extensive tests of such conjectures. We discuss the empirical results in Chapter 6.

3.3 What is the role of international portfolio investment on stock market development?

3.3.1 Equity portfolio investment and stock market development

Previous studies have investigated the effects of financial liberalisation on stock market development. For instance, Levine and Zervos (1998) show that financial liberalisation increases various indicators of stock market development, including market capitalisation to GDP, value traded relative to GDP, turnover ratio, and number of listed companies. They use a sample of 16 countries and find that stock markets become more liquid following stock market liberalisation. Galindo et al. (2001) demonstrate that stock market liberalisation improves capital allocation efficiency of firms in 12 developing countries. Equally, Bekaert and Harvey (2000) show that equity market liberalisation improves risk sharing and thereby reduces cost of capital. Several authors stress the role of foreign investors in stimulating stock market development. Chari and Henry (2004a) provide evidence that financial liberalisation increases risk sharing. Obstfeld (1994) shows that increased risk sharing through international equity diversification improves resource allocation. He finds that financial liberalisation increases the allocation efficiency of domestic investment and raises total factor productivity growth without any need for technological change. However, Chari and Henry (2004b), and Gourinchas and Jeanne (2002) argue that it is not obvious that capital account liberalisation directed at increasing international allocative efficiency would have any effect on domestic allocative efficiency.

Financial theory postulates that financial liberalisation integrates the domestic stock markets with the global stock market and, subsequently, leads to stock market development. For instance, Errunza (2001) argues that financial liberalisation leads to stock market development, which helps to achieve efficient risk sharing, and resource allocations, as well as assists to mobilise and improve the structure of external finance. An efficient stock market, via the pricing mechanism, will differentiate companies and reward well managed and profitable companies with lower cost of capital at the expense of less profitable firms. Several emerging countries have followed policies towards liberalising their stock market in order to attract foreign equity inflows. McKinnon (1973) argues that financial liberalisation relates positively to increased financial development. Galindo et al. (2002), show that financial liberalisation under certain circumstances, promotes financial sector development. Bekaert et al. (2005) analyse the impact of financial liberalisation on stock market development.

In this study, we seek to examine whether home and foreign bias, exhibited by local and foreign investors respectively, have any varying impact on stock market development. We provide inter-related theoretical argument to demonstrate how home and foreign bias affect stock market development.

Current studies suggest that financial liberalisation will lead to a higher flow of foreign equity investment and will help integrate stock markets. Stulz (1999a) argues that both foreign and domestic investors should look to diversify globally to reduce investment risk associated with a country. Large institutional investors increase their investment in emerging markets, primarily to benefit from diversification of risk and to partake in the high economic growth in emerging countries. Errunza (2001) theoretically shows that when countries attract sufficient foreign equity investment and the domestic equity investors diversify investment globally, it improves stock market development.

A review of international equity allocation shows less international portfolio diversification due to the prevalence of the home bias phenomenon. The question is whether home bias will inhibit stock market development? The adverse effect of home bias on stock market development or when domestic investors over-invest in the local stock market in deviation from the theoretical optimal weight suggested by ICAPM, is explained by the fact, that first, the domestic stock market will, to a large extent be segmented, and the risk sharing opportunities between domestic and foreign investors will reduce and will subsequently increase cost of capital. Home bias indicates less participation in the domestic stock market and it will be expensive for investors to trade since investors cannot easily sell their securities to raise funds. This will reduce the incentive to invest in projects of a lengthy duration.

Second, financial theory³⁷ and empirical evidence show a positive relation between home bias and cost of capital, because when home bias persists in a country, few dealers and brokers will be operating in the stock market, and poor trading technology will increase transaction costs. If home bias positively drives cost of capital, it implies that firms will experience reduction in stock prices and stock market valuation. Additionally, it will reduce the value of trading, increase high transaction costs and decrease the significance of the stock market in the economy. Third, the domestic market will be dominated by large institutional

³⁷ For more of the theory, see Stulz (1999a)

investors who will buy-and-hold, and successively create an illiquid stock market which will reduce the participation of domestic investors. Large institutional investors will dominate companies and expropriate minority investors, leading to lower investor confidence and poor stock market development.

Next, we show the transmission mechanism of how foreign bias can impact positively on stock market development. In the case of foreign bias, whereby a country attracts sufficient flow of foreign investment, the stock market will become integrated with the global market and enhance global risk sharing between domestic and foreign investors. Subsequently, it will reduce cost of capital and improve stock market valuation. Adler and Dumas (1983) demonstrate that countries' expected returns depend on the covariance of the country's return, with global stock market return.

Additionally, foreign investors that participate in the domestic stock market will demand better investor protection, boost the domestic investors' confidence and improve stock market liquidity. La Porta et al. (1997) suggest that equity market liberalisation will compel foreign investors from countries with good governance systems to insist on better governance, which will promote stock market development. Hargis and Ramanlal (1997) provide the link between stock market integration and stock market development. They show that stock market integration boosts stock market development via increased market capitalisation and higher stock market liquidity of companies. Intuitively, as more foreign investors participate in the domestic market, they will demand better trading techniques and technology which will, in turn, increase value traded and a decline in transaction costs.

Furthermore, the participation of foreign investors will increase the number of brokers and dealers who will compete amongst themselves for investors. Subsequently, it will lead to a decline in transaction costs (i.e. fees, commissions, market impact). Foreign investors who participate in the domestic market will require timely and quality information, adequate investor protection and trading regulations.

A growing body of literature addresses the general merits of opening stock markets to foreign portfolio investors. Contributing to the positive impact of financial liberalisation on stock market development, Galindo et al. (2002) show that financial liberalisation under certain circumstances, promotes financial sector development. Foreign bias implies that international investors have favourable expectations of the growth prospects in a particular country; they will seek to participate in the growth opportunities in those countries, bid up share prices and will make more financial resources available for investment. Higher participation of foreign investors in the domestic market will have an inverse relation with cost of capital and will, therefore, improve stock market development.

3.3.2 Contributions

Existing studies relate to the impact of financial liberalisation on stock market development and a vast amount of literature has investigated the importance of stock market development in relation to economic growth and development. For instance, earlier research that has found a positive association between stock market development and economic growth, has been examined by (King and Levine, 1993; Levine, 1997; Rajan and Zingales, 1998). Carlin and Mayer (2003) provide compelling evidence that there is a positive relationship between the development of countries' financial systems and economic growth. Bose (2005), in his theoretical model, reveals that stock market development positively relates to economic development.

Despite the importance of stock market development, limited studies have examined factors that influence stock market development. As far as we are aware, no study has investigated the effect of home bias on stock market development. Our study, therefore, contributes to the stock market literature by examining the impact of home bias on stock market development, whilst we control for variables shown in existing studies to drive stock market development.

Second, as far as existing literature is concerned, our work is the first to examine the impact of foreign bias on stock market development. Previous studies focus largely on what determines foreign bias. For example, Chan et al. (2005) investigate factors that influence foreign bias, using mutual fund equity allocation. Chan et al. (2009) study the impact of foreign bias on firm value, whilst Beugelsdijk and Frijns (2010) investigate how culture and cultural distances between stock markets explain foreign bias.

Third, another distinct contribution of our study, relative to existing studies is that our study uses a unique dataset of micro global fund allocation data across 44 countries to examine the impact of global fund foreign bias on stock market development.

Our study is important for policy implication because existing studies show that large and more efficient stock markets lead to economic growth (see Levine and Zervos, 1996, Bencivenga et al, 1996; Rousseau and Wachtel, 1998; Levine et al, 2000; Beck et al, 2000a; Rousseau and Sylla, 2001). However, alternative studies, for instance (see Devereux and Smith, 1994; Stiglitz, 2000; Eichengreen, 2002), show that financial liberalisation has not had the expected positive impact on economic growth.

We thereby provide evidence that cross-country pervasiveness of home bias inhibits financial development. Additionally, we offer evidence that when foreign investors increase their investment towards the global portfolio weight as implied by ICAM, stock market development improves.

We test the following hypotheses:

- **H**₄ Countries with a prevalence for higher home bias, experience poor stock market development.
- **H**₅ Countries with a higher foreign bias, experience better stock market development.
- H_6 Countries with a higher global fund foreign bias, enjoy better stock market development.

3.3.3 Summary of findings

We examine how home and foreign bias affects stock market development. We show that home bias segments stock market and thereby reduce risk sharing between domestic and foreign bias. This study finds robust evidence that domestic investors increasing their bias toward home equity reduce stock market development. The phenomenon is explained by the fact that large domestic institutional investors will buy-and-hold their investment and will adversely affect stock market liquidity and development. Similarly, foreign investors increasing their investment towards the optimal global market capitalisation will improve stock market development. This is due to the fact that the foreign bias tends to integrate the domestic stock market with the global stock market and will increase global risk sharing between domestic and foreign investors and subsequently leads to higher stock market valuation. An endogenous treatment effect provides robust support to the main analysis. The policy implications of our study provide convincing evidence that countries should seek to attract sufficient foreign investment and help domestic investors diversify to enhance stock market development.

3.4 What is the effect of international portfolio investment on investor protection?

3.4.1 Sub-optimal portfolio allocation and investor protection

Current literature indicates the significant role that investor protection standards play in influencing cost of capital, stock market development, stock market returns, and corporations' efforts in raising financial resources, economic growth and development.³⁸ Existing studies show that significant variations in financial systems among countries are shaped by the magnitude of legal protection afforded to minority investors against expropriation by company insiders. Previous literature demonstrates that better investor protection relates to higher stock market valuation (La Porta et al. 1997), more valuations of listed companies relative to assets (Claessens et al. 1999; La Porta et al. 2002), high dividend pay-outs (La Porta et al 2000), lower correlation of ownership and control (La Porta et al. 1999b), lower private benefits of control (Zingales, 1994) and higher correlation between investment opportunities and actual investment (Wurgler, 2000). La Porta et al. (1998) argue that better investor protection increases an investor's willingness to provide finance, which reflects in lower cost of capital and more availability of external financial resources.

Klapper and Love (2004) demonstrate that firm-level governance has a strong positive association with country-level investor protection. For instance, improving the efficiency of a country-level legal system from low to medium value increases the average firm-level governance by about half a standard deviation. A more recent study by Chen et al. (2009) shows firm-level corporate governance and country-level investor protection tend to substitute each other in reducing cost of capital, because when contract enforcement is weak within a country, companies will be unable to overwrite the legal system prevailing in a country and the flexibility of companies to improve corporate governance would be

³⁸ Myriad studies provide the importance of investor protection in explaining cost of capital (Doidge et al. 2004; Chen et al. 2009), stock market returns (Shleifer and Wolfenzon, 2002; Giannetti and Koskinen, 2010; Gompers et al. 2003; Cremers and Nair, 2005; Core et al. 2006), corporations efforts in raising financial resources, economic growth and development (Stulz, 2005; Doidge et al. 2004). See also (Johnson et al. 2000a; Guiso et al. 2003; Levine, 1999; La Porta et al. 2002; Eleswarapu and Venkataraman, 2006).

restricted. Countries that have better investor protection are associated with better firm-level corporate governance. Correspondingly, firm-level corporate governance tends to work well in countries with better country-level investor protection. Chen et al. (2009) argue that minority rights protection includes rights given by law and the effectiveness of enforcement. Better country-level investor protection helps firm-level corporate governance mechanisms to work effectively.

We are primarily motivated to empirically examine this research question by the awareness of recent research which has shown the importance and increased interest in the corporate governance topic among investment banks, rating agencies, and other specialised financial institutions, as country-level investor protection data has recently become available.

To date, limited studies examine factors that determine country level investor protection. Research on institutional investors mainly focuses on the importance of investor protection. In contrast, little direct evidence is presented on the impact of sub-optimal portfolio allocation on investor protection. Better country-level investor protection implies that there are strong institutions within a country that provide rights to investors, and also an effective judicial system that enforces the rights of the investors. We explore the impact of sub-optimal portfolio investment (specifically, via home and foreign bias) on investor protection, which is relevant to international investors and has important policy implications, particularly for emerging countries.

In this study, better investor protection means good governance or quality institutions that protect and enforce the rights of investors. Institutions have sets of rules and norms that shape the social, political and economic interactions among members of the society. Institutional quality is determined by the legal and administrative framework within which investors, corporations, and government institutions operate and interrupt to generate wealth. The role of institutions does not only involve the legal framework, but also includes the attitude of the government towards market freedom and efficiency. Countries with quality institutions have an efficient and independent judiciary that protects the interests of minority investors, property rights and intellectual property protection, efficacy of corporate boards, and less corrupt practices.

The study investigates and tests the proposition of whether a varying degree of sub-optimal portfolio allocation (i.e. home and foreign bias) is important in explaining cross-country variations in investor protection standards. The existence of good governance relates to adequate provision of better investor protection. Specifically, we focus on a particular type of governance that relates to investor protections, which capture two components. First, the investor protection captures the legal rights component, whereby investors are granted legal rights. Second, it captures the enforcement component, whereby the quality of a country's institutions determines the extent to which these rights are respected and enforced legally in court.

Governments throughout the world have a responsibility to achieve sustained high rates of economic growth. They are required to implement policies to reduce unemployment and provide economic opportunities for greater proportions of the population, which can improve the standard of living of the citizens. In most countries, even though much progress has been made towards the stabilisation of the macroeconomic conditions, evidence suggests that sustained economic and national development remain relatively slow, especially, in developing countries, due to weak institutions that fail to protect investors. This has generally led to the acceptance that better investor protection is an essential issue in a development policy. Prior theoretical studies (see Stulz, 1981b; 1999b; Errunza and Losq; 1985) suggest that both domestic and foreign investors should construct a globally diverse equity portfolio to reduce the risk of the portfolio. A study by Oman (2001) shows that the rapid growth of international portfolio investment by the Organisation for Economic Co-operation Development (OECD) and institutional investors, for instance pension funds, has become a driving force for improved corporate governance in emerging markets. Good economic institutions are most likely to flourish in a rent-free environment, whereby small groups are unable to take advantage of a monopoly position of a particular industry or have privileged access to natural resources.

In theory, the issue of investor protection can be explained by globalisation. The advent of financial liberalisation provides equity investors with the opportunity to diversify to reduce portfolio risk. Existing literature suggests that external mechanisms, such as flow of foreign equity investment via institutional investors can help strengthen good governance in the domestic country. Bonaglia et al. (2001) provide a theoretical argument of the link between

globalisation/integration and good governance. We, therefore, provide an argument based on sound theory to explain why home and foreign bias can influence investor protection.

Existing studies suggest that home bias has the consequence of segmenting stock markets, and current literature shows that segmented countries experience poor governance or weak investor protection. The phenomenon of home bias adversely affecting investor protection and is explained by the fact that when domestic investors over-invest in their country, they become controlling shareholders. Subsequently, they will expropriate minority investors, and the country will thereby experience weak investor protection. Errunza (2001) argues that segmented countries experience poor investor protection. The level of a country's investor protection determines whether foreign and local investors will invest in the domestic country (Chan et al. 2005). Weak investor protection is associated with countries with less developed stock markets, and Forbes (2010) shows that foreign investors hold a large proportion of equity in the US, if the domestic stock market is less developed. Janakiramanan (1986) and Stulz (1999a) conclude, that the pervasiveness of home bias in an equity portfolio investment, segments domestic stock markets from the global market. Additionally, home bias reduces the opportunity of domestic investors learning good governance from their foreign counterparts. Therefore, domestic corporations and investors will have no option but to operate in the domestic markets that have poor securities laws. This means that we can hypothesize that domestic investors' over-investing in their local stock market will negatively impact on investor protection standards or good governance.

Next, we provide the transmission mechanism to show how foreign bias impacts on investor protection from the following perspective. Errunza (2001) suggests that the participation of foreign investors in the domestic market will promote good governance. Particularly, competition for foreign equity investment between countries will influence good governance and the regulatory environment. Ceteris paribus, countries that spend resources to improve governance will attract foreign equity investment than those with poor investor protection standards. The argument rests on the theoretical perspective that countries will succumb to pressures from foreign investors to improve investor protection. The justification for the argument is that foreign investors could enjoy better investor protection elsewhere and they will then put pressure on the state to improve investor protection vis-à-vis domestic investors.

When foreign investors allocate more investment towards the host country, studies suggest that it improves investor protection. Rajan and Zingales (2000), and Beck et al. (2000) argue that competition for financial resources comes to bear between countries, and it compels countries to reduce rent seeking activities and rather improve on investor protection. Conceivably, countries will improve on investor protection standards to attract foreign investors. Rent seeking and corrupt activities that reduce shareholders' wealth will come at a cost when several countries are competing for the same foreign investors. Globalisation increases countries' international investment base and foreign investors will demand better institutions to manage risk.

Rajan and Zingales (2003) argue that competition for financial resources becomes harder when foreign investors become involved in the domestic economy. The participation of foreign investors will enhance good governance through the adoption of international accounting rules and regulatory standards. Errunza (2001) assesses the benefits of foreign portfolio investment to the host country. He argues that the involvement of foreign investors in the domestic stock market will lead to good governance and better investor protection. He posits, that foreign investors will demand timely and better access to information on protection of minority rights and quality stock market trading rules and regulations.

Studies suggest that a surge in foreign equity investment inflow will have a positive impact on investor protection. For example, foreign investors from countries with good governance that protect minority investors will export good governance to the host country. Similarly, the domestic investors will learn good governance from their foreign counterparts. Kang and Kim (2010) provide instances where foreign investors engage in governance activities in the host countries in areas such as: hostile takeover threats, proxy contests, expressions of opposition to amend anti-takeover provisions, efforts to seek representatives on the targets board, threat of top executive takeover and demands for asset downsizing.

Doidge et al. (2004) show that financial globalisation helps firms to align themselves to the high level of investor protection prevailing in the host countries. In other words, foreign investors assist in exporting better governance to host countries that have weak investor protection. Smarzynsk and Wei (2000) and Bonaglia et al. (2001) demonstrate that financial globalisation is positively associated with the degree of investor protection granted by a country. They argue that trade openness relates positively to good governance. They attribute

the positive impact to foreign investors exporting and demanding good governance in the host country.

Stulz (2005) notes that financial globalisation lowers the ability to expropriate minority investors, as any attempt of expropriation will compel the investors to move their investments to alternative countries with better investor protection standards. Similarly, foreign investors will avoid investing in the domestic market if it offers weak investor protection. Stulz (2005) suggests that financial globalisation helps controlling shareholders to learn better investors' protection mechanisms from countries with strong investor protection. Globalisation assists companies to gain access to a pool of specialised skills and knowledge. Oman (2001) suggests that international investors' single most important determinant of investment decision is based on a good governance system as it relates to a country's policy credibility. Arndt and Oman (2006) argue that foreign investors drive up the domestic country's government interest in good governance. La Porta et al. (1997) advocate that a better legal environment keeps entrepreneurs from expropriating investors' funds, and thereby increases investors' protection against expropriation by insiders accounts for discrepancies in the nature and effectiveness of stock market development.

The prevailing assumption is that financial globalisation via foreign equity investment inflows will make it difficult for the state to expropriate domestic investors because investors will seek investment outside the domestic country when there is pervasiveness of poor governance. The quality of investor protection prevailing in a country is important for international portfolio investors, due to their investment exposure to risk. We can hypothesize that foreign bias, whereby foreign investors increase portfolio weights towards ICAPM implied weight will have a positive effect on investor protection.

3.4.2 Contributions

Despite the importance of investor protection, limited studies examine factors that influence investor protection. Current studies examine the impact of investor protection on several variables.³⁹ For instance, Wei (2000) shows that corruption negatively relates to an open economy, as resources will be committed to improving institutional quality. Islam and

³⁹ See Chan et al. (2005)

Montenegro (2002) demonstrate that trade openness is robustly positively associated with institutional quality. The IMF (2005) displays that trade openness has a positive impact on both institutional transitions and the quality of economic institutions. Busse and Groning (2009) analyse the importance of trade liberalisation towards good governance. They find that trade openness has a positive influence on good governance.

To the best of our knowledge, there is no single empirical study that examines the implications of home and foreign bias on investor protection. It is likely that investor protection, in addition to factor accumulation and technological change, explains cross-country variations in growth performance. We provide evidence to show that sub-optimal portfolio allocation of investors has a varying impact on four proxies of investor protection we obtain from World Governance Indicators. The study makes several contributions to the existing literature that seeks to understand factors that influence country-level investor protection.

First, we contribute and extend the existing international finance literature by showing that higher degrees of home bias of domestic investors reduce investor protection standards, owing to the fact that the country will become segmented when the stock market is dominated by large domestic institutional investors who will develop close ties with companies and the state, thereby adversely influencing the protection provided to minority investors.

Second, we contribute and demonstrate that foreign bias, where foreign investors allocate investment towards the implied weight recommended by ICAPM, will lead to better investor protection because foreign institutional investors from well governed countries will export and demand good governance and put pressure on the state to improve investor protection. The state trying to maintain a good reputation and efforts to attract sufficient foreign investment, will improve on investor protection. Gillan and Starks (2003) suggest that foreign institutional investors play a significant role in prompting changes in corporate governance practices than domestic money managers. Previous studies fail to examine whether countries' investor protection improves when foreign investors increase their investment in the host country towards the implied world market capitalisation weight.

Third, our study makes additional contribution by using a micro global fund's allocation dataset of 44 countries to examine the effects of global fund foreign bias (GF_FB) on investor protection. The study employs four measures of country level governance indicators to capture protection offered to investors against expropriation, and how the protection is enforced. We use data constructed by World Governance Indicators (described in Chapter 4.3) to test the following hypotheses:

H₇ Higher home bias is related to a lower level of investor protection standards.

 H_8 Higher foreign bias is associated with higher levels of investor protection standards.

H₉ A higher country global fund bias is related to higher investor protection standards.

3.4.3 Summary of findings

Despite the important role of investor protection towards economic growth, as far as we are concerned, no single study has empirically examined the implications of the sub-optimal portfolio allocation (home and foreign bias) phenomenon on investor protection. The study represents the first attempt to empirically examine the effects of home and foreign bias on investor protection. Current studies show that investor protections have positive effects on economic growth and development.

Our findings provide compelling evidence that home bias inhibits investor protection, whilst foreign bias improves investor protection. The robust findings of our study show that about 71-81 per cent of variation in cross-country investor protection is explained by home and foreign bias. The policy implications of the study, particularly for emerging countries, is that encouraging domestic investors to diversify abroad will reduce their controlling influence in firms and will improve investor protection. Similarly, countries can experience better investor protection when they attract sufficient foreign equity investment as a result of foreign investors from better governed countries, demanding and exporting good governance to the host country.

Chapter 4: Data

4.1 Home and foreign bias measures

We use three different datasets to construct the measure of home and foreign bias exhibited by portfolio investors in the country allocations. The first dataset we employ is the standard aggregate country level cross-country equity portfolio holding (in USD millions) data, sourced from the Coordinated Portfolio Investment Survey (*CPIS*) of the International Monetary Fund (IMF). The CPIS collects data on stocks of cross-border holdings of equities for 76 participating countries. For detailed descriptions of these standard data please refer to Bekaert and Wang (2010). We use the annual cross-country portfolio holdings *CPIS* data for the period 2001 to 2010 to construct our equity home bias (*CPIS_HB*) and equity foreign bias (*CPIS_FB*) measures, as described below. Dictated by the availability of data on our key cost of capital proxies and other control variables, we are able to use data on 44 countries out of the list of the 45 highly investable MSCI All Country Index.

The second dataset we use is unique fund level country allocation data from Emerging Portfolio Fund Research (*EPFR*) to create the global fund's foreign bias (*GF_FB*) measure. *EPFR* avails asset allocation data trading in traditional and alternative funds domiciled globally. The aim of the country allocations is to provide a complete and comprehensive picture of fund managers' allocations driving global markets. We use the yearly average (using monthly allocations) country allocations of 122 global equity funds with the total size of all the funds being approximately US\$120 billion. These funds are domiciled across nine countries for the period of 2001 to 2010. As these are purely global funds at a micro level, we should expect the foreign bias to be minimal, compared to the CPIS aggregate data which include various (undisclosed) funds' type and style.

The EPFR is a fairly novel database and had been employed by existing researchers to address different issues (see Jotikasthira et al. 2012; Wei et al. 2010; Broner et al. 2006; Gelos and Wei, 2005). In order to maintain consistency with CPIS data, we incorporate the same sample period of 2001 to 2010 for the EPFR data. Furthermore, as the funds are domiciled only in nine countries (see appendix 1) we are unable to construct a robust measure of home bias, due to the smaller number of observations for our empirical analysis (90

observations only). Appendix 1 provides a detailed summary of information on the yearly average statistics of EPFR funds.

Finally, for the construction of ICAPM benchmark allocation, the third dataset we use is the country level market capitalisation figures of S&P/IFC obtained from the World Development Indicator (*WDI*) of the World Bank. In the following section we describe the construction of our home and foreign bias measures.

4.1.1 Equity home bias

The equity home bias (*EHB*) measure captures the degree to which domestic investors overweight their domestic equity market, relative to the prescription of the ICAPM benchmark. Following the existing literature (e.g. see Ahearne et al. 2004; Chan et al. 2005), we define home bias as:

$$EHB_{lt} = \log\left(\frac{W_{llt}}{W_{lt}^*}\right) \tag{4-1}$$

where w_{llt} represents domestic investors' weightings in the domestic market capitalisation of country *l* for the period *t* and is defined as:

$$w_{llt} = \frac{h_{llt}}{GPH_{lt}} \tag{4-2}$$

where h_{llt} is the stockholdings of domestic investors in their domestic market l and GPH_{lt} is the global holdings of domestic investors for the period t across all 44 countries, including domestic. w_{lt}^* is the ICAPM world benchmark allocation for country l for time period t, which is the same for all investors in all countries and is defined as:

$$w_{lt}^{*} = \frac{cap_{lt}}{\sum_{l=1}^{44} cap_{lt}}$$
(4-3)

 cap_{lt} is the total market capitalisation of country *l* and is obtained from WDI. A value of zero for EHB_{lt} in equation (4-1) indicates that investors have no bias towards their home market, while positive values show the presence of home bias.

It's worth noting that CPIS only reports the bilateral foreign equity portfolio holdings with no investments in domestic markets for each host country *l*. Following the existing literature (Fidora et al. 2007) the construction of domestic holdings (h_{llt}) and global portfolio holdings of domestic investors (GPH_{lt}) are as follows:

$$h_{llt} = MCAP_{lt} - \sum_{k=1}^{43} FPH_{klt, k\neq l}$$

$$\tag{4-4}$$

where $MCAP_{lt}$ is the market capitalisation of equities issued in country l and $FPH_{klt, k\neq l}$ is the holdings of all equities of country l by foreign investors domiciled in country k. The GPH_{lt} is thus constructed as:

$$GPH_{lt} = h_{llt} + \sum_{k=1}^{43} FPH_{lkt, l \neq k}$$
 (4-5)

 FPH_{lkt} is the holdings of foreign securities (k) by investors domiciled in country l at time t. As the EHB_{lt} is constructed using CPIS data, we denote this variable as *CPIS_HB* in our empirical analysis.

4.1.2 Equity foreign bias

Relative to the ICAPM prediction, equity foreign bias implies the disproportionate allocation of investors domiciled in country k into the foreign securities of countries l. Following Chan et al. (2005) we compute the equity foreign bias as

$$EFB_{klt} = \log\left(\frac{W_{klt}}{W_{lt}^*}\right) \tag{4-6}$$

where w_{klt} is the allocation of country k's investors in equities issued by country l for the period t and is defined as

$$w_{klt} = \frac{h_{klt}}{\sum_{l=1}^{43} h_{klt}}$$
(4-7)

where h_{klt} denotes country k's investors' stockholdings of equities in a foreign country l for the period t. w_{lt}^* , as defined in equation (4-3), is the ICAPM benchmark allocation for investing in country l for period t. For each pair countries, i.e, kl, equity foreign bias could be either positive, where foreign investors' (in country k) overweight foreign equity market more than that suggested by the implied global weight, or it could be negative, where foreign investors underweight their investment away from the implied global weight.⁴⁰ For regression analysis we take the average equity foreign bias (*AEFB*_{lt}) exhibited by all source country investors (k=1....n) for the country l for each period t as shown below:

$$AEFB_{lt} = \frac{\sum_{l=1}^{n} EFB_{klt}}{n}$$
(4-8)

Foreign bias measures based on CPIS-IMF and EPFR Global Funds' data are denoted as *CPIS_FB* and *GF_FB* respectively. The number of source countries, i.e. *n*, for *CPIS_FB* is 43 (i.e. the same as host countries, excluding the country for which foreign bias is measured, $k \neq l$) and the *n* for the *GF_FB*, i.e. the number of funds exhibiting foreign bias for each country, is 121, excluding the fund if its country of domicile is the same as the country of allocation, i.e. $k \neq l$.

4.2 Sub-optimal international portfolio allocations and cost of capital

We begin the discussion on data by describing the four different costs of capital measures we incorporate in our primary analysis, followed by a measure of home bias and two measures (using macro and micro data) of foreign biases. Finally, following the existing literature we discuss the control factors that could potentially compete with our measures of home bias and foreign biases. We employ four variables as proxies for country-level cost of capital measures: historical realised market risk premium, sovereign bond credit-risk rating, based on implied cost of capital proxy, Damodaran's (2012) default spread-based country risk premium and, finally, the dividend yield.

4.2.1 Historical realised returns of the market

Following Lau et al. (2010) the first proxy we employ is the historical realised return of the markets (*HRRm*) which is the historical average of excess country equity market return over

 $^{^{40}}$ Note, on aggregate the foreign bias should be negative for each country (*l*) which exhibits home bias. However, given the fact that CPIS does not report the holding all countries in the world, on average, the foreign bias could be positive or negative. Such figures are also reported in the existing literature (see Chan et al. 2005 and Lau et al. 2009).

the risk free rate. For each year, the yearly average stock market returns are computed using the monthly US dollars country stock market indices, sourced from Morgan Stanley Capital International (MSCI). As all returns are denominated in US dollars, we use the yearly average of the monthly return on US Treasury bills as a proxy for the risk free rate for all countries. One of the fundamental assumptions of using the historical risk premium as a proxy for the expected risk premium is that the long term average premium is mean reverting. Although this measure may be prudent for the developed markets where long historical data are available, yielding a lower degree of standard errors, the same luxury could not be accorded for most of the emerging markets which have a relatively shorter history of stock return data. We address this issue by employing alternative average risk premium measures as discussed in the following sections.

4.2.2 Sovereign credit-risk rating measures of cost of capital

As noted above, the disadvantage of using a historical equity risk premium measure, particularly for emerging markets, is that it may yield a higher degree of standard errors, owing to the shorter history of the data used. As such, following the existing literature, we use two additional proxies based on sovereign country credit risk ratings. The first is the proxy of implied cost of capital, estimated using the sovereign credit ratings and is denoted as *rCred*. The second is Damodaran's (2012) country risk premium which is based on sovereign default spread, but adjusted for relative risk of equity versus bond markets' expected return. We denote it as *CERP*. In this section we describe the first measure (*rCred*), followed by the description of the second (*CERP*) in Section 3.1.3.

Following Jewel and Livingston (1998), for each country we use sovereign bond risk rating, denominated in foreign currency, as the proxy of implied cost of capital. The basic idea is that sovereign country credit ratings display fundamental forward looking information on country's risk and, unlike expected return measures based on historical data, do not suffer from the noise of past shocks to a country's growth opportunities. Previous studies show that country credit-risk rating, highly correlates with implied cost of capital, and thereby, could be used as an alternative proxy of cost of capital. For instance, Hail and Leuz (2006) find the sovereign credit ratings measure is highly and significantly correlated (the average correlation coefficient being 0.64) with implied cost of capital. Further, Bhattacharya and Daouk (2002) also suggest that country credit rating is a reliable proxy for *ex ante* risk exposure, particularly for segmented emerging countries. We obtain country credit-risk
ratings of 10-year local currency denominated sovereign bonds from Damodaran's website. Following Reeb et al. (2001) we convert the qualitative credit ratings into numerical values, based on a scale of 1-22. We assign a value of 1 to AAA=1, AA+=2, AA=3....all the way to D=22 and take their natural log into our regressions. We expect a positive (negative) association between *rCred* and home bias (foreign bias) in our estimations.

4.2.3 Country equity risk premium

As an alternative proxy, we employ the country equity risk premium (CERP) constructed and maintained by Damodaran (2012). The CERP measure follows the concept of demanding incremental CERP for investing in a particular market, relative to a matured market as a base country. Damodaran uses the United States as the base country and S&P 500 as the representative stock market. For each country the incremental premium, relative to the base country which reflects the additional country risk premium, is computed by taking the default spread (over the base country) following Moody's risk ratings of sovereign bonds in local currency. The resultant premium is subsequently scaled by the ratio of the country's equity market volatility to bond market volatility. For instance, in calculating the equity risk premium for Brazil, Damodaran first determines the default risk premium spread of a 10-year local currency denominated government bond, over the 10-year US sovereign bond. The resultant premium is subsequently adjusted for the additional risk of the equity market by scaling it with the ratio of standard deviations of Brazil's equity to bond market. The standard deviations of the bond market are estimated, using returns on the 10-year sovereign bonds and those of the local equity market using the national stock index, e.g. Bovespa for Brazil. For more detail see Damodaran (2012).

4.2.4 Dividend Yield

Following its extensive use in the existing literature (see Bekaert and Harvey, 2000; Lau et al. 2010) we employ dividend yield (DY) as an additional proxy of cost of capital. Within the framework of the capital asset pricing model, Lau et al. (2010) suggest that dividend yield is a stable and easily measurable proxy of cost of capital. Furthermore, Bekaert and Harvey (2005) also show that, relative to historical realised returns, dividend yield is a better proxy of cost of capital, particularly for emerging markets where returns are relatively more volatile than those of their developed counterparts. We obtain DY measures for all countries from Thompson Reuters and World Federation of Exchanges.

4.2.5 Control variables

Following the extant literature we use a number of control factors which could potentially compete with our measure of home and foreign bias measures. Although a host of controls are suggested, not all have been shown to have the expected sign on country level cost of capital measures (See Lau et al. 2010). With the view of mitigating omitted variable bias, we use market capitalisation (*MKTCap*) as a proxy of size, market beta (*Beta*) to capture the basic CAPMs systematic risk, book-to-market (*BM*) ratio, one year lagged stock performance (*Retn_1*) to control for momentum effect, exchange rate risk (*Exch*), expected inflation (*Infl_1*), turnover ratio (*Turn*) as a measure of liquidity, stock market integration (*LSMI*), real gross domestic product growth (*RGDPG*), rule of law (*Law*) for variations in legal institutions, country political risk (*PolRisk*), macro-economic risk (*EconRisk*) and macrofinancial risk (*FinRisk*).

Following Kang and Stulz (1997), we employ the log of market capitalisation (*MKTCap*) in USD millions to control for the size and information of the market. Hail and Leuz (2005) note that a larger stock market provides more transparent information which could potentially reduce information costs and, hence, lower cost of capital. We expect *MKTCap* to be negatively related to cost of capital measures. *ICAPM* predicts a positive association between a firm's beta and risk premium. We compute beta for each market as the covariance of the *MSCI* country index return with *MSCI* All Country World index return, using monthly figures for the past five years. The resulting covariance is then divided by the variance of *MSCI* world index return. Following Gebhardt et al. (2001), we use book-to-market ratio (*BM*) to capture differences in growth opportunities. High *BM* captures lower growth opportunities, lower accounting conservatism, and high perceived risk. We calculate *BM* as the log country level ratio of book-to-market and scale it by the total market capitalisation. We source the book value, figures from Worldscope.

Consistent with Jegadeesh and Titman (1993), we use the previous year's stock performance $(Retn_1)$ to capture the momentum effect on the cost of capital. We measure $Retn_1$ as the average *MSCI* monthly index return over the past year. We further employ one year lagged inflation risk $(Infl_1)$ to control for macroeconomic effect. We control for inflation $(Infl_1)$ to ensure our analysis is not driven by variations in expected inflation rates as Brandt and Wang (2003) present evidence that equity risk premium is positively related to the inflation rate. We

obtain one year lagged annual inflation rates for each country from *WDI*. To control for the correlation between exchange return and equity return (*Exch*), for each country we compute a three-year moving average covariance between the monthly stock market index return and the monthly depreciation of the domestic currency with respect to the dollar. The exchange rates are sourced from Thompson Reuters (see Adler and Dumas, 1984).

It is well established in the literature that investors demand liquidity premium for investing in markets with different levels of liquidity (see Jones, 2003; Gibson and Mougeot, 2004; Bekaert et al. 2006; Chordia et al. 2011). We control for market liquidity using the turnover (*Turn*) ratio obtained from *WDI*. Earlier studies (Bekaert and Harvey, 1995; Chan et al. 2005) argue that trade openness may drive the cost of capital. As such, we further control for stock market integration (financial and economic) on the cost of capital by using the log average of a country's annual exports and imports, scaled by *GDP* (*LSMI*). We also expect that countries with higher economic growth would attract foreign investors, implying global potential risk-sharing, owing to growth-reputation, which may reduce the cost of capital. We control for *WDI*.

La Porta et al. (1998) argue that countries with sound and effective legal systems protect outside investors from expropriation risk. We use the International Country Risk Group (*ICRG*) rule of law and order (*Law*) index (0-6) to capture the variations in the quality and observance of legal rules. We obtain the data from Political Risk Services Group's country risk ratings. Further, studies such as that of Erb et al. (1996), also note that differences in country risk ratings may influence equity returns. Following Erb et al. (1996), we use the extensively employed three broad country risk measures, i.e. Political risk (*PolRisk*), economic policy risk (*EconRisk*) and financial policy risk (*FinRisk*) to control their effect on the cost of capital.

The objective of the broad country risk ratings is to provide a common but forward looking platform for assessing the political, economic and financial policy stability in countries covered by ICRG. The *political risk rating* is rated on a scale of 0-100 and comprises 12 components, rated on the basis of pre-set questions for each component. The *political risk rating* captures issues such as, government stability, socioeconomic conditions, investment profile, internal conflict, external conflict, corruption, military in politics, religious tension, law and order, ethnic tensions, democratic accountability and bureaucratic quality. The

economic policy risk is measured on a scale of 0 (low) to 50 (high), incorporates five potential sources of economic risk (GDP per head, Real GDP growth, inflation rate, budget balance as a percentage of GDP, and current account as a percentage of GDP). The *financial policy risk* measure which ranges between 0 (low) and 50 (high), captures five potential sources of financial risk components (foreign debt as a percentage of GDP, exchange rate stability, foreign debt as a percentage of total export and services, current account as a percentage of exports and services, and international liquidity). For all the broad country risk measures, the annual average based on the monthly ratings is used in this study. For further details on the methodology see ICRG (2012).

4.3 International portfolio investment and stock market development

In order to test the effect of home and foreign bias on stock market development hypotheses, we adopt four variables widely used in finance and economic literature; for instance, (Levine and Zervos, 1996; Demirguc-Kunt and Maksimovic, 1996; and Chan et al. 2005) to proxy for stock market development. We also provide the relevance and problems using these proxies. The stock market development measures include market capitalisation as a percentage of GDP (*MGDP*), stock value traded as a percentage of GDP (*TRGDP*), stock turnover ratio (*TURN*) and, finally, transaction cost (*TRCOST*). Our second empirical study focuses on the impact of home and foreign bias on stock market development.

4.3.1 Market capitalisation as a percentage of GDP

The first measure we use as proxy of stock market development, is the ratio of market capitalisation to gross domestic product (*MGDP*). This proxy captures the size and importance of the stock market relative to the entire economy and the extent to which investors can diversify risk. Large stock markets may not function well, and taxes may also inhibit companies from listing. La Porta et al. (1998) demonstrate that *MGDP* captures the breadth of the market, as well as the importance of the stock market in the economy. Levine and Zervos (1996) argue that large size stock markets do not necessarily function effectively, as taxes may distort the incentive to list on the stock exchange. Several studies, for instance (see Levine, 1997; Rousseau and Wachtel, 2000; Durham, 2004), use *MGDP* as an indicator of stock market development. The latter measure also captures the size and market diversification prospect of a country's stock market. A number of studies (La Porta, 1997, 1998, 2000) use this measure as an indicator of market development under the assumption

that stock market size is positively correlated with the ability to mobilise capital and diversify risk. We use this measure under the assumption that this proxy is less arbitrary than other individual measures of stock market development.

4.3.2 Stock value traded as a percentage of GDP

The second stock market development proxy we use is stock value traded during the year as a percentage of GDP (*TRGDP*), which is referred to as the value traded ratio. This proxy is used to complement the market capitalisation to GDP measure as bigger markets may not necessarily be the most active market. *TRGDP* captures stock market liquidity but it does not capture trading cost or the uncertainty that drive trading on a particular market. *TRGDP* positively captures liquidity on the economy. Market capitalization and value traded ratios provide further information about the country's stock market than using a single measure. *TRGDP* is not a direct measure of trading cost or the uncertainty associated with trading on a particular exchange. Although *TRGDP* may not be a direct measure of microstructural sophistication, studies (Levine, 1991 and Bencivenga et al. 1996) motivate the use of value traded to GDP as an indirect measure of liquidity, particularly reflecting liquidity level relative to the size of the economy.

4.3.3 Turnover ratio

The third proxy for stock market development is turnover ratio, which we use to capture the degree of trading activity of the stock market, as some stock markets may be large and inactive. Bekeart and Hodrick (2009) document that market turnover is used as a proxy of liquidity. Other studies suggest that, it is used as an indicator of the rate at which information arrives in the market, that instigates trades and is also related to costs of trading equities because high trading costs cause investors to trade less (see Levine and Zervos, 1998; Amihud and Mendelson, 1986 and Bekeart and Hodrick, 2009). Large stock markets do not indicate liquid market, as a large but inactive market will have large market capitalisation, but small turnover ratio. High turnover ratio indicates low transaction cost. Turnover reflects the ability to trade equity easily and is important to the market by easing tension and providing an asset to savers who can quickly and inexpensively sell when required. Kyle (1984) and Holmstom and Tirole (1993) argue that a liquid stock market can increase incentives for investors to obtain information about firms and improve corporate governance. Turnover ratio does not directly capture the theoretical definition of liquidity. A large but inactive stock market will have a large market capitalisation ratio but small turnover ratio. Whilst value

traded ratio reflects trading relative to the size of the economy, turnover ratio captures trading relative to the size of the stock market. Consequently, it shows that a small and liquid stock market will have a high turnover ratio but a small value traded ratio. Price effect, influences the value traded. To gauge the price effects, we use *TURN* to capture liquidity and is the inverse of transaction cost. High *TURN* implies low transaction cost. Following Demirguc-Kunt and Maksimovic (1999), Megginson et al. (1994), and Boehmer et al. (2005), we measure turnover ratio as the value of stock traded as a percentage of stock market capitalisation, as price affects value traded and market capitalisation.

4.3.4 Transaction cost

Both the above measures of liquidity are indirect factors of market microstructure. We also employ a relatively direct measure of market liquidity, which shows how costly it is to undertake a trade at an average value in a particular market. Transaction costs play a substantial role in shaping the return profile of an investment given a certain level of risk. Solnik and McLeavy (2004) show that transaction cost may materially reduce expected returns and hence potential benefits of diversification. As higher transaction costs are a drain on the expected returns of portfolio, markets with lower transaction cost should attract more investors, relative to markets with higher transaction costs.

Following Chan et al. (2005) and Thapa and Poshakwale (2010), we use a composite measure of market level transaction costs (in basis points), incorporating three different subcomponents of costs related to equity trading.⁴¹ These country level yearly trading cost figures are estimated and maintained by Elkins/McSherry (E/M) and reported in the yearly global stock market fact book of *Standard and Poor (S&P)*. E/M provide analyses of the global trading costs for 150 global institutions (pension funds, investment managers, banks and brokers) by showing estimates of the country level transaction cost figures for international investors. E/M's estimates of market level transaction cost, based on an average transaction in USD, are generated by aggregating three sub-components.

The transaction cost has three components: commission, fees, and market impact. Solnik and McLeavy (2004) show that investors pay commission to brokers for access to brokerage research facilities and services. Investors pay fees for accessing extra services, which include

⁴¹ These are the only aggregate country level proxies for average transaction cost measures sourced from the literature and are available for country level studies in the panel data framework.

post-trade settlement costs. The S&P global stock market fact book (2007) defines market impact, as the difference between the price at which a trade is executed and the average of stock's high, low, opening and closing prices during the trade. In other words, the market impact indicates the influence that large investors, for instance, institutional investors have on stock prices when they purchase or dispose of equity. It is measured as the change in price, divided by the volume of trade.

4.3.5 Control variables

We control for several factors which may have a possible effect on stock market development. Existing literature shows that stock market development is impacted by portfolio flow (*Flow*), savings (*Sav*), private credit (*PCred*), exchange rate (*Exch*), expected inflation (*Infl*), interest rate (*Int*), one year lagged stock performance (*Retn_1*) to control for momentum effect, stock market integration (*LSMI*), gross domestic product per capita (*GDPPC*). We use market capitalisation (*MKTCap*) to capture price effect and law and order (*R_Law*). We use legal origin (*Legal_O*), financial risk (*FinRisk*), and corruption (*Cor*) to allow for variations in institutional quality. We briefly discuss the economic rationale for including these variables in the model.

First, we use *Flow* to capture the extent of flow of foreign direct investments on stock market development, which reflects the degree of capital flows restrictions imposed by a country. Foreign investors are more likely to make direct investments in countries where stock markets are subject to less investment controls on foreign direct investments, and will thereby improve stock market development. For example, Errunza and Rosenberg (1982), show that foreign capital inflows have a positive effect on stock market development. We measure *Flow* as the logarithm of net foreign direct investment as a percentage of GDP. We obtain data from World Bank Indicators (WBI) of the World Bank.

The second control variable is savings (*Sav*), and it reflects the degree to which savings impact on stock market development, as investments are driven by the mobilisation of domestic financial resources. The positive effect of domestic saving on investment is corroborated by Obstfeld and Rogoff (2000). Countries with high levels of savings, experience a greater level of capital flows for investment, which leads to high stock market development. For instance, Garcia and Liu (1999) demonstrate that macroeconomic variables such as the savings rate have a positive impact on stock market development. We measure

Sav as the natural logarithm of a country's gross domestic savings and we expect a positive association with stock market development. We source annual data from World Bank Indicators of the World Bank.

The next control variable is private credit (*PCred*). We use *PCred* to capture financial intermediary development and the activity of commercial banking to provide domestic long-term finance to the private sector (see Bekaert et al. 2005; Claessens et al. 2006). Private credit refers to financial resources provided to the private sector such as: loans, purchase of non-equity securities, trade credits and other accounts receivable that establish a claim for repayment. We expect *PCred* to be positively related to stock market development. Following King and Levine (1993a) and Beck et al. (2000a), we calculate *PCred* as domestic credit to the private sector scale by GDP. We access the data from World Bank Indicators.

The next three control variables we include in the model capture macroeconomic stability. Existing studies, for instance Boyd et al. (2001), demonstrate that macroeconomic stability influences stock market development. Countries with macroeconomic stability will attract significant investors to participate in the stock market, and thereby, help with stock market development. A better macroeconomic environment leads to financial development (see Bencivenga and Smith, 1992; Huybens and Smith, 1999). We use these three indicators: interest rate (*Int*), exchange rate exposure (*Exch*) and one year lagged inflation risk (*Infl_1*) to control for macroeconomic effect on stock market development. Billmeier and Massa (2009) argue that countries with poor domestic economic fundamentals, force investors to use international markets more extensively and have therefore, been blamed for capital flight. Better macroeconomic fundamentals boost domestic investors' confidence and willingness to participate in the stock market, which generates more savings for investment. Boyd et al. (2001) note that, macroeconomic stability promotes stock market development.

High interest rates force investors to reduce their investment in equity, and participate more in bonds and savings with financial institutions, than investment in equity. If companies borrow at high interest rates to expand business, it will increase cost of debt. This can reduce companies' profit and dividend payments, which will result in the reduction of the share price, thereby reducing market capitalisation. We expect real interest rate (*Int*) to be negatively related to stock market development. Exchange rate exposure captures the risk associated with foreign country equity investment. Countries with stable exchange should experience higher flow of foreign investment, thereby improving stock market development. Following Lau et al. (2010), we construct exchange rate (*Exch*) exposure for each country as a three year moving average covariance of the monthly stock market index return, with the monthly depreciation of the domestic currency with respect to the dollar. The exchange rates are sourced from Thompson Reuters (see Adler and Dumas, 1984). If stock returns correlate with exchange rate changes, stocks are implicitly exposed to the exchange rate risk.

We control for inflation (*Infl*) to ensure our analysis is not driven by variations in expected inflation rates, as inflation suggests high consumer prices. This usually slows sales and reduces companies' profit. High inflation leads to high interest rates as the Central Bank will raise interest rates to reduce inflation. The high levels of inflation will bring down stock prices and thereby reduce market capitalisation and value traded. Brandt and Wang (2003) present evidence that equity risk premium is positively related to inflation rate. Stable and moderate inflation levels reflect macroeconomic stability. High levels of volatility of the price will lead to high degrees of uncertainty in stock markets and less incentive for investors to trade in stock markets. The Fisherian hypothesis suggests that the rate of return on equity moves with the rate of inflation. Expected rate of return on equity, consists of real returns, plus the expected rate of inflation. Real return does not move systematically with the rate of inflation. Equity investors will, on average, be fully compensated for the erosion in purchasing power. However, Nelson (1976) and McCarthy et al. (1990) find a negative relation between stock returns and inflation.

Next, we control for return momentum ($Retn_1$). The economic reasoning is that, past positive return will encourage momentum investors to buy more equity, leading to stock market development. Investors are momentum traders or return chasers and they base their equity investment decisions on stock markets' past performance. Investors are more likely to invest in the stock market when past returns are high, which leads to stock market development. Following Jegadeesh and Titman (1993), we use the preceding year's stock performance ($Retn_1$) to capture the momentum effect on stock market development. Consistent with Bohn and Tesar (1996), we expect $Retn_1$ to have a positive effect on stock market development.

We use log stock market integration (*LSMI*) to capture the marginal effects of the degree of trade openness on stock market development. Existing literature shows that trade openness

positively drives stock market development (see Bekaert and Harvey, 2000; 2003; Edison and Warnock, 2003; Henry, 2000b; Kaminsky and Schmukler, 2003; Kim and Singal, 2000; Do and Levchenko, 2004; Huang and Temple, 2005). We calculate stock market integration, by using the natural logarithm average of a country's exports and imports scaled by GDP. Earlier research by Bekaert and Harvey (1995) incorporates time-varying measures of market integration.

Drawing from the existing literature, for instance Bartram and Dufey (2001) who suggest that economic growth and development relate positively to stock market development, we use gross domestic product per capita *GDPPC* to capture income level and overall economic development. With a better economic outlook or higher expected GDP growth with the expectation of economic expansion, stock prices may rise. Investors may buy more stocks with the expectation of future profits and higher prices. If the economic outlook is uncertain, investors may reduce their investment, which reduces market capitalisation. Levine (1997) finds that stock markets tend to develop as income per capita increases and institutional environment improves. We expect income to have a positive effect on stock market capitalisation (*MKTCap*) to capture price effect on firm size, which are also included as a risk proxy that could explain cross-country variation in stock returns (see Fama and French, 1992; 1993). Evidence shows that *MKTCap* has a positive impact on stock market development. We measure *MKTCap* as the logarithm of a country's previous fiscal year-end market capitalisation.

We control for institutional quality, as it may influence stock market development. Quality institutions have sets of rules, practices and norms that shape social and political interactions. Better institutions are linked to more transparent, less corrupt, good governance, better protection of the minority investors and property rights. This subsequently improves investor confidence and thereby leads to stock market development. Karolyi (2004) shows that poorly functioning stock markets, are a result of weak political, legal and other institutional factors. Literature on capital flight suggests that when institutional weakness persists in a country, as argued by Collier et al. (1999), Schineller (1997) and Sheets (1996), domestic investors tend to invest a greater proportion of their investment abroad, which adversely affects stock market development. Domestic investors may be inclined to avoid poor domestic institutional environments and invest in foreign countries where there is better protection of investors.

Weak domestic institutional environments negatively affect stock market development, whilst improved institutional fundamentals, both explicitly and implicitly, positively impact on stock market development. Legal and institutional environments have been found in the existing literature, to have effects on stock market development (Beck et al. 2003; La Porta et al. 1997; 2000a, b; Pistor et al. 2000). We use rule of law (R_Law) to capture institutional quality.

Furthermore, we use legal origin (*Legal_O*), financial risk (*FinRisk*) and corruption (*Cor*) to capture institutional quality effects on stock market development. Existing studies suggest that countries with better legal institutions relate to efficient stock markets (see Lombardo and Pagano, 1999; La Porta et al. 1998; Demirguc-Kunt and Maksimovic, 1998). Ergungor (2008) argues that countries with efficient judicial systems have a positive influence on stock market development. La Porta et al. (1997) argue that the type of legal regime is a good proxy for the degree of investor protection and investors may be reluctant to invest in markets with a poor legal system and high rates of corruption. For instance, La Porta et al. (2002) demonstrate that well-functioning legal institutions lead to higher share prices.

Following La Porta et al. (1998), we use *Legal_O* to control for the type of legal origin on stock market development. La Porta et al. (1998) show that countries with common law legal origin, experience higher levels of investor protection than civil law origin countries. Civil law countries have more concentrated investors, and the underlying assumption is that better shareholder protection will improve stock market development. Shleifer and Vishny (1997) argue that better investor protection reduces the extent of expropriation by corporate insiders, and thereby improves stock market development.

We employ financial risk (*FinRisk*) policy measure of ICRG to capture a country's ability to meet its debt liability. We expect *FinRisk* measure to have negative impact on stock market development. We expect high *FinRisk* to reduce investors' confidence in the economy and will thereby reduce their participation in the stock market. Investors will sell their shares which will in turn, lower share price, and thereby adversely market capitalisation. Karolyi (2004) argues that growth of ADR in emerging countries is the consequence of poor functioning stock markets, which thereby create incentives for firms to leave.

Finally, we use corruption (*Cor*) to capture country-level institutional quality in preventing corrupt practices.⁴² Corruption exposes equity investors to the risk of expropriation through insider trading and economic mismanagement by corrupt government officials. The prevalence and persistence of corruption will reduce investors' participation in the stock market, which will reduce stock market development. Lombardo and Pagano (2000) demonstrate that corruption negatively relates to stock market development. Bekaert et al. (2005) suggest that corruption inhibits the economic and financial environment and reduces the efficiency of government and business by enabling people to assume positions of power through political patronage, rather than ability. We expect corruption to have a negative effect on stock market development. We source annual corruption data from ICRG.

4.4 International portfolio investment and investor protection

Data on governance are available from several sources which are mainly based on surveys. The study employs four measures of investor protection standards or governance indicators constructed by World Governance Indicators (WGI) for the period 2001-2010. The data captures three components of investor protection. First, the legal rights granted to minority investors. Second, the enforcement component, which captures the quality of a country's institutions in determining the extent investor's rights are respected and enforced. Third, the sound business and institutional environment in which corporations operate to create value for shareholders. Since 1996, WGI has employed data from 31 sources, including the Economist Intelligence Unit, Global Insight and Political Risk Services to provide six broad dimensions of governance indicators for 215 countries. WGI constructs the governance measures by averaging together data from several underlying sources that correspond to the concept of governance we seek to capture. WGI's careful construction of the measures, its worldwide coverage, and the suggestions that the data achieves maximum precision, makes the data very attractive.

The WGI dataset have widely been used in existing studies (see Knack, Kugler and Manning, 2003; Neumayer, 2002; Apodaca, 2004; Hart et al. 2005; Llamazares, 2005; Andres, 2006; Das and Andriamananjara, 2006; Jung, 2006; Liu and San, 2006). Other studies use the data to examine the association between governance and growth (see, for instance, Kaufmann et

⁴² Studies show that better legal and institutional environments have positive effects on stock market development (see Beck et al. 2003; La Porta et al. 1997; 1998; 2000; Pistor et al. 2000).

al. 2002; Dollar and Kraay, 2003; Kaufmann and Kraay, 2003; Naude, 2004; Me´on and Sekkat, 2005). We focus on four governance indicators which capture the concept of our study which relate to investor protection; government effectiveness, rule of law, regulatory quality, and control of corruption. WGI data have the advantage of being constructed from aggregate information from various sources, which probably makes it contain less measurement error than other governance data sources. Kaufmann et al. (1999) argue that other existing governance indicators serve as imperfect proxies as they cover much smaller concepts of institutional quality. The governance measures we use in the study are continuous variables by virtue of averaging three years of an ordered variable which takes discrete values that range from 0 to 100.⁴³

4.4.1 Government effectiveness (quality of institutions)

The first investor protection measure we employ is the composite rating referred to as Government effectiveness (Gov_Eff). Government effectiveness captures the quality of government policy formulation and implementation, and the credibility of government's commitment to such policies. It reflects the quality of public institutions and the extent to which they are independent from political pressures, which help create and maintain sound business environment to motivate managers and entrepreneurs to maximise firm's operational efficiency and return on investment (see La Porta et al. 1998 and Bekaert et al. 2007). It does so by ensuring corporations conform to investors' interest and expectations by reducing abuse of power, moral hazard and self-serving behaviour of corporate insiders. Specifically, Gov_Eff manifests institutional effectiveness and the quality of bureaucracy in protecting both domestic and foreign investors. Institutional effectiveness enables the establishment of management's behaviour monitoring tools, warranting accountability and provision of cost effective protection of minority investors. As such, institutional effectiveness facilitates and stimulates the performance of corporations. Excessive bureaucracy and red tape, hinders business activities, and discourages creativity and innovations in the business environment. Less bureaucracy leads to prompt decisions which effectively allow investors and corporations to easily go about their business.

⁴³ Most governance literature, for instance, treat corruption measures as a continuous variable which therefore makes it appropriate to use the OLS estimation method (See Brunetti and Weder, 2003).

 Gov_Eff is rated on a scale of 0 - 100, with countries scoring low (minimum zero) against this measure, reflects weak institutional quality. Similarly, countries scoring high (maximum 100), indicates bureaucratic strength and expertise to govern without drastic changes in policies or interruptions in the provision of services offered by government.

4.4.2 Control of corruption

Control of corruption (*Con_Cor*) reflects the perception of the extent to which public power employed for private benefit is controlled. Wei (2000) notes corruption⁴⁴ reduces the inflow of foreign investment, as it has the same effect as tax. Similarly, Mauro (1995) reports negative association between level of corruption and rates of investment. The prevalence of corruption threatens foreign investments by distorting the economic and financial environment, and lowering the efficiency of government and business activities. Furthermore, as corruption creates an unfavourable business environment, it increases cost of capital for businesses and investors.

WGI measures corruption using variables such as; widespread corruption among government officials, the level of public trust in politicians, corruption between governments and foreign companies, prosecution of public abuse, and the frequency to which companies make extra payments in connection with taxes, customs duty and the judiciary for favour. The WGI measure of control of corruption, ranges from 0 (low control of corruption) to 100 (high control of corruption).⁴⁵ Higher values correspond to better governance or investor protection standards.

4.4.3 Regulatory quality

Regulatory quality (*Reg_Qual*) captures the perceptions of governments' ability in formulating and implementing sound policies and regulations that facilitate and promote private sector development. Few of the key factors used in the construction of the measure includes investment and trade freedom, effectiveness of anti-trust policies and the extent to which the country's legislation is compatible with, and respected by other countries' legal

⁴⁴ The definition of corruption ranges widely from routine tips and quick money to complicated schemes of favour between businessmen and civil servants. Most apparent common form of corruption includes demand for special payments and bribes with regards to import and export licenses, exchange controls, tax assessments, or loans. Corruption is also linked to excessive patronage, nepotism, job reservations, and favour-for-favour, secret party funding, and suspicious close ties between politicians and businesses (see Kaufmann et al. 2010).

⁴⁵ Existing studies modelling the causes of corruption are: (Treisman, 2000; La Porta et al. 1999; Fisman and Gatti, 2002; Brunetti and Weder; 2003 and Persson et al. 2003).

system. The prevalence of trade barriers, discriminatory taxes, competition legislation that prevent fair competition and access barriers to the capital markets, exhibit poor regulatory quality. The variable also captures the extent labour and trade regulations affect the growth of business. *Reg_Qual* is constructed on a scale of 0 to 100, with a higher score reflecting better quality of regulatory environment in which private businesses and investments, including foreign investments, compete in a free and fair atmosphere.

The careful construction of regulatory quality by WGI consists of an investment profile data compiled by the Political Risk Services Group (PRS) which are determined by PRS on the assessment of three sub-components of political risk: contract viability or risk of expropriation, payment delays, and repatriation of profits (see Bekaert et al. 2007).

4.4.4 Rule of law

Rule of law (*Rule_Law*) captures the perception of the extent to which agents' exhibit confidence and abide by the police and courts, particularly quality of contract enforcement, property rights, protection of financial assets and intellectual property. *Rule_Law* also captures the independence of the judiciary and fair administration of justice. In case of foreign investors, an impartial judiciary and, enforceability of contracts and court orders should assist them in pursuit of judicial recourse when their interest is unlawfully affected. WGI employ data from several sources to construct the rule of law measure including factors such as timeliness of the judicial decisions, efficiency of legal framework for challenging regulations and equal treatment of foreign investors before the law as compared to domestic investors. The *Rule_Law* variable ranges from 0 - 100 with lower score reflecting poor and higher score better rule of law.

4.4.5 Control variables

Following the extant literature, we use several control variables that we expect to drive investor protection standards and also have possible impact on home and foreign bias. Table 8-2, columns (2)-(14) show the average values of 13 country level control variables: (1) market capitalization (*MKTCap*), (2) past year return (*Retn_1*), (3) turnover ratio (*Turn*), (4) market capitalisation to GDP (*MGDP*), (5) inflation (*Infl*), (6) log stock market integration (*LSMI*), (7) Tobinq, (8) legal origin (*Legal_O*), (9) press freedom (*Press*), (10) political stability (*PolStab*), (11) conflict (*Confl*), (12) education (*Educ*), (13) gross domestic product

per capita (*GDPPC*) for variations in economic development. We provide a brief justification for their inclusion.

First, we employ logarithmic market capitalisation (*MKTCap*) in USD millions to control for size and information effect on investor protection. Countries that have a larger stock market will have the financial resources to demand better investor protection. La Porta et al. (1998) argue that countries that have better laws that protect investors tend to have larger and deeper capital markets. Greater participation in the stock market, suggests that investors are satisfied with the level of investor protection provided to investors, which leads to deeper stock markets. Following Kang and Stulz (1997), we expect *MKTCap* to be positively related to investor protection.

It is conceivable that better past performance or a prosperous stock market will make financial resources available to investors to demand improved institutions and governance. To provide robustness to our analysis, we include the previous year's stock performance $(Retn_1)$ to capture the momentum effect, which makes extra financial resources available to investors. For instance, Jegadeesh and Titman (1993) show that firms and countries that experience better returns are associated with good governance systems that protect investors. Giannetti and Koskinen (2010) suggest that stocks have lower expected returns when investor protection is weak.

Next, we use Turnover ratio (*TURN*) and market capitalisation to GDP (*MGDP*) to capture stock market development impact on investor protection. Countries that have well developed stock markets have good institutions. Lau et al. (2010) suggest that countries with better governance tend to have larger stock market capitalisation scaled by GDP. *TURN*, captures market liquidity effects on investor protection. Variations in cross-country investor protection can be explained by stock market liquidity, which captures stock market development. Bekaert and Harvey (2000), and Bekaert et al. (2007) show that stock market liquidity, captures cross-sectional variations in market micro structure.

We control for inflation (*Infl*) to ensure that our analysis is not driven by variations in macroeconomic reforms which may further drive regulatory reforms. Neumayer (2002) suggests that low inflation rate reflects sound economic policies, which have positive effect on good governance. Busse and Groening (2009) argue that inflation captures distortions in

macroeconomic fundamentals, lack policy credibility and poor governance. High inflation rates will make companies and government officials engage in corrupt practices. Low inflation captures improvement in governance and we expect inflation to be negatively associated with investor protection.

The IMF (2005) demonstrates that trade openness relates positively to good governance. Additionally, Rajan and Zingales (2003) provide both theoretical and empirical links between trade openness and institutional development.⁴⁶ We, therefore, use the log of stock market integration (*LSMI*) to capture the level of trade openness on investor protection.

We employ *Tobinq* to capture the valuation effect on investor protection. Higher *Tobinq* implies better firm performance and more financial resources to investors to demand good governance. La Porta et al. (2002) provide empirical evidence that higher valuations of firms are associated with countries that have strong investor protection. Studies by Rajan and Zingales (1998), Baker and Wurgler (2000), and La Porta et al. (2000) demonstrate that higher financial resources improve investment protection standards. We employ Tobinq to capture the valuation effect on investor protection.

Coffee (2000) and Johnson et al. (2000b) offer explanations as to why common law countries provide better investor protection than civil law countries. We use legal origin (*Legal_O*) to address the type of a country's legal system or legal origin effect on investor protection. La Porta et al. (2000), Straub (2000), and Chong and Zanforlin (2000) are among those who suggest that common law countries provide better investor protection than civil law countries. Following La Porta et al. (1997), legal origin variable takes a value of 1 if a common law country, otherwise 0.

We use political stability (*PolStab*) and absence of violence (*Confl*) to capture stable countries. Apparently, countries that are politically stable and free from internal/external conflicts, are expected to have quality institutions to enhance good governance and better investor protection. For instance, Busse and Groening (2009) argue that internal and external conflicts have a negative impact on governance and investor protection as stable governments are more accountable. We use political stability (*PolStab*) and absence of violence (*Confl*) to

⁴⁶ See Wei (2000), Lafffont and N'Guessan (1999) for the impact of trade openness on corruption.

capture the effect of a stable government on investor protection. We obtained the rating index (0-100) on political stability from WGI with higher values reflecting higher level of stability. For the conflict factor, we use the aggregated index (0-12) of internal and external sub-component ICRG's political risk index, with higher values signifying lower risk of conflicts.

We employ press freedom (*Press*) to capture the effects of free media and free access to information on investor protection. The IMF (2005) shows that countries that allow adequate press freedom have better investor protection and less corruption. Adequate press freedom will allow journalists to report corrupt practices prevailing in a country. Politicians, business leaders, and the judiciary, will be aware of being exposed by the press and will, therefore, act appropriately. Press freedom serves as accountability of politicians in a country, and greater accountability of politicians relates to policies and institutional reforms that are beneficial for the wider economy. Djankov et al. (2001) show a free media and access to information, permits investors to provide checks and balances on each other's activities and government activities. Press freedom takes a value of 100 (high degree of press freedom), 0 (low degree of press freedom). We expect a high measure of press freedom to be associated with better investor protection. We obtain data from WGI and use the subcomponents of voice and accountability.

We use education (*Educ*) to capture human capital and the concept that, a better educated population will have better effective participation in wider processes of decision making and will demand better governance. Alesina and Perotti (1996) demonstrate the positive effect of education on institutional quality. The education effect on investor protection is examined in the works of Glaeser and Saks (2006), and Rauch and Evans (2000). Following La Porta et al. (1999a) and Haider (2009), we measure education as the total number of new entrants in the last grade of primary education, regardless of age, and expressed as a percentage of the total population of the official primary entrance age. We source the data from WDI.

Finally, we use GDP per capita GDP (*GDPPC*) to capture the effect of wealth and economic development on investor protection standards. We expect countries that have a high income level to have sufficient financial resources to establish strong institutions and to hold government accountable to provide better investor protection. Busse and Groening (2009) use real growth per capita GDP to capture the effects of financial resources in enhancing good governance. Bonaglia et al. (2001) suggest that the level of a country's economic

development influences the cultural attitudes towards investor protection and the resources that may be devoted to monitor public officials. Empirical studies by Gelos and Wei (2005) show that better investor protection is associated with a high level of economic development. Bris and Cabolis (2004) also note that better investor protection exists in countries that experience high economic development. La Porta et al. (1998) establish that gross domestic product per capita determines investor protection. We expect income level to have positive effect on investor protection standards. We obtain the data from WDI.

4.5 Summary of data

Table 4-1: Data description and sources

First Empirical Study: International Portfolio Investment on Cost of Capital: Dependent Variables

Abbreviations	Variables	Description	Source
HRRm	Historical realised return of the market	The historical average of excess country equity market return over the risk free rate. The yearly average stock market returns are computed using the monthly US dollars country stock market indices.	Obtained from Morgan Stanley Capital International (MSCI)
CERP	Country equity risk premium	Country equity risk premium	Constructed and maintained by Damodaran (2012)
rCred	Sovereign credit rating	Natural log of Sovereign bond risk rating, denominated in foreign currency, as the proxy of implied cost of capital. We convert the qualitative credit ratings into numerical values based on a scale of 1-22.	We obtain country credit-risk ratings of 10-year local currency denominated sovereign bonds from Damodaran's website
DY	Dividend yield	Dividend yield calculated as the total amount of dividend yield for a country as a percentage of a country's stock market capitalisation.	We obtain <i>DY</i> measures for all countries from Thompson Reuters and World Federation of Exchanges
Second Empirica	l Study: International Portfolio I	nvestment on Stock Market Development: Dependent Variable	es
MGDP	Market capitalisation as a percentage of GDP	Constructed as the log ratio of market capitalisation as a percentage of GDP to capture the breadth of the market as well as the importance of the stock market in the economy.	World Bank's World Development Indicator
TRGDP	Stock value traded as a percentage of GDP	Stock value traded as a percentage of GDP to complement the market capitalisation ratio and the trading activities of firm's shares.	World Bank's World Development Indicator
TURN	Turnover ratio	Turnover ratio as the value of stock traded as a percentage of stock market capitalisation.	World Bank's World Development Indicator

 TRCOST
 Transaction cost
 A composite measure of market level transaction costs (in basis
 Maintained by Elkins/McSherry and reported in the yearly Standard

 points)
 incorporating three different sub-components of costs
 and Poor Global Stock Markets Factbook.

 related to equity trading: commission, fees, and market impact.
 Impact

Third Empirical Study: International Portfolio Investment on Corporate Governance: Dependent Variables

Gov_Eff	Government effectiveness	Government effectiveness composite measure, captures institutional effectiveness and the quality of bureaucracy in protecting both domestic and foreign investors. WGI award 0 (low score) to countries with weak government effectiveness and 100 (high score) to countries that have government effectiveness.	Maintained by World Bank Governance Indicator - World Bank
Con_Cor	Control of corruption	Control of corruption captures the perception of the extent to which public power is employed for private benefit, which includes petty and large scale corruption. The WGI measure of control of corruption ranges from 0 (low control of corruption) to 100 (high control of corruption).	Maintained by World Bank Governance Indicator - World Bank
Reg_Qual	Regulatory quality	Regulatory quality captures the perceptions of how governments formulate and implement sound policies that promote private sector development and aid foreign investors to control in domestic countries. WGI use data from several sources to construct the regulatory quality measure, which ranges from 0 (low) to 100 (high).	Maintained by World Bank Governance Indicator - World Bank
Rule_Law	Rule of law	Rule of law captures the perception of the extent to which investors have confidence in a country's judicial system. Impartiality of the judiciary, enforceability of contracts and court orders. The rule of law variable, ranges from 0 (low score) to 100 (high score).	Maintained by World Bank Governance Indicator - World Bank

All Three Empirical Chapters: Key Independent Variables

CPIS_HB Home bias Equity home bias (CPIS_HB) captures the degree to which local investors in country *l* over-allocate their local equity market relative to the theoretical suggestion of ICAPM benchmark. It is

		constructed as $EHB_{lt} = \log\left(\frac{W_{llt}}{W_{lt}^*}\right)$ where W_{llt} represents domestic investors' allocations in the domestic equity market of country <i>l</i> for the period <i>t</i> , W_{lt}^* is the ICAPM suggested benchmark allocation for country <i>l</i> for time period <i>t</i> .	
CPIS_FB	Foreign bias	Equity foreign bias (<i>CPIS_FB</i>) implies the disproportionate allocation of investors domiciled in country <i>k</i> into the foreign securities of countries <i>l</i> . We calculate the equity foreign bias as $EFB_{klt} = \log\left(\frac{W_{klt}}{W_{klt}^*}\right)$, where w_{klt} is the allocation of investors in country <i>k</i> 's in equities issued by firms in country <i>l</i> for the period <i>t</i> , W_{klt}^* is the ICAPM suggested benchmark allocation for country <i>l</i> for period <i>t</i> .	Co-ordinated Portfolio Investment Survey of International Monetary Fund and Standard and Poor Global Stock Markets Factbook
GF_FB	Global fund foreign bias	Foreign biases based on EPFR Global Funds' data are denoted as <i>GF_FB</i> . We take the average equity foreign bias (AFB_{jt}) exhibited by all source country investors $(k=1,,n)$ for the country l for each period t as $AEFB_{lt} = \frac{\sum_{k=1}^{n} EFB}{n}$ $k \neq l$, excluding the country for which foreign bias is measured, $k \neq l$.	Unique fund level country allocation data from Emerging Portfolio Fund Research (<i>EPFR</i>) to create the global fund's foreign bias (GF_FB) measure

Control	Variable	for the	three e	mpirical	studies

МКТСар	Market capitalization	The natural logarithm of market capitalisation (MKTCap) of listed companies in USD millions.	Standard and Poor Global Stock Markets Factbook; and World Bank's World Development Indicator
Beta	Beta	Beta is the covariance of the MSCI country's world index return over the past five years divided by the MSCI world index return variance.	Morgan Stanley Capital International (MSCI)
BM	Book to market	BM is the natural logarithm country level ratio of book-to- market; Retn_1 is the average MSCI monthly index return over the past year.	Morgan Stanley Capital International (MSCI)
Retn_1	Return	The average MSCI monthly index returns over the past year.	Morgan Stanley Capital International (MSCI)
Exch	Exchange rate	Three year moving average covariance of the monthly stock market index return with the monthly change of the domestic	Sourced from Thompson Reuters

currency with respect to the dollar.

Infl	Inflation	The one year lagged rate of annual inflation based on the World Bank's World Development Indicator consumer price index.	
Turn	Turnover	The ratio of the total traded volume of stock in a year divided by market capitalisation.	World Bank's World Development Indicator
LSMI	Log stock market integration	The log stock market integration measured as the ratio of a country's annual exports plus imports divided by GDP.	World Bank's World Development Indicator
Law	Law and order	Law represents the law and order rating index of a country.	Political Risk Services Group's ICRG
RGDPG	Real GDP growth	Real gross domestic product growth.	World Bank's World Development Indicator
PolRisk	Political risk	Political risk rating index of a country.	Political Risk Services Group's ICRG
EconRisk	Economic risk	The economic risk rating index of a country.	Political Risk Services Group's ICRG
FinRisk	Financial risk	The financial risk rating index of a country.	Political Risk Services Group's ICRG
Flow	Flow	The log of net foreign direct investment scale by GDP.	World Bank's World Development Indicator
Sav	Savings	The natural logarithm of a country level gross domestic savings.	World Bank's World Development Indicator
PCred	Private credit	Private credit is the log value of domestic credit to the private financial intermediary denominated in USD.	World Bank's World Development Indicator
Int	interest rate	Real interest rate adjusted for inflation as measured by GDP deflator.	World Bank's World Development Indicator
Legal_O	Legal origin	Dummy variable that takes a value of 1 if a common law country or 0 if otherwise.	La Porta (1998)
Cor	Corruption	Index of corruption prevailing in a country. The prevalence and persistence of corruption will reduce investors' participation in the domestic stock market.	Political Risk Services Group's ICRG
GDPPC	GDP per capita	GDPPC is the GDP Per Capita.	World Bank's World Development Indicator
Tobin's Q	Tobin's Q	Tobin's Q is measured as the log market value of equity plus the	Morgan Stanley Capital International and Thompson Reuters

		book value of total assets minus the book value of equity and divided by the book value of total assets of country i .	
PolStab	Political stability	Political instability.	Political Risk Services Group's ICRG
Confl	Conflict	Free from internal conflicts.	Political Risk Services Group's ICRG
Press	Press freedom	Press freedom to capture the effects of free media and free access to information on investor protection. Press freedom takes a value of 100 (high degree of press freedom), 0 (low degree of press freedom).	Maintained by World Bank Governance Indicator - World Bank
Educ	Education	It is measured as the total number of new entrants in the last grade of primary education, regardless of age, and expressed as a percentage of the total population of the official primary entrance age.	World Bank's World Development Indicator

Chapter 5: Methodology

The previous chapter provides the data sources, description of the variables and how they were constructed. This chapter offers the research method and the statistical techniques used in the three empirical analyses. In the research empirical analysis, the study presents a summary analysis of the variables; correlation coefficients, econometric issues, and regression analysis base on pool ordinary least squares (OLS) with a panel modelling.

5.1 Summary analysis

We begin the empirical analysis (in Chapters 6, 7, and 8) with the summary mean statistics of our sample (i.e. the dependent and the key independent variables) constructed on the mean of each country. Equation 5-1 defines how we calculate the mean.

$$\overline{y} = \frac{1}{w} \sum_{j=1}^{n} w_j y_j$$
5-1

Whereby \overline{y} is the mean of the variable we are trying to compute, and y_i is the individual observation on y, where j = 1, ..., n and n is the sample size, w_j denotes the weight.

We further categorise the sample countries into developed and emerging markets, based on MSCI definition. Finally, we analyse the variables in terms of the top 10 countries exhibiting high levels of home and foreign bias, and the bottom 10 countries with less prevalence of home and foreign bias.

5.2 Correlation analysis

In the main analysis in the empirical chapters, we provide bi-variant Spearman's correlation coefficient to show the relationship between the variables used in the study. Through Spearman's correlation coefficient, we are able to provide an early indication to show whether home and foreign bias have a negative or positive relationship with the dependent variables. The correlation coefficient is given by:

$$\rho_{xy} = \frac{Cov(r_x, r_y)}{\sigma_x \sigma_y}$$
 5-2

5.3 Pooled OLS

Selecting the suitable statistical approach using panel data is important in empirical research to ensure consistent and efficient estimation of the parameters.⁴⁷ In this study, we use pooled OLS regression due to the nature of our data, and the use of several time invariant variables that hardly change over time. OLS is the best linear unbiased estimator for the coefficients. The following equation shows the standard linear equation model.

$$y_{it} = a + \beta x_{it} + \varepsilon_{it}$$
 5-3

Where x_{it} denotes K-dimensional vector of the independent variables. The model imposes that the intercept *a* and the slope coefficients in β are identical for all individuals and time periods. The error term in equation 5-3 varies over individuals and time, which captures all unobservable factors that affect the dependent variable y_{it} . Using pooled OLS to estimate the model requires efficiency and unbiasedness. The following condition must be fulfilled.

$$E\{\varepsilon_{it}\} = 0 \tag{5-4}$$

$$E\{x_{it}\varepsilon_{it}\}=0$$
5-5

Ever since pooled OLS have multiple observations for the same units (countries), it is reasonable to assume that the error terms from different periods will be correlated. For instance, econometric literature shows that, a person's wage will be influenced by unobservable characteristics that vary over time. This will make pooled OLS standard errors calculated, based on the assumption of independently and identically distributed (IID) error terms misleading. Furthermore, pooled OLS is more likely to be efficient, compared to an estimator that exploits the correlation over time in ε_{it} .

⁴⁷ Baltagi (2001) suggests three basic approaches (i.e. pooled OLS, random and fixed effects) in examining the relationship within or between each cross section. These approaches can be employ to allow for individual heterogeneity.

In a simple regression model in the empirical corporate finance literature, OLS requires the following important assumptions to produce consistent estimates of the parameters. Gujarati (2003) recommends that, the following important assumptions must be fulfilled when using OLS.

- Sample observations are randomly on Y(and X₁..., X_k). This assumption requires normality, whereby the sample size must be drawn from normal distributed population.
- Mean zero error term(*i.e.*, *E*(*u*) = 0).
 The assumption of independence of error terms suggests that error terms are independent from one another and non-existence of serial correlation.
- The explanatory variables have no linear relationship (i.e., no perfect collinearity so the rank(XX) = K, where X = (1, X₁, ..., X_k).
- 4. There is no correlation between the explanatory variable (i.e., cov(x, u) = 0 for j = 1, ..., K). For unbiased estimates of the parameter, there should be an error term with zero mean, conditional on the explanatory variable (i.e., E(u/x) = 0). There is a problem of inference if the error term correlates with each explanatory variable. It is extremely difficult to empirically test whether the explanatory variable correlates with the error term in the regression, as the error terms are unobservable.

Panel data assume $\varepsilon_{it} = a_i + u_{it}$. Whereby it is assumed that u_{it} is homoscedasticity and does not correlate over time. Additionally, the a_i element is time invariant and homoscedasticity across the individual units.

The assumption $E\{x_{it}, \varepsilon_{it}\} = 0$, implies that the observable regressors in x_{it} are uncorrelated with the unobservable characteristics in both a_i and u_{it} , suggesting that the independent variables are exogenous. In many instances, there is reason to believe that $E\{x_{it}a_i\} = 0$, and the assumption is deemed to be very restrictive. We can have the unobserved heterogeneity in a_i to correlate with one or more of the independent variables. For instance, in a wage equation, a person's unobserved ability perhaps could influence wages (y_{it}) , but also a person's education level (included in x_{it}). In a firm-level investment equation, unobserved firm characteristics may affect investment decision (y_{it}) as well as characteristics in x_{it} (e.g., cost of capital). Consistency of OLS requires the error term $(a_i - a + \varepsilon_{it})$ be uncorrelated with x_{it} . Regressing y_{it} on x_{it} in a pooled OLS will yield a consistent estimate of β if the composite error u_{it} in the pooled OLS model $(y_{it} = a + x_{it}\beta + u_{it})$ is uncorrelated with x_{it} . A violation of the OLS assumptions may result in econometric problems such as inefficient, and biasedness in the estimation of the coefficients, caused by autocorrelation, heteroskedasticity, multicollinearity and endogeneity.

Each country has its own individual characteristics that may or may not influence the predictor variables.⁴⁸ For example, the political system of a country could have some effect on home and foreign bias. In a pooled OLS context, we use several country explicit, rarely changing variables, such as ICRG's, political, economic and financial risk ratings, to allow for country-specific heterogeneity.

In the next section, we discuss autocorrelation and heteroskedasticity problems associated with pooled OLS, which could compromise the consistency of the coefficient estimation. We further provide detailed discussions on how we test and address these consistency and efficiency issues.

5.4 Autocorrelation and heteroskedasticity standard error correction

We show, in this section, econometric issues that affect the efficiency of OLS in estimating the coefficients. We further perform a series of diagnostic tests to check whether or not our model is appropriately specified. Autocorrelation and heteroskedasticity are efficiency problems that are encountered in panel data as a result of the combination of time series and cross-section data. The occurrence of autocorrelation and heteroskedasticity suggest that the error term in the model is no longer independently and identically distributed. By identically distributed, we imply that the residuals are homoskedastic, indicating that they have been obtained from the same population and have a uniform variance. Similarly, by independently distributed, we mean that they are not clustered or serially correlated. The occurrence of autocorrelation and heteroskedasticity make OLS still unbiased, but OLS is no longer efficient, and therefore yield incorrect standard errors and t-statistics.

⁴⁸ The standard econometric technique used in panel data is to control for random and fixed effects. Random effect has the assumption that the individual constant is a group specific disturbance which is similar to the error term, except for each group, whilst the fixed effect assumes that individual constant is a group specific constant term in the regression model (Green, 2007).

$$E\{\varepsilon/x\} = E\{\varepsilon\} = 0$$
 5-6

$$V\{\varepsilon/x\} = V\{\varepsilon\} = \sigma^2$$
 5-7

The above equations imply that the conditional distribution of the errors, given the matrix of the explanatory variables has zero means, constant variances and zero covariance. It indicates that each error has the same variance and two different error terms are uncorrelated. The assumption indicates $E\{\varepsilon_i/x_i\} = 0$. So the model corresponds to the conditional expectations of y_i given x_i .

When estimating such a model, preferably, we will want the error term to be a pure white noise with $E(\varepsilon_{it}, x_{it}) = 0$. Supposing we have controlled for all the time variant variables, the coefficients are unbiased if the $E(a_i, X_{it}) = 0$ for all *i*. In such a case, we assume that the intercepts are different for different individuals but they are random drawings from a distribution with mean μ and σ_a^2 .

5.4.1 Autocorrelation

Panel data includes repeated observation of the same unit (time series); we can therefore expect a violation of $E\{\varepsilon/x\} = E\{\varepsilon\} = 0$, and experience a situation where there is a correlation of time series with its own past and future values. For instance, the likelihood of tomorrow being rainy, is greater if today is rainy than if today is a dry day.

Autocorrelation occurs when the covariance between several error terms is not equal to zero. This implies that two or more consecutive error terms are correlated. The presence of autocorrelation renders OLS unbiased, but OLS becomes inefficient and the standard errors are estimated incorrectly. Error term captures unobserved factors affecting the dependent variable that the model has not accounted for. The persistence of unobserved errors correlating with each other will result in serial correlation. Our panel data contain multiple observations (i.e. time series data) on countries. Therefore, we can expect different error terms of an individual observation to be correlated. Given the model:

$$Y_{it} = b_0 + x_{it} + u_{it} 5-8$$

Autocorrelation indicates a systematic relationship between error terms or the residuals measured at different points in time and results in $Cov(\mu_t \mu_{t-1}) \neq 0$. The systematic relationship will be $\mu_t = \rho \mu_{t-1} + e_t - 1 \le \rho \le 1$.

This suggests that the current value of the residual is related to the last period's value, together with a current period random component e_t . First-order autoregressive process (AR) is the most popular form of autocorrelation, where the error term in $Y_{it} = X_{it}\beta + \varepsilon_{it}$ and is assumed to depend upon on its predecessor as follows $\varepsilon_t = \rho \varepsilon_{t-1} + V_t$ where V_t is an error term with a mean of zero and constant variance σ^2 that exhibits no serial correlation. This indicates that the value of the error term in any observation, is equal to ρ times its value in the previous observation, plus a fresh component V_t which is independent over time.

Autocorrelation is usually seen as a sign of misspecification and inference based on the OLS estimator will be misleading as the standard errors will be based on an incorrect model specification.

We test for the presence of autocorrelation in our model specification by running the Durbin-Watson (1950) test which is the common test for first-order autocorrelation, especially for small sample distribution. The Durbin-Watson test is given by:

$$dw = \frac{\varepsilon_{t=2}^{T} (e_{t-}e_{-1})^2}{\varepsilon_{t=1}^{T} e_t^2}$$
 5-9

Where e_t is the OLS is residual (notice the different indices for summations). Straight forward algebra shows that

$$dw \sim_{\sim} 2 - 2p \qquad \qquad 5-10$$

Consequently, a value of dw close to 2, indicates that the first-order autocorrelation coefficient ρ is close to zero. If dw is much smaller than 2, this suggests a positive autocorrelation ($\rho > 0$); if dw is much larger than 2, the ($\rho < 0$). Even under $H_0: \rho = 0$, the distribution of dw depends not only upon the sample size T and the number of variables K in X_t but also upon the actual values of the X_t . It is possible to compute upper and lower limits for the critical values of dw that depend only upon sample size T and number of variables K in X_t . These values, d_l and d_u , were tabulated by Durbin and Watson (1950)

5.4.2 Heteroskedascity

Linear OLS model is given by⁴⁹

$$y_{it} = \beta_0 + \beta_1 x_{it} + \dots + \beta_k x_{it} + u_{it},$$
 5-11

The assumption is that $Var(u_i/x_i) = \sigma_i^2$ implying that the variance of the error term is constant.

$$Var(u_i/x_i) = \sigma^2 h(x_i)$$
5-12

Heteroscedasticity occurs when equation 5-12 is violated i.e. when the variance of unobserved error u_{it} , conditional on independent variables, is not constant. The variance of the error may be a function of independent variables. Where $Var\{u_{it}/x_{it}\}$ is diagonal, but not equal to σ^2 times the identity matrix. It implies that the error terms are mutually uncorrelated, whilst the variance of u_{it} may vary over the cross section observations. Heteroskedasticity means different error terms do not have identical variances, so the diagonal elements of the covariance matrix are not the same. This implies that the error terms in the model are no longer independently and identically distributed.

Verbeek (2012) illustrates a situation where y_i denotes expenditure on food and x_i consists of a constant and disposable income dpi_{it} . An Engel curve for food needs to be upward sloping, thus on average, higher income corresponds to higher expenditure on food. Nevertheless, we can expect the variation in food expenditures among high income households is much larger than the variations among low income households. This occurs when the variance of u_{it} increases with income. The consequences of heteroskedasticity mean that;

1. The OLS estimators remain unbiased and consistent as a result of the fact that none of the explanatory variables correlate with the error term. Therefore, a correctly specified equation will produce coefficient estimates similar to the real parameters.

⁴⁹ Wooldridge (2009) provide technical details of the consequences of heteroskedasticity and how Breusch-Pagan test can be run to check the presence of heteroskedasticity.

- 2. Heteroskedasticity affects the distribution of the coefficient estimates by increasing the variances of the distribution, which thereby makes the OLS estimates inefficient.
- 3. Heteroskedasticity underestimates the variances of the estimators, which subsequently result in higher values of *t* and *F* statistics.

We use a Breusch-Pagan test for heteroskedasticity which is intended to detect any linear form of heteroskedasticity. This identifies the problem of errors that are not IID. The model tests the null hypothesis that the variances of error terms are equal, versus the alternative, that error term variances are multiplicative functions of one or more variables. The alternative hypothesis states that the error variance increases as the predicted values of *Y* increase, e.g. the bigger the predicted value of Y the bigger the error variance. The null hypothesis of homoskedasticity is given by:

$$H_0:\delta_1 = \delta_2 = \dots = \delta_k = 0.$$
 5-13

The basic approach is to adopt a linear function. The Breusch-Pagan test for heteroskedasticity first uses OLS equation 5-14 to estimate the model to obtain the squared OLS residuals (u^2) for each observation in equation 5.15.

$$y = \beta_0 + \beta_1 x_1 + \dots + \beta_k x_k + u,$$
 5-14

At the second stage, the model runs the following auxiliary regression and keeps the R-squared from the regression, $R_{\hat{\mu}^2}^2$.

$$\hat{u}_t^2 = \delta_0 + \delta_1 x_1 + \delta_2 x_2 + \dots + \delta_k x_k + \nu, \qquad 5-15$$

Where v is an error term with mean zero given the x_j , the dependent variable is the square of the error in equation 5-15 i.e. the OLS model.

Finally, the Breusch-Pagan tests the F statistic and calculates the p-value. If the p-value is sufficiently small, that is, below the chosen significance level, then we reject the null hypothesis of homoscedasticity and conclude there is significant evidence of heteroskedasticity.

5.4.3 Multicollinearity

Multicollinearity occurs when two or more independent variables are highly and positively correlated, implying that the coefficient estimates will tend to be highly and negatively correlated. Greater presence of multicollinearity will result in unreliable estimates with large standard errors and unexpected sign or magnitude, suggesting that one, or more, parameter of interest is estimated highly inaccurately. Essentially, it implies that our sample provides insufficient information about the parameters and will, thereby, render confidence intervals for coefficients to be wide and small t-statistics. The coefficients will have to be larger to be statistically significant.

The study uses variance inflation factors (VIF) to test for the presence of multicollinearity amongst the independent variables. Gujarati (2003) suggests that a VIF value of less than 10 is acceptable. However, Hair et al. (1998) and Kennedy (2008) recommend that a VIF exceeding 10 indicates the presence of harmful multicollinearity. The problem of multicollinearity could be resolved by omitting one or more explanatory variables that are highly correlated. Stata automatically removes one or more of the multicollinearity variables.

5.4.4 Newey-West autocorrelation and heteroskedasticity standard error

We perform a linear regression with Newey-West (1987) standard errors that are robust to both autocorrelation and heteroskedasticity. Green (2012) provides technical details on how the correction is made. Standard OLS regression is given by the following:

$$Y_{it} = X_{it}\beta + U_{it}$$
 5-16

Where in equation 5-12, the variance of $(U_{it}) = 0$ (Omega). If U_{it} are heteroskedastic and / or auto-correlated, then the correct formula for the Variance $(\hat{\beta}_{OLS})$ is

$$\widehat{V}_{ar}(\widehat{\beta}_{OLS}) = (X'X)^{-1}X'\widehat{\Omega}X(X'X)^{-1}$$
5-17

We use linear OLS regression to estimate the coefficients. A lag (0) suggests the absence of autocorrelation. The Newey-West standard errors are computed, conditional on a choice of maximum lag, whilst the variance estimates are computed using the White robust standard error. They are computed from a distributed lag of the OLS residuals, and specify the longest

lag at which auto-covariances are to be calculated. The Newey-West formula involves an expression in squares of the residuals, which is similar to White's robust standard error formula (as well as a second term in the cross-products of the residuals), these robust estimates subsume White's standard error correction.

For lag (m), m > 0, in equation 5-18, the Newey-West is similar to the White formula. However, in equation 5-18 Newey-West add autocorrelation correction to the White formula to compute the variance estimates.

$$X'\hat{\Omega}X_{Newey-West} = X'\hat{\Omega}X_{White}$$
5-18

$$X'\hat{\Omega}X = X'\hat{\Omega}_0 X + \frac{n}{n-k} \sum_{l=1}^m (1 - \frac{1}{m+1}) \sum_{t=l+l}^n \hat{e}_t \hat{e}_{t-l} \left(x'_t x_{t-l} + x'_{t-l} x_t \right)$$
5-19

Where x_t is the row of the X matrix observed at time t. k is number of predictors in the model, l is time lag, and m is maximum time lag. n is the number of observations.

5.5 Specifications issues

Several econometric issues could result in biased estimates of the coefficients. First, omitted variable bias, whereby the omitted variable in the error term correlates with the explanatory variables. Additionally, a reverse causality can occur where the dependent variable influences the key independent variable of interest. Selection bias is caused by the non-randomness of the sample data. There are others such as measurement error, common-method variance, inconsistent inference, and model misspecification that can bias the coefficients. Nevertheless, we do not experience them in this study. In the following sections, we provide these specification issues and how we addressed them.

5.5.1 Omitted variable bias

More broadly, most international finance variables are based on both public and non-public information, suggesting that a number of variables relevant for international finance are not observable. The inability to observe some variables to appear among the explanatory variables, X, these omitted variables appear in the U. If the omitted variable correlates with

the explanatory variables, then there is an endogeneity problem that causes inference to break down.

The failure of $E(u/x_i, ..., x_k) = 0$ renders bias in the OLS estimators; correlation between u and any of $x_i, X_2, ..., x_k$ will basically cause the OLS estimators to be inconsistent. If the error is correlated with any of the independent variables, then OLS is biased and inconsistent.

Ordinary least squares can be represented in a regression equation as

$$y_{it} = a_i + \beta x_{it} + \varepsilon_{it}$$
 5-20

Where y_{it} denotes dependent variable, α_i denotes a constant, βx_{it} denotes the independent variable, and \mathcal{E}_{it} denotes the error term or the unobserved variable, whereby $\mathcal{E}_{it} = \alpha_i + U_{it}$.The constant error term includes both observable and unobservable. In several studies, fixed effect or least square dummy variable (LSDV) is used to address unobserved unit (country) effects. In this study, we use several country specific rarely changing variables such as ICRG's, political, economic and financial risk ratings to allow for country-specific. OLS make the assumption that the error term in the basic OLS regression, indicates the independent variables should vary randomly in the prediction of the dependent variable. The prediction will not be random if the error term correlates with an independent variable. It will thereby, render the estimated coefficients biased and consistent. Other causes of omitted bias are caused by omitted fixed effects, and random effects without checking if variables correlate with fixed or constant effects. Economically, unobserved heterogeneity exists if $E(\eta_1/X_{it}, Z_{it}) \neq 0$. Econometrically, unobservable heterogeneity is a source of endogeneity if the unobservable factors affect either the dependent or the independent variables or both. Failing to include important control variables which correlate with the key independent variables, will make the estimated coefficients biased and inconsistent. We include several control variables to address potential omitted variable bias. For instance, in corporate governance literature, the effects of managerial ability, which is generally difficult to quantify, much less observe, definitely affect performance. Therefore, an OLS regression of performance on board structure, that ignores the unobservable heterogeneity, may find a negative relationship between board independence and performance.

We employ the Ramsey reset to test if our model suffers from omitted variable bias. The Ramsey test generates new variables based on the predicted values, and thereby uses the new variables to test the statistical significance of the variables. The Ramsey test examines whether our model is correctly specified and all important variables are included. In the next, section we test whether our model suffers from reverse causality.

5.5.2 Endogeneity (reverse causality)

Endogeneity as a result of reverse causality is, arguably, an essential and pervasive problem that is encountered in empirical finance research. Reverse causality occurs when the key independent variable is determined simultaneously, along with the dependent variable y. For example, if y_{it} is the cost of capital and x_{it} is home bias, the simultaneity problem arises when home bias is partially determined by cost of capital. Simultaneity bias can be illustrated as follows.

$$y_{it} = x_{it}\beta + \varepsilon_{it}$$
5-21

The model has only explanatory variable x_{it} , which is also determined by y_{it} as follows.

$$x_{it} = \lambda y_{it} + \varepsilon_{it}$$
 5-22

When the assumptions of OLS are violated, the estimated coefficients will be compromised and the results will be uninformative and cannot be causally interpreted. The presence of endogeneity as a result of reverse causality will make OLS generate biased and inconsistent parameter estimates, which makes it virtually impossible to generate reliable inference. Endogeneity is not only pervasive but prevalent across international finance. Failing to acknowledge the presence of endogeneity could be perilous and may have serious implications for inference. Particularly, it may render the coefficient estimates highly biased generating wrong results and erroneous conclusions about the integrity and truthfulness of theory. Additionally, the hypotheses test can be seriously misleading.

Endogeneity comes about from several sources, for instance, omitted variables which produce a correlation between the explanatory variables, simultaneity and selection bias where randomisation is infeasible in the sample selection.
We are concerned that our key independent variables of interest (home and foreign bias) are endogenously determined. Roberts and Whited (2011) show a review article that provides guidance on how to address the endogeneity issue. The methods used to address endogeneity include lagged independent variables and the Heckman selection model to address reverse causality and selection bias respectively. Our attempt to explain the impact of home and foreign bias on cost of capital, stock market development, and investor protection will be biased if there is an issue of endogeneity. For instance, a country with a lower cost of capital, will attract more foreign investors. Similarly, it is conceivable that countries with better stock market development will attract a large percentage of foreign investors. Chan et al. (2005) show countries that have better investor protection, will attract foreign investors. We use Newey-West (1987) to address autocorrelation and heteroskedasticity problems, lagged values to address reverse causality issues, and the Heckman selection model to address selection bias.

5.5.3 Lagged values

Studies suggest that by using panel data, an internal instrument can be generated when there is an endogenous variable. Verbeek, (2012) argues that in dynamic panel data, the endogenous independent variable can be lagged as an instrument to become exogenous. Following several studies that have used lagged values to resolve the endogeneity problem caused by reverse causality (see Yang, 2003: Gelos and Wei, 2005), we use lagged values of the endogenous variables (home and foreign bias) as instrumental variables to address the problem of reverse causality.

$$y_{it} = a_0 + x_{it-1} + z_{it} + \varepsilon_{it}$$
 5-23

Where; y_{it} is the dependent variable, a_0 is the intercept, x_{it-1} is the lagged value of the endogenous variable, z_{it} are the explanatory or control variables, ε_{it} are the error terms. When x_{it} is a predetermine variable but not strictly exogenous, only lagged values of x_{it} is a valid instrument. If x_{it} is strictly exogenous, then current and lagged value of x_{it} is a valid instrument.

The lagged values transform to become exogenous or instrumental variables that strongly correlate with the potentially endogenous variables (home and foreign bias) to influence the

independent variables; however, it is seemingly unlikely to affect the dependent variables except through their effect on the independent variables. Lagged values have the advantage of being very simple to implement, additional data requirements are limited and it is intuitively appealing. Lagged values provide rigour and transparency to the model specification.

5.5.4 Heckman selection model

There is a possibility of the model suffering from selection bias as a result of the nonrandomness of the selection of the countries. Due to unavailability of data for some countries, the study omits several emerging countries and could thereby make our result encounter selection bias. Without correcting for non-randomly selected sample may lead to incorrect conclusion. We, however, use the Heckman selection model to address the selection bias.

The Heckman model offers statistical procedures that address the problem of selection bias and contribute to the programme of resolving selection bias in the following: (i) the model provides a theoretical framework that highlights the importance of modelling the dummy endogenous variable; (ii) The Heckman model was the first attempt that use the probability (i.e. the propensity score) of a participant being in one of the two conditions indicated by endogenous dummy variable, and then use the estimated propensity score model to estimate coefficients of the regression model; (iii) Heckman treats the unobservable selection factors as a problem of specification error or a problem of omitted variables, and correct for bias in the estimation of the outcome equation, by explicitly using information gained from the model of the sample selection; (iv) and he developed a creative two-step procedure by using the simple least squares algorithm.

Heckman (1979) argues that selection bias is a form of omitted variable bias, and his selection model conjectures the following fundamental regression relationship.

$$Y_j = X_j \beta + U_{1j} \tag{5-24}$$

In equation 5-24, the dependent variable is not always observed. However, the dependent variable for observation j is observed when

$$Z_j \Upsilon + U_{2j} > 0 \tag{5-25}$$

Where

$$U_1 \sim N(0, \sigma)$$
$$U_2 \sim N(0, 1)$$
$$Corr \sim (U_1, U_2) = \rho$$

When $\rho \neq 0$, standard regression methods employed in equation 5-24 will produce biased estimates. Nonetheless, the Heckman model will offer a consistent, asymptotically efficient estimate for all the parameters in the model.

Heckman's selection model is based on the following models:

$$Y_1 = \beta' X + U_1 \tag{5-26}$$

$$Y_2 = \Upsilon' Z + U_2 \tag{5-27}$$

Where X is a k-vector of regressors, Z is and m-vector of regressors. The error terms U_1 and U_2 are mutually normally distributed, individually of X and Z, with zero expectations. We are interested in the first model. Y_1 is observed when $Y_2 > 0$. Therefore, the dependent variable is $Y = Y_1$ if $Y_2 > 0$, Y is omitted value if $Y_2 \le 0$.

When we disregard the sample selection bias and regress *Y* on *X* adopting the observed *Y*'s only, it will render the OLS estimator β biased as a result of the fact that

$$E[Y_1/Y_2 > 0, X, Z] = \beta' X + \rho \sigma f(Y'Z) / F(Y'Z), \qquad 5-28$$

F denotes the cumulative distribution function of the standard normal distribution, f represents the corresponding density, σ^2 is the variance of U_1 , whilst ρ is the correlation between U_1 and U_2 . Therefore

$$E[Y_1/Y_2 > 0, X] = \beta' X + \rho \sigma E[f(Y'Z)/X]$$
5-29

Puhani (2000) shows how Heckman's two-step proposal is used to estimate the inverse Mills ratio

$$\times (X'_{2i}\beta_2/\sigma_2) = \frac{\phi(-(X'_{2i}\beta_2/\sigma_2))}{1 - \phi(-(X'_{2i}\beta_2/\sigma_2))}$$
5-30

by way of a probit model and estimate the following equation

$$y_{1i} = X'_{1i}\beta_1 + \frac{\sigma_{12}}{\sigma_2} \times (X'_{2i}\beta_2/\sigma_2) + \varepsilon_1$$
 5-31

in the second stage. Heckman treated sample selection bias as a special case of omitted variable problem with λ being the omitted variable of OLS, we use on the subsample for which $y_1^* > 0$, as long as μ_2 has a normal distribution and ε_1 is independent of λ , Heckman's two step estimator is consistent. However, it is not efficient as ε_1 is heteroskedastic. As the variance of ε_1 is given by

$$Var(\varepsilon_{1i}) = \sigma_1^2 - \frac{\sigma_{12}^2}{\sigma_2^2} \left[\frac{X'_{21}\beta}{\sigma_2} \times \left(\frac{X'_{2i}\beta}{\sigma_2} \right) + \left(\frac{X'_{2i}\beta}{\sigma_2} \right)^2 \right]$$
 5-32

Clearly, $Var(\varepsilon_{1i})$ is not constant, but varies over *i*, as it varies with X_{2i} , in order to obtain a simple and consistent estimate of the asymptotic variance-covariance matrix.

When the endogeneity problem is not accounted for in the estimation process, the OLS will produce biased parameter estimates. Heckman (1976, 1979) offered a two-stage estimation procedure using the inversed-Mills' ratio to take account of the endogeneity bias.⁵⁰

We performed an endogeneity correction for the treatment effects in our empirical chapters. In the first step, a regression to observe a positive outcome of the dependent variable is modelled with a probit model. The estimated parameters are used to calculate the inverse Mills' ratio, which is then included as an additional explanatory variable in the OLS estimation (Green, 1993). Using Heckman's two-stage estimation, we correct the specification for endogeneity and examine whether sub-optimal portfolio allocation has impact on cost of capital, stock market development, and investor protection standards.

⁵⁰ See Chapters 6.6.3, 7.5.2, and 8.5.2.

Chapter 6 First empirical study: Sub-optimal portfolio allocation and cost of capital

Studies show the prevalence of home and foreign bias in international portfolio investment. Studies provide theoretical argument, suggesting that home and foreign bias affect cost of capital. For instance, Stulz (1999) and Errunza (2001) argue that countries that have high prevalence of home bias will experience high cost of capital, due to the lack of international risk sharing between domestic and foreign investors. In light of this evidence, we are motivated to undertake this study as a result of the importance of cost of capital in project evaluation (see Chapter 2.3.1). A lower cost of capital will create more projects, to have positive NPV, which will lead to more investment and economic growth, and create more jobs in the country.

6.1 Empirical analysis

We present the empirical analysis based on the research method in Chapter 5. Current studies, for instance, Stulz (1999) and Errunza (2001) offer theoretical models suggesting that cost of capital reduces when a country attracts sufficient foreign equity investment. Similarly, over-investment in the domestic market by local investors, increases cost of capital. Chan et al. (2009), show that home bias reduces firm valuation as result of high cost of capital being associated with home bias. In this empirical study, our objective is to advance the current literature by providing empirical evidence to answer the research question: "The impact of sub-optimal international portfolio allocation on cost of capital". We test the following hypotheses:

- **H**₁ Countries that exhibit higher home bias are associated with higher cost of capital.
- **H**₂ Countries with higher foreign bias relate to lower cost of capital.
- **H**₃ Pervasiveness with higher global fund foreign bias allocation, is associated with lower cost of capital.

We begin our empirical analysis by presenting the summary statistics of the different cost of capital proxies, CPIS based equity home bias (*CPIS_HB*) and foreign biases (*CPIS_FB*), and the EPFR based global fund foreign bias (*GF_FB*). Subsequently, we report the simple

correlation figures between the dependent variables, i.e., cost of capital proxies and the key independent variables (*CPIS_HB*, *CPIS_FB* and *GF_FB*), econometric issues followed by robust regression analysis. Based on the MSCI definition, our sample country comprises 23 developed countries and 21 emerging countries from 2001 to 2010.⁵¹

6.2 Summary analysis

In Table 6-1, we partition the sample countries into developed and emerging countries. We report the average statistics (mean figures) of the various proxies of the cost of capital (columns 2-5). The sub-optimal international portfolio bias measures are reported in columns 6-8. The same summary figures in panel B are grouped according to level of development, i.e. between emerging and developed markets. Finally, panel C reports the figures for the top 10 and bottom 10, sorted on the basis of *CPIS_HB* measures.

.....Insert Table 6-1 about here.....

Consistent with expectations, the figures in panels A and B show that developed countries exhibit a lower cost of capital, relative to their emerging market counterparts. The lowest cost of capital in terms of historical risk premium is observed for Ireland, followed by the United States, Japan, the United Kingdom, the Netherlands, Switzerland, Belgium, Russia, France, and Italy, as the ten countries with the lowest cost of capital. Similarly, the countries ranking with the highest cost of capital are the Czech Republic, Brazil, Bulgaria, Peru, Thailand, Egypt, Poland, Finland, and the Philippines. With the exception of Finland, they are mainly emerging markets. Focusing on the other three proxies of the cost of capital, a similar pattern can also generally be observed in terms of their ranking, implying that compared to their developing counterparts, investors in developed markets bear a lower cost of capital.

In terms of the three measures of sub-optimal portfolio allocations (i.e. *CPIS_HB*, *CPIS_FB* and *GF_FB*) panels A and B reveal that the top ten countries ranking lowest in terms of home bias measure, i.e. *CPIS_HB*, are mostly developed countries (except China), with the lowest being the United States followed by the United Kingdom, Japan, the Netherlands, Germany, France, Ireland, Canada, and Italy. On the other spectrum, countries ranking highest on the measure of home bias, i.e., investors preferring their home countries' securities rather than

⁵¹ See appendix 2 for the list of countries.

foreign securities, are mainly emerging markets with the highest home bias observed for Bulgaria followed by Romania, Hungary, Peru, the Czech Republic, the Philippines, Egypt, Argentina, Poland, and Indonesia. In terms of *CPIS_FB*, it is unsurprising that European countries predominantly occupy the top ten lists, attributable to being in the same economic union. Nonetheless, we still observe that most developed countries rank higher in terms of foreign bias (positive bias), i.e., countries mostly preferred by international investors, compared to the emerging markets. Correspondingly, observations are almost identical for the *GF_FB* measure.

The above analysis of the summary statistics based on individual countries clearly indicates that countries with the lowest cost of capital seem to be associated with lower home bias by their home investors and higher foreign bias by international investors. Panel C further supports this conjecture, whereby we notice that the lowest home bias figure for the top ten countries is 2.7 (*CPIS_FB* = 1.12 and *GF_FB* = 0.32), compared to the home bias measure of the bottom ten countries of 7.04 (*CPIS_FB* = -4.72 and *GF_FB* = -2.15). When we compare this with the cost of capital measures, we see that the top ten countries' average historical risk premium is 6.10% (*rCred* = 2.00 and *CERP* = 5%), compared to the bottom ten historical premium of 24.50% (*rCred* = 13.78 and *CERP* = 10.10%). Clearly, the analysis strongly suggests that countries with a lower home bias (higher foreign bias) tend to be associated with a lower cost of capital. We address this conjecture in the following section using correlation analysis and robust regression estimations.

We report the summary statistics of the control variables in Table 6-2. *MKTCap*, *Beta*, *BM*, *PolRisk*, *Retn_1* show substantial cross country variations. *MKTCap* is the lowest in Bulgaria (6.62 USD billions) but the highest in the United States (15600 USD billions). *Beta* varies vastly between 0.51 in Switzerland to 1.87 in Brazil. *BM* ranges from 0.22 (Spain) to 1.54 (Mexico). Retn_1 varies widely between 1% (France, Netherland, Belgium, Denmark, Japan, and Hong Kong) and 18% (Brazil). *PolRisk* ranges from 55 in Indonesia to 91.6 in Finland. We provide detailed descriptions of the variables in Chapter 4 Section 2.5.

.....Insert Table 6-2 about here.....

6.3 Correlation analysis

Table 6-3 presents the cross-correlation coefficient matrix between all the variables we use in our analysis.

.....Insert Table 6-3 about here.....

In line with expectations, *CPIS_HB* is positively and statistically significantly correlated with all the cost of capital measures. This indicates that countries with a greater home bias potentially suffer from a higher cost of capital. Such conjecture is again consistent, in line with our summary analysis. Similarly, the *CPIS_FB* measure is negatively correlated with the cost of capital proxies, suggesting that countries which are favoured by foreign investors are associated with a lower cost of capital. Furthermore, the *GF_FB* measure also shows negative and statistically significant correlation coefficients, again providing support to the previous conjecture. Most of the other correlation coefficients display expected signs in terms of their correlation coefficients.

6.4 Econometric issues

Based on section 5.4 and 5.5 of the research methodology, we conduct the following diagnostic test to ensure all econometric problems are detected and accordingly resolved.

6.4.1 Autocorrelation

We test for the presence of autocorrelation by using the Durbin-Watson test and run the regression in STATA: We obtained DW values of 1.779912 which suggests the presence of autocorrelation.

6.4.2 Heteroskedasticity

We use the Breusch-Pagan to test for heteroskedasticity. We obtain a chi-square of 14.06 which indicates the presence of heteroskedascity. We therefore use the Newey-West autocorrelation and heteroskedasticity correction model to address the problem.

Breusch-Pagan test for heteroskedasticity Ho: Constant variance Variables: fitted values of HRRmchi2(1) = 14.06 Prob > chi2 = 0.0002

6.4.3 Multicollinearity

We employ variance inflation factors (VIF) to test for multicollinearity. The mean of VIF for our model is 1.50 which implies that multicollinearity seems not to be a problem in our model.

Variable	VIF	1/VIF
PolRisk	3.06	0.327117
Law	2.73	0.365871
Retn_1	1.45	0.691456
МКТСар	1.36	0.724725
Turn	1.35	0.740455
Infl	1.32	0.771087
Exch	1.28	0.783569
BM	1.25	0.799293
Beta	1.19	0.839133
RGDPG	1.17	0.853348
LSMI	1.17	0.856465
EconRisk	1.14	0.878838
FinRisk	1.05	0.949647
Mean VIF	1.50	

VIF Test Results

6.4.4 Omitted variable bias

We use the Ramsey reset to test for omitted variable bias. The F values are all above 5% which suggests that we have included all relevant explanatory variables and our model does not suffer from omitted variable bias. Ramsey reset test of omitted variables for *CPIS_HB*, *CPIS_FB*, and *GF_FB*. Test of important variables omitted

Ramsey reset test using powers of the fitted values

Ho: model has no omitted variables

Cost of Capital

CPIS_HB	F(3, 422)=1.78	Pro>F=0.15
CPIS_FB	F(3, 422)=1.74	Pro>F=0.16
GF_FB	F(3, 422)=2.64	Pro>F=0.07

6.5 **Regression results**

This section examines whether the cross-sectional and temporal variations in home biases of domestic investors and foreign biases of foreign investors across the world, explain the varying degrees of international differences in the cost of capital. This study uses panel regressions and employs all possible control variables that could potentially be correlated with the different cost of capital measures. Because of the presence of autocorrelation and

heteroskedasticity as outlined in section 6.4.1 and 6.4.2 respectively, all the regression estimations use the Newey-West correction of the standard errors, which are robust to arbitrary autocorrelation and heteroskedasticity. In the following sections, we first discuss the empirical findings between the sub-optimal portfolio allocation measures and cost of capital, reserving the discussion of controls until the end.

6.5.1 Cost of capital and equity home bias

We begin our estimation by assessing the impact of sub-optimal domestic allocation, i.e., home bias (CPIS_HB) on the cost of capital. In Table 6-4, we present the regression results showing the relationship between equity home bias and four costs of capital proxies. The tstatistics are reported in parentheses. The coefficients of CPIS_HB in Table 6-4 are positive and statistically significant, even at the 1% significance level in models 2 to 4, and 5% significance level in Model 1. Consistent with robust international risk sharing theory, the results provide compelling evidence that higher home bias is associated with high cost of capital. The coefficients of 0.696 (t-statistic=2.38), 0.150 (t-statistic=7.21), 0.288 (tstatistic=3.84), and 0.216 (t-statistic=3.20) for HRRm, rCred, CERP, and DY respectively, suggest that an increase in home allocations should also increase the cost of capital. Erb et al. (1996) find that an increase of one unit in the log of a country's credit ratings is associated with the reduction of 10.47% in cost of equity capital. As we use the same measure of country credit rating (i.e. 1-22) and from an economic significance point of view, the statistically significant coefficient of 0.150 in Model 2 indicates that a 10 unit increase in the log of CPIS_HB (i.e. 1%), is related to a rise in the cost of capital by approximately 0.15*10=1.50 basis points.

.....Insert Table 6-4 about here.....

6.5.2 Cost of capital and equity foreign bias

Table 6-5 reports the results of the relationship between cost of capital and *CPIS_FB* measure, including all the country-specific control variables and year effect. Consistent with expectations, the results in Table 6-5 show that the *CPIS_FB* measure is statistically significant at 1% statistical significance level in Models 1 to 3, and 5% statistical significant level in Model 4. The coefficient of -0.812 (t-statistic = -2.19) of Model 1 suggests that the historical risk premium is lower for countries with a higher foreign bias. Model 2

demonstrates a negative coefficient of -0.151 (t-statistic = -8.44) consistent with the view that favourable country allocation by foreign investors is related to favourable country ratings, indicating a lower cost of capital. Similarly, the coefficient of -0.424 (t-statistic = -7.38) also supports the conjecture that countries which attract higher foreign equity portfolio investments are associated with lower country risk premiums. The results in Model 4, further support the theory consistent with the claim, that higher foreign bias is associated with lower cost of capital.

Model 2 indicates that a 10 unit increase in equity foreign bias is related to a fall in cost of capital by approximately by 1.51 basis points (-0.151*10) per year. This is significant because country credit ratings do not move much, in comparison to country stock returns. The coefficient of -0.155 (t-statistic = -1.92) also confirms that countries attracting higher allocations are related to lower dividend yield, suggesting a lower cost of capital.

.....Insert Table 6-5 about here.....

6.5.3 Cost of capital and fund level foreign bias measure

In this section we replicate the regression of Table 6-4 using global fund level data to measure equity foreign bias (GF_FB). As seen from the results in Table 6-5, the signs of the coefficients are negative and statistically significant across all the proxies of cost of capital measures in Models 1-4. The different GF_FB coefficients of -0.983 (t-statistic=-2.29), -0.459 (t-statistic=-6.66), 0.277 (t-statistic=-2.38), and -0.155(t-statistic=-2.22) for *HRRm*, *rCred*, *CERP*, and *DY* respectively, reinforce the findings discussed under Section 3.1. This suggests that higher degrees of foreign bias are inversely associated with differences in the cost of capital across developed and developing countries. The results imply that the risk sharing benefits of international investments, seem to increase, as foreign investors tilt their allocations more towards the implied ICAPM suggestion. This is consistent with the optimal global risk-sharing conjectures of Stulz (1999) and Errunza (2001).

.....Insert Table 6-6 about here.....

6.5.4 Control variables

In terms of the control variables, most of them largely bear expected signs and are statistically significant as reported in Table 6-4. *MKTCap* plays a significant role in

explaining cost of capital. It has statistical and negative effect on cost of capital throughout the models. Infl has a negative effect on cost of capital. With the exception of Model 4, it is statistically significant throughout the models. Turn has a negative impact on cost of capital, but still maintains a correct negative sign and relatively statistically significant when cost of capital is proxy by rCred and CERP. LMSI is negatively related to cost of capital but turns out to be mainly statistically insignificant. The sign is correctly negative. That is, if a country integrates its stock market with the rest of the world, cost of capital falls, due to increase in risk sharing. Its insignificance might indicate that countries have not integrated their stock market sufficiently as financial theory suggests. RGDPG appears not to influence cost of capital, as the coefficients are mainly negative. Law is mainly statistically significant in Models 1 and 2, but it maintains a negative sign in all the models. Such a finding is consistent with existing studies (see Hail and Leuz, 2006). Beta is positively related to the cost of capital and statistically significant. Similarly, BM and Exch are positively associated with the cost of capital but their statistical significance levels are sensitive to different specifications. Such inconsistent behaviour of the control variables is also reported by existing studies (see Lau et al., 2010). PolRisk maintains correct positive sign in Models 1 and 2, and statistically significant at the 1% level.

Other controls such as *PolRisk*, *EconRisk* and *FinRisk* are generally inconsistent with respect to signs and statistical significance to different specifications as a result of the weaknesses of each cost of capital proxy. These control variables have inconsistent sign and statistical significance as a result of fact that; there is a shorter history of stock return data for emerging countries yielding a higher degree of standard error. Further, dividend yield may reflect differences in the countries growth opportunities which may be sensitive to *BM*, *Exch*, *PolRisk* and *FinRisk* and cause them to have inconsistent signs and statistical significance.

6.6 Robustness test

All our above-mentioned empirical results provide strong evidence of the influence of suboptimal international allocations (i.e. home and foreign biases) on the cost of capital, which is consistent with the theory. In this section we run a battery of robustness tests to further test the empirical sensitivity of our results. First, we use *Tobin's Q* as an additional alternative measure of cost of capital. Second, we address the concern of possible endogeneity issues (reverse causality and selection bias) by using two different approaches: first, we deal with reverse causality by employing the pre-determined (exogenous) one year lagged values of home and foreign biases; second, we further use the Heckman selection method to address the sample selection bias. We discuss them in detail in the following sub-sections.

6.6.1 Additional measure of cost of capital: Tobin's Q

As extensively discussed above, we test the impact of $CPIS_HB$, $CPIS_FB$ and GF_FB on the cost of capital using HRRm, CERP, rCred, and DY where we establish that the prevalence of home bias (foreign bias) increases (decreases) the cost of capital. In this section we additionally test whether variations in cross-country aggregate firms' valuations can be explained by differences in home and foreign biases. In order to test the valuation effect and follow the existing literature (Chan et al. 2009), we employ an alternative measure of cost of capital, i.e., *Tobin's Q*. The country level *Tobin's Q* is measured as the log market value of equity plus the book value of total assets minus the book value of equity and divided by the book value of total assets of country *i*. Since there is an inverse relationship between the cost of capital and equity market valuation or firm performance, we expect *CPIS_HB* (*CPIS_FB* and *GF_FB*) to be negatively (positively) associated with Tobin's Q. The results are reported in Table 6-7.

.....Insert Table 6-7 about here.....

We show, in Model 1 of Table 6-7, that the cross-sectional and temporal differences in market valuations are inversely related to *CPIS_HB* as demonstrated by the coefficient of -0.415 (t-statistic = -7.49). This demonstrates that domestic investors' home bias reduces firms' valuations, which in turn implies an increase in the cost of capital. The result is consistent with the conjectures of Errunza and Losq (1985) and Stulz (1999). Home bias reduces firm value because local investors bear a large proportion of risk, as there is inadequate international risk sharing.

In Models 2 and 3 of Table 6-7, the results show that both the measures of foreign bias (*CPIS_FB* and *GF_FB*) are positively associated with higher levels of the *Tobin's Q* measure. The estimated coefficients of *CPIS_FB* and *GF_FB* of 0.171 (t-statistic = 3.47) in Model 2 and 0.294 (t-statistic = 3.94) in Model 3, respectively, implies that as foreign investors increase their allocations towards the suggested ICAPM weight, firms' values increase as a result of the reduction of cost of capital. As such, using the *Tobin's Q* measure, we further

reinforce our previous findings by demonstrating that higher levels of *CPIS_HB* reduce the stock valuations, whereas greater degrees of *CPIS_FB* and *GF_FB* increase the equity valuations.

6.6.2 Reverse causality: Lagged pre-determined variables

Changes in cost of capital may themselves induce foreign investors to invest more, leading to reduced home bias and increased foreign bias. If this is the case, then our estimates may suffer from endogeneity issues arising from reverse causality. We address the potential endogeneity problem using one year lagged values of home and foreign bias, as predetermined exogenous variables. We discuss the results in this section.

To address our concern of endogeneity, we employ lagged *CPIS_HB*, *CPIS_FB* and *GF_FB* in Tables 6-8, 6-9 and 6-10 respectively. Consistent with our expectations, the four coefficients on *CPIS_HB* in Table 6-8 are in line with expected signs and are statistically significant at the 1% level for all the four measures of cost of capital. This confirms and provides robust support to the results reported in Table 6-6 that higher equity home bias is associated with the higher cost of capital, even after addressing the endogeneity problem arising from potential reverse causality.

.....Insert Table 6-8 about here.....

The coefficient estimates of the lagged values of *CPIS_FB* reported in Table 6-9 remain negative and statistically significant across all the proxies of the cost of capital. With the exception of *DY* which has a coefficient of -0.262 (t-statistic = -2.34), the remaining cost of capital measures are statistically significant even at the 1% significance level. These results further support our findings that higher equity foreign biases are associated with lower cost of capital, even after controlling for any potential reverse causality issues.

.....Insert Table 6-9 about here.....

In Table 6-10 we address the possible reverse causality issue using lagged foreign bias measures constructed using the global funds. In line with expectation, all four coefficients of the lagged GF_FB are negative and statistically significant for all four measures of the cost of capital. The statistical significance of GF_FB estimates is consistent with those reported in

Table 6-6. Such results further support the conclusion that with increased levels of foreign investors' allocations, the cost of capital falls.

.....Insert Table 6-10 about here.....

6.6.3 Heckman selection bias

The 44 countries we are able to use are from the 45 all-country index of MSCI and, hence, capture 98 per cent of the highly investable markets. Although we would ideally like to use as many countries as possible, owing to unavailability of data, particularly for smaller emerging and frontier markets, we end up selecting 44 countries. In this section, we use the standard Heckman selection bias method to address the concern of selection biases which may compromise the validity of the results. The results, in general, closely mirror the main result based on OLS.

There is a slight increase in the coefficients and the statistical significance compared to Table 6-4. The results in Models 1-4 of Table 6-11 have a sign similar to our main analysis of Models 1-4 in Table 6-4. This demonstrates that our results are generally not affected by the non-random selection of the countries.

.....Insert Table 6-11 about here.....

The coefficients of *CPIS_FB*, reported in Models 1-4 of Table 6-12 are statistically significant, suggesting a negative and statistically significant effect across all the proxies of cost of capital. The magnitude and degree of statistical significance levels of *CPIS_FB* coefficient estimates are higher, compared with their counterparts from the baseline models of Table 6-5.

.....Insert Table 6-12 about here.....

The results are reported in Models 1-4 of Tables 6-13 of GF_FB . The coefficients and statistical significance of all the regressions improves relative to those reported in previous regressions based on OLS in Table 6-6.

.....Insert Table 6-13 about here.....

Clearly, our extensive sensitivity analysis demonstrates that our findings are indeed robust to different specifications, and use of different estimation methods. The overall results of the empirical analysis provide strong evidence that a higher degree of home bias is related to a higher cost of capital. Correspondingly, foreign investors, increasing their portfolio weight towards the global optimum allocations, reduce the host country's cost of capital.

6.7 Chapter summary

Existing studies provide compelling evidence on the prevalence of home and foreign bias in international portfolio allocations. However, the implications of such biases have not been extensively investigated. The theory notes that relative to the ICAPM, as the magnitude of domestic bias in the equity portfolio allocations decreases, it facilitates the benefits of global risk sharing between foreign and domestic investors. Similarly, following the same ICAPM prescription, the increase in foreign bias towards a particular host country by foreign investors should also positively influence global risk sharing.

We use global macro and fund level micro data on 44 cross-country (developed and emerging) portfolio allocations to construct the home and foreign bias measures. Similarly, following the existing literature, we employ five different proxies of the cost of capital. Applying robust econometric techniques and extensive specifications, our study finds that consistent with the theory, the results reveal that a higher degree of home bias is associated with a higher level of cost of capital. Similarly, a higher degree of foreign bias exhibited by foreign investors in their equity allocations is related to a lower cost of capital for the host countries.

The study has important implications for policy makers, especially, governments in emerging countries. Home and foreign bias matter for cost of capital. Therefore, policy makers in countries such as Bulgaria, Romania, and Indonesia need to encourage domestic investors to diversify their investments internationally. Similarly, policy makers in emerging countries can improve governance and investor protection standards to attract foreign investors.

Panel A: Devel	Panel A: Developed Countries							Emerging Countries							
Country	HRRm (%)	rCred (1-22)	CERP (%)	DY (% of price)	CPIS_HB	CPIS_FB	GF_FB	Country	HRRm (%)	rCred (1-22)	CERP (%)	DY (% of price	CPIS_HB	CPIS_FB	GF_FB
Australia	14	2.4	5	3.88	3.78	-0.37	-0.54	Argentina	19	17.10	14	3.22	6.25	-0.65	-1.64
Austria	12	2.0	5	2.17	4.75	1.13	-0.53	Brazil	28	13.70	10	3.70	4.44	-3.40	-0.51
Belgium	7	2.8	6	3.48	3.62	1.14	-0.54	Bulgaria	27	13.30	9	1.63	9.25	-3.47	-1.87
Canada	13	2.2	5	2.39	3.16	0.12	-0.81	Chile	19	6.60	6	3.09	5.56	-0.34	-2.27
Denmark	15	2.0	5	1.75	4.83	0.74	-0.11	China	11	7.10	6	2.47	3.30	-4.88	-0.26
Finland	12	2.0	5	3.66	4.96	0.28	0.32	Czech Rep	32	6.60	6	4.87	6.97	-0.35	-1.46
France	9	2.0	5	3.39	2.90	0.04	0.07	Egypt	23	10.80	7	4.58	6.58	-2.95	-2.15
Germany	17	2.0	5	2.51	2.88	0.59	0.11	Hungary	18	7.20	6	2.65	7.30	-1.12	-1.26
Greece	13	6.0	6	3.14	5.81	-1.15	-1.11	India	19	12.80	9	1.45	4.39	-6.77	-1.82
Hong Kong	10	5.4	6	2.94	3.56	0.21	0.01	Indonesia	24	15.60	12	2.96	6.13	-5.04	-1.17
Ireland	3	2.2	5	2.57	3.05	2.91	0.60	Korea	12	8.20	6	1.64	4.33	-2.15	0.01
Israel	11	6.8	6	3.66	5.78	-1.33	-0.41	Malaysia	16	8.40	7	3.04	5.30	-2.61	-1.79
Italy	9	3.6	6	4.02	3.30	0.89	-0.19	Mexico	18	9.40	7	1.76	5.23	-3.81	-0.24
Japan	4	5.7	6	1.39	2.35	-0.92	-0.12	Peru	24	11.60	8	3.91	7.04	-3.65	-1.39
Netherlands	7	2.0	5	3.86	2.84	1.08	0.54	Philippines	20	13.30	10	2.16	6.66	-5.06	-1.75
New Zealand	7	2.4	5	4.10	6.05	0.73	0.62	Poland	22	7.40	6	2.61	6.21	-2.32	-3.46
Norway	15	2.0	5	3.10	4.41	1.26	-0.05	Romania	18	13.20	9	2.70	7.98	-3.45	-2.03
Portugal	9	4.3	6	4.27	5.96	0.18	-0.73	Russia	8	12.20	8	2.01	4.61	-5.33	-2.36
Spain	13	2.2	5	3.19	3.70	-0.84	-0.58	South Africa	16	7.80	6	3.08	4.33	-0.52	-1.59
Sweden	12	2.2	5	2.81	3.97	0.73	0.01	Thailand	24	9.40	7	3.24	5.92	-3.43	-0.79
Switzerland	7	2.0	5	1.99	3.37	0.40	0.50	Turkey	19	15.00	12	1.94	5.86	-5.74	-2.01
UK	5	2.0	5	3.15	2.32	0.20	0.40								
US	4	2.0	5	1.85	0.79	-0.79	-0.09								

 Table 6-1: Summary statistics of dependent and key independent variables

Panel B: Averages of the developed and emerging countries.

0	· ·						
Country	HRRm	rCred	CERP	DY	CPIS_HB	CPIS_FB	GF_FB
	(%)	(1-22)	(%)	(% of price)			
Developed	10.30	3.0	5.30	3.00	3.38	0.31	-0.11
Emerging	19.86	10.8	8.14	2.80	5.88	-3.19	-1.51
Panel C: Averages o	f the top and bottom 10 c	ountries.					
Country	HRRm	rCred	CERP	DY	CPIS HB	CPIS FB	GF FB
Country	(%)	(1-22)	(%)	(% of price)			01_10
Top10	6.10	2.0	5.00	1.71	2.70	1.12	0.32
Bottom10	24.50	13.8	10.10	4.10	7.04	-4.72	-2.15

Note: The variables in columns 2-5 are the cost of capital measures. *HRRm* is the historical realised market return measured as the historical average of excess country equity market return over risk free rate. *rCred* is the natural log of numerical values based on Moody's country credit ratings. The qualitative credit ratings are converted into numerical values based on a scale of 1-22. We assigned a value of 1 to AAA=1, AA+=2, AA=3...all the way to D=22. *CERP* is the country equity risk premium based on adding the sovereign default risk premium (scaled by the relative volatility of equity to bond market) to the equity risk premium of a base country (The United States). *DY* is the dividend yield measured as the total amount of stock dividend of a country as a percentage of the market capitalisation of the country. The variables listed in columns 6-8 are the sub-optimal international portfolio allocation bias measures. *CPIS_HB* is the IMF-CPIS based equity home bias and is calculated as the log value of the share of domestic investors in their own country's stock market capitalisation (*l*) relative to the country's world market capitalisation weight. *CPIS_FB* is the IMF-CPIS based equity foreign bias measure computed as the average of the log value of the ratio of foreign allocations from foreign investors domiciled in country *k* investing in equities of country *l* to the benchmark allocation for country *l* ($k \neq l$). *GF_FB* is also an equity foreign bias measure but constructed using EPFR's global micro fund level data.

Panel A: Develo	ped Countries												
Country	MKTCap (in USD billions)	Beta	BM (Ratio)	Retn_1 (%)	Exch (Cov. of % in decimals)	Infl (%)	Turn (% of MKCap)	LSMI (% of GDP)	RGDPG (%)	Law (0-6)	PolRisk (0-100)	EconRisk (0-50)	FinRisk (0-50)
Australia	800	0.97	0.56	4	1.4	3	84.23	40.80	3.07	5.81	87.3	29.1	36.09
Austria	93.6	0.9	0.63	2	0.3	2	44.70	101.49	1.45	5.85	85.1	33.65	38.43
Belgium	251	0.98	0.55	1	0.4	3	44.74	153.17	1.60	4.70	82.1	42.97	27.78
Canada	1430	1.08	0.33	3	1.4	2	75.43	70.33	2.11	5.85	85.6	41.84	29.50
Denmark	168	0.95	0.78	1	1.2	2	81.36	124.87	1.12	5.68	85.2	43.53	41.92
Finland	189	1.62	0.66	6	1.3	2	119.26	55.01	2.05	5.85	91.6	45.22	37.21
France	1740	1.12	0.37	1	1.2	2	101.20	78.37	1.14	4.62	76.0	34.92	30.69
Germany	1280	1.33	0.39	2	5.3	2	141.75	53.34	1.09	4.55	82.3	36.07	26.22
Greece	122	1.07	0.70	9	2.1	7	49.06	78.38	2.40	3.55	75.3	34.77	32.76
Hong Kong	826	1.02	1.08	1	0.3	2	77.91	362.9	3.51	4.66	79.6	43.84	41.38
Ireland	92.7	0.95	0.53	6	1.3	2	52.34	161.14	2.75	5.85	85.7	41.85	35.59
Israel	135	1.03	1.07	5	1.2	3	61.53	76.03	3.43	5.15	65.8	36.16	31.27
Italy	647	0.97	0.51	3	0.4	2	130.19	52.70	0.34	3.79	78.3	35.05	31.70
Japan	3570	0.72	1.03	1	-1.2	0	111.68	26.94	1.14	4.85	81.2	36.28	43.47
Netherlands	581	1.11	0.51	1	0.3	2	147.38	132.82	1.25	5.9	87.5	41.93	29.08
New Zealand	41.5	0.87	0.62	10	1.4	2	46.31	59.34	2.27	5.45	84.8	27.89	26.50
Norway	181	1.29	1.39	4	1.4	2	114.81	71.45	2.21	5.55	84.9	44.60	44.74
Portugal	77.1	0.94	0.29	3	0.3	9	63.47	67.27	0.55	5.05	81.7	34.63	34.60
Spain	1010	1.09	0.22	5	1.3	3	164.33	56.87	1.96	4.66	77.8	38.29	36.77
Sweden	394	1.45	0.35	4	0.4	4	121.81	89.37	2.15	5.98	90.5	44.64	28.42
Switzerland	932	0.51	0.51	3	1.4	1	99.91	88.85	1.43	4.75	86.4	44.93	45.24
UK	2780	0.82	0.43	3	0.4	2	142.71	57.08	1.48	5.30	83.3	34.08	24.27
US	15600	0.88	0.52	2	0.3	2	212.34	25.46	2.00	4.83	76.1	27.80	30.56

 Table 6-2: Summary statistics of control variables

T uner D. Line		0											
	MKTCap				Exch		Turn	LSMI					
Country	(in USD	Beta	BM	Retn_1	(Cov. of % in	Infl	(% of	(% of	RGDPG	Law	PolRisk	EconRisk	FinRisk
	billions)		(Ratio)	(%)	decimals)	(%)	MKTCap)	GDP)	(%)	(0-6)	(0-100)	(0-50)	(0-50)
Argentina	77.4	1.26	0.58	7	2.2	10	10.40	40.41	4.47	3.21	70.2	32.43	31.15
Brazil	673	1.87	1.25	18	1.2	7	48.15	25.81	3.16	2.33	66.5	34.98	32.63
Bulgaria	6.62	1.29	1.31	9	4.3	5	18.62	116.48	3.86	3.89	74.6	31.08	32.30
Chile	151	0.94	0.83	13	1.3	4	15.37	69.21	3.66	4.85	76.5	40.14	25.74
China	2430	1.18	0.76	9	1.4	3	122.53	58.66	9.27	3.93	65.4	37.37	46.50
Czech Rep	37	0.90	1.31	10	2.3	2	60.96	58.66	3.06	5.15	80.7	36.91	31.03
Egypt	68.7	0.75	1.17	13	1.4	9	36.26	93.56	4.75	3.92	65.0	34.50	33.46
Hungary	26	1.29	0.55	7	1.2	5	77.24	146.36	2.03	4.30	80.9	34.87	35.64
India	754	1.01	0.81	11	1.3	7	115.82	39.90	7.66	3.70	57.9	33.53	37.38
Indonesia	125	1.23	1.20	8	1.4	8	53.16	56.92	4.99	2.92	55.0	36.83	29.54
Korea	633	1.52	0.28	3	1.2	3	222.26	82.12	4.52	4.76	76.1	41.64	34.19
Malaysia	220	1.18	0.62	9	1.3	4	31.36	191.51	4.43	3.27	74.5	35.78	36.95
Mexico	254	1.22	1.54	4	1.3	4	27.84	54.33	1.74	3.42	73.2	38.38	38.97
Peru	48.7	0.91	1.01	6	2.3	2	12.97	43.06	5.59	2.75	63.6	39.06	31.58
Philippines	63.4	1.10	0.94	8	1.4	5	18.51	89.85	4.75	2.70	67.2	29.84	35.77
Poland	103	0.71	0.82	4	0.4	3	37.55	75.52	4.15	4.45	79.6	36.48	36.19
Romania	20.5	0.67	0.59	7	1.4	11	12.97	75.08	3.88	3.75	66.7	31.71	35.21
Russia	607	1.50	0.56	6	2.3	10	58.94	55.18	4.74	4.29	68.9	37.73	43.92
South Africa	499	0.97	0.96	13	1.4	5	69.52	59.34	3.01	2.70	71.5	35.07	25.91
Thailand	130	1.16	0.90	9	2.3	12	94.00	135.01	4.17	3.17	64.7	34.20	33.74
Turkey	151	1.33	0.51	14	1.4	9	152.80	49.15	4.96	4.50	65.9	32.59	32.02

Panel B: Emerging Countries

MKTCap is the country market capitalisation; Beta is the covariance of *MSCI* country's' world index return over past five years divided by *MSCI* world index return variance; *BM* is the log country level ratio of book-to-market; *Retn_1* is the average *MSCI* monthly index return over the past year; *Exch* is the three year moving average covariance of the monthly stock market index return with the monthly depreciation of the domestic currency with respect to the dollar; *Infl* is the following year's percentage change in the consumer price index; Turn is the ratio of the total traded volume of stock in a year divided by market capitalisation; *LSMI* is the stock market integration measured as the ratio of a country's annual exports plus imports divided by *GDP*; *RGDPG* is the real growth rate in the domestic product; Law represents the rule of law index of a country; *PolRisk* is the political risk index of a country; *EconRisk* represents the economic risk index of a country.

	1	2	3	4	5	6	7	8	9	10	11	12	13	14	15	16	17	18	19	20
HRRm (1)	1																			
rCred (2)	0.47*	1																		
CERP (3)	0.48*	0.61	1																	
DY (4)	0.13*	0.21	0.08	1																
CPIS_HB (5)	0.14*	0.52	0.46*	0.10*	1															
CPIS_FB (6)	-0.21	-0.60	-0.57	-0.13	-0.39	1														
GF_FB (7)	-0.1	-0.43	-0.37	-0.08	-0.46	0.46*	1													
MKTCap (8)	-0.37*	-0.03	-0.02	-0.22	-0.25	-0.12	0.03	1												
Beta (9)	0.30*	0.26	0.26*	0.04	0.05	-0.21	0.02	0.06	1											
BM (10)	0.16*	0.27	0.24*	0.03	0.34*	-0.25	-0.12	0.04	0.02	1										
Retn_1 (11)	0.04	0.28	0.28*	0.09	0.24*	-0.26	-0.17	-0.03	0.28	0.19	1									
Exch (12)	0.15*	0.06	0.07	0.09	0.03	-0.10	-0.13	0.04	0.02	-0.10	-0.32	1								
Infl (13)	-0.25*	-0.41	-0.35*	-0.02	0.26*	-0.29	-0.29	-0.04	0.15	0.02	0.20	0.19*	1							
Turn (14)	-0.20	-0.48*	-0.45	-0.11	-0.60	0.16*	0.40*	0.30*	0.18	-0.28	0.03	-0.05	-0.18	1						
LSMI (15)	-0.11	-0.20*	-0.25	-0.08	0.15*	0.28*	0.06	-0.17	-0.10	0.02	-0.02	-0.03	-0.07	-0.03	1					
RGDPG (16)	-0.10*	-0.27	-0.28*	-0.12	0.17*	-0.46	-0.24	-0.03	0.07	-0.03	-0.05	0.25*	0.13*	-0.03	-0.02	1				
Law (17)	-0.35	-0.61*	-0.50	-0.04	-0.33	0.56*	0.35*	-0.04	-0.10	-0.31	-0.21	-0.02	-0.27	0.37*	0.19*	-0.19	1			
PolRisk (18)	-0.25	0.67*	-0.54	-0.03	-0.33	0.71*	0.39*	-0.09	-0.13	-0.30	-0.30	-0.05	-0.37	0.26*	0.26*	-0.25	0.77*	1		
EconRisk (19)	0.02	0.28*	-0.25	-0.05	-0.16	0.21*	0.19*	0.09	0.04	-0.01	-0.07	0.02	-0.14	0.15*	0.26*	-0.03	0.18*	0.23*	1	
FinRisk (20)	-0.10	0.05	-0.07	-0.17	-0.05	-0.09	0.06	-0.03	-0.10	0.10	-0.08	0.03	-0.09	0.03	0.04	0.09	0.07	0.08	0.06	1

Table 6-3: Pearson's pairwise correlation coefficient between the dependent and independent variables

Note: The variables labelled 1- 4 are the cost of capital measures and 5-7 are the sub-optimal international portfolio allocation bias measures. They are described in Table 6-1. The other variables include *MKTCap* as the log country market capitalisation; *Beta* is the covariance of *MSCI* all country's world index return over the past five years divided by the *MSCI* world index return variance; *BM* is the log country level ratio of book-to-market; Retn_1 is the average *MSCI* monthly index return over the past year; *Exch* is the three year moving average covariance of the monthly stock market index return with the monthly change of the domestic currency with respect to the dollar; *Infl* is the one year lagged rate of inflation based on the consumer price index; *Turn* is the ratio of the total traded volume of stock in a year divided by market capitalisation; *LSMI* is a measure of market integration measured as the ratio of a country's annual exports plus imports divided by *GDP*; *RGDPG* is the real growth rate in the domestic product; *Law* represents the rule of law rating index of a country; *PolRisk* is the political risk rating index of a country; *FinRisk* is the financial risk rating index of a country. Statistical significance is reported against 10% (*), 5% (**) and 1% (***) significance levels.

$coc_{lt} - a$	+ p_1 CFIS_ID+ p_2	Controls + Country	y and Teal effects	$f + \varepsilon_{it}$
	Model (1)	Model (2)	Model (3)	Model (4)
	HRRm	rCred	CERP	DY
CPIS_HB	0.696**	0.150***	0.288***	0.216***
	(2.38)	(7.21)	(3.84)	(3.20)
MKTCap	-0.739***	-0.596***	-0.882**	-0.121***
-	(-7.02)	(-5.44)	(-2.41)	(-4.92)
Beta	0.655***	0.194***	0.138***	0.846
	(4.64)	(2.99)	(4.24)	(0.42)
BM	0.180*	0.471	-0.129	-0.136
	(1.94)	(0.78)	(-0.56)	(-0.07)
Retn_1	0.264	2.978	0.143	28.24***
	(0.78)	(0.92)	(1.16)	(2.64)
Exch	0.238	0.738	0.399	3.438
	(0.23)	(1.31)	(1.31)	(1.31)
Infl	-0.116**	-0.738***	-0.462**	-2.577*
	(-2.01)	(-3.96)	(-2.04)	(-1.87)
Turn	-0.278***	0.110***	-0.645***	0.393
	(-6.65)	(8.30)	(-3.05)	(0.30)
LSMI	-0.606	-0.488***	-0.710***	-0.110
	(-0.11)	(-2.85)	(-3.59)	(-0.76)
RGDPG	-0.107	-0.726**	-0.174***	-0.541**
	(-1.02)	(-2.22)	(-3.93)	(-2.16)
Law	-0.153***	-0.334***	-0.166	0.534
	(-4.02)	(-2.85)	(-1.07)	(0.51)
PolRisk	0.148***	0.122***	-0.372*	-0.948
	(3.04)	(8.36)	(-1.92)	(-0.65)
EconRisk	-0.246	0.201*	-0.268*	0.709
	(-0.68)	(1.69)	(-1.78)	(0.71)
FinRisk	-0.702	0.631***	-0.166	-0.222***
	(-0.24)	(2.99)	(-0.22)	(-3.26)
Constant	-0.927**	1.637***	0.125***	5.511***
	(-2.17)	(6.92)	(7.72)	(4.05)
Number of Observations	390	424	440	424
Adj. R-squared	0.449	0.737	0.521	0.251
Country effects	Yes	Yes	Yes	Yes
Year fixed effects	Yes	Yes	Yes	Yes

 Table 6- 4: The relation between equity home bias and four cost of capital proxies

 This table shows regression results of the following model:

 $CoC_{lt} = \alpha + \beta_1 CPIS_HB + \beta_2 Controls + Country and Year effects + \varepsilon_{it}$

Note: The dependent variables are cost of capital (CoC_{lt}) measures of country *i* at time *t* (*HRRm*, *rCred*, *CERP* and *DY*) as described in Table 6-1 above. The key independent is *CPIS_HB* also described in Table 6-1 above. The controls include *MKTCap*, the log country market capitalisation; *Beta* is the covariance of the *MSCI* country's world index return over the past five years divided by the *MSCI* world index return variance; *BM* is the log country level ratio of book-to-market; *Retn_1* is the average *MSCI* monthly index return over the past year; *Exch* is the three year moving average covariance of the monthly stock market index return with the monthly change of the domestic currency with respect to the dollar; *Infl* is the one year lagged rate of inflation based on the consumer price index; *Turn* is the ratio of the total traded volume of stock in a year divided by market capitalisation; *LSMI* is a measure of market integration measured as the ratio of a country's annual exports plus imports divided by *GDP*; *RGDPG* is the real growth rate in the domestic product; *Law* represents the rule of law rating index of a country; *FinRisk* is the financial risk rating index of a country. All t-statistics reported are based on Newey-West autocorrelation and heteroskedasticity-corrected standard errors. For tractable interpretation, all the coefficients are reported as elasticity and the statistical significance is reported against 10% (*), 5% (**) and 1% (***) significance levels.

$coc_{lt} - a +$	p_1 CI IS_I'D+ p_2 C	unuois + Counu	y and Teal effect	stc _{it}
	Model (1)	Model (2)	Model (3)	Model (4)
	HRRm	rCred	CERP	DY
CPIS_FB	-0.812***	-0.151***	-0.424***	-0.221**
	(-2.19)	(-8.44)	(-7.38)	(-2.05)
MKTCap	-0.372**	-0.948***	-0.374**	-0.541***
	(-2.35)	(-2.69)	(-2.01)	(-5.37)
Beta	0.705	0.97***	0.500***	0.373
	(1.22)	(6.85)	(3.54)	(1.45)
BM	0.0478	0.116	0.182	-0.412
	(0.08)	(0.44)	(1.41)	(-1.61)
Retn_1	-0.296	-1.801	-0.159**	-24.01**
	(-1.35)	(-1.24)	(-2.21)	(-2.59)
Exch	0.280	0.308	0.555***	1.382
	(0.54)	(0.83)	(2.97)	(0.66)
Infl	-0.661**	-0.807***	-0.239**	-1.645
	(-2.04)	(-3.82)	(-2.50)	(-1.16)
Turn	-0.875	-0.146***	-0.430***	0.103
	(-0.23)	(-11.01)	(-6.56)	(0.63)
LSMI	-0.603	-0.398	-0.526	0.942
	(-0.40)	(-0.20)	(-0.56)	(1.52)
RGDPG	-0.165**	-0.479	-0.449**	-0.770***
	(-2.19)	(-1.10)	(-2.01)	(-2.96)
Law	-0.117***	0.203	-0.469	0.255
	(-3.12)	(1.44)	(-0.69)	(1.53)
PolRisk	0.122***	0.103***	-0.101	-0.394**
	(2.67)	(5.33)	(-1.05)	(-1.97)
EconRisk	-0.165	0.462***	-0.946	0.184*
	(-0.74)	(3.35)	(-1.41)	(1.80)
FinRisk	-0.369*	-0.183	-0.135**	0.068
	(-1.93)	(-0.16)	(-2.55)	(0.80)
Constant	0.142**	1.513***	0.825***	6.879***
	(2.28)	(9.35)	(10.27)	(2.60)
Number of Observations	440	440	364	417
Adj. R-squared	0.888	0.660	0.669	0.346
Country effects	Yes	Yes	Yes	Yes
Year fixed effects	Yes	Yes	Yes	Yes

 Table 6- 5: The relation between equity foreign bias and four cost of capital proxies

 This table shows regression results of the following model:

 $CoC_{lt} = \alpha + \beta_1 \text{CPIS}_FB + \beta_2 \text{Controls} + \text{Country and Year effects} + \varepsilon_{ii}$

Note: The dependent variables are cost of capital (CoC_{lt}) measures of country *i* at time *t* (*HRRm*, *rCred*, *CERP* and *DY*) as described in Table 6-1 above. The key independent is *CPIS_FB* also described in Table 6-1 above. All the controls are the same as described in Table 6-4 above. All t-statistics reported are based on Newey-West autocorrelation and heteroskedasticity-corrected standard errors. For tractable interpretation, all the coefficients are reported as elasticity and the statistical significance is reported against 10% (*), 5% (**) and 1% (***) significance levels.

Table 6- 6: The relation between equity foreign bias using global fund and four different cost of capital proxies

This table shows regression results of the following model:

	Model (1)	Model (2)	Model (3)	Model (4)
	HRRm	rCred	CERP	DY
GF_FB	-0.983***	- 0.459***	-0.277**	-0.155**
	(-2.29)	(-6.66)	(-2.38)	(-2.22)
MKTCap	-0.627***	-0.197***	-0.256***	-1.049***
-	(-4.93)	(-2.65)	(-4.29)	(-4.96)
Beta	0.700***	0.197***	0.156***	0.762
	(6.47)	(2.65)	(5.87)	(0.32)
BM	0.380***	0.238***	-0.927	0.685
	(4.31)	(3.31)	(-0.41)	(0.21)
Retn_1	-0.268	-1.348	-0.176	-5.881
	(-0.59)	(-0.61)	(-1.30)	(-0.72)
Exch	0.130	0.235	0.543	6.713***
	(1.10)	(0.53)	(1.36)	(3.25)
Infl	-0.259***	-1.148*	-0.447**	0.576
	(-4.03)	(-1.83)	(-2.35)	(0.31)
Turn	-0.436***	-0.247***	-0.112***	-0.153
	(-11.12)	(-8.37)	-(8.35)	(-1.06)
LSMI	-0.305	0.506	-0.590***	-0.946*
	(-0.52)	(0.22)	(-3.42)	(-1.66)
RGDPG	-0.899	-0.635	-0.196***	-0.219
	(-0.75)	(-1.36)	(-5.57)	(-0.73)
Law	-0.302***	-0.546***	0.358	-0.304
	(-5.08)	(-3.20)	(0.22)	(-0.20)
PolRisk	0.248***	0.119***	-0.454***	-0.205
	(4.60)	(5.74)	(-2.87)	(-1.19)
EconRisk	-0.330	0.496***	-0.241*	0.565
	(-0.78)	(3.04)	(-1.93)	(0.68)
FinRisk	-0.446	-0.959	0.285	0.346
	(-0.13)	(-0.73)	(0.03)	(0.49)
Constant	-0.217***	2.654***	0.136***	23.30***
	(-3.46)	(6.01)	(8.77)	(5.91)
Number of Observations	380	410	380	400
Adj. R-squared	0.463	0.559	0.535	0.579
Country effects	Yes	Yes	Yes	Yes
Year fixed effects	Yes	Yes	Yes	Yes

 $CoC_{lt} = \alpha + \beta_1 \text{GF}_F\text{B} + \beta_2 \text{Controls} + \text{Country and Year effects} + \varepsilon_{it}$

Note: The dependent variables are cost of capital (CoC_{lt}) measures of country *i* at time *t* (*HRRm*, *rCred*, *CERP* and *DY*) as described in Table 6-1 above. The key independent is *GF_FB* also described in Table 6-1 above. All the controls are the same as described in Table 6-4 above. All t-statistics reported are based on Newey-West autocorrelation and heteroskedasticity-corrected standard errors. For tractable interpretation, all the coefficients are reported as elasticity and the statistical significance is reported against 10% (*), 5% (**) and 1% (***) significance levels.

Table 6-7: Relation betw	veen foreign bias, ho	ome bias, country bias	and Tobin's Q
	Model (1)	Model (2)	Model (3)
	Tobin's Q	Tobin's Q	Tobin's Q
CPIS_HB	-0.415***		
	(-7.49)		
CPIS_FB		0.717***	
		(3.47)	
GF_FB			0.294***
			(3.94)
MKTCap	-0.449***	-0.419***	-0.491***
-	(-19.10)	(-16.66)	(-18.68)
Beta	-0.202	-0.160	-0.0137
	(-1.08)	(-0.80)	(-0.07)
BM	-0.172	-0.501***	-0.725***
	(-0.95)	(-2.67)	(-3.95)
Retn_1	16.11	5.845	2.690
	(1.63)	(0.60)	(0.26)
Exch	3.245	4.895**	3.021
	(1.32)	(2.11)	(1.13)
Infl	-1.294	-0.307	-1.965
	(-0.91)	(-0.21)	(-1.31)
Turn	-0.105	0.346***	0.145
	(-1.01)	(3.70)	(1.25)
LSMI	-0.663***	-0.923***	-1.014***
	(-4.84)	(-6.52)	(-7.34)
RGDPG	-0.424	-0.510	-0.231
	(-1.62)	(-0.20)	(-0.78)
Law	0.172*	0.202**	0.0913
	(1.83)	(2.02)	(0.95)
PolRisk	0.601***	0.412***	0.448***
	(5.15)	(3.03)	(3.49)
EconRisk	-0.207**	-0.121	-0.106
	(-2.24)	(-1.24)	(-1.10)
FinRisk	-0.123*	-0.447	0.368
	(-1.68)	(-0.56)	(0.48)
Constant	12.90***	11.63***	12.86***
	(12.19)	(10.23)	(11.67)
Number of Observations	440	440	410
Adj. R-squared	0.565	0.502	0.567
Country effects	Yes	Yes	Yes
Year fixed effects	Yes	Yes	Yes

Note: The dependent variable *Tobin's Q* is measured as the log market value of equity plus the book value of total assets minus the book value of equity and divided by the book value of total assets of country *i*. The key independent variables are *CPIS_HB, CPIS_FB* and *GF_FB*, also described in Table 6-1 above. All the controls are the same as described in Table 6-4 above. All t-statistics reported are based on Newey-West autocorrelation and heteroskedasticity-corrected standard errors. For tractable interpretation, all the coefficients are reported as elasticity and the statistical significance is reported against 10% (*), 5% (**) and 1% (***) significance levels.

	Model (1)	Model (2)	Model (3)	Model (4)
	HRRm	rCred	CERP	DY
CPIS HB 1	0.829***	0.305***	0.336***	0.196***
	(3.41)	(3.82)	(4.67)	(3.38)
MKTCap	-0.808***	-0.160***	-0.101***	-0.105***
L	(-7.69)	(-4.89)	(-3.31)	(-4.39)
Beta	0.721***	0.214***	0.143***	0.662
	(8.57)	(7.97)	(5.73)	(0.34)
BM	0.194**	0.408	-0.131	-0.0277
	(2.50)	(1.62)	(-0.55)	(-0.15)
Retn 1	-0.603	-1.065	0.275**	-30.36***
—	(-1.38)	(-0.73)	(2.02)	(-2.87)
Exch	0.577	0.263	0.753**	2.217
	(0.57)	(0.77)	(2.33)	(0.90)
Infl	0.0761	0.731***	0.263	2.486*
	(1.36)	(3.84)	(1.45)	(1.77)
Turn	-0.268***	-0.119***	-0.555***	0.700
	(-6.14)	(-8.09)	(-4.05)	(0.65)
LSMI	0.292	0.429**	-0.642***	-0.317
	(0.52)	(2.27)	(-3.57)	(-0.23)
RGDPG	0.107	-0.728**	-0.177***	-0.374
	(0.99)	(-2.04)	(-5.21)	(-1.44)
Law	-0.160***	0.344***	-0.265**	-0.183
	(-4.04)	(2.65)	(-2.13)	(-0.19)
PolRisk	0.128**	0.121***	-0.317**	0.154
	(2.55)	(7.48)	(-2.05)	(0.13)
EconRisk	-0.232	0.221*	-0.265**	0.346
	(-0.61)	(1.68)	(-2.15)	(0.36)
FinRisk	-0.701	-0.161	0.371	-0.242***
	(-0.24)	(-1.61)	(0.39)	(-3.33)
Constant	-0.111**	1.632***	0.112***	4.178***
	(-2.52)	(11.17)	(8.06)	(3.94)
Number of Observations	354	383	396	382
Adj. R-squared	0.472	0.720	0.539	0.248
Country effects	Yes	Yes	Yes	Yes
Year fixed effects	Yes	Yes	Yes	Yes

Table 6-8: Using one-year lag CPIS HB

Note: The dependent variables are cost of capital measures (*HRRm*, *rCred*, *CERP* and *DY*) as described in Table 6-1 above. The key independent is one year lagged *CPIS_HB*, also described in Table 1 above. All the controls are the same as described in Table 6-4 above. All t-statistics reported are based on Newey-West autocorrelation and heteroskedasticity-corrected standard errors. For tractable interpretation, all the coefficients are reported as elasticity and the statistical significance is reported against 10% (*), 5% (**) and 1% (***) significance levels.

8 4	Model (1)	Model (2)	Model (3)	Model (4)
	HRRm	rCred	CERP	DY
CPIS_FB_1	-0.931***	-0.286***	-0.428***	-0.262**
	(-3.69)	(-3.43)	(-9.29)	(-2.34)
MKTCap	- 0.922**	-0.986***	-0.329*	-0.993***
	(-2.00)	(-2.59)	(-1.80)	(-6.70)
Beta	0.610	0.206***	0.529***	0.402
	(0.92)	(6.86)	(3.51)	(1.40)
BM	-0.524	0.170	0.181	0.825***
	(-0.01)	(0.06)	(1.34)	(2.70)
Retn_1	-0.505**	-0.635	-0.880	-22.19**
	(-1.98)	(-0.39)	(-1.08)	(-2.06)
Exch	0.116	0.456	0.706***	1.720
	(0.20)	(1.13)	(3.42)	(0.71)
Infl	-0.354	-0.590**	-0.200*	-0.162
	(-0.97)	(-2.50)	(-1.87)	(-0.10)
Turn	-0.482	-0.144***	-0.422***	-0.556***
	(-0.99)	(-9.85)	(-5.93)	(-2.70)
LSMI	-0.183	-0.268	0.451	-1.390**
	(-1.09)	(-0.13)	(0.45)	(-2.00)
RGDPG	-0.166*	-0.592	-0.444*	-0.102***
	(-1.97)	(-1.28)	(-1.84)	(-3.64)
Law	-0.107**	0.172	-0.220	-0.352*
	(-2.52)	(1.15)	(-0.30)	(-1.94)
PolRisk	0.116**	0.961***	-0.142	-0.211
	(2.04)	(4.43)	(-1.34)	(-0.90)
EconRisk	-0.133	0.503***	-0.502	0.205*
	(-0.05)	(3.38)	(-0.68)	(1.86)
FinRisk	-0.397*	0.762	-0.135**	0.130
	(-1.92)	(0.06)	(-2.45)	(1.45)
Constant	0.709	1.593***	0.838***	7.559**
	(0.94)	(8.90)	(9.48)	(2.52)
Number of Observations	396	396	331	376
Adj-squared	0.884	0.659	0.665	0.362
Country effects	Yes	Yes	Yes	Yes
Year fixed effects	Yes	Yes	Yes	Yes

Table 6-9:	Using	one-vear	lag	CPIS	FB
\mathbf{I} u \mathbf{U} \mathbf{U} \mathbf{U} \mathbf{U}	COME	une rear	IUC.		• •

Note: The dependent variables are cost of capital measures (*HRRm*, *rCred*, *CERP* and *DY*) as described in Table 6-1 above. The key independent is one year lagged *CPIS_FB*, also described in Table 1 above. All the controls are the same as described in Table 6-3 above. All t-statistics reported are based on Newey-West autocorrelation and heteroskedasticity-corrected standard errors. For tractable interpretation, all the coefficients are reported as elasticity and the statistical significance is reported against 10% (*), 5% (**) and 1% (***) significance levels.

8	Model (1)	Model (2)	Model (3)	Model (4)
	HRRm	rCred	CERP	DY
GF_FB_1	-0.147***	-0.567***	-0.324***	-0.216***
	(-4.09)	(-5.50)	(-2.86)	(-2.88)
MKTCap	-0.749***	-0.372	-0.154***	-0.971***
	(-6.94)	(-0.98)	(-4.34)	(-9.60)
Beta	0.684***	0.166***	0.164***	0.297
	(7.50)	(5.17)	(5.79)	(1.24)
BM	0.415***	-0.482	-0.137	0.158
	(4.90)	(-1.63)	(-0.05)	(0.70)
Retn_1	-0.396	-0.136	-0.341**	-6.301
	(-0.08)	(-0.08)	(-2.25)	(-0.68)
Exch	0.291	0.212	0.682*	6.311***
	(0.24)	(0.51)	(1.85)	(2.97)
Infl	-0.189***	-0.989***	-0.406**	-0.556
	(-2.93)	(-4.35)	(-2.07)	(-0.34)
LSMI	-0.453	-0.0175	-0.480***	-0.882
	(-0.75)	(-0.80)	(-2.62)	(-1.37)
RGDPG	-0.643	-0.354	-0.191***	-0.184
	(-0.52)	(-0.80)	(-5.08)	(-0.55)
Law	-0.291***	-0.598***	-0.419	-0.566
	(-6.72)	(-3.92)	(-0.30)	(-0.36)
PolRisk	0.224***	0.124***	-0.404**	-0.271
	(3.99)	(6.35)	(-2.38)	(-1.48)
EconRisk	-0.218	0.219	-0.245*	0.517
	(-0.49)	(1.40)	(-1.81)	(0.57)
FinRisk	-0.480	-0.534	0.596	-0.189
	(-0.14)	(-0.44)	(0.55)	(-0.26)
Constant	-0.232***	1.786***	0.124***	22.54***
	(-4.93)	(10.47)	(8.59)	(7.84)
Number of Observations	342	369	342	360
Adj. R-squared	0.474	0.622	0.548	0.587
Country effects	Yes	Yes	Yes	Yes
Year effects	Yes	Yes	Yes	Yes

Table	6-10:	Using	one-vear	lag	GF	FB
Lanc	0-10.	USHIE	Unc-vcar	IUL	UL.	ĽD

Note: The dependent variables are cost of capital measures (*HRRm*, *rCred*, *CERP* and *DY*) as described in Table 6-1 above. The key independent is one year lagged GF_FB , also described in Table 6-1 above. All the controls are the same as described in Table 6-3 above. All t-statistics reported are based on Newey-West autocorrelation and heteroskedasticity-corrected standard errors. For tractable interpretation, all the coefficients are reported as elasticity and the statistical significance is reported against 10% (*), 5% (**) and 1% (***) significance levels.

Table 0-11. CI 15_11D Tegi	coston based on	the netwinan tw	o stage il catille	it mouel
Heckman two-stage model	Model (1)	Model (3)	Model (2)	Model (4)
-	HRRm	rCred	CERP	DY
CPIS_HB	0.775***	0.332***	0.288***	0.134**
	(3.34)	(8.14)	(4.28)	(3.43)
МКТСар	-0.796***	-0.619***	-0.824***	-0.108***
-	(-8.06)	(-5.51)	(-3.09)	(-4.62)
Beta	0.525***	-0.214***	0.149***	-0.681
	(6.01)	(-7.39)	(6.05)	(-0.54)
BM	0.167	0.387	0.174	0.162
	(1.45)	(1.10)	(0.59)	(0.82)
Retn_1	-0.371**	-0.934	0.128	-0.265***
	(-2.57)	(-0.65)	(1.20)	(-2.89)
Exch	0.345	0.637	0.402	0.279
	(0.30)	(0.67)	(1.34)	(1.14)
Infl	-0.241***	-0.708***	-0.487**	-0.703
	(-4.21)	(-3.45)	(-2.10)	(-0.50)
Turn	- 0.325***	-0.116***	-0.671***	0.337
	(-4.32)	(-7.78)	(-5.15)	(0.33)
LSMI	-0.642	-0.356*	-0.698***	-0.116
	(-0.37)	(-1.80)	(-4.28)	(-0.85)
RGDPG	-0.124	-0.810**	-0.153***	-0.640***
	(-0.14)	(-2.40)	(-5.47)	(-2.84)
Law	-0.238***	-0.240*	-0.183	0.605
	(-5.13)	(-1.77)	(-1.45)	(0.65)
PolRisk	0.177***	0.134***	0.317***	0.867
	(3.03)	(7.20)	(2.62)	(1.30)
EconRisk	0.279	0.238***	0.272**	0.819
	(1.11)	(3.29)	(2.40)	(1.00)
FinRisk	0.787	0.613	0.174	0.217***
	(1.11)	(1.08)	(0.19)	(2.99)
Lambda (inverse Mills' ratio)	-0.534***	-0.614***	-0.487***	-0.527***
	(-16.24)	(-14.57)	(-12.36)	(-12.48)
Wald Chi-square	308.85	376.35	299.74	227.61
Constant	0.607***	1.541***	0.125***	6.276***
	(4.27)	(10.08)	(9.74)	(5.96)

Table 0-11: UPIN IID regression based on the neckinal two stage treatment mod	e treatment mod	o stage 1	two	ckman	he Ho	on	based	regression	HB	CPIS	5-11:	Table (
---	-----------------	-----------	-----	-------	-------	----	-------	------------	----	------	-------	---------

Note: This table presents the coefficients of the estimates from Heckman two-stage treatment effects models. In the first stage, we run the probit model. We include Lambda (inverse Mills' ratio) in the second stage with control variables. The dependent variables are cost of capital measures (*HRRm*, *rCred*, *CERP* and *DY*) as described in Table 6-1 above. The key independent variable is *CPIS_HB*, also described in Table 6-1 above. All the controls are the same as described in Table 6-3 above. All t-statistics reported are in parentheses. For tractable interpretation, all the coefficients are reported as elasticity and the statistical significance is reported against 10% (*), 5% (**) and 1% (***) significance levels.

Heckman two-stage model	Model (1)	Model (2)	Model (3)	Model (4)
Heekman two stage model	HRRm	rCred	CERP	DY
CDIS ER	0.682***	0.220***	0.456***	0.270**
Cr15_1B	(2.21)	(9.19)	(4.22)	(2.10)
MKTCon	(-3.31)	(-0.10)	(-4.22)	(-2.10)
WIKTCap	-0.430^{11}	-0.801	-0.340^{11}	-0.073^{+++}
Data	(-2.54)	(-2.70)	(-2.24)	(-7.55)
Beta	(1, 25)	0.19/	0.131^{***}	0.450*
DM	(1.55)	(7.14)	(5.73)	(1.80)
BM	0.137	0.107	0.204	0.393
D	(0.26)	(0.45)	(0.05)	(1.43)
Retn_1	-0.248	-1.755	-0.204*	-23.41***
	(-1.62)	(-1.28)	(-1.74)	(-2.67)
Exch	0.361	0.324	0.497*	1.280
	(0.75)	(0.85)	(1.65)	(0.66)
Infl	-0.724**	-0.782***	-0.348***	-1.231
	(-2.19)	(-3.93)	(-2.62)	(-0.93)
Turn	-0.844	-0.177***	-0.508***	-0.146
	(-0.41)	(-11.32)	(-8.38)	(-0.57)
LSMI	0.617	0.358	-0.412**	0.784
	(0.83)	(0.20)	(-2.53)	(1.23)
RGDPG	-0.160**	-0.398	-0.422***	-0.699**
	(-2.27)	(-1.13)	(-3.46)	(-2.50)
Law	-0.193***	-0.153	-0.430	-0.249
	(-2.78)	(-1.48)	(-1.14)	(-1.56)
PolRisk	0.111**	0.114***	-0.146	0.308***
	(2.56)	(5.48)	(-0.94)	(2.78)
EconRisk	0.153	0.521***	0.880**	0.203**
	(0.73)	(3.45)	(2.51)	(2.14)
FinRisk	0.484**	0.177	0.127	0.097
	(2.10)	(0.17)	(1.29)	(0.89)
Lambda (inverse Mills' ratio)	0.347***	0.394***	0.489***	0.513***
	(21.45)	(18 67)	(15 33)	(17.25)
Wald Chi-square	491.55	438.77	523.78	434.28
Constant	0.130**	1 513***	0.122***	10 24***
	(2.21)	(9.61)	(9.34)	(4.24)

Table 6-12: CPIS	FB	regression	based	on the	Heckman	two stage	treatment	model
		ICLICOBION	nabea	on the	iicciman	the buge	u cuuntu	mouci

Note: This table presents the coefficients of the estimates from Heckman two-stage treatment effects models. In the first stage, we run the probit model. We include Lambda (inverse Mills' ratio) in the second stage with control variables. The dependent variables are cost of capital measures (*HRRm*, *rCred*, *CERP* and *DY*) as described in Table 6-1 above. The key independent variable is *CPIS_FB*, also described in Table 6-1 above. All the controls are the same as described in Table 6-4 above. All t-statistics reported are in parentheses. For tractable interpretation, all the coefficients are reported as elasticity and the statistical significance is reported against 10% (*), 5% (**) and 1% (***) significance levels.

Heckman two-stage model	Model (1)	Model (2)	Model (3)	Model (4)
	HRRm	rCred	CERP	DY
GF_FB	-0.531***	-0.659***	-0.327***	-0.123***
	(-4.62)	(-11.20)	(-6.32)	(-5.11)
МКТСар	-0.770***	- 0.266***	-0.424**	-0.910***
	(-7.23)	(-7.74)	(-2.56)	(-4.64)
Beta	0.563***	0.190***	0.146***	0.816
	(6.37)	(3.96)	(6.29)	(0.44)
BM	0.243***	0.581	0.629	-0.792
	(2.87)	(1.26)	(0.29)	(-0.44)
Retn_1	-0.300**	-2.943	0.253**	-20.29**
	(-1.98)	(-1.16)	(2.10)	(-2.08)
Exch	0.216	0.458	0.466	3.475
	(0.70)	(1.04)	(1.20)	(1.40)
Infl	0.334***	0.497	0.434**	-0.188
	(4.61)	(1.37)	(2.48)	(-0.13)
Turn	-0.345***	-0.125***	-0.559***	-0.632
	(-5.38)	(-2.71)	(-7.59)	(-1.39)
LSMI	-0.410	-0.471	-0.529***	-0.792***
	(-0.07)	(-1.34)	(-3.24)	(-3.22)
RGDPG	-0.759	-0.543***	-0.183***	-0.423**
	(-0.59)	(-8.10)	(-5.61)	(-2.36)
Law	-0.283***	-0.431***	-0.241	-0.297
	(-5.99)	(-5.45)	(-1.22)	(-0.74)
PolRisk	0.177***	0.222***	0.419***	0.186
	(3.00)	(7.42)	(2.90)	(1.59)
EconRisk	0.315	0.249	0.313***	0.532
	(0.71)	(1.05)	(2.75)	(0.61)
FinRisk	0.525	0.584***	0.330	0.226***
	(1.47)	(3.13)	(1.02)	(3.10)
Lambda (inverse Mills' ratio)	0.613***	0.486***	0.493***	0.628***
	(19.32)	(17.55)	(21.03)	(16.23)
Wald Chi-square	562.37	523.74	496.25	477.49
Constant	-0.150**	2.715***	0.143***	7.499***
	(2.03)	(7.38)	(6.34)	(5.14)

Table 6-13: GF	FB	regression	based (on the	Heckman	two-stage	treatment	effect model

Note: The dependent variables are cost of capital measures (*HRRm*, *rCred*, *CERP* and *DY*) as described in Table 6-1 above. The key independent is *GF_FB*, also described in Table 6-1 above. All the controls are the same as described in Table 6-4 above. All t-statistics reported are based on Newey-West autocorrelation and heteroskedasticity-corrected standard errors. For tractable interpretation, all the coefficients are reported as elasticity and the statistical significance is reported against 10% (*), 5% (**) and 1% (***) significance levels.

Chapter 7 Second empirical study: International portfolio investment and stock market development

Stock markets are increasingly seen as an important source of financial resources for investment in both developed and emerging countries (see Chapter 2 Section 4.1). Current studies have provided compelling evidence of the prevalence of home and foreign bias across both developed and emerging markets. Studies provide theoretical argument that domestic and foreign investors over or underinvestment will have implications on stock market development. For instance, Errunza (2001) provides theoretical suggestions that sub-optimal portfolio allocations could have varying effects on stock market development. Countries that experience significant home bias will have high illiquid stock market, which will inhibit stock market development. In light of this theoretical argument and the importance of stock market development, we are motivated to undertake this study to examine whether sub-optimal portfolio allocation (home and foreign bias) have implications for stock market development. We use four measures of stock market development to test the following hypothesis:

- H₄ Countries with a prevalence of higher home bias experience poor stock market development.
- H₅ Countries with a higher foreign bias experience better stock market development.
- **H**₆ Countries with a higher global fund foreign bias enjoy better stock market development.

7.1 Summary analysis

We start our empirical analysis by providing summary statistics of several measures of stock market development. ⁵² Consistent with the definition provided by MSCI, the sample countries used in our study include 23 developed countries and 21 emerging countries from 2001 to 2010. We subdivided our sample into developed and emerging countries and present the summary figures of the four stock market development measures at country level in Table

⁵² See Table 6.1 for the average statistics of CPIS_HB, CPIS_FB, and GF_FB.

7-1. In panel B, we group the same summary figures into developed and emerging markets. Subsequently, we present in panel C, the figures for both the top 10 and the bottom 10 countries arranged in line, with sub-optimal portfolio allocation. Successively, we report the correlation matrix between the four stock market development proxies and the three key independent variables (*CPIS_HB*, *CPIS_FB* and *GF_FB*), then proceed with our robust regression analysis.

.....Insert Table 7-1 about here.....

As reported in Table 7-1, we partition our sample countries in developed and emerging countries. Developed countries tend to experience well developed stock markets as compared to emerging markets. Most of the countries that rank in the top ten against the measure of stock market capitalisation as a percentage of GDP (MGDP), are developed markets (Hong Kong, Switzerland, South Africa, Malaysia, United Kingdom, United States, Australia, Canada, Chile, and Sweden). On the other hand, the bottom ten countries classified as least developed in terms of stock market development are mostly emerging markets, except Austria and New Zealand (Romania, Bulgaria, Hungary, Czech Republic, Mexico, Poland, Turkey, and Indonesia). A similar picture emerges when we compare the top and bottom ten rankings against the TRGDP and TURN factors, whereby the majority of markets occupying the top positions are developed with the bottom ten being emerging markets.⁵³ When we compare the markets on the basis of lowest to highest transaction cost (TRCOST), the top ten countries with the least transaction costs are all developed markets (Japan, United States, France, Germany, Switzerland, Belgium, Netherlands, Italy, Sweden and Norway). Similarly countries with the highest transaction costs are all emerging markets (Indonesia, Argentina, Egypt, South Africa, Peru, Romania, Philippines, Russia, Chile, and Poland). Clearly, the ranking based on all the four stock market development measures, convincingly supports the view that compared to the smaller emerging markets, developed markets are bigger in size, more highly diversified, exhibit higher level of liquidity, and reflect lower transaction costs.

To further substantiate the initial country-wise analysis, we present the same summary figures in Panel B of Table 7-1 but group the countries on basis of developed and emerging

⁵³ The top ten countries against the measure of *TRGDP* are Hong Kong, United States, Switzerland, United Kingdom, South Korea, Netherlands, Spain, Finland and South Africa. Similarly the bottom ten are Czech Republic, Indonesia, Austria, Poland, Philippines, Mexico, Bulgaria, Argentina, Peru and Romania.

markets. The average stock market development figures of the developed markets (MGDP = 97.05, TRGDP = 99.67, TURN = 99.47 and TRCOST = 31.51) reflect a much higher level of stock market development, relative to the figures for emerging markets (MGDP = 58.97, TRGDP = 33.62, TURN = 60.70 and TRCOST = 54.68).

We observe that the top 10 countries average figures (MGDP = 170.00, TRGDP = 177.68, TURN = 156.31 and TRCOST = 30.70) again reflect much higher level of stock market development, compared to the bottom ten figures for emerging markets (MGDP = 24.49, TRGDP = 8.08, TURN = 21.59 and TRCOST = 76.01)

To summarise, the extensive analysis of the summary statistics on stock market development and sub-optimal portfolio investments impart strong indications that countries that exhibit lower home bias and are favoured more by foreign investors have more developed stock markets. This suggests that markets where domestic (foreign) investors' exhibit lower (higher) levels of home bias (foreign bias) have more developed stock markets. We further substantiate such signals, using correlation and more robust regression analysis in the following sections.

We present a summary statistics of the control variables in Table 7-2.⁵⁴ *Flow, Int, FinRisk,* and *Cor* show substantial cross country variations. *Flow* ranges from 0.191 (Belgium) to 0.002 (Japan). *Int* varies between 40.5 (Brazil) and 1.12 (United States). China has the highest financial risk (*FinRisk=46.5*) and lowest is United Kingdom (*FinRisk=24.27*). *Cor* is between 5.55 (Finland) and 1.72 (Russia). Detailed description of the control variables are presented in Section 4.3.5.

.....Insert Table 7-2 about here.....

7.2 Correlation coefficients

Table 7-3 reports the cross-correlation coefficient matrix between all variables under scrutiny. Not surprisingly, Table 7-3 shows positive correlations among the first three measures of stock market development (*i.e. MGDP*, *TRGDP* and *TURN*). Further, in line with expectation, *TRCOST* is negatively correlated with the other three proxies of market

⁵⁴ To avoid a repetition of summary statistics of control variables used in the previous empirical chapter, see Table 6.2 for a summary statistics of control variables such as *Exch*, *Infl*, *Retn_1*, *LSMI*, *MKTCap*, and *R_Law*.

development. These estimates show that markets which are bigger and more diversified are also more liquid and price efficient. In analysing the measures of stock market development against the measures of sub-optimal portfolio allocations (*CPIS_HB, CPIS_FB* and *GF_FB*), consistent with our theoretical expectations, we find that the home bias (*CPIS_HB*) carries statistically significant negative correlation coefficients with the first three stock market development indicators i.e. *MGDP*, *TRGDP*, *TURN* and positive with *TRCOST*. This provides further support to the conjecture that the tendency of investors to overweight their domestic stocks in their portfolio has a negative impact on the development of domestic stock markets. The measures of foreign bias (i.e., *CPIS_FB* and *GF_FB*) bear statistically significant and positive correlation coefficients with the three measures of stock market development (*MGDP*, *TRGDP* and *TURN*) and negative with *TRCOST*. Again, the estimates suggest a positive association between foreign equity portfolio allocation and the level of stock market development. The coefficients of correlations of most other variables have expected signs.

.....Insert Table 7-3 about here.....

7.3 Econometric issues

As reported in Section 5.4 of the research method, we perform diagnostic tests to find if our models have economic issues, and accordingly resolved them.

7.3.1 Autocorrelation

We use the Durbin-Watson test to check for the presence of autocorrelation by running the following regression in STATA. We obtained DW values of 1.183272 which indicates the presence of autocorrelation.

7.3.2 Heteroskedasticity

We use Breusch-Pagan to test for heteroskedasticity. A small chi-square value of 2.80 implies there is no heteroskedasticity problem and the variances of the error terms are equal.

Breusch-Pagan test for heteroskedasticity

Ho: Constant variance Variables: fitted values of MGDP chi2(1) = 2.80Prob > chi2 = 0.0943

7.3.3 Multicollinearity

We use variance inflation factors (VIF) to test for multicollinearity. The mean of VIF for our model is 1.71 which suggests that our model has no multicollinearity problem.

Variable	VIF	1/VIF
MKTCap	1.82	0.54824
R_Law	1.62	0.61878
Retn_1	1.51	0.66198
PCred	1.49	0.67186
Cor	1.48	0.67349
Legal_O	1.33	0.75094
Exch	1.30	0.77216
LSMI	1.21	0.82736
Infl	1.21	0.82846
GDPPC	1.15	0.86953
Int	1.12	0.89126
Flow	1.09	0.91624
FinRisk	1.08	0.92534
Sav	1.03	0.97215
Mean VIF	1.71	

7.3.4 Omitted variable bias

We employ Ramsey reset to test for omitted variable bias. The F values are all above 5% and therefore indicate that we have included all relevant explanatory variables and our model does not suffer from omitted variable bias.

Ramsey reset test of omitted variables for CPIS_HB, CPIS_FB, and GF_FB. Test of important variables omitted. Ramsey reset test using powers of the fitted values

Ho: model has no omitted variables

Cost of Capital		
CPIS_HB	F(3, 422)=1.78	Pro>F=0.15
CPIS_FB	F(3, 422)=1.74	Pro>F=0.16
GF_FB	F(3, 422)=2.64	Pro>F=0.06

7.4 **Regression results**

This section documents further evidence on whether the cross-sectional and temporal differences in home and foreign biases of equity portfolio investors, can explain the variations in the levels of stock market development. In all pooled OLS regressions, the
standard errors are corrected for arbitrary autocorrelation and heteroskedasticity using the Newey-West method. We also deal with the potential issue of estimation biases using alternative specifications. In subsequent sections, we initially discuss the empirical results between the home and foreign bias measures and stock market development. We preserve the discussion of the control variables at the end.

7.4.1 Stock market development and equity home bias

We explore, in this section, does home bias affect the level of stock market development? We address this question by presenting the results of four regressions (models 1-4) as reported in Table 7-4.⁵⁵ In all the specifications (models 1-4) the key explanatory variable of interest, is home bias (*CPIS_HB*), with *MGDP*, *TRGDP*, *TURN* and *TRCOST* as dependent variables respectively. All the specifications include the controls, country and year fixed effects.

Models 1-3 (columns 2-4) show that the coefficients of equity home bias (*CPIS_HB*) measure, bears the expected negative signs against *MGDP*(-0.276), *TRGDP*(-0.662) and *TURN*(-0.380) variables. Similarly, the *CPIS_HB* coefficient in Model 4 (column 5) records a significant positive sign against *TRCOST* (0.145). All the coefficients are statistically significant at 1% significance level. These figures suggest, that on average, a one percentage decrease in home bias, increases the size and diversity of the stock market by nearly 0.3% of GDP (*MGDP* = -0.276). Similar increase in home bias, negatively affects the liquidity of the market by 0.7% (*TRGDP* = -0.662), 0.4% (*TURN* = -0.380), and 0.1% (*TRCOST* = 0.145) against the proxies reflecting the degree of market liquidity. Even though, as in any empirical work, we have to exercise caution in quantitatively interpreting the results from a sample. Our study provides a qualitative and statistically strong indication of the adverse connotations of over-weighting home markets on the development level of local stock market.

.....Insert Table 7-4 about here.....

7.4.2 Stock market development and equity foreign bias

In Table 7-5, we examine the effect of foreign bias (*CPIS_FB*) on the four stock market development proxies by running four regressions (Models 1-4) which also include control variables and year fixed effects. In line with economic basis, the positive and statistically

⁵⁵ The *t*-statistics are reported in parentheses.

significant coefficients on the stock market development measures of Models 1-3 provide convincing evidence that the higher the foreign bias (*CPIS_FB*), the more developed are the stock markets. The coefficient estimates of 0.093, 0.205, and 0.096 for *MGDP*, *TRGDP*, and *TURN* respectively suggest that markets which are able to attract a higher level of foreign investors' fund (i.e. higher foreign bias) benefit from developing the depth and breadth of their local stock market. Similarly, the coefficient estimates in Model 4 of -0.087, implies that foreign investors' favourable bias towards a country, helps render the market more competitive in terms of reducing transaction costs.

.....Insert Table 7-5 about here.....

7.4.3 Stock market development and fund level foreign bias measures

We further investigate whether equity foreign bias (*GF_FB*), measured by cross-country allocations of global funds, have a similar influence on the prospects of stock market development. As reported in Table 7-6, with the exception of Model 1, where the coefficient estimates are statistically significant at the 10% level, the remaining coefficients reported in Models 2-4 are statistically significant at the 1% level. The reported coefficients of 0.063, 0.394, 0.323 and -0.112 for *MGDP*, *TRGDP*, *TURN*, and *TRCOST* respectively lend strong support to our earlier results. This clearly suggests that market competition, size and diversity contributions of foreign portfolio investors, are imperative to develop the local stock markets.

.....Insert Table 7-6 about here.....

7.4.4 Control variables

Our findings are robust to control variables where most exhibit the expected signs. Among the control variables, the theoretically consistent and statistically significant variables include, flow of foreign direct investment (*Flow*), growth of private sector credit (*PCred*), recent stock market returns (*Retn_1*), market capitalisation (*MKTCap*), law and order in the country (*R_Law*) and legal origin (*Legal_O*). As expected, a higher level of inward foreign direct investment flows, have a significant positive influence on the development of the depth and breadth of local stock markets. Similarly, the growth in private sector credit (*PCred*) lends support to the development of equity markets. The estimates also show that the development of the stock market is positively associated with the recent stock returns (*Retn_1*). As the stock markets (*MKTCap*) grow bigger, they further drive the competition, diversity and liquidity of the markets. Finally, both the investor protection measures (rule of law and legal origin), carry expected signs and show that quality of legal institutions are imperative to the development of the stock markets, and countries with common law based legal traditions, are associated with higher stock market development.

Interest rate (*Int*) has a significant negative effect on stock market development. Other control variables show less statistically significant impact on stock market development. *Sav, Exch, Infl* and *FinRisk* measures have the expected sign but play virtually no role in explaining stock market development. Stock market integration largely has a statistically significant positive impact on stock market development. Finally, *Cor* negatively related to stock market development, and is mainly significant. This suggests that countries that experience a high level of corruption will experience poor stock market development.

7.5 Robustness tests

Our main analysis provided in the previous section, offers compelling evidence that is consistent with financial literature and theory; thus home and foreign bias have a varying impact on stock market development. In this section, we extend our analysis of the relationship between sub-optimal international equity allocations (i.e. home and foreign bias) and stock market development, by running a battery of robustness tests to provide support to our main analysis. First, we use lagged values to address the concern of endogeneity, caused by reverse causality. Second, we employ the Heckman selection model to address the endogeneity problem as a result of selection bias. We will discuss them in detail in the following sub-sections.

7.5.1 Reverse causality: Lagged pre-determined variables

It is arguable that the relation between the biases and stock market development are driven by some unobserved country characteristics which could influence the biases (i.e. home and foreign bias) and stock market development; for instance, investors and mutual funds, preferring to hold stocks of countries with developed stock market. We, thereby, conduct a series of robustness tests to examine whether our main result is sensitive to endogeneity problems, by including lagged pre-determined values of home and foreign bias as instrumental variables. It is quite conceivable that the level of a country's stock market development could influence foreign investors to increase their equity allocation. This will subsequently lead to a higher foreign bias. The result of this will render our main coefficient

estimates to suffer from endogeneity problems as a result of reverse causality. Under the international capital asset pricing model risk sharing hypothesis, we expect an improvement in stock market development when a country attracts sufficient foreign investment. We use lagged values to address our concern of reverse causality.

In Table 7-7, we replicate our main regression in Table 7-4 by using lagged value (*CPIS_HB_1*) to provide robustness. The coefficients reported in Table 7-7 are -0.250 (t-statistics=-10.55), -0.545 (t-statistics=-16.43), -0.275 (t-statistics=-9.17), and 0.110 (t-statistics=6.77) for *MGDP*, *TRGDP*, *TURN*, and *TRCOST* respectively. This suggests that regardless of using lagged values, our main result is similar to those reported in Table 7-4.

.....Insert Table 7-7 about here.....

As reported in Table 7-8, introducing lagged values does not materially affect our results reported in Table 7-5. *CPIS_FB_1* exhibits significantly positive association with three stock market development measures. As expected, *CPIS_FB_1* increases *TRCOST*. As reported in Model 4, the significance level of *CPIS_FB* marginally improves in Table 7-8 as compared with Model 4 of Table 7-5. This provides strong support that our main result does not suffer from reverse causality.

.....Insert Table 7-8 about here.....

The coefficient estimates of lagged values of GF_FB_1 reported in Table 7-9, as in line with our expectations, are positively related to stock market development. The coefficient estimate of GF_FB_1 is negative and the statistical significance is marginally better than those reported in Model 4 of Table 7-6. The result provides compelling support to our main analysis. This suggests that countries favoured by foreign investors, experience improved stock market development, after we have addressed the concern of reverse causality using lagged values.

.....Insert Table 7-9 about here.....

7.5.2 Heckman selection bias

In this section, we re-estimate Tables 7-4 to 7-6 using the Heckman two-stage selection model, as we consider the possibility of our main result being influenced by selection bias. One might argue that the coefficient estimates in our main analysis are driven by endogeneity cause by selection bias. Ideally, we would have preferred to use more countries, but due to unavailability of data, especially for emerging and frontier markets, we limited our sample to 44 countries. This may give rise to an endogeneity problem caused by selection bias, and thereby, may invalidate the statistical inferences.

The overall findings in Table 7-10 confirm our main analysis of *CPIS_HB* on stock market development. Even after addressing endogeneity using Heckman selection, the magnitude and degree of statistical significance marginally improves. The results reported in Table 7-10, further show no material change in inferences of our main findings. This implies that strong preferences for domestic equities are exhibited by investors in international markets, despite the well-documented gains from international diversification inhibiting stock market development.

.....Insert Table 7-10 about here.....

Table 7-11 replicates the report in the main analysis in Table 7-5. The coefficient and the statistical significance of *CPIS_FB* improve as compared to Table 7-5. They are statistical significant at the 1% level.

.....Insert Table 7-11 about here.....

In Table 7-12, the statistical significance of GF_FB is marginally higher, compared to the main analysis in Table 7-6. There is an increase in the coefficient estimates compared to the main estimates using OLS in Table 6-6. This further corroborates our main evidence that foreign investors, by tilting their investment towards a country, improve stock market development.

.....Insert Table 7-12 about here.....

7.6 Chapter summary

This chapter reports the results of the empirical findings on the impact of sub-optimal portfolio allocation on stock market development. There is a body of research that shows the importance of stock market development towards economic growth and development. Finance theory demonstrates that financial liberalisation, via foreign investors, will improve stock market development. A review of the existing literature indicates that there are limited studies examining factors that determine stock market development.

In this study, we calculate home and foreign bias by employing two sets of a comprehensive data from 44 countries from 2001 to 2010, to evaluate their implication on stock market development. We report two main findings. First, we show that countries where home bias is more pronounced, experience less stock market development as a result of international under-diversification by domestic investors and a reduction of risk sharing between domestic and foreign investors. Further explanation for the result is that home bias suggests that domestic institutional investors will engage in buy-and-hold activities which will adversely affect stock market liquidity and development. Second, we show that foreign bias improves stock market development. This is explained by the fact that the country will be integrated with the global stock market, which will enhance global risk sharing between domestic and foreign investors, and therefore, reduce cost of capital and increase stock market valuation. Additionally, foreign investors will demand improved governance in the host country, which will boost the participation of domestic investors in the stock market.

The study offers several policy implications for policy makers in emerging countries. First, the negative effect of home bias on stock market development, suggests that policy makers and governments from countries such as Bulgaria, Romania, Hungary, Peru, Czech Republic, Philippines, Egypt, Argentina, Poland, and Indonesia where home bias is not only prevalent, but also persistent, could provide incentives to the domestic investors to diversify internationally to reduce risk. Second, our empirical results also show that foreign bias or countries that are favoured by foreign investors, experience an enhanced stock market. The policy implication of this study is that, countries, particularly emerging markets contemplating to develop the depth and breadth of their local stock markets, should initiate effective policy measures by encouraging international diversification of their domestic investors. For instance, policy makers should implement policies that can attract foreign

investors by improving good governance and macroeconomic fundamentals to attract foreign investors.

Developed Cou	eveloped Countries				Emerging Countries				
-	MGDP	TRGDP	TURN	TRCOST		MGDP	TRGDP	TURN	TRCOST
Country	(% of	(% of	(% of	(Basis	Country	(% of	(% of	(% of	(Basis
	GDP)	GDP)	MKTCap)	points)		GDP)	GDP)	MKTCap	points)
Australia	119.16	94.35	84.22	31.31	Argentina	38.67	3.16	10.41	67.98
Austria	28.95	13.56	44.70	30.47	Brazil	54.96	25.61	48.16	43.06
Belgium	65.83	28.98	44.75	28.16	Bulgaria	17.52	3.17	18.61	60.21
Canada	114.47	82.78	75.42	30.28	Chile	107.1	15.69	15.36	NA
Denmark	63.31	50.19	81.36	32.04	China	69.07	86.97	122.53	46.50
Finland	98.60	124.71	119.26	37.72	Czech Rep	25.33	15.13	60.97	56.37
France	80.67	81.52	101.21	24.74	Egypt	55.67	21.88	36.26	68.15
Germany	45.71	65.40	141.75	25.60	Hungary	24.53	18.61	77.26	51.00
Greece	51.91	24.77	49.06	54.34	India	68.12	62.30	115.82	59.06
Hong Kong	421.17	349.97	77.92	39.22	Indonesia	30.03	14.52	53.16	65.30
Ireland	46.77	25.00	52.33	31.24	Korea	73.81	145.02	227.27	55.05
Israel	84.72	47.91	61.53	37.36	Malaysia	137.03	40.62	31.37	51.21
Italy	37.54	50.46	130.20	29.15	Mexico	28.33	7.46	27.84	35.71
Japan	77.97	89.44	111.68	19.38	Peru	47.15	2.95	6.97	71.24
Netherland	91.33	139.12	147.37	28.45	Philippines	48.26	8.42	18.51	88.02
New Zealand	36.38	15.63	45.63	34.58	Poland	28.56	10.55	37.55	NA
Norway	55.67	62.57	114.81	30.21	Romania	16.31	1.92	12.97	73.12
Portugal	39.39	25.67	63.45	31.83	Russia	61.59	33.32	58.94	NA
Spain	86.70	136.66	164.33	46.82	South Africa	215.23	97.61	48.01	68.54
Sweden	104.08	124.5	121.80	28.60	Thailand	62.06	50.92	94.00	53.14
Switzerland	229.24	225.24	99.90	27.10	Turkey	28.94	40.20	152.80	51.10
UK	128.47	178.77	142.71	50.02					
US	124.09	255.15	212.32	21.73					

 Table 7-1: Summary statistics of the dependent variables

Panel B: Averages of the developed and emerging countries.

Country group	MGDP	TRGDP	TURN	TRCOST
	(% of GDP)	(% of GDP)	(% of MKTCap)	(Basis points)
Developed	97.05	99.67	99.47	31.51
Emerging	58.97	33.62	60.70	54.68

Panel C: Averages of the top and bottom 10 countries.

Country group	MGDP	TRGDP	TURN	TRCOST
	(% of GDP	(% of GDP)	(% of MKTCap	(Basis points)
Top 10	170.00	177.68	156.31	30.70
Bottom 10	26.49	8.08	21.59	76.01

Note: The dependent variables are the stock market development measures. MGDP is market capitalisation as a percentage of GDP; TRGDP is stock value traded as a percentage of GDP; TURN is value of stock traded as a percentage of stock market capitalization; TRCOST is transaction cost and is an inverse of turnover ratio.⁵⁶

⁵⁶ As in chapter 6, we use *CPIS_HB*, *CPIS_FB* and *GF_FB* as sub-optimal portfolio allocation. Summary analyses of these variables are provided in chapter 6.2.

Panel A: Developed Countries									
Country	Flow (% of GDP)	Sav (log of USD millions)	PCred (log of USD millions)	Int (%)	GDPPC (in USD)	Legal_O (0-1)	RiskRisk (0-50)	Cor (0-6)	
Australia	0.047	21.4	4.85	2.13	34705	1	36.09	4.667	
Austria	0.028	25.4	4.87	2.74	37845	0	38.43	4.842	
Belgium	0.191	23.4	4.72	2.67	36323	0	27.78	3.817	
Canada	0.006	21.8	5.29	1.83	35335	1	29.50	4.854	
Denmark	0.025	24.9	5.22	1.50	47736	0	41.92	5.267	
Finland	0.033	23.8	4.42	2.13	38092	0	37.21	5.550	
France	0.003	24.8	4.75	2.17	34014	0	30.69	3.238	
Germany	0.020	22.0	4.91	1.51	34333	0	26.22	4.325	
Greece	0.115	17.5	4.76	7.54	22032	0	32.76	2.775	
Hong Kong	0.011	24.5	5.02	3.95	27859	1	41.38	3.817	
Ireland	0.031	22.8	5.11	2.14	46103	1	35.59	3.288	
Israel	0.102	25.8	4.43	5.34	21799	1	31.27	3.075	
Italy	0.003	24.3	4.83	3.56	30279	0	31.70	2.467	
Japan	0.002	24.9	5.75	3.31	35857	0	43.47	3.258	
Netherland	0.046	21.4	5.22	1.32	39882	0	29.08	5.083	
New Zealand	0.014	23.4	4.90	1.75	25219	1	26.50	5.233	
Norway	0.047	17.8	4.39	1.74	66658	0	44.74	5.000	
Portugal	0.033	26.4	5.11	5.42	18380	0	34.60	3.600	
Spain	0.041	27.0	5.12	5.64	25992	0	36.77	3.858	
Sweden	0.042	21.2	4.80	2.74	41035	0	28.42	5.192	
Switzerland	0.038	17.0	5.17	2.15	54237	0	45.24	4.163	
United Kingdom	0.061	27.9	5.20	1.99	35933	1	24.27	4.171	
United States	0.016	29.9	5.41	1.12	42341	1	30.56	4.033	

 Table 7-2: Summary statistics of control variables

Country	Flow	Sav	PCred	Int	GDPPC	Legal_O	RiskRisk	Cor
	(% of GDP)	(of USD millions)	(log of USD millions)	(%)	(in USD)	(0-1)	(0-50)	(0-6)
Argentina	0.026	26.6	3.56	7.37	5918	0	31.15	2.608
Brazil	0.030	25.2	4.42	40.50	5834	0	32.63	2.313
Bulgaria	0.157	22.7	3.77	4.68	4273	0	32.30	2.500
Chile	0.084	19.5	4.34	4.20	8116	0	25.74	3.496
China	0.004	20.0	4.93	1.43	2299	0	46.50	1.854
Czech Rep	0.114	23.5	3.90	4.71	13843	0	31.03	3.146
Egypt	0.009	24.7	4.42	3.59	1639	0	33.46	2.246
Hungary	0.080	22.0	4.23	4.48	10655	0	35.64	3.467
India	0.004	21.6	4.13	4.83	830	1	37.38	2.021
Indonesia	0.059	25.3	3.76	5.60	1594	1	29.54	2.088
Korea	0.004	21.1	4.56	3.48	16657	1	34.19	2.471
Malaysia	0.041	21.8	4.86	2.80	6036	1	36.95	2.754
Mexico	0.028	22.2	3.61	4.24	7959	0	38.97	2.142
Peru	0.064	24.4	2.94	18.20	3319	0	31.58	2.517
Philippines	0.013	23.1	3.92	4.27	1428	1	35.77	2.033
Poland	0.046	22.5	3.91	8.31	8802	0	36.19	2.688
Romania	0.068	28.4	3.34	5.13	5306	0	35.21	2.238
Russia	0.047	18.7	3.38	5.27	6375	0	43.92	1.725
South Africa	0.012	25.1	5.19	4.20	4868	1	25.91	2.671
Thailand	0.043	20.9	4.87	3.30	3021	1	33.74	1.742
Turkey	0.027	23.9	4.01	6.31	6991	0	32.02	2 392

Panel B: Emerging Countries

Flow is the log of net foreign direct investment scale by GDP; *Sav* is gross domestic savings as a percentage of GDP; *PCred* is the domestic credit to the private financial sector, scaled by GDP; *Int* is annual interest rate; *GDPPC* is the GDP Per Capita; *Legal_O* is a dummy variable that takes a value of 1 if a common law country or 0 if otherwise; *FinRisk* is the financial risk of a country; *Cor* is the corruption level prevailing in a country.

	1	2	3	4	5	6	7	8	9	10	11	12	13	14	15	16	17	18 1	19	20	21
MGDP(1)	1																				
TRGDP (2)	0.76*	1																			
TURN (3)	0.33*	0.85*	1																		
TRCOST (4)	-0.24	-0.51*	-0.51*	1																	
CPIS_HB (5)	-0.61*	-0.77*	-0.61*	0.65*	1																
CPIS_FB (6)	0.24*	0.27*	0.17*	-0.55*	-0.39*	1															
GF_FB (7)	0.19*	0.40*	0.42*	-0.46*	-0.46*	0.46*	1														
Flow (8)	0.06*	0.11*	0.22*	-0.16*	-0.18*	0.06*	0.07*	1													
Sav (9)	0.09*	0.06*	0.03	-0.03	0.02	0.03*	0.03*	0.06	1												
PCred (10)	0.37*	0.40*	0.29*	0.05*	-0.43*	0.04*	0.12*	0.06	-0.07	1											
Int (11)	-0.07*	-0.18*	-0.19*	0.10*	0.12*	-0.16*	-0.05*	0.02	0.05*	-0.10*	1										
Exch (12)	-0.16*	-0.01	-0.06*	0.07*	0.02	-0.10*	-0.13*	-0.02	-0.03	-0.02	0.19*	1									
Infl (13)	-0.14*	-0.22*	-0.18*	0.23*	0.27*	-0.30*	-0.29*	-0.04*	0.04*	-0.14*	0.06*	0.19*	1								
Retn_1 (14)	0.27*	0.14*	0.04	-0.12*	0.24*	-0.26*	-0.17*	-0.04*	0.03	-0.08	0.15*	-0.32*	0.20*	1							
LSMI (15)	0.12*	0.06	0.02	-0.10*	0.15*	0.28*	0.05*	-0.18*	-0.10*	0.18*	-0.24*	-0.02	-0.08*	-0.02	1						
GDPPC (16)	0.05*	0.04*	0.06*	-0.44*	0.17*	-0.46*	-0.24*	-0.08*	-0.02	0.04*	-0.02	0.25*	0.13*	-0.05*	-0.02*	1					
MKTCap (17)	0.30*	0.35*	0.29*	-0.06*	-0.25*	-0.12*	0.02	0.07*	-0.02	0.44*	-0.02	0.04*	-0.03	-0.04	-0.17*	-0.02	1				
R_Law (18)	0.36*	0.52*	0.46*	-0.56*	-0.47*	0.78*	0.47*	0.12*	-0.03	0.1	-0.21*	-0.1	-0.36*	-0.25	0.25*	-0.37	0.02	1			
Legal_O (19)	0.35*	0.28*	0.13*	-0.24*	-0.18*	-0.07*	0.10*	-0.23*	0.05*	0.36*	-0.12*	0.02	-0.02	-0.07*	0.12*	0.12*	0.34*	0.25* 1	l		
FinRisk (20)	-0.06*	-0.03*	-0.02	0.10*	-0.05*	-0.09*	0.06*	-0.04*	-0.04*	0.02*	-0.08*	0.03	-0.09*	-0.08*	0.04*	0.09*	-0.02	-0.14* -	0.14*	1	
Cor (21)	-0.4	-0.26	-0.13	0.31	0.29	-0.29	-0.22	-0.03	-0.02	-0.13	0.11	-0.18	0.24	0.51	-0.09	-0.11	-0.05	0.16 -	0.16	-0.09	1

Table 3: Pearson's pairwise correlation coefficients between the dependent and independent variables

Note: The variables labelled 1- 4 are the stock market development measures and 5-7 are the sub-optimal international portfolio allocation bias measures. They are described in Table 7-1. The other variables include *Flow* as the log of net foreign direct investment scale by GDP; *Sav* is gross domestic savings as a percentage of GDP; *PCred* is the domestic credit to the private financial sector, scaled by GDP; *Int* is annual interest rate; *Exch* is the three year moving average covariance of the monthly stock market index return with the monthly depreciation of the domestic currency with respect to the dollar; *Infl* is the one year lagged rate of inflation based on the consumer price index; *Retn_1* is the average *MSCI* monthly index return over the past year; *LSMI* is the log stock market integration measured as the ratio of a country's annual exports plus imports divided by *GDP*; *GDPPC* is the GDP Per Capita; *MKTCap* is the log of country market capitalisation; *R_Law* represents the law and order index of a country; *Legal_O* is a dummy variable that takes a value of 1 if a common law country or 0 if otherwise; *FinRisk* is the financial risk of a country; *Cor* is the corruption level prevailing in a country. Statistical significance is reported against 10% (*), 5% (**) and 1% (***) significance levels.

Table 7-4: The relation between equity home bias and four measures of stock market development

This table shows regression results of the following model:

	Model (1)	Model (2)	Model (3)	Model (4)
	MGDP	TRGDP	TURN	TRCOST
CPIS_HB	-0.276***	-0.662***	-0.380***	0.145***
	(-11.11)	(-18.38)	(-10.58)	(7.58)
Flow	0.450**	0.171	0.675***	-0.098
	(2.48)	(0.66)	(2.88)	(-0.74)
Sav	0.634*	0.586	0.101	-0.156
	(1.88)	(1.29)	(0.24)	(-0.68)
PCred	0.787	0.649**	0.574**	-0.136
	(0.33)	(1.98)	(2.48)	(-0.98)
Int	-0.610***	-0.546	-0.128**	0.383*
	(-2.99)	(-1.06)	(-2.43)	(1.68)
Exch	-0.270***	-0.221	-0.105	0.126
	(-2.96)	(-1.57)	(-0.78)	(0.18)
Infl	-0.118	-0.843	-1.045	0.166
	(-0.23)	(-0.83)	(-0.89)	(0.30)
Retn_1	0.288	0.186***	0.232***	-0.402
	(0.79)	(2.89)	(3.26)	(-1.51)
LSMI	0.341***	0.599***	0.262**	-0.144
	(4.73)	(6.04)	(2.58)	(-0.24)
GDPPC	0.330*	0.478**	0.270	-0.324*
	(1.85)	(2.19)	(1.03)	(-1.70)
МКТСар	0.651***	0.108***	0.460**	-0.399
-	(4.43)	(4.96)	(2.33)	(-0.31)
R_Law	0.430**	0.179***	0.133***	-0.523***
	(2.03)	(7.15)	(5.08)	(-4.10)
Legal_O	0.246***	0.194	0.051	-0.183***
-	(2.71)	(1.64)	(0.51)	(-2.65)
FinRisk	-0.119	-0.281	-0.170	0.156
	(-0.35)	(-0.67)	(-0.37)	(0.70)
Cor	-0.150	-0.328	-0.331	0.220**
	(-1.07)	(-0.16)	(-0.15)	(1.98)
Constant	4.381***	3.565***	3.596***	2.412***
	(6.14)	(3.69)	(3.78)	(5.75)
Number of Observations	440	440	440	410
Adj. R-squared	0.61	0.71	0.48	0.61
Country effects	Yes	Yes	Yes	Yes
Year fixed effects	Yes	Yes	Yes	Yes

 $SMD_{lt} = \alpha + \beta_1 CPIS_HB + \beta_2 Controls + Country and Year effects + \varepsilon_{it}$

Note: The dependent variables are the stock market development (SMD_{lt}) measures of country *l* at time *t*. *MGDP* is market capitalisation as a percentage of GDP; *TRGDP* is stock value traded as a percentage of GDP; *TURN* is value of stock traded as a percentage of stock market capitalisation; *TRCOST* is transaction cost and is an inverse of turnover ratio; *CPIS_HB* is equity home bias and is calculated as the log share of domestic investors in their country's stock market capitalisation relative to the country's world market capitalisation weight; *Flow* is the log of net foreign direct investment scale by GDP; *Sav* is gross domestic savings as a percentage of GDP; *PCred* is the domestic credit to the private financial sector, scaled by GDP; *Int* is interest rate; *Exch* is the three year moving average covariance of the monthly stock market index return with the monthly depreciation of the domestic currency with respect to the dollar; *Infl* is the one year lagged rate of inflation based on the consumer price index; *Retn_1* is the average *MSCI* monthly index return over the past year; *LSMI* is the GDP Per Capita; *MKTCap* is the log of country market capitalisation; *R_Law* represents the law and order index of a country; *Legal_O* is a dummy variable that takes a value of 1 if a common law country or 0 if otherwise; *FinRisk* is the financial risk of a country; *Cor* is the corruption level prevailing in a country. All t-statistics reported are based on Newey-West autocorrelation and heteroskedasticity-corrected standard errors. For tractable interpretation all the coefficients are reported as elasticity and the statistical significance is reported against 10% (*), 5% (**) and 1% (***) significance level.

Table 7-5: The relation between equity equity foreign bias and four measures of stock market development

This table shows regression results of the following model:

	Model (1)	Model (2)	Model (3)	Model (4)
	MGDP	TRGDP	TURN	TRCOST
CPIS_FB	0.093***	0.205***	0.096***	-0.087***
	(3.96)	(4.63)	(2.85)	(-6.02)
Flow	0.136	0.092***	0.111***	-0.161
	(0.63)	(2.69)	(4.38)	(-1.07)
Sav	0.829**	0.102*	0.313	-0.729
	(2.17)	(1.65)	(0.66)	(-0.27)
PCred	0.745***	0.134***	0.578**	-0.228*
	(3.05)	(3.03)	(2.05)	(-1.95)
Int	-0.429	-0.176**	-0.204***	0.599**
	(-1.23)	(-2.07)	(-2.93)	(2.58)
Exch	-0.247**	-0.149	-0.446	0.579
	(-2.16)	(-0.65)	(-0.25)	(0.74)
Infl	-0.326	-0.199	-0.179	0.439
	(-0.53)	(-1.50)	(-1.38)	(0.85)
Retn_1	0.167	0.207*	0.235***	-0.351***
	(0.31)	(1.88)	(2.61)	(-2.76)
LSMI	0.350	0.285*	0.225**	-0.244***
	(0.33)	(1.74)	(2.22)	(-5.00)
GDPPC	0.372	0.507	0.208	-0.267
	(1.49)	(1.38)	(0.61)	(-1.49)
MKTCap	0.768***	0.135***	0.617**	-0.149
	(3.48)	(3.40)	(2.40)	(-0.13)
R_Law	0.114***	0.390***	0.284***	-0.611***
	(3.85)	(7.98)	(8.17)	(-4.95)
Legal_O	0.416***	0.595***	0.170	-0.368
	(4.45)	(3.54)	(1.26)	(-0.53)
FinRisk	-0.501	-0.113*	-0.583	0.500*
	(-1.28)	(-1.67)	(-0.99)	(1.87)
Cor	-0.296*	-0.401	-0.200	0.236**
	(-1.68)	(-1.41)	(-0.87)	(2.43)
Constant	2.069**	-2.027	0.330	3.523***
	(2.31)	(-1.49)	(0.34)	(7.85)
Number of Observations	440	440	440	410
Adj. R-squared	0.49	0.46	0.42	0.24
Country effects	Yes	Yes	Yes	Yes
Year fixed effects	Yes	Yes	Yes	Yes

 $SMD_{lt} = \alpha + \beta_1 CPIS_FB + \beta_2 Controls + Country and Year effects + \varepsilon_{it}$

Note: The dependent variables are the stock market development (SMD_{lt}) measures of country l at time t. MGDP is market capitalisation as a percentage of GDP; TRGDP is stock value traded as a percentage of GDP; TURN is value of stock traded as a percentage of stock market capitalisation; TRCOST is transaction cost and is an inverse of turnover ratio; CPIS_FB is equity foreign bias and is the log deviation of investors in country l in stockholdings for each host country k ($k \neq l$) from the rest of the world market capitalization weight of country l, calculated as the ratio of total market capitalisation of a country's domestic equity held by foreign investors divided by the country's total world market capitalisation weight. Flow is the log of net foreign direct investment scale by GDP; Sav is gross domestic savings as a percentage of GDP; PCred is the domestic credit to the private financial sector, scaled by GDP; Int is annual interest rate; Exch is the three year moving average covariance of the monthly stock market index return with the monthly depreciation of the domestic currency with respect to the dollar; Infl is the one year lagged rate of inflation based on the consumer price index; Retn_l is the average MSCI monthly index return over the past year; LSMI is the log stock market integration measured as the ratio of a country's annual exports plus imports divided by GDP; GDPPC is the GDP Per Capita; MKTCap is the log of country market capitalisation; R_Law represents the law and order index of a country;; Legal_O is a dummy variable that takes a value of 1 if a common law country or 0 if otherwise; FinRisk is the financial risk of a country; Cor is the corruption level prevailing in a country. All t-statistics reported are based on Newey-West autocorrelation and heteroskedasticity-corrected standard errors. For tractable interpretation all the coefficients are reported as elasticity and the statistical significance is reported against 10% (*), 5% (**) and 1% (***) significance level.

Table 7-6: The relation between equity country bias and four measures of stock market development

This table shows regression results of the following model:

	Model (1)	Model (2)	Model (3)	Model (4)
	MGDP	TRGDP	TURN	TRCOST
GF_FB	0.063*	0.394***	0.323***	-0.112***
	(1.84)	(6.44)	(6.08)	(-3.97)
Flow	0.166	0.075**	0.096***	-0.119
	(0.75)	(2.15)	(4.03)	(-0.71)
Sav	0.697*	0.858	0.305	-0.499
	(1.73)	(1.37)	(0.69)	(-0.19)
PCred	0.779***	0.133***	0.531**	-0.180
	(3.31)	(3.29)	(2.23)	(-1.28)
Int	-0.136	-0.238***	-0.233***	0.092***
	(-0.37)	(-2.93)	(-3.66)	(3.61)
Exch	-0.194*	-0.155	-0.114	0.158
	(-1.71)	(-0.77)	(-0.77)	(0.20)
Infl	-0.515	-1.565	-1.144	0.358
	(-0.82)	(-1.22)	(-1.02)	(0.57)
Retn_1	0.482	0.165*	0.230***	-0.524
	(0.93)	(1.77)	(3.17)	(-1.63)
LSMI	0.623	0.384	0.092	-0.156***
	(0.59)	(0.26)	(1.03)	(-2.64)
GDPPC	0.087	0.145	0.178	-0.390**
	(0.41)	(0.51)	(0.72)	(-2.26)
MKTCap	0.784***	0.150***	0.746***	-0.235
	(3.65)	(4.36)	(3.60)	(-0.19)
R_Law	0.135***	0.317***	0.179***	-0.714***
	(6.45)	(8.89)	(6.72)	(-6.33)
Legal_O	0.355***	0.367**	0.121	0.150**
	(3.65)	(2.33)	(0.11)	(2.19)
FinRisk	-0.101	-0.310	-0.116	0.486
	(-0.26)	(-0.06)	(-0.26)	(0.21)
Cor	-0.358**	-0.405	-0.131	0.327***
	(-2.01)	(-1.59)	(-0.67)	(3.23)
Constant	1.895**	-2.132	0.430	3.306***
	(2.07)	(-1.59)	(0.49)	(6.90)
Number of Observations	440	440	440	410
Adj. R-squared	0.43	0.46	0.36	0.54
Country effects	Yes	Yes	Yes	Yes
Year fixed effects	Yes	Yes	Yes	Yes

 $SMD_{lt} = \alpha + \beta_1 GF_FB + \beta_2 Controls + Country and Year effects + \varepsilon_{it}$

Note: The dependent variables are the stock market development (SMD_{lt}) measures of country *l* at time *t*. *MGDP* is market capitalisation as a percentage of GDP; *TRGDP* is stock value traded as a percentage of GDP; *TURN* is value of stock traded as a percentage of stock market capitalisation; *TRCOST* is transaction cost and is an inverse of turnover ratio; *GF_FB* is equity country bias using global fund; *Flow* is the log of net foreign direct investment scale by GDP; *Sav* is gross domestic savings as a percentage of GDP; *PCred* is the domestic credit to the private financial sector, scaled by GDP; *Int* is interest rate; *Exch* is the three year moving average covariance of the monthly stock market index return with the monthly depreciation of the domestic currency with respect to the dollar; *Infl* is the one year lagged rate of inflation based on the consumer price index; *Retn_1* is the average *MSCI* monthly index return over the past year; *LSMI* is the GDP Per Capita; *MKTCap* is the log of country market capitalisation; *R_Law* represents the law and order index of a country; *Legal_O* is a dummy variable that takes a value of 1 if a common law country or 0 if otherwise; *FinRisk* is the financial risk of a country; *Cor* is the corruption level prevailing in a country. All t-statistics reported are based on Newey-West autocorrelation and heteroskedasticity-corrected standard errors. For tractable interpretation all the coefficients are reported as elasticity and the statistical significance is reported against 10% (*), 5% (**) and 1% (***) significance level.

<u></u>	<u>car and equity in</u>			
	Model (1)	Model (2)	Model (3)	Model (4)
	MGDP	TRGDP	TURN	TRCOST
CPIS_HB_1	-0.250***	-0.545***	-0.275***	0.110***
	(-10.55)	(-16.43)	(-9.17)	(6.77)
Flow	0.467**	0.171	0.665***	-0.724
	(2.51)	(0.66)	(2.84)	(-0.61)
Sav	0.654*	0.890*	0.328	-0.288
	(1.80)	(1.75)	(0.71)	(-1.25)
PCred	0.175	0.472**	0.299	-0.519
	(1.17)	(2.26)	(1.59)	(-0.51)
Int	-0.121***	-0.188	-0.969*	0.129
	(-2.85)	(-0.03)	(-1.81)	(0.51)
Exch	-0.320***	-0.461***	-0.257**	0.121
	(-3.13)	(-3.23)	(-1.99)	(0.19)
Infl	-0.238	-0.308	-0.192	0.086
	(-0.42)	(-0.39)	(-0.03)	(0.25)
Retn_1	0.130	0.197***	0.265***	-0.274
	(0.03)	(3.33)	(4.96)	(-0.92)
LSMI	0.271***	0.342***	0.0691	0.0739
	(3.99)	(3.60)	(0.81)	(1.45)
GDPPC	0.481***	0.932***	0.457***	0.200***
	(4.03)	(5.60)	(3.04)	(2.70)
MKTCap	0.425***	0.879***	0.525***	0.573
	(3.87)	(5.74)	(3.79)	(0.76)
R_law	0.414**	0.180***	0.125***	-0.523***
	(2.41)	(7.52)	(5.75)	(-5.30)
Legal_O	0.227***	0.261***	0.997	-0.165***
	(3.18)	(2.62)	(1.11)	(-3.77)
FinRisk	-0.158	-0.713*	-0.608	0.352*
	(-0.52)	(-1.68)	(-1.58)	(1.95)
Cor	-0.330***	-0.562	-0.207	0.158*
	(-2.62)	(-0.56)	(-1.31)	(1.86)
Constant	4.649***	2.075**	1.960**	3.001***
	(7.48)	(2.39)	(2.50)	(7.68)
Number of Observations	396	396	396	369
Adj. R-squared	0.59	0.69	0.45	0.61
Country effects	Yes	Yes	Yes	Yes
Vear fixed effects	Ves	Ves	Ves	Ves

Table 7-7: Using one-year lag equity home bias

Note: The dependent variables are the stock market development measures. *MGDP* is market capitalisation as a percentage of GDP; *TRGDP* is stock value traded as a percentage of GDP; *TURN* is value of stock traded as a percentage of stock market capitalisation; *TRCOST* is transaction cost and is an inverse of turnover ratio; *CPIS_HB_1* is one year lag equity home bias and is calculated as the log share of domestic investors in their country's stock market capitalisation relative to the country's world market capitalisation weight; *Flow* is the log of net foreign direct investment scale by GDP; *Sav* is gross domestic savings as a percentage of GDP; *PCred* is the domestic credit to the private financial sector, scaled by GDP; *Int* is annual interest rate; *Exch* is the three year moving average covariance of the monthly stock market index return with the monthly depreciation of the domestic currency with respect to the dollar; *Infl* is the one year lagged rate of inflation based on the consumer price index; *Retn_1* is the average *MSCI* monthly index return over the past year; *LSMI* is the log stock market integration measured as the ratio of a country's annual exports plus imports divided by *GDP*; *GDPPC* is the GDP Per Capita; *MKTCap* is the log of country market capitalisation; *R_Law* represents the law and order index of a country; *Legal_O* is a dummy variable that takes a value of 1 if a common law country or 0 if otherwise; *FinRisk* is the financial risk of a country; *Cor* is the corruption level prevailing in a country. All t-statistics reported are based on Newey-West autocorrelation and heteroskedasticity-corrected standard errors. For tractable interpretation all the coefficients are reported as elasticity and the statistical significance is reported against 10% (*), 5% (**) and 1% (***) significance level.

Table 7-0. Using one	-year rag equit	y ioi cigii Dias		
	Model (1)	Model (2)	Model (3)	Model (4)
	MGDP	TRGDP	TURN	TRCOST
CPIS_FB_1	0.085***	0.195***	0.091***	-0.082***
	(4.61)	(5.72)	(3.53)	(-6.78)
Flow	0.151	0.099***	0.113***	-0.171
	(0.74)	(2.63)	(3.96)	(-1.30)
Sav	0.086**	0.128*	0.493	-0.183
	(2.10)	(1.71)	(0.87)	(-0.71)
PCred	0.070***	0.129***	0.555***	-0.263**
	(4.67)	(4.68)	(2.66)	(-2.40)
Int	-0.535	-0.160*	-0.184***	0.554**
	(-1.14)	(-1.84)	(-2.81)	(1.99)
Exch	-0.251**	-0.277	-0.145	0.416
	(-2.20)	(-1.32)	(-0.91)	(0.58)
Infl	-0.298	-1.628	-1.291	0.524
	(-0.48)	(-1.42)	(-1.50)	(1.38)
Retn_1	0.265	0.233***	0.274***	-0.629**
	(0.55)	(2.64)	(4.06)	(-2.43)
LSMI	0.257	0.244**	0.193**	-0.236***
	(0.39)	(1.99)	(2.09)	(-4.89)
GDPPC	0.370***	0.345	0.488	-0.291***
	(2.70)	(1.37)	(0.03)	(-3.48)
MKTCap	0.762***	0.130***	0.631***	-0.482
_	(5.59)	(5.15)	(3.33)	(-0.50)
R_law	0.244***	0.333***	0.108**	0.666***
	(6.52)	(4.82)	(2.06)	(3.18)
Legal_O	0.384***	0.596***	0.263**	-0.388
	(4.85)	(4.08)	(2.39)	(-0.78)
FinRisk	-0.291	-0.119*	-0.865*	0.554***
	(-0.84)	(-1.86)	(-1.80)	(2.66)
Cor	-0.512***	-0.617**	-0.283	0.262***
	(-3.73)	(-2.44)	(-1.48)	(2.75)
Constant	2.197***	-2.212*	0.127	3.697***
	(3.32)	(-1.82)	(0.14)	(8.38)
Number of Observations	396	396	396	369
Adj. R-squared	0.47	0.42	0.24	0.56
Country effects	Yes	Yes	Yes	Yes
Vear fixed effects	Ves	Ves	Ves	Ves

Table 7-8: Using one-year lag equity foreign bias

Note: The dependent variables are the stock market development measures. MGDP is market capitalisation as a percentage of GDP; TRGDP is stock value traded as a percentage of GDP; TURN is value of stock traded as a percentage of stock market capitalisation; TRCOST is transaction cost and is an inverse of turnover ratio; CPIS_FB_1 is one year lag equity foreign bias and is the log deviation of investors in country l in stockholdings for each host country k $(k \neq l)$ from the rest of the world market capitalisation weight of country l calculated as the ratio of total market capitalisation of a country's domestic equity held by foreign investors divided by the country's total world market capitalisation weight. Flow is the log of net foreign direct investment scale by GDP; Sav is gross domestic savings as a percentage of GDP; PCred is the domestic credit to the private financial sector, scaled by GDP; Int is annual interest rate; Exch is the three year moving average covariance of the monthly stock market index return with the monthly depreciation of the domestic currency with respect to the dollar; Infl is the one year lagged rate of inflation based on the consumer price index; Retn_l is the average MSCI monthly index return over the past year; LSMI is the log stock market integration measured as the ratio of a country's annual exports plus imports divided by GDP; GDPPC is the GDP Per Capita; MKTCap is the log of country market capitalisation; R_{Law} represents the law and order index of a country; $Legal_O$ is a dummy variable that takes a value of 1 if a common law country or 0 if otherwise; FinRisk is the financial risk of a country; Cor is the corruption level prevailing in a country. All t-statistics reported are based on Newey-West autocorrelation and heteroskedasticity-corrected standard errors. For tractable interpretation all the coefficients are reported as elasticity and the statistical significance is reported against 10% (*), 5% (**) and 1% (***) significance level.

Table 7-7. Using one-	year rag equity (Junity blas		
	Model (1)	Model (2)	Model (3)	Model (4)
	MGDP	TRGDP	TURN	TRCOST
GF_FB_1	0.061**	0.366***	0.290***	-0.097***
	(2.03)	(6.97)	(7.61)	(-4.91)
Flow	0.178	0.084**	0.101***	-0.161
	(0.85)	(2.27)	(3.76)	(-1.17)
Sav	0.076*	0.131*	0.640	-0.175
	(1.82)	(1.79)	(1.20)	(-0.65)
PCred	0.072***	0.125***	0.483**	-0.178
	(4.71)	(4.62)	(2.46)	(-1.57)
Int	-0.344	-0.212**	-0.213***	0.779***
	(-0.72)	(-2.50)	(-3.46)	(2.70)
Exch	-0.202*	-0.184	-0.110	0.334
	(-1.74)	(-0.90)	(-0.75)	(0.45)
Infl	-0.562	-1.686	-1.032	0.728*
	(-0.89)	(-1.51)	(-1.27)	(1.85)
Retn_1	0.330	0.161*	0.240***	-0.392
	(0.15)	(1.89)	(3.88)	(-1.11)
LSMI	0.535	-0.506	0.097	0.163***
	(0.81)	(-0.43)	(1.15)	(3.25)
GDPPC	0.136	0.110	0.424	-0.394***
	(1.06)	(0.49)	(0.26)	(-4.71)
МКТСар	0.761***	0.141***	0.747***	-0.779
	(5.42)	(5.71)	(4.17)	(-0.08)
R_law	0.269***	0.406***	0.151***	-0.693***
	(7.03)	(6.03)	(3.09)	(-3.16)
Legal_O	0.337***	0.426***	0.151	-0.135***
	(4.15)	(2.97)	(1.46)	(-2.65)
FinRisk	-0.559	-0.198	-0.299	0.193
	(-0.16)	(-0.32)	(-0.67)	(0.91)
Cor	-0.581***	-0.543**	-0.127	0.338***
	(-4.12)	(-2.19)	(-0.71)	(3.46)
Constant	2.083***	-2.193*	0.282	3.470***
	(3.08)	(-1.84)	(0.33)	(7.57)
Number of Observations	396	396	396	369
Adj. R-squared	0.45	0.44	0.32	0.52
Country effects	Yes	Yes	Yes	Yes
Year fixed effects	Ves	Ves	Ves	Ves

Table 7-9: Using one-year lag equity country bias

Note: The dependent variables are the stock market development measures. *MGDP* is market capitalisation as a percentage of GDP; *TRGDP* is stock value traded as a percentage of GDP; *TURN* is value of stock traded as a percentage of stock market capitalisation; *TRCOST* is transaction cost and is an inverse of turnover ratio; *GF_FB_1* is one year lag of equity country bias measured using global fund; *Flow* is the log of net foreign direct investment scale by GDP; *Sav* is gross domestic savings as a percentage of GDP; *PCred* is the domestic credit to the private financial sector, scaled by GDP; *Int* is annual interest rate; *Exch* is the three year moving average covariance of the monthly stock market index return with the monthly depreciation of the domestic currency with respect to the dollar; *Infl* is the one year lagged rate of inflation based on the consumer price index; *Retn_1* is the average *MSCI* monthly index return over the past year; *LSMI* is the log stock market integration measured as the ratio of a country's annual exports plus imports divided by *GDP*; *GDPPC* is the GDP Per Capita; *MKTCap* is the log of country market capitalisation; *R_Law* represents the law and order index of a country; *Legal_O* is a dummy variable that takes a value of 1 if a common law country or 0 if otherwise; *FinRisk* is the financial risk of a country; *Cor* is the corruption level prevailing in a country. All t-statistics reported are based on Newey-West autocorrelation and heteroskedasticity-corrected standard errors. For tractable interpretation all the coefficients are reported as elasticity and the statistical significance is reported against 10% (*), 5% (**) and 1% (***) significance level.

Heckman two-stage model	Model (1)	Model (2)	Model (3)	Model (4)	
C	MGDP	TRGDP	TURN	TRCOST	
CPIS_HB	-0.355***	-0.721***	-0.432***	0.132***	
	(-13.24)	(-21.56)	(-12.88)	(11.44)	
Flow	0.427***	0.148	0.684***	-0.145	
	(2.35)	(0.75)	(4.22)	(-0.72)	
Sav	0.638**	0.487	0.292	-0.144	
	(2.09)	(1.25)	(0.71)	(-0.37)	
PCred	0.769	0.654***	0.677***	-0.269	
	(0.63)	(3.72)	(3.66)	(-1.85)	
Int	-0.231**	-0.541	-0.185***	0.387*	
	(-2.34)	(-1.32)	(-3.73)	(1.97)	
Exch	-0.286***	-0.257	-0.211**	0.143	
	(-3.21)	(-1.40)	(-2.24)	(0.57)	
Infl	-0.122	-0.830	-0.657	0.204	
	(-0.18)	(-1.36)	(-0.72)	(0.64)	
Retn_1	0.303	0.232***	0.321***	-0.546	
	(0.91)	(3.28)	(4.79)	(-1.42)	
LSMI	0.476***	0.648***	0.388***	-0.263	
	(6.25)	(7.61)	(3.67)	(-0.42)	
GDPPC	0.435***	0.487***	0.226	-0.411***	
	(3.54)	(3.81)	(1.48)	(-4.72)	
MKTCap	0.683***	0.233***	0.563***	-0.431	
	(5.95)	(6.18)	(3.61)	(-0.78)	
R_law	0.233***	0.309*	0.524	-0.194	
	(4.61)	(1.97)	(1.24)	(-0.76)	
Legal_O	0.256***	0.241***	0.490	-0.198***	
	(3.67)	(3.45)	(0.81)	(-4.57)	
FinRisk	-0.158	-0.322	-0.496	0.162	
	(-0.64)	(-0.56)	(-1.25)	(0.93)	
Cor	-0.186*	-0.427	-0.562	0.292***	
	(-1.99)	(-0.64)	(-0.48)	(3.38)	
Lambda (inverse Mills' ratio)	0.623***	0.644***	0.563***	-0.611***	
	(26.12)	(23.88)	(18.57)	(-20.46)	
Wald Chi-square	478.24	396.75	388.49	437.86	
Constant	4.255***	4.33***	3.288***	3.862***	
	(7.48)	(4.70)	(4.98)	(6.83)	

 Table 7-10: CPIS_HB regression based on the Heckman two-stage treatment effect model

Note: This table presents the coefficients of the estimates from Heckman two-stage treatment effects models. In the first stage, we run the probit model. We include Lambda (inverse Mills' ratio) in the second stage with control variables. The dependent variables are the stock market development measures. MGDP is market capitalisation as a percentage of GDP; TRGDP is stock value traded as a percentage of GDP; TURN is value of stock traded as a percentage of stock market capitalisation; TRCOST is transaction cost and is an inverse of turnover ratio; CPIS_HB is equity home bias and is calculated as the log share of domestic investors in their country's stock market capitalisation relative to the country's world market capitalisation weight; Flow is the log of net foreign direct investment scale by GDP; Sav is gross domestic savings as a percentage of GDP; PCred is the domestic credit to the private financial sector, scaled by GDP; Int is annual interest rate; Exch is the three year moving average covariance of the monthly stock market index return with the monthly depreciation of the domestic currency with respect to the dollar; Infl is the one year lagged rate of inflation based on the consumer price index; Retn_1 is the average MSCI monthly index return over the past year; LSMI is the log stock market integration measured as the ratio of a country's annual exports plus imports divided by GDP; GDPPC is gross domestic product per capita; MKTCap is the log of country market capitalisation; R_Law represents the law and order index of a country; Legal_O is a dummy variable that takes a value of 1 if a common law country or 0 if otherwise; FinRisk is the financial risk of a country; Cor is the corruption level prevailing in a country. All t-statistics reported are in parentheses. For tractable interpretation all the coefficients are reported as elasticity and the statistical significance is reported against 10% (*), 5% (**) and 1% (***) significance level.

Heckman two-stage model	Model (1)	Model (2)	Model (3)	Model (4)
C	MGDP	TRGDP	TURN	TRCOST
CPIS_FB	0.734***	0.281***	0.183***	-0.120***
	(5.35)	(7.12)	(3.32)	(-8.56)
Flow	0.231	0.142***	0.274***	-0.242*
	(0.70)	(3.43)	(4.13)	(-1.98)
Sav	0.892**	0.133	0.467	-0.813
	(2.50)	(1.67)	(0.71)	(-0.84)
PCred	0.793***	0.281***	0.586***	-0.247***
	(5.18)	(5.27)	(3.61)	(-2.84)
Int	-0.522	-0.184***	-0.230***	0.647***
	(-0.97)	(-3.44)	(-2.64)	(2.90)
Exch	-0.332**	0.155	0.478	-0.611
	(2.27)	(0.93)	(1.22)	(-1.31)
Infl	-0.423	-0.256**	-0.224	0.534
	(-0.54)	(-2.12)	(-1.83)	(1.53)
Retn_1	0.170	0.248***	0.283***	-0.374***
	(0.39)	(2.31)	(4.68)	(-3.53)
LSMI	0.382	0.297**	0.245**	-0.366***
	(0.55)	(2.33)	(2.20)	(-5.83)
GDPPC	0.377***	0.624**	0.480	-0.253***
	(2.81)	(2.18)	(0.19)	(-3.84)
МКТСар	0.791***	0.562***	0.259***	-0.278
	(5.91)	(5.14)	(3.73)	(-0.81)
R_law	0.239***	0.418***	0.297**	-0.643***
	(7.05)	(10.84)	(11.35)	(-7.61)
Legal_O	0.427***	0.610***	0.188***	0.394
	(5.55)	(4.76)	(2.98)	(0.80)
FinRisk	-0.566	-0.172*	-0.617**	0.527***
	(-1.49)	(-1.88)	(-2.36)	(2.97)
Cor	-0.311**	-0.456*	-0.297	0.275***
	(-2.45)	(-1.86)	(-1.50)	(3.87)
Lambda (inverse Mills' ratio)	-0.348***	-0.453***	-0.387***	0.563***
	(-23.11)	(-18.16)	(-13.68)	(12.27)
Wald Chi-square	544.81	486.42	423.88	497.76
Constant	2.245***	-2.358*	0.140	3.637***
	(3.23)	(-1.61)	(0.26)	(9.31)

 Table 7-11: CPIS_FB regression based on the Heckman two-stage treatment effect

 model

Note: This table presents the coefficients of the estimates from Heckman two-stage treatment effects models. In the first stage, we run the probit model. We include Lambda (inverse Mills' ratio) in the second stage with control variables. The dependent variables are the stock market development measures. MGDP is market capitalisation as a percentage of GDP; TRGDP is stock value traded as a percentage of GDP; TURN is value of stock traded as a percentage of stock market capitalisation; TRCOST is transaction cost and is an inverse of turnover ratio; CPIS_FB is equity foreign bias and is the log deviation of investors in country l in stockholdings for each host country k $(k \neq l)$ from the rest of the world market capitalisation weight of country l, calculated as the ratio of total market capitalisation of a country's domestic equity held by foreign investors divided by the country's total world market capitalisation weight. Flow is the log of net foreign direct investment scale by GDP; Sav is gross domestic savings as a percentage of GDP; PCred is the domestic credit to the private financial sector, scaled by GDP; Int is annual interest rate; Exch is the three year moving average covariance of the monthly stock market index return with the monthly depreciation of the domestic currency with respect to the dollar; Infl is the one year lagged rate of inflation based on the consumer price index; Retn_1 is the average MSCI monthly index return over the past year; LSMI is the log stock market integration measured as the ratio of a country's annual exports plus imports divided by GDP; GDPPC is the GDP Per Capita; MKTCap is the log of country market capitalisation; R_Law represents the law and order index of a country; Legal O is a dummy variable that takes a value of 1 if a common law country or 0 if otherwise; FinRisk is the financial risk of a country; Cor is the corruption level prevailing in a country. All t-statistics reported are in parentheses. For tractable interpretation all the coefficients are reported as elasticity and the statistical significance is reported against 10% (*), 5% (**) and 1% (***) significance level.

Heckman two-stage model	Model (1)	Model (2)	Model (3)	Model (4)
	MGDP	TRGDP	TURN	TRCOST
GF_FB	0.124***	0.402***	0.396***	-0.164***
	(3.64)	(8.31)	(7.97)	(-5.55)
Flow	0.178	0.128**	0.122***	-0.161
	(0.94)	(2.64)	(5.97)	(-0.86)
Sav	0.713**	0.926	0.474	-0.540
	(2.03)	(1.44)	(1.41)	(-0.33)
PCred	0.898***	0.157***	0.544**	-0.197*
	(5.56)	(5.82)	(2.44)	(-1.82)
Int	-0.187	-0.258***	-0.245***	0.124***
	(-0.47)	(-3.37)	(-3.72)	(3.60)
Exch	0.217*	0.166	0.143	0.179
	(1.97)	(1.12)	(1.20)	(0.62)
Infl	-0.563	-1.621	-1.411	0.406
	(-0.77)	(-1.64)	(-1.23)	(0.86)
Retn_1	0.467	0.135**	0.241***	-0.567*
	(1.26)	(2.47)	(4.63)	(-1.90)
LSMI	0.645	-0.395	-0.110	0.177***
	(0.84)	(-0.53)	(-1.47)	(3.74)
GDPPC	0.113	0.162	0.273	0.405***
	(0.67)	(0.86)	(0.91)	(3.77)
МКТСар	0.821***	0.162***	0.752***	-0.326
	(5.24)	(6.11)	(4.88)	(-0.63)
R_law	0.286***	0.375***	0.189***	-0.742***
	(7.19)	(9.37)	(7.64)	(-6.88)
Legal_O	0.351***	0.386***	0.177*	-0.193***
	(4.51)	(3.67)	(1.70)	(-3.67)
FinRisk	-0.152	-0.365	-0.142	0.502
	(-0.44)	(-0.17)	(-0.83)	(0.63)
Cor	-0.367***	-0.427*	-0.156	0.341***
	(-2.89)	(-1.90)	(-0.96)	(3.68)
Lambda (inverse Mills' ratio)	-0.611***	-0.467***	-0.296***	0.329***
	(-12.46)	(-8.24)	(-11.47)	(15.22)
Wald Chi-square	363.96	438.67	476.91	432.18
Constant	2.356***	-2.347*	0.286	3.247***
	(254)	(-1.97)	(0.56)	(8 56)

Table 7-12: GF FB regression based on the Heckman two-stage treatment effect model

Note: This table presents the coefficients of the estimates from Heckman two-stage treatment effects models. In the first stage, we run the probit model. We include Lambda (inverse Mills' ratio) in the second stage with control variables. The dependent variables are the stock market development measures. MGDP is market capitalisation as a percentage of GDP; TRGDP is stock value traded as a percentage of GDP; TURN is value of stock traded as a percentage of stock market capitalisation; TRCOST is transaction cost and is an inverse of turnover ratio; GF_FB is equity country bias using global fund; Flow is the log of net foreign direct investment scale by GDP; Sav is gross domestic savings as a percentage of GDP; PCred is the domestic credit to the private financial sector, scaled by GDP; Int is annual interest rate; Exch is the three year moving average covariance of the monthly stock market index return with the monthly depreciation of the domestic currency with respect to the dollar; Infl is the one year lagged rate of inflation based on the consumer price index; Retn_l is the average MSCI monthly index return over the past year; LSMI is the log stock market integration measured as the ratio of a country's annual exports plus imports divided by GDP; GDPPC is the GDP Per Capita; MKTCap is the log of country market capitalisation; R Law represents the law and order index of a country; Legal O is a dummy variable that takes a value of 1 if a common law country or 0 if otherwise; FinRisk is the financial risk of a country; R_Law represents the law and order index of a country; Cor is the corruption level prevailing in a country. All t-statistics reported are in parentheses. For tractable interpretation all the coefficients are reported as elasticity and the statistical significance is reported against 10% (*), 5% (**) and 1% (***) significance level.

Chapter 8 Third empirical study: International portfolio investment and investor protection

Henry (2003) contends that financial globalisation may affect factor productivity of a country, by promoting better corporate governance and signalling higher quality of state governance.⁵⁷ In other words, financial globalisation, i.e. encouraging international portfolio diversification, which integrates the local with world capital markets, may have a lasting effect on the improvement of investor protection standards. However, the significant departure from the theoretical benchmarking in international equity portfolio diversification, referred to as home and foreign bias, is a ubiquitous phenomenon and is well documented in the finance literature.

A number of researchers and policy-makers have argued and provided compelling evidence that investor protection standards play a significant role in the reduction of cost of capital, stock market development, economic growth and development (see Chapter 2 Section 5.1). Studies argue that better investor protection standards lead to attracting foreign investors. Further, foreign investors avoid investing in countries with weak investor protection standards (Chan et al., 2005). Other studies provide strong argument that, countries that experience poor investor protection standards attract foreign investors from well governed countries as a result of high expected returns. Subsequently, these foreign investors from the host countries. With knowledge of this argument and the existing studies that suggest sub-optimal portfolio allocation (home and foreign bias) could have a varying impact on investor protection (see Errunza, 2001). We are therefore motivated to embark on this study to investigate the implications of home and foreign bias on investor protection standards as improving investor protection standards is an important policy priority in many emerging countries.

In past years, there was unavailability of cross-border equity allocation data which has hindered research in these areas. We therefore make use of current available data of the Coordinated Portfolio Investment Survey of the IMF and EPFR data to embark on this study.

⁵⁷ In modern finance, it is axiomatic that investor protection regulations and practices are pivotal for the welfare of corporates and countries (La Porta et, al, 1997, 1998, 2000, 2002).

Consistent with our discussion in Chapter 3 (see Section 3.4), we employ four measures of investor protection, which we obtain from the World Governance Indicators. We test the following hypotheses in this chapter:

- **H**₇ Higher home bias is related to a lower level of investor protection standards.
- H₈ Higher foreign bias is associated with higher levels of investor protection standards.
- **H**₉ A higher country global fund bias is related to higher investor protection standards.

8.1 Summary statistics

Before we start the analysis of the regression results, we provide summary statistics which offer an important overview and information of the relationship between sub-optimal portfolio allocation and investor protection. In line with the classification provided by MSCI, our sample country includes 23 developed countries and 21 emerging countries for the period 2001 to 2010. We partition our sample countries into developed and emerging countries and report the sample average statistics in panel A of Table 8-1 for the several measures of investor protection (columns 2 to 5).⁵⁸ Corresponding statistics are reported in panel B for developed as compared to emerging markets. Finally, panel C presents the sample mean for the top 10 and bottom 10 countries.

.....Insert Table 8-1 about here.....

In line with our anticipations, the figures reported in panels A and B indicate that developed countries display high investor protection in comparison to their emerging country equivalents, we detected the highest investor protection in terms of government effectiveness in Finland followed by Denmark, Sweden, Switzerland, Norway, Netherland, Canada, Australia, Austria and New Zealand (10 highest investor protection). Equally, the bottom investor protection ranking countries are Indonesia, Egypt, Russia, Peru, Argentina, Romania, Philippines, India, China and Brazil. A similar trend is observed in the other measures of investor protection.

⁵⁸ See columns 6-8 of Table 6-1 for the summary statistics of home and foreign bias measures i.e. *CPIS_HB*, *CPIS_FB*, and *GF_FB*.

Panel C of Table 8-1 demonstrates that the top 10 countries' average government effectiveness is 87.51 (*Con_Cor* = 85.68, *Reg_Qual* = 86.72, *Rule_Law* = 86.17) compared to the bottom 10 government effectiveness of 56.19 (*Con_Cor* = 50.13, *Reg_Qual* = 58.87, *Rule_Law* = 50.66). As such, the analysis is strongly indicates that countries with a lower home bias (higher foreign bias) tend to be associated with a higher investor protection standards or good governance.

Table 8-2 presents the summary statistics of the control variables.⁵⁹ *PolStab, Confl, Press,* and *Educ* demonstrate significant cross country variations. *PolStab* varies widely between 40.86 in Egypt to 99.62 in Denmark. *Press* is the lowest in China (6.96) but highest in Denmark (98.75). Detailed descriptions of the control variables are presented in Section 4.4.5.

.....Insert Table 8-2 about here.....

8.2 Correlation analysis

We report in Table 8-3, the correlation coefficient between all the variables we use in our analysis. Consistent with our expectations, $CPIS_HB$ is negatively and significantly correlated with all the investor protection proxies. This suggests that, from a simple univariate perspective, countries with a greater home bias experience poor investor protection. This assumption is again consistent with our summary analysis. Correspondingly, the $CPIS_FB$ measure has a positive correlation with the investor protection measures, indicating that countries that attract sufficient foreign investors are associated with better investor protection. Additionally, the GF_FB measure also shows positive and statistically significant correlation coefficients, offering further support to the previous assumption. Several of the control variables correlation coefficients display the expected signs.

.....Insert Table 8-3 about here.....

8.3 Econometric issues

We perform the following diagnostic test to resolve econometric problems.

⁵⁹ See Table 6-2 for the summary statistics of *MKTCap*, *Retn_1*, *Turn*, *Infl*, *LSMI*, and Table 7-2 for the summary statistics of *Legal_O*, and *GDPPC*.

8.3.1 Autocorrelation

We use the Durbin-Watson test to check for the presence of autocorrelation by running Durbin-Watson regression (Regdw) in STATA. We obtained DW values of 1.976017 which suggests the presence of autocorrelation.

8.3.2 Heteroskedasticity

We use Breusch-Pagan to test for heteroskedasticity. A large chi-square value of 200.50 indicates the presence of a heteroscedasticity problem. This suggests that the variance of the error terms is not equal. Breusch-Pagan tests of the null hypothesis that the error variances are equal, are rejected. In our main empirical analysis, we employ Newey-West autocorrelation and heteroskedasticity standard error correction which addresses autocorrelation and heteroskedasticity.

Breusch-Pagan test for heteroskedasticity Ho: Constant variance Variables: fitted values of Gov_Eff chi2(1) = 200.50Prob > chi2 = 0.0000

8.3.3 Multicollinearity

We adopt variance inflation factors (VIF) to test for multicollinearity. The mean of VIF for our model is 1.25 which implies that multicollinearity seems not to be a problem in our model.

	VIF Test Result	
Variable	VIF	1/VIF
МКТСар	1.61	0.622237
Legal_O	1.52	0.659116
Confl	1.31	0.760486
MGDP	1.30	0.768587
Turn	1.28	0.779293
Press	1.28	0.784259
Retn_1	1.19	0.839531

Infl	1.18	0.847324
Tobinq	1.16	0.860835
PolStab	1.15	0.867942
GDPPC	1.11	0.898337
LSMI	1.08	0.922136
Educ	1.08	0.924289
Mean VIF	1.25	

8.3.4 Omitted variable bias

We employ Ramsey reset to test for omitted variable bias. The F values are all below 5% which indicates that our model does suffer from omitted variable bias, and we therefore include several relevant explanatory variables; Ramsey reset test of omitted variables for *CPIS_HB*, *CPIS_FB*, and *GF_FB*. Test of important variables omitted. Ramsey reset test using powers of the fitted values.

Ho: model has no omitted variables

Investor protection:

CPIS_HB	F(3, 422)=9.57	Pro>F=0.000
CPIS_FB	F(3, 422)=8.51	Pro>F=0.001
GF_FB	F(3, 422)=7.58	Pro>F=0.001

8.4 **Regression results**

We examine, in this section, whether cross-sectional and temporal variations in home biases of domestic investors and foreign biases of foreign investors across the world explain the international differences in investor protection standards. To this end, we employ pooled OLS with panel dataset regressions, using several control variables that could potentially be correlated with the different investor protection measures, including country and year effects. All the regression estimations use robust Newey-West autocorrelation and heteroskedasticity standard error correction.

8.4.1 Investor protection and equity home bias

We start our analysis by examining the effect of non-optimal domestic allocation, i.e. home bias (*CPIS_HB*) on country level investor protection standards. In Table 8-4, we present the regression results, showing the association between equity home bias and four different investor protection measures. We present *t-statistics* in parentheses. Throughout the models,

the coefficient on *CPIS_HB* is negatively and statistically significant, even at the 1% level in Models 1 to 4. The results offer convincing evidence that domestic investors' overinvestment in the local stock market, relate negatively to investor protection standards. This is consistent with the theoretical assumption by Errunza (2001) that home bias reduces country level good governance (investor protection). On average, one standard deviation increase in home bias reduces investor protection by Gov_Eff (0.26), Con_Cor (0.28), Reg_Qual (0.24), and (0.25). The R² reported in Table 8-4 ranges from 74% in Model 4 to 79% in Model 1. This confirms the relative importance of home bias in explaining investor protection.

.....Insert Table 8-4 about here.....

8.4.2 Investor protection and equity foreign bias

Table 8-5 presents the results of the association between investor protection and *CPIS_FB*. In line with our anticipations, the estimated coefficient on *CPIS_FB* is positively and statistically significant at a minimum of 1% level, in all the four models. The coefficient of 0.472 (t-statistic = 14.13) of Model 1, suggests that the government effectiveness improves for countries with a higher foreign bias. Model 2 exhibits a positive coefficient of 0.652 (t-statistic = 14.14) which is consistent with the view that favourable country allocation by foreign investors is associated with favourable effective control of corruption, indicating a better investor protection. Equally, the coefficient of 0.494 (t-statistic = 11.91) provides support to the conjecture that countries that attract higher foreign equity portfolio investments are associated with better regulatory quality. The coefficient of 0.627 (t-statistic=12.34) in Model 4, further supports the theory that higher foreign bias is associated with improved rule of law.

Model 2 shows that a one standard deviation increase in foreign bias is associated with 0.65% improvement in control of corruption and 0.49% enhancement in regulatory quality in Model 3. The R² ranges from 73\% in Model 4 to 82% in Model 2.

.....Insert Table 8-5 about here.....

8.4.3 Investor protection and fund level foreign bias measures

Table 8-6 presents replicated panel regression estimates of Table 8-5 by employing global fund level data to proxy equity foreign bias (GF_FB). We show in Table 8-6 that the signs of the coefficients are positively and statistically significant across all the measures of investor protection. The different GF_FB coefficients estimates of 0.262 (t-statistic = 3.55), 0.371 (t-statistic = 3.76), 0.318 (t-statistic = 3.40), and 0.307 (t-statistic = 3.11) for Gov_Eff , Con_Cor , Req_Qual , and $Rule_Law$ respectively, provide robustness to the findings discussed above. This indicates that foreign bias, whereby a country attracts sufficient foreign investment, relates positively to investor protection across developed and developing countries. The results are consistent with the theoretical assumption by Errunza (2001) that foreign investors from well governed countries will demand and export good governance to the host countries in which they invest.

.....Insert Table 8-6 about here.....

8.4.4 Control variables

In line with our expectations, several control variables mainly bear the expected signs and are statistically significant as presented in Tables 8-4 to 8-6. *Turn, MGDP, LMSI, Tobinq, Educ, and GDPPC* are positive and statistically significantly related to the investor protection standards. Such findings are consistent with existing studies (see Kang and Stulz, 1997; Lau et al. 2010; Rajan and Zingales, 2003; La Porta et al. 2002). *Retn_1* and *Infl* are inversely related to investor protection and are mainly statistically significant. Similarly, *MKTCap, Confl,* and *Press* are positively associated with investor protection, but their statistical significance levels are sensitive to different specifications. Such inconsistent behaviour of the control variables is also reported by existing studies (see Lau et al. 2010). Other controls such as *PolStab* and (*Legal_O*) have a positive and (negative) relation with investor protection measures, but are generally inconsistent with respect to statistical significance to different specifications.

8.5 Robustness tests

The evidence from univariate, correlations, as well as regression analyses, offer compelling empirically and theoretically consistent suggestion, that equity portfolio investors' fund allocations play a significant role in investor protection standards. Further tests are undertaken in this section to ensure that the above findings are robust.

We run a series of sensitivity analyses in this section to further evaluate the robustness of our main empirical results. Initially, we address the concern of possible endogeneity issues (reverse causality and selection bias) by employing two alternative statistical techniques. First, we treat reverse causality by using the pre-determined (exogenous), one year lagged values, of home and foreign biases. Finally, we adopt the Heckman selection method to address the concern of sample selection bias.

8.5.1 Reverse causality: Lagged pre-determined variables

There is growing evidence that international equity portfolio investors avoid investing in companies or countries that display weak corporate governance. For example, foreign investors may be attracted to countries with better investor protection or good governance. For instance, employing US data Agarwal et al. (2005) show that US investors tend to allocate more funds to countries with better investor protection. Similarly, Chan et al. (2005) argue that foreign investors avoid investing in countries that have poor investor protection standards. The occurrence of this situation will render our estimates to suffer from endogeneity issues arising from reverse causality. We use lagged values statistical techniques to address the potential reverse causality concern. Following McKnight and Weir (2009), in Tables 8-7 to 8-9, we use one-year lagged values of home and foreign bias as instrumental variables or pre-determined exogenous variables.

To address our concern of endogeneity, we employ lagged *CPIS_HB*, *CPIS_FB* and *GF_FB* as instrumental variables in Table 8-7 to 8-9. Consistent with our expectations, the alternative coefficient estimates on *CPIS_HB* in Table 8-7 have the expected negative sign and are statistically significant at the 1% level in all cases. This provides robust support to the result presented in Table 8-4, that higher equity home bias is associated with poor investor protection standards, even after addressing the endogeneity problem arising from possible reverse causality.

.....Insert Table 8-7 about here.....

The coefficient estimates of the lagged values of *CPIS_FB* presented in Table 8-8 remain positive and statistically significant at the 1% significance level throughout all the measures of investor protection. These results further provide compelling robustness to our findings that higher equity foreign bias relates to better investor protection, even after controlling for any likely reverse causality issues. The result demonstrates that by employing different relevant statistical techniques show that the findings of the study are robust.

.....Insert Table 8-8 about here.....

In Table 8-9 we use lagged foreign bias measures constructed using the global funds to address the concern of a potential reverse causality problem. Consistent with our expectation, all four coefficients of the lagged GF_FB are positive and statistically significant at the 1% level for all four proxies of investor protection. The statistical significance of GF_FB estimates are similar to those presented in Table 8-6. Such results further provide robust support to the conclusion that country level investor protection standard improves with increased levels of foreign investor allocation.

.....Insert Table 8-9 about here.....

8.5.2 Heckman selection bias

We use data of 44 countries which are from the 45 all-country index of MSCI and hence, capture 98 per cent of the highly investable markets. Even though, we would preferably like to use as many countries as possible, due to unavailability of data, for the most part for smaller emerging and frontier markets, we result in selecting 44 countries. In this section, we employ the standard robust econometric methodology, i.e., Heckman two-stage selection bias model to resolve the concern of selection biases which may affect the validity of the results. In Tables 8-10 to 8-12, we use the Heckman two-stage selection model to address the issue of selection bias.

The results in Models 1-4 of Table 8-10 are slightly similar to our main analysis of Models 1-4 in Table 8-4. The signs and significance of the coefficients marginally improved. This suggests that endogeneity problems do not generally influence our results.

.....Insert Table 8-10 about here.....

The coefficients of *CPIS_FB*, reported in Models 1-4 of Table 8-11 are positive and statistically significant across all the proxies of investor protection. The magnitude and degree of statistical significance levels of *CPIS_FB* is higher compared to their counterparts from the baseline models of Table 8-5.

.....Insert Table 8-11 about here.....

The results of GF_FB are reported in Model 1-4 of Table 8-12 and are unaffected by using the Heckman selection model. The coefficients are positive and are quantitatively higher than those presented in the previous regressions in Table 8-6. There is a marginal increase in the statistical significance level as compared to Table 8-6.

.....Insert Table 8-12 about here.....

Essentially, our extensive sensitivity analyses demonstrate that our results are certainly robust to different specifications and to the use of alternative estimation methods. The entire analysis of the empirical result offers compelling evidence that a higher degree of home bias is associated with weak investor protection. Correspondingly, foreign investors increasing their portfolio allocation weight towards the global optimum allocations, lead to better investor protection standards.

8.5.3 Alternative investor protection measures

We provide further robustness to our results by using two alternative measures of investor protection. The first measure is International Country Risk Guide (ICRG) investor profile (*ICRG_InvFile*) which captures government attitudes towards inward investment. ⁶⁰ The variable is provided by ICRG and has widely been used by academics. It is a sub-component of ICRG political risk rating which consists of (i) contract viability or risk of expropriation, (ii) payment delays, and (iii) repatriation of profits. The second proxy of investor protection is from firm level survey data provided by World Bank Doing Business (WBDB) investor

⁶⁰ See Bekaert et al. 2007.

protection (*WBDB_InvPro*). We use four components which capture investor protection (i) strength of investor protection, (ii) extent of disclosure index, (iii) ease of shareholder suit, and (iv) credit strength of legal rights index.

Table 8-13 provides the regression results for each of the investor protection measures. In all cases with the exception of GF_FB , the investor protection measures are statistically significant at the 5% level and have the expected sign. The regression result shows that when ICRG investor protection (*ICRG_InvFile*) measure is used, all three sub-optimal portfolio allocation measures are statistically significant at the 5% level and also bear the expected sign. Home bias (*CPIS_HB*) negatively relate to *ICRG_ InvFile*, whilst foreign bias (*CPIS_FB* and *GF_FB*) have a positive association with *ICRG_InvFile*. However, there is a mixed result when investor protection is proxy by data obtained from the WBDB. *CPIS_HB* and *CPIS_FB* have the expected sign and are statistically significant at the 5% and 10% level respectively. However, *GF_FB* is not significant when we employ the *WBDB_InvPro* and it bears an opposite sign. This is could be due to a small observation as the data covers only six years, since WBDB started collecting data from 2006. The evidence presented, suggests that the result is robust to alternative specification and is not sensitive to the choice of particular source of governance data.

.....Insert Table 8-13 about here.....

8.6 Chapter summary

Researchers argue that investor protection plays an important role towards economic growth and development. Therefore, this suggests that improving investor protection is an important policy priority, particularly in several emerging countries. As far as we are concerned, no study has empirically examined the implications of home and foreign bias phenomenon on investor protection. Current literatures mainly examine the impact of investor protection standards on economic growth. This study represents the first attempt to empirically examine the effects of home and foreign bias on investor protection. We hypothesise that; the predominance of home bias will segment the domestic market and will thereby impede the improvement in country-level investor protection standards. Similarly, when a country attracts sufficient international investment, foreign investors from well governed countries will demand good governance and will positively contribute towards the improvement of investor protection standards.

Despite our understanding of what causes home and foreign biases in international equity portfolio diversifications, studies investigating their implications are highly limited. Economic conjecture advocates that financial globalisation that promotes greater international portfolio diversifications may improve factor productivity of a country by promoting better investor protection standards. In a financially open economy, domestic and foreign investors should hold the equities issued by corporates of the country following the ICAPM benchmark. However, defying the normative suggestions, equity investors exhibit different degrees of home and foreign bias in their cross-country diversifications. Lower home and higher foreign bias implies a higher degree of financial integration or globalisation, which further suggests greater presence of foreign investors in the domestic market. A higher presence of foreign investors, demands corporates to adopt a higher standard of governance practices. Similarly, in a financially more integrated market, it becomes difficult for the state itself to expropriate investors, as they risk losing the much needed foreign investments if they do not heed to the demands of foreign investors. The growing interest of foreign investors, thus, drives reform in the domestic investor protection regulations, and the state itself has to improve their own IPS.

Our study is the first attempt to empirically examine the following question: Do the phenomena of the widely observed home and foreign biases exhibited by equity portfolio investors have any implications for investor protection standards? Our study directly uses pooled OLS with panel modelling to examine how sub-optimal portfolio allocation via home and foreign bias, influence country level investor protection standards. We offer a distinct test of our hypothesis using CPIS and EFPR global fund data sample covering a cross-section of 23 developed and 21 developing countries worldwide for the period 2001 to 2010. Consistent with economic justifications, our empirical investigations show that encouraging international portfolio diversification plays an influential role in the development of IPS of a country.

The result provides compelling evidence that home and foreign bias have varying significant influence on investor protection. First, we find that markets where investors observe a higher degree of home bias are associated with poor quality of corporate and state investor protection standards. The explanation for the negative impact is that, large domestic investors will develop close ties with managers and the state, which makes it easier to expropriate minority investors. Our result is consistent with the theoretical assumption by Errunza (2001) that home bias segments the market and leads to weak investor protection. Similarly, relative to more closed markets (lower foreign bias), countries that allow greater foreign portfolio investments (greater foreign bias) reflect higher levels of investor protection standards. Specifically, we find that closer the degree of domestic and foreign investors' international equity investments relative to the theory, higher is the quality of investor protection standards. This suggests that foreign institutional investors from well governed countries will export and demand better investor protection in the host country. Our findings support the notion that financial globalisation has a significant positive effect in the improvement of state and corporate governance.

We also find that most developed countries report stronger positive foreign bias, i.e. these countries are preferred by international investors, compared to the emerging markets. These findings are consistent with existing studies (see Bekaert and Wang, 2010 and Chan et al., 2005). We further add to this literature by providing new evidence on the observance of biases in the cross-country allocations by the sophisticated global fund managers, whose sole objective is to create a broad based optimal global diversification. We uncover that the manifestation of investment biases is not only reserved to the aggregate and macro data which may include country, regional or global funds, but such biases are also observed by the most sophisticated global fund managers.

The policy implications of this study are that policy makers should provide incentives to local investors to invest abroad and, similarly, embark on policies that can attract foreign investors by improving macroeconomic fundamentals. Literature advances sound economic arguments, as we explain that encouraging international equity portfolio diversification, particularly allowing inward foreign equity portfolio investments, should improve the quality of investor protection standards (see Smarzynsk and Wei, 2000 and Bonaglia et al. 2001). Consistent with this economic reasoning, our empirical results reveal that countries where investors span their equity portfolio in line with the theoretically optimal international benchmark, are associated with higher quality of investor protection standards. Such positive implications may arise through indirect or 'collateral' benefits of greater financial integration. Regarding

financially open economies, foreign equity portfolio investors play a catalytic role in improving the quality of investor protection standards.

Developed Countries				Emerging Countries					
	Gov_Eff	Con_Cor	Reg_Qual	Rule_Law		Gov_Eff	Con_Cor	Reg_Qual	Rule_Law
Country	(0-100)	(0-100)	(0-100)	(0-100)	Country	(0-100)	(0-100)	(0-100)	(0-100)
Australia	94.82	94.9	95.3	95.1	Argentina	50.60	42.8	30.4	34.4
Austria	94.48	95.1	93.9	97.9	Brazil	56.87	55.8	57.8	43.8
Belgium	91.68	89.5	87.7	88.6	Bulgaria	58.18	55.0	67.1	51.9
Canada	95.74	95.1	93.6	95.2	Chile	86.69	90.2	92.3	87.9
Denmark	98.79	99.8	98.0	98.6	China	54.44	38.3	43.4	41.5
Finland	98.93	99.8	97.4	99.5	Czech Rep	77.73	68.3	82.5	76.2
France	90.47	90.1	84.8	90.0	Egypt	40.12	36.7	40.2	51.8
Germany	91.63	93.5	92.6	93.2	Hungary	78.86	73.4	84.0	77.7
Greece	75.38	70.5	76.6	74.2	India	54.36	44.2	43.6	56.9
Hong Kong	90.98	91.4	99.1	87.9	Indonesia	39.38	21.5	36.9	25.9
Ireland	91.34	91.8	96.6	93.2	Korea	79.47	68.8	72.6	77.0
Israel	82.23	81.6	81.0	78.5	Malaysia	80.74	66.2	66.0	64.3
Italy	73.59	69.1	78.6	67.6	Mexico	60.95	49.0	64.3	39.7
Japan	86.44	86.2	80.3	88.4	Peru	43.22	47.3	61.2	30.0
Netherland	96.72	96.9	98.1	95.5	Philippines	52.71	35.5	53.4	39.6
New Zealand	93.80	98.9	97.2	97.2	Poland	70.38	68.6	73.0	67.0
Norway	96.86	96.1	89.0	99.1	Romania	50.77	50.9	60.0	52.1
Portugal	84.03	83.9	84.0	85.3	Russia	42.83	19.6	36.7	20.4
Spain	86.71	86.1	87.4	85.9	South Africa	72.66	65.6	66.6	55.8
Sweden	97.82	98.2	93.7	97.7	Thailand	62.64	50.3	61.1	57.2
Switzerland	97.68	96.6	95.3	97.8	Turkey	59.62	51.5	59.2	54.1
United Kingdom	93.66	94.3	97.1	93.7					
United States	90.95	90.6	93.6	91.8					

 Table 8-1: Summary statistics of dependent variables

Note: The dependent variables are investor protection measures. Gov_Eff is government effectiveness; Con_Cor is control of corruption; Reg_Qual is regulatory quality; Rule_Law is rule of law.
Panel B: Averages of the developed and emerging countries.

Country group	Gov_Eff	Con_Cor (0.100)	Reg_Qual	Rule_Law
Developed	96.56	97.14	96.80	97.36
Emerging	48.53	38.52	46.16	37.90

Panel C: Averages of the top and bottom 10 countries.

Country group	Gov_Eff	Con_Cor	Reg_Qual	Rule_Law
	(0-100)	(0-100)	(0-100)	(0-100)
Top10	87.51	85.68	86.72	86.17
Bottom10	56.19	50.13	58.87	50.66

Note: The dependent variables are the investor protection measures. *Gov_Eff* is government effectiveness; *Con_Cor* is control of corruption; *Reg_Qual* is regulatory quality; *Rule_Law* is rule of law.

Developed Countries				Emerging Countries									
Country	MGDP (% of GDP)	Tobinq	PolStab (1-100)	Confl (1-12)	Press (0-100)	Educ (0-100)	Country	MGDP (%of GDP)	Tobinq	PolStab (1-100)	Confl (1-12)	Press (0-100)	Educ (0-100)
Australia	119.16	5.83	94.93	11.26	94.29	NA	Argentina	38.67	5.76	52.70	9.50	56.75	102.18
Austria	28.95	5.35	95.47	11.79	92.51	100.32	Brazil	54.96	5.01	55.90	9.90	59.34	109.13
Belgium	65.83	5.09	93.37	11.43	93.04	90.06	Bulgaria	17.52	5.25	58.07	10.74	64.43	98.72
Canada	114.47	5.67	96.49	11.37	96.02	95.75	Chile	107.1	0.21	85.34	10.63	78.11	97.33
Denmark	63.31	3.72	99.62	10.90	98.75	100.82	China	69.07	4.18	57.24	10.52	6.96	NA
Finland	98.60	6.71	99.30	11.32	98.32	99.63	Czech Rep	25.33	3.33	79.84	10.86	78.44	97.99
France	80.67	5.59	90.89	9.53	88.24	NA	Egypt	55.67	5.39	40.86	9.17	18.63	95.19
Germany	45.71	5.42	91.42	11.05	93.61	101.65	Hungary	24.53	0.92	77.38	11.22	84.89	97.31
Greece	51.91	5.77	73.44	10.77	79.03	100.87	India	68.12	2.70	54.28	7.62	59.43	83.92
Hong Kong	421.17	3.69	93.90	10.83	58.75	97.65	Indonesia	30.03	-1.81	44.54	7.82	39.16	98.77
Ireland	46.77	6.13	90.86	11.51	93.04	95.24	Korea	73.81	-0.72	82.04	9.83	68.89	100.31
Israel	84.72	4.47	81.95	6.60	67.74	103.67	Malaysia	137.03	4.98	83.08	10.41	37.26	95.12
Italy	37.54	5.22	70.69	10.14	80.89	101.81	Mexico	28.33	4.11	61.19	9.28	53.14	101.32
Japan	77.97	1.29	88.30	11.04	77.96	102.18	Peru	47.15	5.17	40.95	7.50	47.33	100.13
Netherland	91.33	6.12	96.07	10.55	98.32	100.02	Philippines	48.26	2.66	53.21	7.44	49.79	94.36
New Zealand	36.38	6.22	94.54	11.41	98.08	NA	Poland	28.56	4.93	69.21	9.90	78.73	96.86
Norway	55.67	4.41	96.50	11.34	98.13	100.39	Romania	16.31	3.49	49.77	10.15	60.20	96.40
Portugal	39.39	5.02	82.15	10.18	90.93	NA	Russia	61.59	2.66	41.83	8.57	28.67	94.36
Spain	86.70	6.00	84.61	9.52	87.09	101.02	South Africa	215.23	4.84	71.08	8.95	69.42	95.28
Sweden	104.08	3.48	97.84	11.37	97.98	99.29	Thailand	62.06	2.96	63.68	8.71	47.16	NA
Switzerland	229.24	5.21	97.95	11.99	96.83	96.00	Turkey	28.94	5.26	61.92	8.73	41.72	97.57
UK	128.47	6.04	92.84	10.45	91.74	NA							
US	124.09	6.50	90.47	10.88	88.50	100.85							

Table 8-2: Summary statistics of control variables

MGDP is market capitalisation scaled by GDP; *Tobin's q* is measured as the log market value of equity plus the book value of total assets minus the book value of equity and divided by the book value of total assets of country i.; *Pop* is population; *PolStab* is political stability; *Confl* is threat of conflict; *Press* is press freedom; *Educ* is education per capita.

	1	2	3	4	5	6	7	8	9	10	11	12	13	14	15	16	17	18	19	20
Gov_Eff (1)	1																			
Con_Cor (2)	0.96*	1																		
Reg_Qual (3)	0.92*	0.94*	1																	
Rule_Law (4)	0.95*	0.96*	0.92*	1																
CPIS_HB (5)	-0.52*	-0.47*	-0.42*	-0.47*	1															
CPIS_FB (6)	0.79*	0.82*	0.77*	0.78*	-0.39*	1														
GF_FB (7)	0.50*	0.49*	0.47*	0.47*	-0.46*	0.46*	1													
MKTCap (8)	0.13	0.12	0.09	0.06	-0.25*	-0.12*	0.03	1												
Retn_1 (9)	-0.29*	-0.27*	-0.21*	-0.25*	0.24*	-0.26*	-0.17*	-0.04	1											
Turn (10)	0.44*	0.35*	0.34*	0.43*	-0.60*	0.16*	0.40*	0.30*	0.05*	1										
MGDP (11)	0.43*	0.38*	0.35*	0.36*	-0.61*	0.24*	0.19*	0.30*	-0.27*	0.34*	1									
Infl (12)	-0.39*	-0.40*	-0.37*	-0.36*	0.27*	-0.29*	-0.29*	-0.03	0.20*	-0.18*	-0.14*	1								
LSMI (13)	0.28*	0.22*	0.31*	0.25*	0.15*	0.28*	0.05*	-0.17*	-0.02	-0.02	0.12*	-0.08*	1							
Tobinq (14)	0.28*	0.35*	0.30*	0.27*	-0.25*	0.33*	0.17*	-0.58*	-0.13*	0.03	0.20*	-0.08*	-0.06*	1						
Legal_O (15)	-0.07*	-0.03	-0.03*	-0.04*	-0.18*	-0.07*	0.10*	0.33*	-0.07*	0.16*	0.35*	-0.03	0.12*	-0.11*	1					
PolStab (16)	0.07*	0.07*	0.13*	0.03	-0.07*	0.04*	-0.12*	0.16*	0.04*	0.05*	0.06*	0.17*	-0.12*	-0.08*	-0.20*	1				
Confl (17)	0.24*	0.24*	0.20*	0.23*	-0.05*	0.24*	0.13*	-0.20*	-0.14*	-0.04*	0.05*	-0.17*	0.07*	0.14*	-0.31*	0.04*	1			
Press (18)	0.09*	0.10*	0.15*	0.05*	0.06*	-0.15*	0.05*	-0.16*	0.07*	0.14	-0.22*	0.12*	-0.14*	-0.16*	-0.17*	0.03*	-0.21*	1		
Educ (19)	0.19*	0.20*	0.18*	0.13*	-0.09*	0.24*	0.19*	0.03	-0.11*	-0.03	0.04	-0.14*	-0.09*	0.12*	-0.07*	-0.02	0.16*	-0.09*	1	
GDPPC (20)	0.20*	0.23*	0 24*	0.23*	0.04	0.27*	0.16*	0.05*	-0.13*	-0.04*	-0.11*	-0.03	-0.08*	0.09*	-0.02	-0.08*	0.13*	0.09*	0.13*	1

Table 8-3: Pearson's pairwise correlation coefficients between the dependent and independent variables

Note: The variables labelled 1- 4 are the investor protection measures as defined in Table 8-1 and 5-7 are the sub-optimal international portfolio allocation bias measures as described in Table 6-1. The other control variables⁶¹ include *MKTCap* as the country market capitalisation; *Retn_1* is the average *MSCI* monthly index return over the past year; *Turn* is turnover ratio; *MGDP* is market capitalisation scaled by GDP; *Infl* is the one year lagged rate of inflation based on the consumer price index; *LSMI* is a measure of market integration measured as the ratio of a country's annual exports plus imports divided by *GDP*; *Tobinq q* is measured as the log market value of equity plus the book value of total assets minus the book value of equity and divided by the book value of total assets of country i; *Legal_O* is a dummy variable that takes a value of 1 if a common law country or 0 if otherwise; *Pop* is population; *PolStab* is political stability; *Confl* is threat of conflict; *Press* is press freedom; *Educ* is education; *GDPPC* is gross domestic product per capita. Statistical significance is reported against 10% (*), 5% (**) and 1% (***) significance.

⁶¹ Several control variables are correlated. However, following the existing literature, we need to control for their impact of these variables on investor protection standards. This has resulted in most of the control variables exhibiting insignificant t-statistics in Tables 8.4-8.12.

Table 8-4: The relation between equity home bias and four investor protection proxies This table shows regression results of the following model:

	Model (1)	Model (2)	Model (3)	Model (4)
	Gov_Eff	Con_Cor	Req_Qual	Rule_Law
CPIS_HB	-0.260***	-0.278***	-0.236***	-0.254***
	(-3.86)	(-3.10)	(-3.34)	(-2.96)
MKTCap	0.668	0.942	0.119*	0.835
-	(1.21)	(1.18)	(1.93)	(1.07)
Retn_1	-0.323***	-0.312***	-0.175	-0.371***
	(-3.97)	(-3.02)	(-1.63)	(-3.27)
Turn	0.539***	0.468***	0.423***	0.775***
	(4.82)	(3.13)	(3.13)	(5.25)
MGDP	0.295*	0.377*	0.203	0.188
	(1.91)	(1.83)	(1.24)	(0.89)
Infl	-0.510***	-0.786***	-0.566***	-0.656***
	(-2.85)	(-3.10)	(-2.65)	(-2.81)
LSMI	0.133***	0.140***	0.157***	0.152***
	(8.40)	(7.36)	(8.94)	(7.49)
Tobinq	0.202***	0.334***	0.288***	0.271***
	(2.77)	(3.12)	(3.33)	(2.65)
Legal_O	-0.396**	-0.701***	-0.683***	-0.450
	(-1.99)	(-2.62)	(-2.90)	(-1.64)
PolStab	0.114	0.123	0.278**	0.072
	(1.02)	(0.83)	(2.05)	(0.04)
Confl	0.190*	0.165	0.456	0.255
	(1.70)	(1.03)	(0.36)	(1.59)
Press	0.415	0.235	0.614	0.506
	(0.58)	(0.23)	(0.73)	(0.52)
Educ	0.433***	0.521***	0.417**	0.246
	(3.49)	(3.13)	(2.56)	(1.46)
GDPPC	0.616***	0.870***	0.777***	0.882***
	(4.51)	(4.60)	(4.62)	(4.65)
Constant	-2.280***	-2.789***	-2.041***	-1.817**
	(-3.98)	(-3.55)	(-2.67)	(-2.26)
Number of Observations	440	440	440	440
Adj. R-squared	0.79	0.81	0.75	0.74
Country effects	Yes	Yes	Yes	Yes
Year fixed effects	Yes	Yes	Yes	Yes

 $IP_{lt} = \alpha + \beta_1 \text{CPIS}_{\text{HB}} + \beta_2 \text{Controls} + \text{Country and Year effects} + \varepsilon_{it}$

Note: The dependent variables are the investor protection (IP_{lt}) measures of country *l* at time *t* (*Gov_Eff, Con_Cor, Req_Qual, Rule_Law*) as described in Table 8-1 above. The key independent variable is *CPIS_HB* also described in Table 6-1. The other independent variables include *MKTCap* as the log country market capitalisation; *Retn_1* is the average MSCI monthly index return over the past year; *Turn* is turnover ratio; *MGDP* is market capitalisation scaled by GDP; *Infl* is the one year lagged rate of inflation based on the consumer price index; *LSMI* is a measure of market integration measured as the ratio of a country's annual exports plus imports divided by *GDP*; *Tobinq q* is measured as the log market value of equity plus the book value of total assets minus the book value of equity and divided by the book value of total assets of country i; *Legal_O* is a dummy variable that takes a value of 1 if a common law country or 0 if otherwise; *PolStab* is political stability; *Confl* is threat of internal conflict; *Press* is press freedom; *Educ* is education; *GDPPC* is gross domestic product per capita. Statistical significance is reported against 10% (*), 5% (**) and 1% (***) significance.

Table 8-5: The relation between equity foreign bias and four investor protection proxies This table shows regression results of the following model:

	Model (1)	Model (2)	Model (3)	Model (4)
	Gov_Eff	Con_Cor	Req_Qual	Rule_Law
CPIS_FB	0.472***	0.652***	0.494***	0.627***
	(14.13)	(14.44)	(11.91)	(12.34)
MKTCap	0.597*	0.818*	0.110***	0.712
-	(1.74)	(1.67)	(2.66)	(1.42)
Retn_1	-0.153**	-0.497	-0.147	-1.135
	(-2.42)	(-0.66)	(-0.16)	(-1.32)
Turn	0.535***	0.396***	0.388***	0.694***
	(6.46)	(3.66)	(3.20)	(6.17)
MGDP	0.395***	0.439***	0.273**	0.235
	(3.69)	(3.08)	(2.10)	(1.57)
Infl	-0.302***	-0.470***	-0.335**	-0.347**
	(-2.87)	(-3.44)	(-2.54)	(-2.54)
LSMI	0.463***	0.243*	0.679***	0.414**
	(3.62)	(1.67)	(4.34)	(2.38)
Tobinq	0.095**	0.175**	0.171***	0.116*
-	(1.99)	(2.51)	(2.82)	(1.67)
Legal_O	-0.242	-0.182	-0.292*	-0.500
-	(-0.18)	(-1.02)	(-1.66)	(-0.26)
PolStab	0.147*	0.171	0.313**	0.393
	(1.72)	(1.58)	(2.41)	(0.33)
Confl	0.176**	0.145	0.311	0.236**
	(2.29)	(1.36)	(0.34)	(2.21)
Press	0.105**	0.110*	0.445	0.134**
	(2.24)	(1.73)	(0.08)	(2.14)
Educ	0.875	0.380	0.528	-0.220
	(0.85)	(0.30)	(0.42)	(-1.48)
GDPPC	0.126	0.199	0.267*	0.238
	(1.21)	(1.41)	(1.82)	(1.55)
Constant	-0.327	0.231	0.545	0.909
	(-0.67)	(0.04)	(0.09)	(1.25)
Number of Observations	440	440	440	440
Adj. R-squared	0.78	0.82	0.75	0.73
Country effects	Yes	Yes	Yes	Yes
Year fixed effects	Yes	Yes	Yes	Yes

 $IP_{lt} = \alpha + \beta_1 \text{CPIS}_FB + \beta_2 \text{Controls} + \text{Country and Year effects} + \varepsilon_{it}$

Note: The dependent variables are the investor protection (IP_{lt}) measures of country *l* at time *t* (*Gov_Eff, Con_Cor, Req_Qual, Rule_Law*) as described in Table 8-1 above. The key independent variable is *CPIS_FB* also described in Table 6-1. All the controls are the same as described in Table 8-4 above. All t-statistics reported are based on Newey-West autocorrelation and heteroskedasticity-corrected standard errors. For tractable interpretation, all the coefficients are reported as elasticity and the statistical significance is reported against 10% (*), 5% (**) and 1% (***) significance levels.

Table 8-6: The relation between equity foreign bias using global fund and four different investor protection proxies

This table shows regression re	esults of the following model:
--------------------------------	--------------------------------

	Model (1)	Model (2)	Model (3)	Model (4)
	Gov_Eff	Con_Cor	Req_Qual	Rule_Law
GF_FB	0.262***	0.371***	0.318***	0.307***
	(3.55)	(3.76)	(3.40)	(3.11)
МКТСар	0.816	0.112	0.134**	0.993
	(1.49)	(1.43)	(2.23)	(1.29)
Retn_1	-0.359***	-0.333***	-0.192*	-0.396***
	(-4.50)	(-3.30)	(-1.81)	(-3.56)
Turn	0.611***	0.496***	0.444***	0.818***
	(5.57)	(3.70)	(3.25)	(5.75)
MGDP	0.530***	0.624***	0.413**	0.415**
	(3.51)	(3.06)	(2.47)	(2.02)
Infl	-0.518***	-0.766***	-0.548**	-0.648***
	(-2.84)	(-3.05)	(-2.57)	(-2.76)
LSMI	0.115***	0.119***	0.139***	0.133***
	(7.16)	(6.09)	(7.93)	(6.36)
Tobinq	0.225***	0.355***	0.305***	0.291***
	(3.15)	(3.42)	(3.60)	(2.91)
Legal_O	-0.462**	-0.789***	-0.758***	-0.525**
	(-2.44)	(-3.11)	(-3.42)	(-1.97)
PolStab	0.768	0.732	0.235*	0.489
	(0.69)	(0.51)	(1.77)	(0.31)
Confl	0.166	0.130	0.152	0.226
	(1.55)	(0.87)	(0.13)	(1.47)
Press	0.294	0.543	-0.770	0.358
	(0.42)	(0.05)	(-0.96)	(0.38)
Educ	0.350***	0.397**	0.310*	0.146
	(2.66)	(2.31)	(1.94)	(0.82)
GDPPC	0.534***	0.761***	0.684***	0.791***
	(3.92)	(4.07)	(3.97)	(4.13)
Constant	-2.070***	-2.367***	-1.674**	-1.501*
	(-3.35)	(-2.91)	(-2.23)	(-1.76)
Number of Observations	440	440	440	440
Adj. R-squared	0.76	0.80	0.74	0.71
Country effects	Yes	Yes	Yes	Yes
Year fixed effects	Yes	Yes	Yes	Yes

$IP_{lt} = \alpha + \beta_1 \text{GF}_F$	$B + \beta_2 Controls +$	Country and	Year effects + ε_{it}
--	--------------------------	-------------	-----------------------------------

Note: The dependent variables are the investor protection (IP_{lt}) measures of country *l* at time *t* (*Gov_Eff, Con_Cor, Req_Qual, Rule_Law*) as described in Table 8-1 above. The key independent variable is *GF_FB* also described in Table 6-1. All the controls are the same as described in Table 8-4 above. All t-statistics reported are based on Newey-West autocorrelation and heteroskedasticity-corrected standard errors. For tractable interpretation, all the coefficients are reported as elasticity and the statistical significance is reported against 10% (*), 5% (**) and 1% (***) significance levels.

	Model (1)	Model (2)	Model (3)	Model (4)
	Gov Eff	Con Cor	Reg Qual	Rule Law
CPIS HB 1	-0.294***	-0.322***	-0.248***	-0.291***
	(-5.16)	(-4.12)	(-3.63)	(-3.64)
MKTCap	0.616	0.100*	0.114**	0.828
•	(1.63)	(1.94)	(2.52)	(1.57)
Retn_1	-0.292***	-0.313***	-0.165	-0.357***
	(-3.35)	(-2.63)	(-1.58)	(-2.93)
Turn	0.536***	0.441***	0.469***	0.771***
	(5.71)	(3.43)	(4.18)	(5.88)
MGDP	0.260**	0.342**	0.171	0.169
	(2.17)	(2.09)	(1.20)	(1.01)
Infl	-0.532***	-0.763***	-0.643***	-0.670***
	(-4.30)	(-4.50)	(-4.35)	(-3.87)
LSMI	0.139***	0.146***	0.160***	0.158***
	(10.53)	(8.12)	(10.18)	(8.60)
Tobinq	0.187***	0.335***	0.274***	0.267***
-	(3.74)	(4.88)	(4.57)	(3.82)
Legal_O	-0.333**	-0.730***	-0.673***	-0.464**
-	(-2.12)	(-3.40)	(-3.59)	(-2.12)
PolStab	0.748	0.114	0.255**	0.226
	(0.89)	(0.99)	(2.54)	(0.19)
Confl	0.201**	0.165	0.544	0.254**
	(2.22)	(1.33)	(0.50)	(2.00)
Press	0.328	0.733	-0.786	0.406
	(0.56)	(0.09)	(-1.13)	(0.50)
Educ	0.438***	0.546***	0.400**	0.298
	(3.16)	(2.87)	(2.41)	(1.53)
GDPPC	0.632***	0.895***	0.779***	0.879***
	(5.75)	(5.94)	(5.92)	(5.71)
Constant	-2.320***	-2.906***	-1.993**	-2.067**
	(-3.55)	(-3.25)	(-2.55)	(-2.26)
Number of Observations	396	396	396	396
Adj. R-squared	0.80	0.81	0.76	0.74
Country effects	Yes	Yes	Yes	Yes
Year fixed effects	Yes	Yes	Yes	Yes

Note: The dependent variables are the investor protection measures (*Gov_Eff, Con_Cor, Req_Qual, Rule_Law*) as described in Table 8-1 above. The key independent variable is *CPIS_HB* also described in Table 6-1. All the controls are the same as described in Table 8-4 above. All t-statistics reported are based on Newey-West autocorrelation and heteroskedasticity-corrected standard errors. For tractable interpretation, all the coefficients are reported as elasticity and the statistical significance is reported against 10% (*), 5% (**) and 1% (***) significance levels.

	Model (1)	Model (2)	Model (3)	Model (4)
	Gov Eff	Con Cor	Reg Qual	Rule Law
CPIS FB 1	0.471***	0.656***	0.492***	0.628***
	(17.78)	(18.79)	(14.56)	(16.86)
MKTCap	0.608**	0.975**	0.112***	0.801**
	(2.11)	(2.57)	(3.05)	(1.98)
Retn_1	-0.470	-0.107	-0.598	-0.607
	(-1.46)	(-0.12)	(-0.70)	(-0.65)
Turn	0.554***	0.390***	0.437***	0.707***
	(8.87)	(4.74)	(5.48)	(8.05)
MDGP	0.353***	0.390***	0.213**	0.198*
	(4.31)	(3.61)	(2.03)	(1.71)
Infl	-0.349***	-0.474***	-0.429***	-0.386***
	(-3.70)	(-3.82)	(-3.57)	(-2.91)
LSMI	0.521***	0.311**	0.734***	0.490***
	(5.01)	(2.27)	(5.53)	(3.34)
Tobinq	0.900**	0.189***	0.165***	0.125**
-	(2.34)	(3.73)	(3.36)	(2.31)
Legal_O	-0.536	-0.183	-0.263*	-0.623
-	(-0.44)	(-1.14)	(-1.69)	(-0.36)
PolStab	0.822	0.129	0.266***	0.769
	(1.28)	(1.52)	(3.25)	(0.09)
Confl	0.183***	0.143	0.376	0.233**
	(2.66)	(1.57)	(0.43)	(2.40)
Press	0.103**	0.106*	-0.479	0.135**
	(2.30)	(1.80)	(-0.08)	(2.16)
Educ	0.981	0.592	0.361	-0.172
	(0.91)	(0.42)	(0.26)	(-1.13)
GDPPC	0.139	0.211*	0.265**	0.223*
	(1.58)	(1.82)	(2.36)	(1.80)
Constant	-0.447	-0.150	0.0637	0.605
	(-0.88)	(-0.22)	(0.10)	(0.84)
Number of Observations	396	396	396	396
Adj. R-squared	0.77	0.81	0.75	0.72
Country effects	Yes	Yes	Yes	Yes
Year fixed effects	Yes	Yes	Yes	Yes

Table	8-8:	Using	one-vear	lag	CPIS	FR
Lanc	0-0.	USINE	Unc-ycar	Iaz		гр

Note: The dependent variables are the investor protection measures (*Gov_Eff, Con_Cor, Req_Qual, Rule_Law*) as described in Table 8-1 above. The key independent variable is *CPIS_FB* also described in Table 6-1. All the controls are the same as described in Table 8-4 above. All t-statistics reported are based on Newey-West autocorrelation and heteroskedasticity-corrected standard errors. For tractable interpretation, all the coefficients are reported as elasticity and the statistical significance is reported against 10% (*), 5% (**) and 1% (***) significance levels.

	Model (1)	Model (2)	Model (3)	Model (4)
	Gov_Eff	Con_Cor	Req_Qual	Rule_Law
GF_FB_1	0.266***	0.402***	0.337***	0.346***
	(4.50)	(5.08)	(4.89)	(4.26)
MKTCap	0.688*	0.109**	0.121***	0.907*
•	(1.81)	(2.13)	(2.71)	(1.73)
Retn 1	-0.343***	-0.347***	-0.186*	-0.391***
	(-3.99)	(-3.01)	(-1.85)	(-3.30)
Turn	0.651***	0.509***	0.507***	0.842***
	(7.54)	(4.39)	(5.02)	(7.07)
MGDP	0.520***	0.620***	0.383***	0.421***
	(4.82)	(4.28)	(3.04)	(2.83)
Infl	-0.577***	-0.784***	-0.653***	-0.694***
	(-4.67)	(-4.72)	(-4.52)	(-4.07)
LSMI	0.116***	0.119***	0.139***	0.134***
	(9.02)	(6.91)	(9.24)	(7.58)
Tobing	0.206***	0.350***	0.283***	0.281***
1	(4.12)	(5.19)	(4.84)	(4.07)
Legal_O	-0.389**	-0.803***	-0.731***	-0.528**
0 =	(-2.46)	(-3.78)	(-3.95)	(-2.42)
PolStab	0.276	0.485	0.201**	0.796
	(0.32)	(0.42)	(2.02)	(0.68)
Confl	0.168*	0.118	0.161	0.213*
	(1.84)	(0.96)	(0.15)	(1.69)
Press	0.139	0.195	0.101	0.174
	(0.24)	(0.25)	(1.46)	(0.21)
Educ	0.371***	0.425**	0.296*	0.196
	(2.61)	(2.23)	(1.78)	(1.00)
GDPPC	0.555***	0.782***	0.684***	0.781***
	(4.96)	(5.20)	(5.23)	(5.06)
Constant	-2.189***	-2.501***	-1.616**	-1.742*
	(-3.29)	(-2.80)	(-2.07)	(-1.90)
Number of Observations	396	396	396	396
Adj. R-squared	0.75	0.80	0.75	0.70
Country effects	Yes	Yes	Yes	Yes
Year fixed effects	Yes	Yes	Yes	Yes

 Table 8-9: Using one-year lag GF FB

Note: The dependent variables are the investor protection measures (Gov_Eff , Con_Cor , Req_Qual , $Rule_Law$) as described in Table 8-1 above. The key independent variable is GF_FB also described in Table 6-1. All the controls are the same as described in Table 8-4 above. All t-statistics reported are based on Newey-West autocorrelation and heteroskedasticitycorrected standard errors. For tractable interpretation, all the coefficients are reported as elasticity and the statistical significance is reported against 10% (*), 5% (**) and 1% (***) significance levels.

Heckman two-stage model	Model (1)	Model (2)	Model (3)	Model (4)
č	Gov_Eff	Con_Cor	Req_Qual	Rule_Law
CPIS_HB	-0.310***	-0.325***	-0.277***	-0.308***
	(-4.14)	(-3.61)	(-4.48)	(-3.74)
МКТСар	0.762*	0.968**	0.133***	0.921*
	(1.63)	(2.11)	(2.80)	(1.66)
Retn_1	-0.329***	-0.334***	-0.245***	-0.425***
	(-4.14)	(-3.76)	(-3.96)	(-3.88)
Turn	0.557***	0.564***	0.521***	0.823***
	(6.07)	(4.24)	(4.26)	(6.88)
MGDP	0.342***	0.487***	0.348**	0.231
	(3.61)	(3.49)	(3.45)	(1.47)
Infl	-0.579***	-0.811***	-0.667***	-0.694***
	(-3.11)	(-4.62)	(-5.49)	(-4.77)
LSMI	0.279***	0.154***	0.249***	0.174***
	(10.28)	(8.63)	(10.08)	(9.44)
Tobinq	0.384***	0.472***	0.342***	0.375***
	(4.42)	(5.36)	(5.26)	(4.23)
Legal_O	-0.434***	-0.825***	-0.774***	-0.483**
	(-2.78)	(-3.52)	(-3.71)	(-3.34)
PolStab	0.248	0.282	0.322***	0.147
	(1.45)	(1.08)	(4.27)	(0.48)
Confl	0.312***	0.288	0.547	0.474***
	(2.98)	(1.61)	(0.69)	(3.67)
Press	0.576	0.269	0.737	0.578
	(0.62)	(0.68)	(1.06)	(0.56)
Educ	0.547***	0.633***	0.584***	0.438
	(3.88)	(3.67)	(3.69)	(1.77)
GDPPC	0.732***	0.948***	0.881***	0.934***
	(6.40)	(5.84)	(5.66)	(6.95)
Lambda (inverse Mills' ratio)	0.543***	0.479***	0.751***	0.623***
	(12.55)	(17.08)	(14.87)	(15.33)
Wald Chi-square	548.03	845.79	681.53	790.35
Constant	-2.352***	-4.675***	-4.355***	-2.564***
	(-3.88)	(-4.59)	(-3.46)	(-3.67)

Table 8-10: CPIS_HB regression ba	ed on the Heckman two-stage treatment effect
model	

Note: This table presents the coefficients of the estimates from Heckman two-stage treatment effects models. In the first stage, we run the probit model. We include Lambda (inverse Mills' ratio) in the second stage with control variables. The dependent variables are the investor protection measures (*Gov_Eff, Con_Cor, Req_Qual, Rule_Law*) as described in Table 8-1 above. The key independent variable is *CPIS_HB* also described in Table 6-1. All the controls are the same as described in Table 8-4 above. All t-statistics reported are in parentheses. For tractable interpretation, all the coefficients are reported as elasticity and the statistical significance is reported against 10% (*), 5% (**) and 1% (***) significance levels.

Heckman two-stage model	Model (1)	Model (2)	Model (3)	Model (4)
-	Gov_Eff	Con_Cor	Req_Qual	Rule_Law
CPIS_FB	0.532***	0.537***	0.527***	0.764***
	(17.33)	(17.56)	(14.45)	(15.52)
МКТСар	0.656***	0.833**	0.364***	0.818**
	(3.28)	(2.45)	(3.60)	(1.51)
Retn_1	-0.325***	-0.408	-0.344	-0.402
	(-3.22)	(-0.76)	(-0.87)	(-1.83)
Turn	0.621***	0.486***	0.424***	0.734***
	(9.77)	(4.52)	(6.25)	(8.47)
MGDP	0.421***	0.658***	0.492***	0.433***
	(5.34)	(3.96)	(3.46)	(2.92)
Infl	-0.352***	-0.578***	-0.377***	-0.395***
	(-3.28)	(-6.23)	(-3.67)	(-3.72)
LSMI	0.624***	0.455**	0.756***	0.624***
	(4.21)	(2.35)	(5.28)	(3.84)
Tobinq	0.132***	0.244***	0.203***	0.271**
	(2.87)	(3.73)	(3.81)	(2.29)
Legal_O	-0.361	-0.251	-0.382***	-0.587
	(-0.87)	(-1.65)	(-3.87)	(-0.64)
PolStab	0.255***	0.263***	0.488^{***}	0.462
	(3.64)	(3.31)	(4.27)	(0.87)
Confl	0.234***	0.250*	0.439	0.372***
	(3.64)	(1.97)	(0.64)	(3.44)
Press	0.277***	0.247***	0.545	0.262***
	(3.76)	(2.73)	(0.18)	(3.98)
Educ	0.884	0.466	0.633	0.324*
	(1.43)	(0.68)	(0.64)	(1.95)
GDPPC	0.268***	0.209*	0.283***	0.347**
	(2.90)	(1.99)	(2.83)	(2.73)
Lambda (inverse Mills' ratio)	-0.345***	-0.641***	-0.482***	-0.632***
	(-19.46)	(-11.48)	(-13.28)	(-14.73)
Wald Chi-square	762.29	587.66	652.59	546.39
Constant	-0.483	0.354	0.563	0.783
	(-0.77)	(0.28)	(0.63)	(1.67)

 Table 8-11: CPIS_FB regression based on the Heckman two-stage treatment effect

 model

Note: This table presents the coefficients of the estimates from Heckman two-stage treatment effects models. In the first stage, we run the probit model. We include Lambda (inverse Mills' ratio) in the second stage with control variables. The dependent variables are the investor protection measures (Gov_Eff , Con_Cor , Req_Qual , $Rule_Law$) as described in Table 8-1 above. The key independent variable is $CPIS_FB$ also described in Table 6-1. All the controls are the same as described in Table 8-4 above. All t-statistics reported are in parentheses. For tractable interpretation, all the coefficients are reported as elasticity and the statistical significance is reported against 10% (*), 5% (**) and 1% (***) significance levels.

			0	
Heckman two-stage model	Model (1)	Model (2)	Model (3)	Model (4)
-	Gov_Eff	Con_Cor	Req_Qual	Rule_Law
GF_FB	0.308***	0.394***	0.432***	0.467***
	(5.12)	(5.25)	(4.97)	(4.63)
МКТСар	0.836***	0.186***	0.227***	1.132***
	(3.27)	(3.62)	(3.78)	(3.86)
Retn_1	-0.454***	-0.367***	-0.274***	-0.487***
	(-4.88)	(-3.89)	(-3.78)	(-3.77)
Turn	0.674***	0.586***	0.487***	0.895***
	(7.07)	(5.33)	(4.97)	(7.11)
MGDP	0.542***	0.685***	0.532***	0.424***
	(4.32)	(4.21)	(4.87)	(3.68)
Infl	-0.554***	-0.836***	-0.562***	-0.769***
	(-3.97)	(-5.88)	(-4.34)	(-4.83)
LSMI	0.237***	0.254***	0.367***	0.188***
	(8.74)	(7.68)	(9.35)	(7.63)
Tobinq	0.366***	0.383***	0.446***	0.314***
	(4.37)	(6.76)	(5.78)	(4.03)
Legal_O	-0.539***	-0.823***	-0.811***	-0.578***
	(-3.11)	(-4.94)	(-5.36)	(-3.45)
PolStab	0.792	0.766	0.368***	0.535
	(1.27)	(0.96)	(3.76)	(1.24)
Confl	0.241***	0.228***	0.297	0.276*
	(2.86)	(2.87)	(0.77)	(1.88)
Press	0.313	0.573	-0.784*	0.488
	(0.76)	(0.14)	(-1.97)	(0.68)
Educ	0.413***	0.433**	0.427**	0.286
	(3.56)	(2.04)	(2.62)	(1.21)
GDPPC	0.681***	0.863***	0.694***	0.807***
	(6.74)	(4.62)	(4.28)	(5.76)
Lambda (inverse Mills' ratio)	-0.677***	-0.543***	-0.368***	-0.462***
· · · · · · · · · · · · · · · · · · ·	(-17.24)	(-16.25)	(-11.72)	(-14.93)
Wald Chi-square	765.72	615.04	578.67	558.82
Constant	-2.421***	-3.856***	-2.322**	-2.635***
	(-4.28)	(-3.69)	(-2.77)	(-2.94)

Table 8-12: GF	FB	regression	based (on the	Heckman	two-stage	treatment	effect model

Note: This table presents the coefficients of the estimates from Heckman two-stage treatment effects models. In the first stage, we run the probit model. We include Lambda (inverse Mills' ratio) in the second stage with control variables. The dependent variables are the investor protection measures (Gov_Eff , Con_Cor , Req_Qual , $Rule_Law$) as described in Table 8-1 above. The key independent variable is GF_FB also described in Table 6-1. All the controls are the same as described in Table 8-4 above. All t-statistics reported are in parentheses. For tractable interpretation, all the coefficients are reported as elasticity and the statistical significance is reported against 10% (*), 5% (**) and 1% (***) significance levels.

	ICRG_InvFile	ICRG_InvFile	ICRG_InvFile	WBDB_InvPro	WBDB_InvPro	WBDB_ InvPro
CPIS_HB	-0.097***			-1.156***		
	(-9.17)			(-2.82)		
CPIS_FB		0.140***			0.712**	
		25.85			2.21	
GF_FB			0.154***			-0.24
			10.24			(-1.33)

Table 0 12.	Llaina	altamativa	investor	matation	
1 able 0-15:	USINg	alternative	Investor	protection	measures

Note: *ICRG_InvFile* is investor profile measure provided by International Country Risk Guide (ICRG); *WBDB_InvPro* is investor protection measure provided by World Bank Doing Business (WBDB).

Chapter 9 Summary, conclusions and limitations

9.1 Introduction

This chapter summarises the thesis and key findings. The study consists of three research questions (R.Q).

R.Q 1: What is the impact of sub-optimal international portfolio allocation on cost of capital?

R.Q 2: What is the role of international portfolio investment on stock market development?

R.Q 3: What is the effect of international portfolio investment on investor protection?

9.2 Restatement of the research questions

Countries, through financial liberalisation, have removed the explicit barriers to international capital inflows and outflows. Furthermore, despite the widely apparent, recognised diversification benefits from international investment through risk sharing (see Grubel, 1968; Solnik, 1974; Errunza, 1998; DeSantis and Gerard, 1997) evidence demonstrates that due to the phenomenon of home and foreign bias, domestic investors still allocate disproportionately more investment in the domestic market and are not holding internationally diversified portfolios as recommended by ICAPM. Similarly, foreign investors are not investing adequately in the domestic market. Reasons for the home and foreign bias phenomenon include deviation from purchasing power parity, information asymmetries, poor accounting standards, hedging motives and behavioural biases. Studies suggest that home bias segments the domestic market, whilst when foreign investors allocate investment towards the implied weight recommended by ICAPM, it integrates the market.

Therefore, if the prevalence of home and foreign bias segments and integrates the market, then these biases will have important implications. For instance, Errunza (2001) argues that the inflow of foreign equity investment will have impact on cost of capital, stock market development, and investor protection. Our study aims to answer three important research questions which include: What is the impact of home and foreign bias on cost of capital? What is the impact of home and foreign bias on stock market development? Does home and foreign bias have any influence or determine country level investor protection standards?

9.3 Hypothesis, data and methodology

Generally, we derived the following nine hypotheses. We use the first three hypotheses to answer the first research question: What is the impact of sub-optimal portfolio allocations on the cost of capital?

- **H**₁ Countries that exhibit higher home bias are associated with higher cost of capital.
- **H**₂ Countries with higher foreign bias relate to lower cost of capital.
- **H**₃ Pervasiveness with higher global fund foreign bias allocation is associated with lower cost of capital.

We developed the following three hypotheses to answer our second research question: What is the role of international portfolio investment on stock market development?

- H₄ Countries with a prevalence for higher home bias, experience poor stock market development.
- **H**₅ Countries with a higher foreign bias, experience better stock market development.

 H_6 Countries with a higher global fund foreign bias, enjoy better stock market development.

We use the last three hypotheses to answer the third research question. What is the effect of international portfolio investment on investor protection?

H₇ Higher home bias is related to a lower level of investor protection standards.

 H_8 Higher foreign bias is associated with higher levels of investor protection standards.

H₉ A higher country global fund bias is related to higher investor protection standards.

Our study employs three different datasets to calculate our measures of home and foreign bias. First, following Chan et al. (2005) we use recently available, cross-country bilateral equity data, for a period of 10 years (2001-2010) which we obtained from Coordinated Portfolio Investment Survey (CPIS) to construct home and foreign bias. Second, we employ a distinctive fund level country allocation dataset, provided by Emerging Portfolio Fund Research (EPFR) to construct the global fund foreign bias measure. Finally, we use country level market capitalisation sourced from World Development Indicator (WDI) to construct ICAPM benchmark allocation. In our empirical studies, we use five proxies of cost of capital, four measures of stock market development, and four investor protection measures which have widely been used in several academic literatures.

Consistent with previous studies, in the first empirical study, we use historical realised return of the market (see Lau et al. 2010), sovereign credit rating (Reeb et al. 2001), country equity risk premium (Damodaran, 2012), and dividend yield (Bekaert and Harvey, 2005) to proxy for cost of capital. We further use *Tobin's Q* to capture valuation effect or firm performance. Following existing studies, in the second empirical study, we use four measures of stock market development. Particularly, we use market capitalisation as a percentage of GDP (Levine and Zervos, 1996), stock value traded as a percentage of GDP (Bencivenga et al. 1995), turnover ratio (Boehmer et al. 2005), and transaction cost (Chan et al. 2005) to proxy for stock market development. Finally, in the third empirical study, drawing on existing literature, we use four investor protection measures, sourced from World Governance Indicators, which has largely been employed by several academics (see Neumayer, 2002; Anders, 2006; Jung, 2006).

Generally, we test those hypotheses using pooled OLS, with panel setup and Newey-West autocorrelation and heteroskedasticity corrected standard errors. We performed several diagnostic tests and addressed our concern of endogeneity issues using lagged values, and the Heckman selection model.

9.4 Research findings

The first empirical study examines the impact of sub-optimal (home and foreign bias) international portfolio allocation on cost of capital. The study uses historical realised market risk premium, sovereign credit risk rating, country equity risk premium, and dividend yield as proxies of cost of capital. The results in Chapter 6 demonstrate that home bias increases cost of capital, whilst foreign bias reduces cost of capital as a result of increased risk sharing between domestic and foreign investors. We provide robustness to our main result by

examining the impact of home and foreign bias on firm valuation via *Tobin's Q*. We find that home and foreign bias reduces and increases firm performance respectively. The result further shows that home and foreign bias reduces and increases cost of capital respectively. This is as a result of the inverse relation between cost of capital and firm valuation. Additionally, we perform sensitivity analysis by using lagged values, and the Heckman selection model to address our concern of endogeneity. The result provides compelling evidence that our main result is robust.

The second empirical study investigates the impact of home bias (domestic investors overweighting the domestic markets) and foreign bias (under- or over-weighting the foreign markets) allocation on stock market development. Existing studies largely examine the importance of stock market development. We use four measures (market capitalisation as a percentage of GDP, stock value traded as a percentage of GDP, turnover ratio, and transaction cost) as proxies of stock market development. The research findings show the effects of home bias with the exception of transaction cost, reduces market capitalisation as a percentage of GDP, stock value traded as a percentage of GDP, and turnover ratio. The explanation is that the prevalence of home bias, segments stock markets, and therefore, reduces risk sharing and liquidity.

However, the results further show that foreign bias increases the stock market development measures, apart from transaction cost, which we expect to have a negative relation with foreign bias. The coefficients of the stock market development measures are all positive and statistically significant at the one percentage level. This is explained by the fact that foreign investors provide capital, expertise, and competition, which could spur financial deepening of the local market, particularly in countries with weaker financial development. Furthermore, foreign investors not only infuse the much needed capital for investment and growth but also promote competition, raise the bar of corporate governance and transparency, and promote the domestic firms to the international financial markets. We use lagged values, and the Heckman selection model to address the concern of endogeneity. The sensitivity analysis provides support that our main result is robust.

The third empirical study examines the impact of home and foreign bias on investor protection standards. We use four measures of investor protection (government effectiveness, control of corruption, regulatory quality, and rule of law) which we source from the World Governance Indicators. The investor protection measures capture the rights given to minority investors, how these rights are enforced at the courts, and the sound business and institutional environment in which corporations operate to create value for shareholders. The results suggest that, even after controlling for variables that affect investor protection, home bias leads to weak investor protection standards. This corroborates with the theory that domestic institutional investors that become controlling shareholders in companies, will engage in activities that will lead to poor governance. Similarly, the results also show that foreign bias improves investor protection standards. This, in part, is due to foreign investors from well governed countries exporting and demanding good governance from the host country that has weak investor protection.

We perform several sensitivity analyses. First, we use alternative measures of investor protection from two governance sources to check the robustness of our result. The results largely provide support to our main analysis that home bias leads to weak investor protection, whilst foreign bias improves country level investor protection. We recognise that foreign investors may allocate investment in countries with better investor protection, which will create reverse causality issues. We, therefore, use lagged values and the Heckman selection model to address the endogeneity problem. The result provides robust evidence to support our main analysis that home and foreign bias have a negative and positive impact on country-level investor protection.

9.5 Key contributions

We provide a brief summary of our main research contributions as we have shown the detail contributions in Chapters 1 and 3. Evidence provided in the existing literature of the prevalence of home and foreign bias in equity investment allocation, motivates us to carry out this study. We make use of recent available cross-country portfolio allocation provided by CPIS and EPFR to make important contributions to how sub-optimal portfolio (i.e. equity home and foreign bias) allocation impacts on cost of capital, stock market development, and investor protection.

In the first empirical analysis, we contribute to the current literature by showing that home bias exhibited by domestic investors increases cost of capital due to inadequate risk sharing. Conversely, we also show that foreign bias reduces cost of capital as a result of increased risk sharing between domestic and foreign investors.

Second, there are limited direct empirical studies that have examined factors that influence or determine stock market development. This study makes contributions to the stock market development literature, by examining the impact of home and foreign bias on stock market development. We show that home bias impedes on stock market development, whilst foreign bias enhances stock development.

Strands of literatures have investigated the importance of investor protection standards. As far as we are concerned, scant studies have examined factors that influence or determine investor protection. We, therefore, make contributions to the governance literature by providing empirical evidence on how home and foreign bias impact on investor protection. Our result shows that home bias leads to poor investor protection standards, whilst higher foreign bias, enhances investor protection standards.

9.6 Potential limitations of the research and avenues for future research

We made substantial efforts to answer the research questions to achieve our research objectives. Nonetheless, there are some potential limitations to this study and we, therefore, advice readers to exercise caution when interpreting the findings of the research. The findings of this study are limited to 44 countries, due to unavailability of data on all other countries. Also, we use predetermined criteria to select the 44 countries. The non-random sample of the countries will subject our study to sample selection bias which affects the validity of our research findings. However, we reduce the limitations by using the Heckman selection model to provide robustness to the main result.

There is unavailability of data on several emerging and frontier markets. Analysis of the research, based on developed and emerging countries would have provided more insight into the study if there were data on emerging markets. Future research can examine the impact of home and foreign bias on emerging countries' cost of capital, stock market development, and investor protection when emerging countries' data become available.

Countries are attracting multinational companies, which imply that domestic investors can have international diversification, based on the multinational companies in the domestic market. This may have ramifications on the suggestion by ICAPM. Therefore, future research can examine the extent that multinational companies affect the home and foreign bias phenomenon. Additionally, future research can examine if domestic investors are able to diversify risk domestically, through multinational companies in the domestic countries.

A further limitation of the study is that, domestic and foreign investors may avoid countries with high cost of capital, less developed stock markets, and weak investor protection. This creates reverse causality which may affect the validity of the study. However, the potential reverse causality is addressed empirically by using lagged values.

The Emerging Portfolio Fund Research (EPFR) data are global funds domiciled in nine countries focusing on specific regions or markets. Due to the objectives of the fund, they do not allocate investment in their home countries. As a result, we are unable to use EPFR data to construct a home bias measure. Future research can use mutual funds' data to extensively investigate the impact of home bias on stock market development and investor protection. The study can be extended in the future by examining the impact of home and foreign bias on cash flow.

9.7 Policy implications of the research

Apparently, evidence provided in the study, suggests that stock market liberalisation has failed to encourage investors engage in optimal portfolio allocation, leading to home and foreign bias. We show that cross-country variations in home and foreign bias matters for cost of capital, stock market development, and investor protection, and therefore, have several important policy implications. For instance, existing studies show that reduction in cost of capital is important for project evaluation and investment, whilst improvement in stock market development and better investor protection matters for economic growth and development. The policy implications of the study are that, policy makers in countries that experience high degrees of home bias, need to encourage domestic investors to diversify investment abroad. Similarly, policy makers can embark on strategies to attract foreign investors to participate in the domestic stock market.

9.8 Concluding remarks

This chapter summarises and concludes the research. The chapter first presents a restatement of the research questions. It then proceeds to highlight a summary of the research hypothesis, data, and methodology and findings. The key contributions of the research are presented prior to highlighting the potential limitations of the research. The chapter provides the policy implications of the research and avenues for future study.

Consistent with economic justification, the research findings show that home bias increases cost of capital, inhibit stock market development and leads to weak investor protection standards and practices. Similarly, the study finds overwhelming evidence that foreign bias reduces cost of capital improves stock market development and enhances investor protection standards. The evidence presented in the study demonstrates that foreign equity investment, increases liquidity and risk sharing between domestic and foreign investors. It also shows that the participation of foreign investors from well governed countries also demand good governance from the host countries.

The study might interest academics, researchers, corporate managers, market participants and governments in emerging countries. The research primary, contributes to the existing literature by providing direct empirical evidence of the role that home and foreign bias play on cost of capital, stock market development, and investor protection. The findings of the study will help policy-makers understand the need to encourage domestic investors to diversify investment abroad. Furthermore, policy-makers should develop programmes to attract foreign investors. Particularly, governments in emerging countries need to improve macroeconomic fundamentals and good governance to attract and retain foreign investors.

Appendix

Appendix 1

Emerging Portfolio Fund Research Global fund data

Fund management	Main country of	Fund manager	Fund focus	ETE	Fund size (USD million)	Percentage of the total funds
United States	United States	Franklin Templeton Investment Management	Unassigned	Y	21,012	0.175498
United States	United States	Franklin Templeton Investment Management	Unassigned	Y	611	0.005103
United States	United States	Franklin Templeton Investment Management	Unassigned	Y	7,336	0.061272
United States/Canada	United States	Franklin Templeton Investment Management	Unassigned	Y	998	0.008336
United Kingdom	United Kingdom	Henderson Global Investors	Unassigned	Y	12	0.0001
United Kingdom	United States	JPMorgan Asset Management	Unassigned	Y	174	0.001453
United Kingdom	United States	JPMorgan Asset Management	Unassigned	Y	133	0.001111
France	United Kingdom	Halbis Capital Management	Unassigned	Y	119	0.000994
United Kingdom	United Kingdom	Black Rock Investment Management	Unassigned	Y	54	0.000451
United Kingdom	United Kingdom	Invesco Asset Management	Unassigned	Y	114	0.000952
United Kingdom	United Kingdom	Aberdeen Asset Management	Unassigned	Y	85	0.00071
United Kingdom	United States	JPMorgan Asset Management	Long Funds	Y	49	0.000409
United Kingdom	United States	JPMorgan Asset Management	Unassigned	Y	200	0.00167
United States	United Kingdom	Morgan Stanley Investment Management	Unassigned	Y	357	0.002982
United States	United Kingdom	Invesco Asset Management	Unassigned	Y	392	0.003274

United States	United States	Morgan Stanley Investment Management	Unassigned	Y	78	0.000651
United States	United States	Black Rock Investment Management	Unassigned	Y	118	0.000986
United Kingdom/Canada	United States	Franklin Templeton Investment Management	Unassigned	Y	3,151	0.026318
United Kingdom	United States	Thread needle Investment Management	Unassigned	Y	461	0.00385
United States	Netherlands	Trade winds NWQ Global Investors	Unassigned	Y	212	0.001771
United States	United States	Putnam Investment Management	Unassigned	Y	952	0.007951
United States	United States	Putnam Investment Management	Unassigned	Y	2,197	0.01835
United Kingdom	United Kingdom	Aberdeen Asset Management	Unassigned	Y	409	0.003416
United Kingdom	United Kingdom	Invesco Asset Management	Unassigned	Y	83	0.000693
United Kingdom	United Kingdom	Invesco Asset Management	Unassigned	Y	19	0.000159
Canada	United Kingdom	Invesco Asset Management	Unassigned	Y	118	0.000986
United Kingdom	United States	JPMorgan Asset Management	Unassigned	Y	397	0.003316
United Kingdom	United States	JPMorgan Asset Management	Unassigned	Y	1,454	0.012144
United Kingdom	United Kingdom	Invesco Asset Management	Unassigned	Y	2,242	0.018726
United Kingdom	United Kingdom	Invesco Asset Management	Unassigned	Y	152	0.00127
United Kingdom	United Kingdom	Invesco Asset Management	Unassigned	Y	141	0.001178
United Kingdom	United Kingdom	Deutsche Asset Management	Unassigned	Y	375	0.003132
Germany	Germany	WestLB Asset Management	Unassigned	Y	28	0.000234
France	France	Amundi Investment Solutions	Unassigned	Y	148	0.001236
United States	United States	Pine Bridge Investments LLC	Unassigned	Y	430	0.003591
United States	France	BNP Paribas Investment Partners	Unassigned	Y	310	0.002589
United States	United Kingdom	Invesco Asset Management	Unassigned	Y	97	0.00081

United States	United States	Pine Bridge Investments LLC	Large Cap	Y	101	0.000844
United Kingdom	Germany	Allianz Dresdner Asset Management	Unassigned	Y	21	0.000175
United Kingdom	Germany	RCM Capital Management	Unassigned	Y	71	0.000593
France	France	Comgest S.A.	Unassigned	Y	71	0.000593
Germany	Germany	Deutsche Asset Management	Unassigned	Y	44	0.000367
United States	Germany	Allianz Global Investors	Unassigned	Y	41	0.000342
Switzerland	United States	Capital Research & Management	Unassigned	Y	1,367	0.011418
United States	United States	Black Rock Investment Management	Unassigned	Ν	879	0.007342
United States	United States	Alliance Bernstein Capital Management	Unassigned	Ν	1,489	0.012437
United Kingdom	United Kingdom	Aberdeen Asset Management	Unassigned	Ν	36	0.000301
United Kingdom	United Kingdom	Gartmore Investment Limited	Unassigned	Ν	22	0.000184
Denmark	Denmark	Jyske Invest	Unassigned	Ν	28	0.000234
United Kingdom	United Kingdom	Gartmore Investment Limited	Unassigned	Ν	42	0.000351
France	France	Comgest S.A.	Unassigned	Ν	14	0.000117
United States	United States	Marathon Asset Management	Unassigned	Ν	3,887	0.032465
United Kingdom	United Kingdom	Aberdeen Asset Management	Unassigned	Ν	909	0.007592
United States	United States	Fayez Sarofim & Co.	Unassigned	Ν	438	0.003658
United Kingdom	Australia	First State Investments	Unassigned	Ν	59	0.000493
United Kingdom	Australia	First State Investments	Unassigned	Ν	8	6.68E-05
United Kingdom	United Kingdom	Aviva Investors	Unassigned	Ν	52	0.000434
Canada	United Kingdom	HSBC Asset Management	Unassigned	Ν	99	0.000827
United Kingdom	United Kingdom	Baillie Gifford	Unassigned	Ν	198	0.001654

United Kingdom	United Kingdom	Gartmore Investment Limited	Unassigned	Ν	148	0.001236
United Kingdom	United Kingdom	Gartmore Investment Limited	Unassigned	Ν	306	0.002556
United Kingdom	United Kingdom	Henderson Global Investors	Unassigned	Ν	239	0.001996
United States	United States	Batterymarch Financial Management	Unassigned	Ν	69	0.000576
United Kingdom	United Kingdom	Schroder Investment Management	Unassigned	Ν	421	0.003516
United Kingdom	United Kingdom	Schroder Investment Management	Unassigned	Ν	535	0.004468
United Kingdom	United Kingdom	Schroder Investment Management	Unassigned	Ν	60	0.000501
United Kingdom	United Kingdom	Henderson Global Investors	Unassigned	Ν	337	0.002815
United Kingdom	United Kingdom	Martin Currie Investment Management	Unassigned	Ν	81	0.000677
Germany	Germany	Deutsche Asset Management	Unassigned	Ν	948	0.007918
Germany	Germany	Deutsche Asset Management	Unassigned	Ν	4,101	0.034253
Germany	Germany	Deutsche Asset Management	Unassigned	Ν	1,909	0.015944
Germany	Germany	Deutsche Asset Management	Unassigned	Ν	7,514	0.062759
Germany	Germany	Allianz Global Investors	Unassigned	Ν	124	0.001036
United Kingdom	United Kingdom	M&G Investment Management	Unassigned	Ν	5,791	0.048368
United States	United States	First Eagle Investment Management, LLC	Unassigned	Ν	15,232	0.127222
United States	France	BNP Paribas Investment Partners	Unassigned	Ν	51	0.000426
Netherlands	France	BNP Paribas Investment Partners	Unassigned	Ν	940	0.007851
United Kingdom	United Kingdom	Aberdeen Asset Management	Unassigned	Ν	32	0.000267
United Kingdom	Switzerland	Credit Suisse Asset Management	Unassigned	Ν	1,569	0.013105
United Kingdom	United States	Black Rock Investment Management	Unassigned	Ν	290	0.002422
United Kingdom	Switzerland	Pictet Asset Management	Unassigned	Ν	86	0.000718

Switzerland	Switzerland	Pictet Asset Management	Unassigned	Ν	322	0.002689
United Kingdom	United Kingdom	Schroder Investment Management	Unassigned	Ν	180	0.001503
United Kingdom	United States	JPMorgan Asset Management	Large Cap	Ν	351	0.002932
United Kingdom	United States	JPMorgan Asset Management	Large Cap	Ν	10	8.35E-05
United Kingdom	United States	JPMorgan Asset Management	Large Cap	Ν	1,378	0.011509
Luxemburg	Denmark	Linde Partners Asset Management	Unassigned	Ν	355	0.002965
Finland	Finland	Glitnir Asset Management	Unassigned	Ν	105	0.000877
United Kingdom	United Kingdom	Aberdeen Asset Management	Unassigned	Ν	235	0.001963
Luxemburg	Germany	Union Investment GmbH	Unassigned	Ν	230	0.001921
Germany	United States	Black Rock Investment Management	Large Cap Blend	Ν	67	0.00056
United States	United States	Batterymarch Financial Management	Unassigned	Ν	142	0.001186
United States	United States	William Blair & Co.	Unassigned	Ν	38	0.000317
United States	United States	Artisan Partners	Unassigned	Ν	27	0.000226
Singapore	Singapore	Fullerton Fund Management	Unassigned	Ν	417	0.003483
United States	United States	Black Rock Investment Management	Unassigned	Ν	938	0.007834
United Kingdom	United Kingdom	Martin Currie Investment Management	Unassigned	Ν	97	0.00081
United States	United States	Global Currents Investment Management	Unassigned	Ν	1	8.35E-06
United Kingdom	United States	Newton Investment Management	Unassigned	Ν	4	3.34E-05
France	United States	State Street Global Advisors	Unassigned	Ν	559	0.004669
United States	United States	Institutional Capital LLC	Unassigned	Ν	48	0.000401
United Kingdom	United Kingdom & France	AXA Framlington Investment Management	Unassigned	Ν	340	0.00284

United Kingdom	United Kingdom	M&G Investment Management	Unassigned	Ν	91	0.00076
Canada	Canada	Goodman & Company, Investment Counsel	Unassigned	Ν	613	0.00512
Canada	Canada	Goodman & Company, Investment Counsel	Unassigned	Ν	739	0.006172
United States	United States	The Boston Company Asset Management	Unassigned	Ν	29	0.000242
United Kingdom	United Kingdom	Baillie Gifford	Unassigned	Ν	172	0.001437
United Kingdom	United States	Walter Scott & Partners	Unassigned	Ν	84	0.000702
Netherlands	France	BNP Paribas Investment Partners	Unassigned	Ν	38	0.000317
United Kingdom	Switzerland	Global Asset Management	Unassigned	Ν	815	0.006807
United Kingdom	United States	Assenagon Asset Management	Unassigned	Ν	87	0.000727
United Kingdom	United Kingdom	M&G Investment Management	Unassigned	Ν	3,554	0.029684
United Kingdom	United Kingdom	Aberdeen Asset Management	Unassigned	Ν	77	0.000643
United Kingdom	United Kingdom	Baillie Gifford	Unassigned	Ν	2,909	0.024297
United Kingdom	United Kingdom	Baillie Gifford	Unassigned	Ν	1,487	0.01242
United States	United States	John Hancock Asset Management	Unassigned	Ν	832	0.006949
United States	United States	AQR Capital Management	Unassigned	Ν	405	0.003383
Norway	Norway	DnB NOR Asset Management	Large Cap Blend	Ν	165	0.001378
United States	United States/Australia	Trilogy Global Advisors	Unassigned	Ν	28	0.000234
AUnited Statestria	Austria	Kepler-Fonds KAGm.b.H.	Large Cap Value	Ν	65	0.000543
Norway	Norway	SKAGEN AS	Large Cap Value	Ν	5,595	0.046731
Norway	Norway	DnB NOR Asset Management	Large Cap Blend	Ν	222	0.001854

ETF is exchange-traded fund

_

Append	lix	2
--------	-----	---

Developed	Emerging
Australia	Argentina
Austria	Brazil
Belgium	Bulgaria
Canada	Chile
Denmark	China
Finland	Czech Republic
France	Egypt
Germany	Hungary
Greece	India
Hong Kong	Indonesia
Ireland	Korea
Israel	Malaysia
Italy	Mexico
Japan	Peru
Netherlands	Philippines
New Zealand	Poland
Norway	Romania
Portugal	Russia
Spain	South Africa
Sweden	Thailand
Switzerland	Turkey
United Kingdom	
United States	

Developed and Emerging Markets based on MSCI definition

References

- Abiad, A., Detragiache, E. and Tressel, T. (2008), "A New Database of Financial Reform", *IMF Working Paper* 08/266.
- Adler, M. and Dumas, B. (1983), "International Portfolio Choice and Corporation Finance: A Synthesis", *Journal of Finance*, vol. 38, no. 3, pp. 925-984.
- Adler, M. and Dumas, B. (1984), "Exposure to currency risk: definition and measurement", *Financial Management*, vol. 13, no. 2, pp. 41–50.
- Agarwal, P. (2007), "Institutional ownership and stock liquidity", *Working Paper*, Cornell University.
- Aggarwal, R., Erel, I., Ferreira, M. and Matos, P. (2011), "Does governance travel around the world? Evidence from institutional investors", *Journal of Financial Economics*, vol. 100, no.1, pp. 154-181.
- Ahearne, A., Griever, W. and Warnock, F. (2004), "Information costs and home bias: an analysis of US holdings of foreign equities", *Journal of International Economics*, vol. 62, no. 2, pp. 313-336.
- Alesina, A. and Perotti, R. (1996), "Income distribution, political instability and investment", *European Economic Review*, vol. 40, no. 6, pp. 1203-1228.
- Ali, A., Durtschi, C., Lev, C. B. and Trombley, M. (2004), "Changes in Institutional Ownership and Subsequent Earnings Announcement Abnormal Returns", *Journal of Accounting, Auditing, and Finance*, vol. 19, no. 3, pp. 221–248.
- Amihud, Y. and Mendelson, H. (1986), "Asset pricing and bid-ask spread", *Journal of Financial Economics*, vol. 17, pp, 223-249.
- Andres, A.R. (2006), "Software piracy and income inequality", *Applied Economics Letters*, vol. 13, no. 2, pp. 101-105.
- Apodaca, C. (2004), "The rule of law and human rights", *Judicature*, vol. 87, no. 6, pp. 292-299.
- Arndt, C. and Oman, C. (2006), "Uses and abuses of governance indicators", Paris: OECD Development Studies Centre.
- Atje, R., and Jovanovic, B. (1993), "Stock markets and development", *European Economic Review* vol. 37, no. 2-3, pp. 632-640.
- Back, K., Cao, C. H. and Willard, G. A. (2000), "Imperfect competition among informed traders", *Journal of Finance*, vol. 55, no. 5, pp. 2117-2155.

- Bae, K.H., Warren, B. and Mao, C. X. (2006), "Stock market liberalization and the information environment", *Journal of International Money and Finance*, vol. 25, no. 3, pp. 404-428.
- Bailey, W. and Jagtiani, J. (1994), "Foreign ownership restrictions and stock prices in the Thai capital market", *Journal of Financial Economics*, vol. 36, no. 1, pp. 57-87.
- Baker, M. and Wurgler, J. (2000), "The equity share in new issues and aggregate stock returns", *Journal of Finance*, vol. 55, no. 5, pp. 2219-2257.
- Baltagi, B. H. (2001), "Econometric Analysis of Panel Data", Wiley and Sons Ltd.
- Bartram, S. M. and Dufey, G. (2001), "International portfolio investment: theory, evidence, and institutional framework", *Financial Markets, Institutions & Instruments*, vol. 10, no. 3, pp. 85-155.
- Baxter, M. and Jermann, U. J. (1997), "The International Diversification Puzzle is Worse than You Think", *American Economic Review*, vol. 87, no. 1, pp. 170-180.
- Bebchuk, L. A. and Cohen, A. (2005), "The Costs of Entrenched Boards", *Journal of Financial Economics*, vol. 78, no. 2, pp. 409-433.
- Bebchuk, L., Cohen, A. and Ferrell, A. (2009), "What Matters in Corporate Governance?" *Review of Financial Studies*, vol. 22, no. 2, pp. 783-827.
- Becht, M., Franks, J. and Grant, J. (2008), "The returns to public and private activism in Europe", *Unpublished working paper*, London Business School.
- Beck, T., Demirguc-Kunt, A. and Levine, R. (2000a), "A new database on financial development and structure", *World Bank Economic Review*, vol. 14, no. 3, pp. 597-605.
- Beck, T., Demirgüç-Kunt, A. and Levine, R. (2003), "Law, Endowments, and Finance", *Journal of Financial Economics*, vol. 70, no. 2, pp. 137-181.
- Beck, T., Levine, R. and Loayza, N. (2000b), "Finance and the sources of growth", *Journal* of Financial Economics, vol. 58, no. 1-2, pp. 261-300.
- Beck, T., Lundberg, M. and Majnoni, G. (2006), "Financial intermediary development and growth volatility: do intermediaries dampen or magnify shocks?" *Journal of International Money and Finance*, vol. 25, no. 7, pp. 1146-1167.
- Bekaert, G. (1995), "Market integration and investment barriers in emerging equity markets", *The World Bank Economic Review*, vol. 9, no. 1, pp. 75-107.
- Bekaert, G. and Harvey, C. R. (1995), "Time-Varying World Market Integration", *Journal of Finance*, vol. 50, no. 2, pp. 403-444.
- Bekaert, G. and Harvey, C. R. (1998), "Capital flows and the behavior of emerging market equity returns", *Unpublished Working Paper 6669*. National Bureau of Economic Research, Cambridge, MA.

- Bekaert, G. and Harvey, C. R. (2000), "Foreign speculators and emerging equity markets," *Journal of Finance*, vol. 55, no. 2, pp. 565-613.
- Bekaert, G. and Harvey, C. R. (2005), "Chronology of important economic, financial and political events in emerging markets", http://www.duke.edu/~charvey/chronology.htm.
- Bekaert, G., Harvey, C. R. and Lumsdaine, R. L. (2002), "Dating the integration of world equity markets", *Journal of Financial Economics*, vol. 65, no. 2, pp. 203-247.
- Bekaert, G., Harvey, C. R. and Lundblad, C. (2001), "Emerging equity markets and economic growth", *Journal Development Economics*, vol. 66, no. 2, pp. 465–504.
- Bekaert, G., Harvey, C. and Lundblad, R. (2005), "Does Financial Liberalization Spur Growth?" *Journal of Financial Economics*, vol. 77, no. 1, pp. 3-56.
- Bekaert, G., Harvey, C. R. and Lundblad, C. (2006), "Growth volatility and financial liberalization", *Journal of International Money and Finance*, vol. 25, no. 3, pp. 370-403.
- Bekaert, G., Harvey, C. R. and Lundblad, C. (2007), "Liquidity and expected returns: Lessons from emerging markets", *Review of Financial Studies*, vol. 20, no. 6, pp. 1783-1831.
- Bekaert, G. and Hodrick, R. (2009). International financial management. Pearson Prentice Hall. New Jersey.
- Bekaert, G., Hodrick, R. and Zhang, X. (2009), "International stock return co-movements", *Journal of Finance*, vol. 64, no.6, pp. 2591-2626.
- Bekaert, G. and Urias, M. (1996), "Diversification, integration, and emerging market closedend funds", *Journal of Finance*, vol. 51, no. 3, pp. 835-869.
- Bekaert, G. and Wang, X. (2010), "Inflation risk and the inflation premium", *Economic policy*, vol. 25, no. 64, pp. 755-806.
- Bencivenga, V. R. and Smith, B. D. (1992), "Deficits, inflation, and the banking system in developing countries: The optimal degree of financial repression", *Oxford Economic Papers*, vol. 44, no. 4, pp. 767-790.
- Bencivenga, V. R., Smith, B. D. and Starr, R. M. (1996), "Liquidity of secondary capital markets: Allocative efficiency and the maturity composition of the capital stock", *Economic Theory*, vol. 7, no. 1, pp. 19-50.
- Beugelsdijk, S. and Frijns, B. (2010), "A cultural explanation of the foreign bias in international asset allocation", *Journal of Banking and Finance*, vol. 34, no. 9, pp. 2121-2131.
- Bhattacharya, U. and Daouk, H. (2002), "The world price of insider trading", *Journal of Finance*, vol. 57, no. 1, pp. 75–108.

- Billmeier, A. and Massa, I. (2009), "What drives stock market development in emerging markets- institutions, remittances, or natural resources?" *Emerging Markets Review*, vol. 10, no. 1, pp. 23-35.
- Boehmer, E., Saar, G. and Yu, L. (2005), "Lifting the veil: an analysis of pre-trade transparency at the NYSE", *Journal of Finance*, vol. 60, no. 2, pp. 783–815.
- Bohn, H. and Tesar, L. (1996), "U.S. equity investment in foreign markets: Portfolio rebalancing or return chasing?" *American Economic Review*, vol. 86, no. 2, pp. 77-81.
- Bolton, P. and Thadden, V. (1998), "Block, Liquidity and Corporate Governance", *Journal of Finance*, vol. 53, no. 1, pp. 1-25.
- Bonaglia, F., de Macedo, J. B. and Bussolo, M. (2001). "How globalisation improves governance", *Technical Papers No. 181. OECD Development Centre* (Paris: OECD).
- Bose, N. (2005), "Endogenous Growth and the Emergence of Equity Finance, *Journal of Development Economics*", vol. 77, no. 1, pp. 173-188.
- Boyd, J. H., Levine, R. and Smith, B. (2001), "The impact of inflation on financial sector performance", *Journal of Monetary Economics*, vol. 47, no. 2, pp. 221–248.
- Boyer, B. H., Kumagai, T. and Yuan, K. (2006), "How Do Crises Spread? Evidence from Accessible and Inaccessible Stock Indices", *Journal of Finance*, vol. 61, no. 2, 957-1003.
- Brandt, M. W. and Wang, K. Q. (2003), "Time-varying risk aversion and unexpected inflation", *Journal of Monetary Economics*, vol. 50, no. 7, pp. 1457-1498.
- Bris, A. and Cabolis, C. (2004), "Adopting Better Corporate Governance: Evidence from Cross- Border Mergers", *Yale International Center for Finance*, working paper.
- Brockman, P. and Yan, X. (2009), "Block ownership and firm-specific information", *Journal* of Banking and Finance, vol. 33, no. 2, pp. 208-316.
- Broner, F., Gelos, R. G. and Reinhart, C. (2006), "When in peril, retrench: Testing the portfolio channel of contagion", *Journal of International Economics, vol.* 69, no. 1, pp. 203-230.
- Brunetti, A., Kisunko, G. and Weder, B. (1998), "Credibility of rules and economic growth: evidence from a worldwide survey of the private sector", *World Bank Economic Review*, vol. 12, no. 3, pp. 353-384.
- Brunetti, A. and Weder, B. (2003). "A Free Press is Bad News for Corruption", *Journal of Public Economics*, vol. 87, no. 7-8, pp. 1801-1824.
- Bushee, B. and Goodman, T. (2007), "Which Institutional Investors Trade Based on Private Information about Earnings and Returns?" *Journal of Accounting Research*, vol. 45, no. 2, pp. 289-321.

Business Week. (2006). Special Report: Fidelity's Divided Loyalties.

- Busse, M. and Gröning, S. (2007), "Does Trade Liberalisation Lead to Better Governance? An Analysis of the Proposed ACP/EU Economic Partnership Agreements", *Hamburg Institute of International Economics* (HWWI).
- Busse, M. and Gröning, S. (2009), "Does foreign aid improve governance?" *Economics Letters*, vol. 104, no. 2, pp. 76-78.
- Calderon C. and Liu, L. (2002), "The direction of causality between financial development and economic growth", Working Papers, Central Bank of Chile.
- Cameron, R. (1967), *Banking in the Early Stages of Industrialization*. New York: Oxford University Press.
- Carlin, W. and Mayer, C. (2003), "Finance, investment, and growth", *Journal of Financial Economics*, vol. 69, no. 1, pp. 191–226.
- Carrieri, F., Errunza, V. and Majerbi, B. (2006), "Does emerging market exchange risk affect global equity prices?" *Journal of Financial and Quantitative Analysis*, vol. 41, no. 3, pp. 511-540.
- Castro, R., Clementi, G. L. and MacDonald, G. (2004), "Investor protection, optimal incentives, and economic growth", *The Quarterly Journal of Economics*, vol. 119, no. 3, pp. 1131-1175.
- Chan, K., Covrig, V. and Ng, L. (2005), "What determines the domestic bias and foreign bias? Evidence from mutual fund equity allocations world- wide", *Journal of Finance*, vol. 60, no. 3, pp. 1495-1534.
- Chan, K., Covrig, V. and Ng, L. (2009), "Does home bias affect firm value? Evidence from holdings of mutual funds worldwide, "*Journal of International Economics*. vol. 78, no. 2, pp. 230-241.
- Chan, K. C., Karolyi, G. A, and Stulz, R. M, (1992), "Global financial markets and the risk premium on U.S. equity", *Journal of Financial Economics*, vol. 32, no. 2, pp. 137-167.
- Chari, A. and Henry, P. (2004a), Is the Invisible Hand Discerning or Indiscriminate? Investment and Stock Prices in the Aftermath of Capital Account Liberalizations. NBER Working Paper No. 10318.
- Chari, A. and Henry, P. (2004b), "Risk sharing and asset prices: Evidence from a natural experiment", *Journal of Finance*, vol. 59, no. 3, pp. 1295-1324.
- Chen, K. C. W., Chen, Z. and Wei, K. C. J. (2009), "Legal protection of investors, corporate governance, and the cost of equity capital", *Journal of Corporate Finance*, vol. 15, no. 3, pp. 273–289.
- Chen, X., Harford, J. and Li, K. (2007), "Monitoring: Which institutions matter?" *Journal of Financial Economics*, vol. 86, no. 2, pp. 279–305.

- Choe, H., Kho, B. C. and Stulz, R. M. (1999), "Do Foreign Investors Destabilize Stock Markets?" *Journal of Financial Economics*, Vol. 54, no. 2, pp. 227-264.
- Choe, H., Kho, B. C. and Stulz, R. M. (2000), "Do domestic investors have more valuable information about individual stocks than foreign investors?" *Working paper*, Seoul National University and Ohio State University.
- Choe, H., Kho, B. C. and Stulz, R. M. (2005), "Do domestic investors have an edge? The trading experience of foreign investors in Korea", *Review of Financial Studies*, vol. 18, no. 3, pp. 795-829.
- Chong, A. and Zanforlin, L. (2000), "Law tradition and institutional quality", *Journal of International Development*, vol. 12, no. 8, pp. 1057–1068.
- Chordia, T., Roll, R. and Subrahmanyam, A. (2011), "Recent trends in trading activity and market quality", *Journal of Financial Economics*, vol. 101, no. 2, pp. 243-263.
- Chou, R. K., Ho, K. Y. and Weng, P. S. (2012), Liquidity, market returns, and foreign institutional trading. [Online] Available from: http://asianfa2012.mcu.edu.tw/fullpaper/10459.pdf [Accessed 24 March 2013].
- Claessens, S., Djakov, S., Fan, J. and Lang, L. (1999), "Expropriation of minority shareholders: evidence from East Asia", *Policy Research paper 2088*. World Bank, Washington DC.
- Claessens, S., Djankov, S. and Lang, L. H. P. (2000), "The separation of ownership and control in East Asian corporations", *Journal of Financial Economics*, vol. 58, no. 1–2, pp. 81-112.
- Claessens, S., Klingebiel, D. and Scmukler, S. L. (2006), "Stock market development and internationalization: Do economic fundamentals spur both similarly?" *Journal of Empirical Finance*, vol. 13, no. 3, pp. 316-350.
- Coffee, J. C. (1999), "The future as history: the prospects for global convergence in corporate governance and its implications", *Northwestern University Law Review*, vol. 93, pp. 641–707.
- Coffee, J. C. (2000). "Privatization and Corporate Governance: The Lessons from Securities Market Failure", *Unpublished working paper*, 158, Columbia Law School, New York.
- Collier, P., Hoeffler, A. and Pattillo, C. (1999), "Flight Capital as a Portfolio Choice", *IMF Working Paper 99/171*, full text on www.imf.org.
- Cooper, I. A. and Kaplanis, E. (1986), Costs to cross-border investment and international equity market equilibrium. In: Edwards, A. (Ed.), *Recent Advances in Corporate Finance*. Cambridge University Press.
- Cooper, I. A. and Kaplanis, E. (1994), "Home Bias in Equity Portfolios, Inflation Hedging, and International Capital Market Equilibrium", *Review of Financial Studies*, vol. 7, no. 1, pp. 45-60.

- Core, J. E., Guay, W. R. and Rusticus, T. O. (2006), "Does weak governance cause weak stock returns? An examination of firm operating performance and investors' expectations", *Journal of Finance, vol.* 61, no. 2, pp. 655-687.
- Cremers, K. J. M. and Nair, V. B, (2005), "Governance Mechanisms and Equity Prices", *Journal of Finance*, vol. 60, no. 6, pp. 2859-2894.
- Dahlquist, M., Pinkowitz, L., Stulz, R. M. and Williamson, R. (2003), "Corporate governance and the home bias", *Journal of Financial & Quantitative Analysis*, vol. 38, no. 1, pp. 87-110.
- Dahya, J., Dimitrov, O. and McConnell, J. J. (2008), "Dominant shareholders, corporate boards and corporate value: a cross-country analysis", *Journal of Financial Economics*, vol. 87, no. 1, pp. 73-100.
- Damodaran, A. (2012), Equity risk premium (ERP): Determinants, estimation and implications, 2012 Edition. [Online] Available from: (http://papers.ssrn.com/sol3/papers.cfm?abstract_id=2027211 [Accessed: 21 May 2012].
- Daouk, H., Lee, C. M. C. and Ng, D. (2006), "Capital market governance: How do security laws affect market performance?" *Journal of Corporate Finance*, vol. 12, no. 3, pp. 560 –593.
- Das, G. G. and Andriamananjara, S. (2006), "Hub-and-spokes free trade agreements in the presence of technology spillovers: An application to the western hemisphere", *Review of World Economics*, vol. 142, no. 1, pp. 33–66.
- Davis, G. and Kim, E. H. (2007), "Business Ties and Proxy Voting by Mutual Funds", *Journal of Financial Economic*, vol. 85, no. 2, pp. 552–70.
- Demirguc-Kunt, A. and Maksimovic, V. (1996), "Stock market development and financial choices of firms", *World Bank Economic Review*, vol. 10, no. 2, pp. 341-369.
- Demirguc-Kunt, A. and Maksimovic, V. (1998), "Law, finance, and firm growth", *Journal of Finance*, vol. 53, no. 6, pp. 2107-2137.
- Demirguc-Kunt, A. and Maksimovic, V. (1999), "Institutions, financial markets and firm debt maturity", *Journal of Financial Economic Review*, vol. 54, no. 3, pp. 295-336.
- De Jong, F. and de Roon, F. (2005), "Time-varying market integration and expected returns in emerging markets", *Journal of Financial Economics*. vol. 78, no. 3, pp. 583-613.
- De La Torre, A., Gozzi, J. C. and Schmukler, S. L. (2007), "Stock market development under globalization: Whither the gains from reforms?" *Journal of Banking and Finance*, vol. 31, no. 6, pp. 1731–1754.
- Dennis, P. and Weston, J. (2001), "Who's Informed? An Analysis of Stock Ownership and Informed Trading", Working paper, University of Virginia.

- De Roon, F. A., Nijman, T. E. and Werker, B. J. M. (2001), "Testing for mean variance spanning with short sales constraints and transaction costs: The case of emerging markets", *Journal of Finance*, vol. 56, no. 2, pp. 721–742.
- DeSantis, G. and Gerard, B. (1997), "International asset pricing and portfolio diversification with time-varying risk", *Journal of Finance*, vol. 52, no. 5, pp. 1881-1913.
- Devereux, M. B. and Smith, G. W. (1994), "International Risk Sharing and Economic Growth", *International Economic Review*, vol. 35, no. 3, pp. 535-50.
- Diamond, D. W. and Verrecchia, R. E. (1982), "Optimal managerial contracts and equilibrium security prices", *Journal of Finance*, vol. 37, no. 2, pp. 275-287.
- Dimson, E., Marsh, P. and Staunton, M. (2002), "Triumph of the Optimists: 101 Years of Global Investment Return", Princeton University Press.
- Djankov, S., Hart, O. D., McLiesh, C. and Shleifer, A. (2008), "Debt enforcement around the world", *Journal of Political Economy* 116, 1105-1149.
- Djankov, S., McLiesh, C., Nenova, T. and Schleifer, A. (2001). "Who Owns the Media?" Mimeo, background paper prepared for the WDR 2001 on Market Institutions.
- Djankov, S., Caralee McLiesh, C. and Andrei Shleifer. A. (2007), "Private Credit in 129 Countries", *Journal of Financial Economics*, vol. 84, no. 2, 299–329.
- Do, Q. T. and Levchenko, A. A. (2004), "Trade and Financial Development", Paper presented at LACEA, San José, Costa Rica.
- Doidge, C., Karolyi, G.A. and Stulz, R.M. (2004), "Why are countries so important for the corporate governance?" *Working paper*. Ohio State University.
- Doidge, C., Karolyi, G. A. and Stulz, R. M. (2007), "Why do countries matter so much for corporate governance?" *Journal of Financial Economics* 86, no. 1, pp. 1–39.
- Dollar, D. and Kraay, A. (2003), "Institutions, trade and growth", *Journal of Monetary Economics*, vol. 50, no. 1, pp. 133-162.
- Domowitz, I., Glen, J. and Madhavan, A. (1998), "International cross-listing and order flow migration: evidence from an emerging market", *Journal of Finance, vol. 53*, no. 6, pp. 2001–2027.
- Douma, S., George, R. and Kabir, R. (2006), "Foreign and domestic ownership, business groups and firm performance", *Strategic Management Journal*, vol. 27, no. 7, pp. 637–657.
- Driessen, J. and Laeven, L. (2007), "International portfolio diversification benefits: Crosscountry evidence from a local perspective", *Journal of Banking and Finance*, vol. 31, no. 6, 1693-1712.
- Durbin, J. and Watson, G. S. (1950), "Testing for serial correlation in least squares regression", *Biometrika*, vol. 37, nos. 3-4, pp. 409-428.
- Durham, J. B. (2004), "Absorptive Capacity and the Effects of Foreign Direct Investment and Equity Foreign Portfolio Investment on Economic Growth", *European Economic Review*, vol. 48, no. 2, pp. 285-306.
- Durnev, A. and Kim, E. H. (2005), "To Steal or Not to Steal: Firm Attributes, Legal Environment, and Valuation", *Journal of Finance*, vol. 60, no. 3, pp. 1461-1493.
- Dyck, A. and Zingales, L. (2002), "The Corporate Governance Role of the Media", in R. Islam ed. The right to tell: The role of the Media in Development," *The World Bank*, Washington DC.
- Dyck, A, and Zingales, L. (2004), "Private Benefits of Control: An International Comparison", *Journal of Finance*, vol. 59, no. 2, pp. 537-600.
- Economist, (2008), German corporate governance: raising their voices. March 19.
- Edison, H. J. and Warnock, F. E. (2008), "Cross-border listings, capital controls, and equity flows to emerging markets", *Journal of International Money and Finance*, vol. 27, no. 6, pp. 1013-27.
- Eichengreen, B. (2002), "Capital account liberalization: what do the cross-country studies tell us?" *World Bank Economic Review*, vol. 15, no. 3, pp. 341-365.
- Eleswarapu, V. R. and Venkataraman, K. (2006), "The impact of legal and political institutions on equity trading costs: A cross-country analysis", *Review of Financial Studies*, vol. 19, pp. 1081-1111.
- Engelen, P. J. and Van Essen, M. (2010), "Underpricing of IPOs: Firm-, issue- and country-specific characteristics", *Journal of Banking and Finance*, vol. 34, no. 8, pp. 1958-1969.
- Erb, C., Harvey, C. R. and Viskanta, T. (1996), "Expected returns and volatility in 135 countries", *Journal of Portfolio Management*, vol. 22, no. 3, pp. 46-58.
- Ergungor, E. (2008), "Financial system structure and economic growth: Structure matters", *International Review of Economics and Finance*, vol. 17, no. 2, pp. 292-305.
- Errunza, V. (2001), "Foreign portfolio equity investments, financial liberalization, and economic development", Review of International Economics, vol. 9, no. 4, pp. 703-726.
- Errunza, V., Hogan, K. and Hung, M.W. (1999), "Can the gains from international diversification be achieved without trading abroad?" *Journal of Finance*, vol. 54, no. 6, pp. 2075–2108.
- Errunza, V. and Losq, E. (1985), "International asset pricing and mild segmentation: Theory and test", *Journal of Finance*, vol. 40, no. 1, pp. 105-124.

- Errunza, V. and Losq, E. (1989), "Capital flow controls, international asset pricing, and investors' welfare: A multi-country framework", *Journal of Finance*, vol. 44, no. 4, pp. 1025-1038.
- Errunza, V. and Miller. D. P. (1998), "Market segmentation and the cost of capital in international equity markets", *Working paper*, McGill University and Texas A&M University.
- Errunza, V. and Miller, D. P. (2000), "Market segmentation and the cost of capital in international equity markets", *Journal of Financial and Quantitative Analysis*, vol. 35, no. 4, pp. 577-600.
- Errunza, V. and Rosenberg, B. (1982), "Investment risk in developed and less developed countries", *Journal of Financial and Quantitative Analysis*, vol. 17, no. 5, pp. 741-762.
- Eun, C. S. and Janakiramanan, S. (1986), "A model of international asset pricing with a constraint on the foreign equity ownership", *Journal of Finance*, vol. 41, no. 4, pp. 897-914.
- Eun, C. S. and Resnick, B. (1994), "International Diversification of Investment Portfolios: US and Japanese Perspective", *Management Science*, vol. 40, no. 1, pp. 140-161.
- Evans, J. L. and Archer, S. H. (1968), "Diversification and the Reduction of Dispersion: An Empirical Analysis", *Journal of Finance*, vol. 23, no. 5, pp. 761-767.
- Fama, E. F. and French, K. R. (1992), "The cross-section of expected stock returns", *Journal* of *Finance*, vol. 47, no. 2, pp. 427-465.
- Fama, E. F. and French, K. R. (1993), "Common risk factors in the returns on stocks and bonds", *Journal of Financial Economics*, vol. 33, no. 1, pp. 3-56.
- Ferreira, M., Massa, M. and Matos, P. (2010), "Shareholders at the gate? institutional investors and cross-border mergers and acquisitions", *Review of Financial Studies*, vol. 23, pp. 601-644.
- Fidora, M., Fratzscher, M. and Thimann, C. (2007), Home bias in global bond and equity markets; The role of real exchange rate volatility. *Journal of International Money and Finance*, vol. 26, no. 4, pp. 631-655.
- Ferreira, M. and Matos, P. (2008), "The colors of investors' money: the role of institutional investors around the world", *Journal of Financial Economics*, vol. 88, no. 3, pp. 499-533.
- Firth, M., Fung, P. M. Y. and Rui, O. M. (2008), "Ownership, governance mechanisms and agency costs in China listed firms", *Journal of Asset Management*, vol. 9, pp. 90-101.
- Fischer, S. (1998), "Capital Account Liberalization and the Role of the IMF", *Princeton Essays in International Finance*, pp. 1-10.

- Fischer, S. (2003), "Globalization and its challenges", *American Economic Review*, vol. 93, no. 2, pp. 1-30.
- Fisman, R. J. and Gatti, R. (2002),"Decentralization and Corruption: Evidence across Countries", *Journal of Public Economics*, vol. 83: no. 3, pp. 325-345.
- Fletcher, J. and Marshall, A. (2005), "An empirical examination of the benefits of international diversification", *Journal of International Markets, Institutions and Money*, vol. 15, no. 5, pp. 455-468.
- Foerster, S. and Karolyi, G. A. (1999), "The effects of market segmentation and investor recognition on asset prices: evidence from foreign stocks listing in the United States", *Journal of Finance*, vol. 54, no, 3, pp. 981-1014.
- Forbes, K. (2010), "Why do Foreigners Invest in the United States?" *Journal of International Economics*, vol. 80, no. 1, pp. 3-21.
- Foster, F. D. and Viswanathan, S. (1996), "Strategic trading when agents forecast the forecasts of others", Journal of Finance, vol. 51, no. 4, pp. 1437-1478.
- French, K. and Poterba, J. (1991), "Investor Diversification and International Equity Markets", *American Economic Review, vol.* 81, no. 2, pp. 222-226.
- Friedman, E., Johnson, S. and Mitton, T. (2003), "Intra-group capital flow and propping", *Journal of Comparative Economics*, vol. 31, pp. 732-750.
- Froot K. and Ramadorai T. (2008), "Institutional Portfolio Flows and International Investments", *The Review of Financial Studies*, vol. 21, no. 2, pp. 937-971.
- Galindo, A., Micco, A. and Ordoñez, G. (2002), "Financial Liberalization: Does it Pay to Join the Party?" *Economia*, vol. 3, no. 1, pp. 231-261.
- Galindo, A., Schiantarelli, F. and Weiss, A. (2001), "Does Financial Liberalization Improve the Allocation of Investment", *Unpublished working paper*, Boston College.
- Garcia, V. F. and Liu, L. (1999), "Macroeconomic determinants of stock market development", *Journal of Applied Economics*, vol. 2, no. 1, pp. 29-59.
- Garretsen, H., Lensink, R. and Sterken, E. (2004), "Growth, financial development, societal norms and legal institutions", *Journal of International Financial Markets, Institutions and Money*, vol. 14, no. 2, pp. 165-183.
- Gebhardt, W.R., Lee, C. M. and Swaminathan, B. (2001), "Toward an implied cost of capital", *Journal of Accounting Research*," vol. 39, no. 1, pp. 135-176.
- Gelb A. H. (1989), "Financial Policies, Growth, and Efficiency", Policy Planning, and Research", *Working Paper*, No. 202, World Bank.
- Gelos, R. G. and Wei, S. J. (2005), "Transparency and international portfolio holdings", *Journal of Finance*, vol. 60, no. 6, pp. 2987-3020.

- Ghani, E. (1992), "How financial markets affect long-run growth: A cross-country study", WPS 843, World Bank, operations evaluation department, Washington, D. C. Processed.
- Giannetti, M. and Koskinen, Y. (2004), "Investor protection and the demand for equity", SSE/EFI Working paper series in Economics and Finance No. 526.
- Giannetti, M. and Koskinen, Y. (2010), "Investor protection, equity returns and financial globalization", *Journal of Financial and Quantitative Analysis*, vol. 45, no. 1, pp. 135-168.
- Giannetti, M and Laeven, L. (2009), "Pension reform, ownership structure, and corporate governance: Evidence from a Natural Experiment", *Review of Financial Studies*, vol. 22, no. 10, pp. 4091-4127.
- Giannetti, M. and Simonov, A. (2006), "Which investors fear expropriation? evidence from investors' portfolio choices", *Journal of Finance*, vol. 61, no. 3, pp. 1507–1547.
- Gibson, R. and Mougeot, N. (2004), "The pricing of systematic liquidity risk: Empirical evidence from the US stock market", *Journal of Banking and Finance*, vol. 28, no. 1, pp. 157-178.
- Gillan, S. and Starks, L. (2003), "Corporate Governance, Corporate Ownership, and the Role of Institutional Investors: A Global Perspective", *Journal of Applied Finance, vol.* 13, pp. 4-22.
- Glaeser, E. L. and Saks, R. E. (2006), "Corruption in America", Journal of Public Economics, vol. 90, no. 6-7, pp. 1053-1072.
- Glosten, L. R. and Milgrom, P. R. (1985), "Bid, ask and transaction prices in a specialist market with heterogeneously informed traders", *Journal of Financial Economics*, vol. 14, no. 1, pp. 71-100.
- Goetzmann, W., Ingersoll, J., Spiegel, M. and Welch, I. (2002), "Sharpening Sharpe Ratios", Working paper, NBER.
- Gompers, P., Ishii, J. and Metrick, A. (2003), "Corporate Governance and Equity Prices", *Quarterly Journal of Economics*, vol. 118, pp. 107–55.
- Gourinchas, P. and Jeanne, O. (2002), "On the Benefits of Capital Account Liberalization", *Princeton University Working Paper*.
- Green, W. (2007), "Discrete Choice Modeling", Handbook of Econometrics, vol. 2, Part 4.2, edited by T. Mills and K. Patterson London: Palgrave.
- Griffin, J. M., Nardari, F. and Stulz, R. M. (2002), "Daily cross-border equity flows: Pushed or pulled?" *Working paper*, National Bureau of Economic Research, no. 900.
- Grinblatt, M. and Keloharju, M. (2000), "The investment behaviour and performance of various investor types: A study of Finland's unique data set", *Journal of Financial Economics*, vol. 55 no. 1, pp. 43-67.

- Grinblatt, M. and Keloharju, M. (2004), "Tax-loss trading and wash sales", *Journal of Financial Economics*, vol. 71, no. 1, pp. 51-76.
- Grossman, S. J. and Hart, O. D. (1980), "Disclosure laws and takeover bids", *Journal of Finance*, vol. 35, pp. 323-334.
- Grossman, S. J. and Stiglitz, J. E. (1980), "On the impossibility of informationally efficient markets", *American Economic Review*, vol. 70, no. 3, pp. 393-408.
- Grubel, H. G. (1968), "Internationally diversified portfolios", *The American Economic Review*, vol. 58, no. 5, pp. 1299-1314.
- Guercio, D. D. and Hawkins, J. (1999), "The motivation and impact of pension fund activism", *Journal of Financial Economics*, vol. 52, no. 3, pp. 293-340.
- Guiso, L., Haliassos, M. and Jappelli, T. (2001), "Household portfolios", MIT Press Cambridge, Mass.
- Guiso, L., Haliassos, M. and Jappelli, T. (2003), "House stockholding in Europe: Where do we stand and where do we go?" *Economic Policy*, vol. 18, no. 36, pp. 123-170.
- Gujarati, D. (2003), Basic Econometrics (4th ed). New York: McGraw-Hill.
- Gul, F. A., Kim, J. B. and Qiu, A. (2010), "Ownership concentration, foreign shareholding, audit quality, and stock price synchronicity: Evidence from China", *Journal of Financial Economics*, vol. 95, no. 3, pp. 425-442.
- Gupta, N. and Yuan, K. (2004), "On the growth effect of stock market liberalizations", University of Michigan mimeo.
- Hargis, K., and Ramanlal, P. (1997), "The internationalization of emerging equity markets: Domestic market development or market retardation", Working paper, University of South Carolina.
- Haidar, J. I. (2009), "Investor protections and economic growth", *Economics Letters*, vol. 103, no. 1, pp. 1-4.
- Hail, L. and Leuz, C. (2005), "Cost of Capital and Cash Flow Effects of U.S. Cross-Listings", Working paper, University of Pennsylvania.
- Hail, L. and Leuz, C. (2006), "International differences in the cost of equity capital: do legal institutions and securities regulation matter?" *Journal of Accounting Research*, vol. 44, no. 3, pp. 485–531.
- Hail, L. and Leuz, C. (2009), "Cost of capital and changes in growth expectations around U.S cross-listings", *Journal of Financial Economics*, vol. 93, no. 3, pp. 428-454.
- Hair, J. F., Anderson, R. E., Tatham, R. C. and Black, W. C. (1998), *Multivariate Data Analysis*. Upper Saddle River. New Jersey: Prentice-Hall.

- Hart, D., Atkins, R. and Youniss, J. (2005), "Knowledge, youth bulges, and rebellion", Psychological Science, vol. 16, no. 8, pp. 661-662.
- Harvey, C. R. (1991), "The world price of covariance risk", *Journal of Finance*, vol. 46, no. 1, pp. 111-157.
- Harvey, C. R. (1995), "Predictable risk and returns in emerging markets", *Review of Financial Studies*, vol. 8, no. 3, pp. 773-816.
- Hau, H. (2001), "Location matters: An examination of trading profits", *Journal of Finance*, vol. 56, no. 5, pp. 1959-1983.
- Heckman, J. (1979), "Sample selection bias as a specification error", *Econometrica*, vol. 47, no. 1, 153-162.
- Heflin, F. and Shaw, K. W. (2000), "Blockholder ownership and market liquidity", *Journal of Financial and Quantitative Analysis*, vol. 35, no. 4, pp. 621-633.
- Henry, P. B. (1998), "Stock market liberalization, economic reform, and emerging market equity prices", *Research Paper 1505R*, Stanford, Calif.: Stanford University.
- Henry, P. B. (2000a), "Stock market liberalization, economic reform and emerging market equity prices", *Journal of Finance*, vol. 55, no. 2, pp. 529-564.
- Henry, P. B. (2000b), "Do Stock Market Liberalizations Cause Investment Booms?" *Journal* of Financial Economics, vol. 58, Nos. 1-2, pp. 301-334.
- Henry, P. B. (2003), "Capital account liberalization, the cost of capital, and economic growth", *American Economic Review*, vol. 93, pp. 91-96.
- Hentschel, L. and Long, J. B. (2004), "Numeraire portfolio measures of the size and source of gains from international diversification", Working Paper, University of Rochester.
- Heston, S. L. and Rouwenhorst, K. G. (1994), "Does Industrial Structure explain the benefits of international diversification?" *Journal of Financial Economics*, vol. 36, no. 1, pp. 3-27.
- Hietala, P. (1989), "Asset pricing in partially segmented markets: Evidence from Finnish market", *Journal of Finance*, vol. 44, no. 3, pp. 697-715.
- Holden, C. W. and Subrahmanyam, A. (1992), "Long-lived private information and imperfect Competition", *Journal of Finance*, vol. 47, no. 1, pp. 247-270.
- Holmstrom, B. and Tirole, J. (1993), "Market liquidity and performance monitoring", *Journal of Political Economy*, vol. 101, no. 3, pp. 678-709.
- Horn, N. (2001), Cross-border Mergers and Acquisitions and the Law: A General Introduction. In Norbert Horn, (Ed.). *Cross-Border Mergers and Acquisitions and the Law*. Great Britain: Kluwer Law International.

- Huang, Y. and Temple, J. R. W. (2005), "Does external trade promote financial development?" CEPR discussion paper no. 5150.
- Huang, W., Zhu, T., (2012), Foreign institutional investors and corporate governance in emerging markets: Evidence from split share reform in China. [Online] Available from: <u>http://papers.ssrn.com/sol3/papers.cfm?abstract_id=2022984</u> [Accessed: 23 April 2013].
- Huberman, G. and Kandel, S, (1987), "Mean-Variance Spanning", *Journal of Finance*, vol. 42, no. 2, pp. 873-888.
- Hunter, D. M. (2006), "The evolution of stock market integration in the post-liberalization period A look at Latin America", *Journal of International Money and Finance*, vol. 25, no. 5, pp. 795-280.
- Huybens, E. and Smith, B. D. (1999), "Inflation, Financial Markets and Long-run Real Activity", *Journal of Monetary Economics*, vol. 43, no.2, pp. 283–315.

International Monetary Fund. (2005), World Economic Outlook (April), Chapter II.

- International Country Risk Guide. (2012). [Online] Available from: <u>http://www.eui.eu/Research/Library/ElectronicResources/E-Resources/2012/12-10-</u> InternationalCountryRiskGuide.aspx [Accessed 15 November 2012].
- Islam, R. and Montenegro, C.E. (2002), "What determines the quality of institutions?" *World Bank Policy Research Working Paper No* 2764. World Bank, Washington, DC.
- Janakiramanan, S. (1986), "A Model of Inter-national Asset Pricing with a Constraint on the Foreign Equity Ownership", *Journal of finance*, vol. 41, no. 4, pp. 897-914.
- Jegadeesh, N. Titman, S. (1993), "Returns to buying winners and selling losers: Implications for stock market efficiency", *Journal of Finance*, vol. 48, no. 1, pp. 65-91.
- Jensen, M. C. and Meckling, W. (1976), "Theory of the firm: Managerial behaviour, agency costs and capital structure", *Journal of Financial Economics*, vol. 3, pp. 305-360.
- Jensen, M. C. and Murphy, K. J. (1990), "Performance pay and top-management", Journal of Political Economy, vol. 98, no. 2, pp. 225-264.
- Jewel, J. and Livingston, M. (1998), "Split Ratings, Bond Yields, and Underwriter Spreads", *Journal of Financial Research*, vol. 21, no. 2, pp. 185-204.
- Johnson, S., Boone, P., Breach, A. and Friedman, E. (2000a), "Corporate governance in the Asian financial crisis", *Journal of Financial Economics*, vol. 58, nos. 1-2, pp. 141-186.
- Johnson, S., La Porta, R., Lopez-de-Silanes, F. and Shleifer, A. (2000b), "Tunneling", *American Economic Review Papers and Proceedings*, vol. 90, no. 2, pp. 22-27.

- Jones, C.S. (2003), "The dynamics of stochastic volatility; evidence from underlying and options markets", *Journal of Econometrics*, vol. 116, nos. 1-2, pp. 181-224.
- Jotikasthira, C., Lundblad, C. and Ramadorai, T. (2012), "Asset Fire Sales and Purchases and the International Transmission of Funding Shocks", *Journal of Finance*, vol. 67, no 6, 2015-2050.
- Jung, M. (2006), "Host country attractiveness for CDM non-sink projects", *Energy Policy*, vol. 34, no. 15, pp. 2173–2184.
- Kahn, C. and Winton, A. (1998), "Ownership structure, speculation, and shareholder intervention", *Journal of Finance*, vol. 53, no. 1, pp. 99-129.
- Kaminsky, G. and Schmukler, S. (2003), "Short-run pain, long-run gain: the effects of financial liberalization", NBER Working Paper 9787.
- Kang, J. K. and Kim, J. M. (2010), "Do foreign investors exhibit a corporate governance disadvantage? An information asymmetry perspective", *Journal of International Business Studies*, vol. 41, pp. 1415-1438.
- Kang, J. K. and Stulz, R. M. (1997), "Why is there a home bias? An analysis of foreign portfolio equity ownership in Japan", *Journal of Financial Economics*, vol. 46, no. 1, pp. 3-28.
- Kar, P. (2001), "Corporate governance in India", In Corporate Governance in Asia A Comparative Perspective. OECD.
- Karolyi, A. (2004), "The role of ADRs in the development and integration of emerging equity markets", *Review of Economics and Statistics*, vol. 86, no. 3, pp. 670-690.
- Karolyi, G. A. and Stulz, R. M. (2003), Are Financial Assets Priced Locally or Globally? Handbook of the Economics of Finance, Volume 1B: Asset pricing, George M. Constantinides, Milton Harris, and René M. Stulz, eds., Elsevier North Holland, Amsterdam, The Netherlands.
- Kaufmann, D. and Kraay, A. (2003), "Governance and growth: Causality which way? Evidence for the world, in brief", Washington DC: World Bank.
- Kaufmann, D., Kraay, A. and Mastruzzi, M. (2010)," The World Governance Indicators-Methodology and Analytical issues", Washington DC: World Bank.
- Kaufmann, D., Kraay, A. and Zoido-Lobaton, P. (1999)," Aggregating governance indicators", Washington DC: World Bank.
- Kaufmann, D., Kraay, A. and Zoido-Lobaton, P. (2002), "Governance matters II: Updated indicators for 2000/01", Washington DC: World Bank.
- Kaya, I., Lyubimov, K. and Miletkov, M. (2012), "To liberalize or not to liberalize: Political and economic determinants of financial liberalization", *Emerging Markets Review*, vol. 13, no.1, pp. 78-99.

- Ke, B. and Petroni, K. (2004), "How informed are actively trading institutional investors? Evidence from their trading behaviour before a break in a string of consecutive earnings increases", *Journal of Accounting Research*, vol. 42, no. 5, pp. 895-927.
- Kennedy, P. (2008), A guide to econometrics. 5th ed. Cambridge, UK: The MIT Press.
- Khanna, T. and Palepu, K. (2000), Emerging market business groups, foreign intermediaries, and corporate governance. In: Morck, R. K. (Ed.), *Concentrated Corporate Ownership*. Chicago: University of Chicago Press.
- Kim, E. H. and Singal, V. (2000). "Stock market openings: experience of emerging economies", *Journal of Business*, vol. 73, no. 1, pp. 25-66.
- King, R. G. and Levine, R. (1993a), "Finance and growth: Schumpeter might be right", *Quarterly Journal of Economics*, vol. 108, pp. 717-738.
- Klapper, L. F. and Love, I. (2004). "Corporate governance, investor protection, and performance in emerging markets", *Journal of Corporate Finance*, vol. 10, no. 5, pp. 723-728.
- Knack, S., Kugler, M. and Manning, N. (2003), "Second-generation governance indicators", *International Review of Administrative Services*, vol. 69, no. 3, pp. 345-364.
- Kyle, A. S. (1984), "Market Structure, Information, Futures Markets, and Price Formation," in G. G. Storey, A. Schmitz, and A. H. Sarris (ed.), *International Agricultural Trade: Advanced Readings in Price Formation, Market Structure, and Price Instability*, Westview Press: Boulder, Colo., pp. 45-64.
- Kyle, A. S. (1985), "Continuous auctions and insider trading", *Econometrica*, vol. 53, no. 6, pp. 1315-1335.
- Laeven, L. (2003), "Does financial liberalization reduce financing constraints?" *Financial Management*, vol. 32, no.1, pp. 5-34.
- Laeven, L. and Majnoni, G. (2005), "Does judicial efficiency lower the cost of credit?" *Journal of Banking and Finance*, vol. 29, no. 2, pp. 1791-1812.
- Laffont, J. J. and N'Guessan, T. J. (1999), "Competition and corruption in an agency relationship", *Journal of Development Economics*, vol. 60, no. 2, pp. 271-295.
- Laffont, J. J. and Tirole, J. (1988), "Adverse selection and renegotiation in procurement", Mimeo. (MIT).
- Lam, S. S., Koh, F. and Chia, Y. K. (1990), "Singapore Airlines: Price effects of restriction on foreign ownership", *Securities Industry Review*, vol. 16, pp. 57-66.
- La Porta, R., Lopez-de-Silanes, F., and Shleifer. A. (2002), "Government Ownership of Banks", *Journal of Finance*, vol. 57, no. 1, pp. 265–301.

- La Porta, R., Lopez-de-Silanes, F., Shleifer, A. and Vishny, R. (1997), "Legal Determinants of External Finance", *Journal of Finance*, vol. 52, no. 3, pp. 1131-1150.
- La Porta, R., Lopez-de-Silanes, F., Shleifer, A. and Vishny, R. (1998), "The Quality of Government", National Bureau of Economic Research, Working paper 6727. Cambridge, MA.
- La Porta, R., Lopez-de-Silanes, F., Shleifer, A. and Vishny, R. (1999a). "The Quality of Government", *Journal of Law, Economics, & Organization*, vol. 15, no. 1, pp. 222-279.
- La Porta, R., Lopez-de-Silanes, F., Shleifer, A. and Vishny, R. (1999b), "Investor protection and corporate valuation", NBER Working paper 7403.
- La Porta, R., Lopez-de-Silanes, F., Shleifer, A. and Vishny, R. (2000a), "Agency problems and dividend policies around the world", *Journal of Finance*, vol. 55, no. 1, pp. 1-33.
- La Porta, R., Lopez-de-Silanes, F., Shleifer, A. and Vishny, R. (2000b), "Investor protection and corporate governance", *Journal of Financial Economics*, vol. 58, no. 1, pp. 3-27.
- Lau, S. T., Ng, L. and Zhang, B. (2010), "The world price of home bias", *Journal of Financial Economics*, vol. 97, no. 2, pp. 191-217.
- Lessard, D. R. (1973), "International portfolio diversification: A multivariate analysis for a group of Latin American countries", *Journal of Finance*, vol. 28, no. 3, pp. 619-633.
- Lessard, D. R. (1976), "World, country, and industry relationship in equity returns: Implications for risk reduction through international diversification", *Financial Analysts Journal*, vol. 32, no. 1, pp. 32-38.
- Leuz, C., Lins, L. V. and Warnock, F. E. (2010), "Do foreigners invest less in poorly governed firms?" *Review of Financial Studies*, vol 23, no. 3, pp. 3245-3285.
- Leuz, C., Nanda, D. and Wysocki, P. (2003), "Earnings management and investor protection: an international comparison", *Journal of Financial Economics*, vol. 69, no. 3, pp. 505-527.
- Levine, R. (1991), "Stock markets, growth, and tax policy", *Journal of Finance*, vol. 46, no. 4, pp. 1445-1465.
- Levine, R. (1997), "Financial Development and Economic Growth: Views and Agenda", *Journal of Economic Literature*, vol. 35, no. 2, pp. 688-726.
- Levine, R. (1999), "Law, finance, and economic growth", *Journal of Financial Intermediation*, vol. 8, nos. 1-2, pp. 36-67.
- Levine, R., Loayza, N. and Beck, T. (2000), "Financial intermediation and growth: causality and causes", *Journal of Monetary Economics*, vol. 46, no. 1, pp. 31-77.

- Levine, R. and Zervos, S. (1995), "Policy, stock market development and economic growth," Paper presented at the World Bank Conference on Stock Markets, Corporate Finance and Economic Growth", *The World Bank*, Washington, DC, February 16-17.
- Levine, R. and Zervos, S. (1996), "Stock market development and long-run growth", *World Bank Economic Review*, vol. 10, no. 2, pp. 323-339.
- Levine, R. and Zervos, S. (1998), "Capital control liberalization and stock market development", *World Development*, vol. 26, no. 7, pp. 1169-1184.
- Levy, H. and Sarnat, M. (1970), "International diversification of investment portfolios", *American Economic Review*, vol. 60, no. 4, pp. 668-675.
- Lewis, K. K. (1999), "Explaining home bias in equities and consumption", *Journal of Economics. Literature*, vol. 37, no. 2, pp. 571-608.
- Li, M., Ramaswamy, K. and Petitt, B. (2006), "Business groups, market failures: a focus on vertical and horizontal settings", *Asia Pacific Journal of Management*, vol. 23, no, 4, pp. 439-452.
- Li, K., Sarkar, A. and Wang, Z. (2003), "Diversification benefits of emerging markets subject to portfolio constraints", *Journal of Empirical Finance*, vol. 10, nos. 1-2, pp. 57–80.
- Liu, M.C. and San, G. (2006), "Social learning and digital divides: A case study of internet technology diffusion", *International Review for Social Sciences*, vol. 59, no. 2, pp. 307-321.
- Llamazares, I. (2005), "Patterns in contingencies: The interlocking of formal and informal political institutions in contemporary Argentina", *Social Forces*, vol. 83, no. 4, pp. 1671-1695.
- Lombardo, D. and Pagano, M. (2000), Law and Equity Markets: A Simple Model", forthcoming in Convergence and diversity of corporate governance regimes and capital markets, edited by Luc Renneboog, Joe McCahery, Pieter Moerland and Theo Raaijmakers. Oxford University Press.
- Markowitz, H. (1952), "Portfolio selection", Journal of Finance, vol. 7, no. 1, pp. 77-91.
- Markowitz, H. (1959), Portfolio selection: Efficient diversification of investments, New York: Wiley.
- Martin, P. and Rey, H. (2000), "Financial integration and asset returns", *European economic Review*, vol. 44, no. 7, pp. 1327-1350.
- Mauro, P. (1995), "Corruption and Growth", *Quarterly Journal of Economics*, vol. 110, no. 3, pp. 681–712.
- Mauro, P. (1996), "The effects of corruption on growth, investment and government expenditure", *Journal of Public Economics*, vol. 69, no. 2, pp. 263-279.

- Mayshar, J. (1979), "Transaction Cost in a Model of Capital Market Equilibrium," *Journal of Political Economy*, vol. 87, no.4, pp. 673-700.
- McCarthy, J., Najand, M. and Seifert, B. (1990), "Empirical tests of the proxy hypothesis," *Financial Review*, vol. 25, no. 2, pp. 251-63.
- McKinnon, R. I. (1973), *Money and capital in economic development*. Washington, DC: Brookings Institution.
- McKinsey and Company (2003), "New horizons multinational company investment in developing countries, Chapter 3, auto sector case study", McKinsey Global Institute.
- McKnight, P. J., and Weir, C. (2009), "Agency Costs, Corporate Governance Mechanisms and Ownership Structure in Large UK Publicly Quoted Companies: A Panel Data Analysis," *Quarterly Review of Economics and Finance*, vol. 49, pp. 139-158.
- Mendelson, H. and Tunca, T. I. (2004), "Strategic trading, liquidity, and information acquisition," *Review of Financial Studies*, vol. 17, no. 2, pp. 295-337.
- Megginson, W., Nash, R. and Van Randenborgh, M. (1994), "The Financial and Operating Performance of Newly Privatized Firms: An International Empirical Analysis", *Journal of Finance*. Vol. 49, no. 2, pp. 403-452.
- Me'on, P. G. and Sekkat, K. (2005), "Does corruption grease or sand the wheels of growth?" *Public Choice*, vol. 122, nos. 1–2, pp. 69–75.
- Merton, R.C. (1987), "A Simple Model of Capital Market Equilibrium with Incomplete Information", *Journal of Finance*, vol. 42, no. 3, pp. 483-510
- Miller, D. P. (1999), "The impact of international market segmentation on securities prices: evidence from Depositary Receipts", *Journal of Financial Economics*, vol. 51, no. 1, pp. 103-123.
- Mitton, T. (2002), "A cross-firm analysis of the impact of corporate governance on the East Asian financial crisis", *Journal of Financial Economics*, vol. 64, no. 2, pp. 215-241.
- Mitton, T. (2004), "Corporate governance and dividend policy in emerging markets". *Emerging Markets Review*, vol. 5, no. 4, pp. 409-426.
- Mitton, T. (2006), "Stock market liberalization and operating performance at the firm level", *Journal of Financial Economics*, vol. 81, no. 3, pp. 625-647.
- Monks, R. and Minow, N. (1995), "Corporate governance, Blackwell Business", Cambridge, MA.
- Naude, W. A. (2004), "The effects of policy, institutions and geography on economic growth in Africa: An econometric study based on cross-section and panel data", *Journal of International Development*, vol. 16, no. 6, pp. 821-849.

- Nelson, C. R. (1976), "Inflation and rates of return on common stock", *Journal of Finance*, vol. 31, no. 2, pp. 471-483.
- Nenova, T. (2000), "The value of a corporate vote and private benefits: A cross-country analysis", Manuscript, Harvard University.
- Nesbitt, S. L. (1994), "Long-term rewards from shareholder activism: a study of the 'CalPERS effect", *Journal of Applied Corporate Finance*, vol. 6, no. 4, pp. 75-80.
- Neumayer, E. (2002), "Do democracies exhibit stronger international environmental commitment? A cross-country analysis," *Journal of Peace Research*, vol. 39, no. 2, pp. 139-164.
- Newey, W. K. and West, K. D. (1987), "A Simple, Positive Definite, Heteroskedasticity and Autocorrelation Consistent Covariance Matrix," *Econometrica*, vol. 55, no. 3, pp. 703–708.
- Ng, L., Wu, F. and Zhang, B. (2011), "The role of foreign block holders in stock liquidity: A cross-country analysis", Working paper.
- Obstfeld, M. (1994), "Risk-taking, Global Diversification, and Growth", *American Economic Review*, vol. 84, no. 5, pp. 1310-1329.
- Obstfeld, M. and Rogoff, K. (2000), "New directions for stochastic open economy models", *Journal of International Economics*, vol. 50, pp. 117-153.
- Odier, P., Solnik, B. and Zucchinetti, S. (1995), "Global Optimization for Swiss Pension Funds", *Financial Markets and Portfolio Management*, vol. 9, pp. 210-231.
- Oman, C. P. (2001). "Corporate Governance and National Development", *Development Centre Technical Papers*, Number 180. Paris: OECD.
- Persson, T., Tabellini, G. and Trebbi, F. (2003),"Electoral Rules and Corruption", *Journal of the European Economic Association*, vol. 1, no. 4, pp. 958-989.
- Pistor, K., Raiser, M. and Gelfer, S. (2000), "Law and finance in transition economies", *Economics of Transition*, vol. 8, no. 2, pp. 325-368.
- Puhani, P. (2000). "The Heckman correction for sample selection and its critique", vol.14, no.1, pp. 53-68.
- Rajan, M. and Friedman, J., (1997), "An examination of the impact of country risk on the international portfolio selection", *Global Finance Journal*, vol. 8, no. 1, pp. 55-70.
- Rajan, R. G. and Zingales, L. (1998), "Which Capitalism: Lessons from the Asian Crisis", *Journal of Applied Corporate Finance*, vol. 11, no. 3, pp. 40-48.
- Rajan, R. G. and Zingales, L. (2000), *The governance of the New Enterprise. In Corporate Governance, Theoretical and Empirical Perspectives.* Xaxier Vives ed. Cambridge: Cambridge University Press.

- Rajan, R. G. and Zingales. L. (2003), "The Great Reversals: The Politics of Financial Development in the Twentieth Century", *Journal of Financial Economics*, vol. 69, no. 1, pp. 5-50.
- Rauch, J. E. and Evans. B. (2000), "Bureaucratic Structure and Bureaucratic Performance in Less Developed Countries", *Journal of Public Economics*, vol. 75, no. 1, pp. 49-71.
- Reeb, D. M., Sattar, A. M. and Alle, J. M. (2001), "Firm internationalization and the cost of debt financing. Evidence from non-provisional publicly traded debt", *The Journal of Financial and Quantitative Analysis*, vol. 36, no. 3, pp. 395-414.
- Rhee, S. G. and Wang, J. (2009), "Foreign institutional ownership and stock market liquidity: Evidence from Indonesia", *Journal of Banking and Finance*, vol. 33, no. 7, pp. 1312-1324.
- Roberts, M. and Whited, T. (2011), "Endogeneity in corporate finance", Working paper, Wharton.
- Rogoff, K. S. (1999), "Institutions for Reducing Global Financial Instability", *Journal of Economic Perspectives*, vol. 13, no. 4, pp. 21-42.
- Rousseau, P. L. and Sylla, R. (2001), "Financial system, economic growth, and globalization", National Bureau of Economic Research Working paper No. 8323.
- Rousseau, P. L. and Wachtel, P. (1998), "Financial intermediation and economic performance: Historical evidence from five industrialized countries", *Journal of Money, Credit and Banking*, vol. 30, no. 4, pp. 657–678.
- Rousseau, P. L. and Wachtel, P. (2000), "Equity markets and growth: Cross country evidence on timing and outcomes, 1980-95", *Journal of Banking and Finance*, vol. 24, no. 12, pp. 1933-1957.
- Rubin, A. (2007) "Ownership Level, Ownership Concentration and Liquidity", *Journal of Financial Markets*, vol. 10, no. 3, pp. 219-248.
- Sarin, A., Shastri, K. A. and Shastri, K. (2000), "Ownership structure and stock market liquidity", *Working paper*, University of Pittsburgh.
- Scharfstein, D. (1988), "The disciplinary role of takeovers", *Review of Economic Studies*, vol. 55, no. 2, pp. 185-200.
- Schineller, L. (1997), "An Econometric Model of Capital Flight from Developing Countries", *Federal Reserve Board, International Finance Discussion Paper* No. 579.
- Schmukler, S. and Vesperoni, E. (2003), "Financial globalization and debt maturity in emerging economies," *Working paper, World Bank and International Monetary Fund.*

Schumpeter. J. A. (1911), The Theory of Economic Development.

- Seasholes, M. S. (2000), "Smart foreign traders in emerging markets", Working paper, Harvard University.
- Seasholes, M. S. (2004), "Re-examining information asymmetries in emerging stock markets", Working paper, University of California-Berkeley.
- Shaw, E. S. (1973), *Financial deepening in economic development*. New York: Oxford University Press.
- Shawky, H. A., Kuenzel, R. and Mikhail, A. D. (1997), "International portfolio diversification: A synthesis and an update", *Journal of International Financial Markets*, *Institutions and Money*, vol. 7, no. 4, pp. 303-327.
- Sheets, N. (1996), "Capital Flight from the Countries in Transition: Some Empirical Evidence", *Journal of Policy Reform*, vol. 1, no. 3, pp. 259-277.
- Shleifer, A. and Vishny, R. (1986), "Large shareholders and corporate control", *Journal of Political Economy*, vol. 95, no. 3, pp. 461–488.
- Shleifer, A. Vishny, R. (1997), "A survey of corporate governance", *Journal of Finance*, vol. 52, no. 2, pp. 737-783.
- Shleifer, A. Wolferzon, D. (2002), "Investor protection and equity markets", *Journal of Financial Economics*, vol. 66, no. 1, pp. 187-214.
- Smarzynsk, B. and Wei, S. J. (2000), "Corruption and composition of foreign direct investment: Firm-level evidence", NBER WP w7969.
- Smith, M. (1996), "Shareholder activism by institutional investors: Evidence from CalPERS", *Journal of Finance*, vol. 51, no. 1, pp. 227-252.
- Solnik, B. H. (1974), "Why not diversify international rather than domestically?" *Financial Analyst Journal*, vol. 30, no. 4, pp. 48-54.
- Solnik, B. and McLeavey, D. (2004), *International Investments*, Fifth Edition, Boston: Pearson Addison Wesley.
- Spamann, H. (2010), "The 'anti director rights index' revisited", *Review of Financial Studies* vol. 23, no. 2, pp. 467-486.
- Stapleton, R. and Subrahmanyam, M. (1977), "Market imperfections, capital market, and equilibrium, and corporate finance", *Journal of Finance*, vol. 32, no. 2, pp. 307-319.
- Stiglitz, J. E. (2000), "Capital market liberalization, economic growth and instability", *World Development*, vol. 28, no. 6, pp. 1075-1086.
- Stoll, H. R. (2000), "Friction", Journal of Finance, vol. 55, no. 4, pp. 1479-1514.
- Straub, S. (2000), "Empirical Determinants of Good Institutions, Do we know anything? Inter-American Development Bank", *Research Department*, Working Paper No 423.

- Stulz, R. M. (1981b), "On the Effects of Barriers to International Investment", *Journal of Finance*, vol. 36, no. 4, pp. 923-934.
- Stulz, R. M. (1999a), International portfolio flows and security markets, in Martin Feldstein, ed.: *International Capital Flows*. Chicago: University Chicago Press.
- Stulz, R. M. (1999b), "Globalization, corporate finance, and the cost of capital", *Journal of Applied Corporate Finance*, vol. 12, no. 3, pp. 8-25.
- Stulz, R. M. (2005), "Limits of Financial Globalization," *Journal of Finance*, vol. 60, no. 4, pp. 1595-1638.
- Stulz, R., Wasserfallen, W. (1995), "Foreign equity investment restrictions, capital flight and shareholder wealth maximization: Theory and evidence", *Review of Financial Studies*, vol. 8, no. 4, pp. 1019-1057.
- Subrahmanyam, A. (1991), "Risk aversion, market liquidity, and price efficiency", *Review of Financial Studies*, vol. 4, no. 3, pp. 417-441.
- Summers, L. H. (2000), "International Financial Crises: Causes, Prevention, and Cures", *American Economic Review*, vol. 90, no. 2, pp. 1-16.
- Tesar, L. and Werner, I. (1995), "Home bias and high turnover", *Journal of International Money and Finance*, vol. 14, no. 4, pp. 467-493.
- Tesar, L. L. and Werner, I. (1998), "The Internationalization of Securities Markets since the 1987 Crash." In *Brookings–Wharton Papers on Financial Services*, ed. R. Litan and A. Santomero. Washington, D.C.: Brookings Institution Press, 281–349.
- Thapa, C. and Poshakwale, S. S. (2010), "International equity portfolio allocations and transaction costs", *Journal of Banking and Finance*, vol. 34, no. 11, pp. 2627-2638.
- Treisman, D. (2000), "The Causes of Corruption: A Cross-National Study", *Journal of Public Economics*, vol. 76, no. 3, pp. 399-457.
- Vagias, D. and van Dijk, M. (2011), "International Capital Flows and Liquidity", Erasmus University Working Paper.
- Verbeek, M. (2012), "A guide to modern econometrics (4th Ed.)", John Wiley and Sons, New York.
- Warnock, F. E. (2002), "Home bias and high turnover reconsidered", *Journal of International Money and Finance, vol.* 21, no. 6, pp. 795-805.
- Watts, R. L. and Zimmerman, J. L. (1986), *Positive accounting theory*. New Jersey: Prentice Hall.
- Wei, S. J. (2000), "How taxing is corruption on international investors?" *The Review of Economics and Statistics*, vol. 82, no. 1, pp. 1-11.

- Wei, S. J., Zhang, Z. and Du, Q. (2010), "Does the global fireman inadvertently add fuel to the fire? New evidence from institutional investors' responses to IMF program announcements", *Journal of International Money and Finance*, vol. 29, no. 4, pp. 728-741.
- Wei, Z. F. X. and Zhang, S. (2005), "Ownership structure and firm value in 34 China's privatized firms: 1991-2001", *Journal of Financial and Quantitative Analysis*, vol. 40, no. 1, pp. 87-108.
- Wooldridge, J. M. (2009). Introductory Econometrics: A Modern Approach, 4e. Mason, OH: South-Western CENGAGE Learning.
- Wurgler, J. (2000), "Financial markets and allocation of capital", *Journal of Financial Economics*, vol. 58, nos. 1-2, pp. 187-214.
- Yang, H. Y. (2003), "A note on the causal relationship between energy and GDP in Taiwan", *Energy Economics*, no. 22, pp. 309-317.
- Yermack, D. (2006), "Flights of Fancy: Corporate Jets, CEO Perquisites and Inferior Shareholder Returns", *Journal of Financial Economics*, vol. 80, no. 1, pp. 211-242.
- Zingales, L. (1994), "The value of the voting right: A study of the Milan stock exchange", *Review of Financial Studies*, vol. 7, no. 1, pp. 125-148.